# **Environmental Security Technology Certification Program**

# (ESTCP)

# In-situ Substrate Addition to Create Reactive Zones for Treatment of Chlorinated Aliphatic Hydrocarbons: Hanscom Air Force Base

Contract #41624-99-C-8032



April 4, 2003 Final Report

Report Documentation Page			Form Approved OMB No. 0704-0188		
Public reporting burden for the collection of information is estimated to average 1 hour per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding this burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to Washington Headquarters Services, Directorate for Information Operations and Reports, 1215 Jefferson Davis Highway, Suite 1204, Arlington VA 22202-4302. Respondents should be aware that notwithstanding any other provision of law, no person shall be subject to a penalty for failing to comply with a collection of information if it does not display a currently valid OMB control number.					
1. REPORT DATE 04 APR 2003	REPORT DATE     2. REPORT TYPE		3. DATES COVERED 00-00-2003 to 00-00-2003		
4. TITLE AND SUBTITLE			5a. CONTRACT NUMBER		
In-situ Substrate Addition to Create Reactive Zones for Treatment of			reatment of	5b. GRANT NUMBER	
Chlorinated Aliphatic Hydrocarbons: Hanscom Air Force Base			e Base	5c. PROGRAM ELEMENT NUMBER	
6. AUTHOR(S)				5d. PROJECT NUMBER	
				5e. TASK NUMBER	
				5f. WORK UNIT NUMBER	
7. PERFORMING ORGANIZATION NAME(S) AND ADDRESS(ES) ARCADIS G&M, Inc,4915 Prospectus Dr., Suite F,Durham,NC,27713			8. PERFORMING ORGANIZATION REPORT NUMBER		
9. SPONSORING/MONITORING AGENCY NAME(S) AND ADDRESS(ES) 10. SPONSOR/MONITOR'S ACRONYM(			ONITOR'S ACRONYM(S)		
			11. SPONSOR/MONITOR'S REPORT NUMBER(S)		
12. DISTRIBUTION/AVAILABILITY STATEMENT Approved for public release; distribution unlimited					
13. SUPPLEMENTARY NC	DTES				
14. ABSTRACT					
15. SUBJECT TERMS					
16. SECURITY CLASSIFICATION OF: 17. LIMITATION OF			18. NUMBER	19a. NAME OF	
a. REPORT unclassified	b. ABSTRACT unclassified	c. THIS PAGE unclassified	Same as Report (SAR)	<b>431</b>	RESI UNSIDLE FERSUN

Standard Form 298 (Rev. 8-98) Prescribed by ANSI Std Z39-18

# Final Report: In-situ Substrate Addition to Create Reactive Zones for Treatment of Chlorinated Aliphatic Hydrocarbons: Hanscom Air Force Base

Prepared for: Air Force Center for Environmental Excellence (AFCEE) 3207 Sydney Brooks Brooks AFB, Texas 78235-5344 POC: Mr. Jerry Hansen

and

Environmental Security Technology Certification Program (ESTCP) 901 North Stuart Street, Suite 303 Arlington, Virginia 22203 POC: Dr. Andrea Leeson

Prepared by: ARCADIS G&M, Inc. 4915 Prospectus Dr., Suite F Durham, North Carolina 27713 Christopher C. Lutes, Victor D'Amato P.E., Angela Frizzell P.G., Michael Hansen P.E., Glen Gordon P.E., Peter Palmer P.E., P.G. and Suthan Suthersan Ph.D., P.E.

## April 4, 2003

Under Contract #41624-99-C-8032

## CONTENTS

LIS	T OF ACRONYMS AND ABBREVIATIONS	V
ACH	KNOWLEDGEMENTS	XIII
PRE	EFACE	XIV
EXE	ECUTIVE SUMMARY	1
1.	INTRODUCTION	
1.1	Background	5
1.2	Objectives of the Demonstration	6
1.2	2.1 Objectives	6
1.3	Regulatory Drivers	6
1.4	Stakeholder/End-User Issues	7
2.	TECHNOLOGY DESCRIPTION	9
2.1	Technology Development and Application	9
2.	1.1 General Technology Description and History	9
2.	1.2 Design Criteria	
2.2	Previous Testing of the Technology	
2.3	Factors Affecting Cost and Performance	
2.4	Advantages and Limitations of the Technology	14
3.	SITE/FACILITY DESCRIPTION	
3.1	Performance Objectives	17
3.2	Selecting Test Site	
3.3	Test Site Description	19
3.3	3.1 Site/Facility Description	
3.3	3.2 Site/Facility Characteristics	
	3.3.2.1 Geology and Hydrogeology	
	3.3.2.2 Climatology	
	3.3.2.3 Distribution of CAHs and Current Pumping Remedy	
	3.3.2.3.1 Existing Groundwater Remediation	
3.3	3.3 Site/Facility Maps and Photographs	
3.3	3.4 Conceptual Site Model	
3.4	Pre-Demonstration Testing and Analysis	
3.5	Testing and Evaluation Plan	
3.5	5.1 Demonstration Installation and Startup	

3.	.5.2	Period	of Operation	29
3.	.5.3	Amou	nt/Treatment Rate of Material Treated	29
3.	.5.4	Residu	als Handling	29
3.	.5.5	Operat	ting Parameters for the Technology	
3.	.5.6	Experi	mental Design	
3.	.5.7	Sampl	ing Plan	
	3.5.7.	1 5	Sample Collection	
	3.5	5.7.1.1	Well Installation Procedures	31
	3.5	5.7.1.2	Groundwater Sampling	
	3.5	5.7.1.3	Saturated Soil Sampling	35
	3.5	5.7.1.4	Sample Shipment and Labeling	35
	3.5.7.	2 5	Sample Analysis	
	3.5.7.	3 I	Experimental Controls	
	3.5.7.	4 I	Data Quality Parameters	
	3.5.7.	5 I	Data Quality Indicators	
	3.5.7.	6 (	Calibration Procedures, Quality Control Checks, and Corrective Action	
3.	.5.8	Demol	bilization	
3.6	Se	lection	of Analytical/Testing Methods	
3.7	Se	lection	of Analytical/Testing Laboratory	
4.	PER	FOR	MANCE ASSESSMENT	39
4.1	Pe	erforma	ance Criteria	
4.2	Pe	erforma	ance Confirmation Methods	
4.3	Da	ata Ana	alysis, Interpretation and Evaluation	
4.	.3.1	Perfor	mance Overview	
4.	.3.2	Proces	s Monitoring Results and System Operational Adjustments	40
	4.3.2.	1 I	Injection Rates and Field Parameter Observations	40
	4.3.2.	2 0	Changes in Groundwater Flow	41
	4.3.2.	.3 .	TOC, DOC, and Other Measures of Reactive Zone Influence	41
	4.3.2.	4 1	Bromide Tracer Data	42
4	.3.3	CAH	Freatment	43
	4.3.3.	1 5	Summary of CAH Treatment Results	43
	4.3.3.	2 1	Methodologies Used in CAH Data Analysis	44
	4.3	3.3.2.1	Dilution Calculations Using Tracer	45
	4.3	3.3.2.2	Bar-Line Plots of Substrate Dose vs. CAH concentration	46
	4.3	3.3.2.3	MCL Index	46
	4.3	3.3.2.4	Rate Calculations	47
	433	3 1	Detailed Discussion of Observations at Individual Wells	47

4.3.3.3.	CAH Data at Well IRZ-1	47
4.3.3.3.2	2 CAH Data at Well RAP1-6T	
4.3.3.3.	3 CAH Data at the Injection Well IRZ-INJ	
4.3.3.3.4	4 Upgradient Monitoring Well B239 CAH Data	
4.3.3.3.	5 CAH Data at all Other Monitoring Wells	
4.3.3.3.0	5 MCL Index	51
4.3.3.3.2	7 Statistical Analysis of CAH Data	
4.3.3.4	Analysis of CAH Data: Conditions Required for Enhanced Biodegradation	
4.3.3.5	Calculated Biodegradation Rates	
4.3.4 Hydr	ogeology Discussion	
4.3.4.1	Changes in Groundwater Flow Direction due to Changes in Pumping Rates	56
4.3.4.2	Groundwater Velocity Estimation	
4.3.5 Seco	ndary Water Quality Issues	
4.3.5.1	Arsenic - Field and Bench Scale Observations from Another DoD IRZ Site	
4.3.5.2	Metals in Molasses	60
4.3.6 Othe	r Observations	61
4.3.6.1	Reactive Zone Size and Residence Time	61
4.3.6.2	Utility of Hydrogen Data	
4.3.6.3	Fatty Acids	65
4.3.6.4	Microbial Population Characterization	65
4.3.7 Com	parison of Results with Objectives	
4.3.7.1	Primary Objectives	66
4.3.7.2	Secondary Objectives	67
4.3.8 Tech	nology Comparison	
4.3.8.1	Groundwater Extraction and Treatment (Pump & Treat)	70
4.3.8.2	Aquifer Sparging	71
4.3.8.3	Chemical Oxidation	71
4.3.8.4	ERD	71
4.3.9 Less	ons Learned	71
4.3.9.1	Substrate Dosing Required for Successful Treatment	
4.3.9.2	Microbial Acclimation / The Role of Bioaugmentation	
4.3.9.3	Fermentation and Byproduct Formation	
4.3.9.4	Pilot Test Design	74
4.3.9.5	Application in Areas of High Constituent Concentration/DNAPL	
4.3.9.6	Secondary Water Quality Impacts	75
5. COST A	SSESSMENT	
5.1 Cost Re	porting	76

5.2	Cost Analysis
5.2.1	Cost Comparison
5.2	2.1.1 Cost Comparisons at Commercial Sites vs. Pump & Treat76
5.2	2.1.2 Cost Comparisons at Commercial Sites vs. Other Innovative Technologies
5.2.2	Cost Basis
5.2.3	Cost Drivers
5.2.4	Life-Cycle Costs
6. IN	MPLEMENTATION ISSUES
6.1	Environmental Checklist
6.2	Other Regulatory Issues
6.3	End-User Issues
6.3.1	Secondary Water Quality and Gas Production Issues
6.3.2	Procurement and Implementation Issues
6.3.3	Transition
7. R	EFERENCES
8. P	OINTS OF CONTACT

Appendices

Appendix A-1	IRZ/ERD Bibliography
Appendix A-2	Soil Sieve Analyses
Appendix A-3	Standard Operating Procedure: Field Screening of Water & Vapor Samples for Volatile Organic Compounds by Gas Chromatography
Appendix A-4a	Boring Logs and Well Construction Diagrams
Appendix A-4b	Authorization to Leave Monitoring Wells in Place
Appendix A-5	Survey Data
Appendix A-6a	Additional Discussion of Data Quality
Appendix A-6b	Validation Memos
Appendix A-6c	QC Notes
Appendix A-7	Additional Discussion of Hydrogeologic Evaluations
Appendix A-8	Additional Discussion of Secondary Water Quality Issues
Appendix A-9	Data Archiving and Demonstration Plans
Appendix A-10	Microbial Characterization Report
Appendix C	Quality Assurance Project Plan (QAPP)

# List of Acronyms and Abbreviations

1,2-DCE	1,2-Dichloroethene
1,2-DCA	1,2-Dichloroethane
1,1,1-DCA	1,1,1-Dichloroethane
1,1,1-TCA	1,1,1-Trichloroethane
AFB	Air Force Base
AFCEE	Air Force Center for Environmental Excellence
bgs	below ground surface
BOD	Biochemical Oxygen Demand
BRAC	Base Realignment and Closure
CAH CEPCLA	Chlorinated Aliphatic Hydrocarbons
CERCLA	Chloroform
cis 1 2 DCE	ciis 1.2 Dichloroethene
<i>cis</i> -1,2-DCE	cis-1,2-Dichloroemene
COC	Constituents of Concern
COD	Chemical Oxygen Demand
COTS	Commercially Off-The-Shelf
СТ	Carbon Tetrachloride
DCE	Dichloroethene
DNAPL	Dense, Non-Aqueous Phase Liquid
DO	Dissolved Oxygen
DOC	Dissolved Organic Carbon
DoD	Department of Defense
EARP	Enhanced Anaerobic Reductive Precipitation
ERD	Enhanced Reductive Dechlorination
ESTCP	Environmental Security Technology Certification Program
GC	Gas Chromatograph
H <sub>2</sub>	Hydrogen
IRZ	In-situ Reactive Zone
Κ	Hydraulic Conductivity
L	Liters
MADEP	Massachusetts Department of Environmental Protection

MC	Methylene Chloride
MCL	Maximum Contaminant Level
mg/L	milligrams per Liter
MNA	Monitored Natural Attenuation
NAS	National Academy of Science
ORP	Oxidation Reduction Potential
PBC	Performance Based Contract
PCE	Tetrachloroethene
PCP	Pentachlorophenol
psig	per square inch gauge pressure
PVC	Polyvinyl Chloride
RCRA	Resource Conservation and Recovery Act
STL	Severn Trent Laboratories
TBP	Tributyl Phosphate
TCA	Trichloroethane
TCE	Trichloroethene
TDS	Total Dissolved Solids
TOC	Total Organic Carbon
TPH	Total Petroleum Hydrocarbon
trans-DCE	trans-1,2-Dichloroethene
TVOC	Total Volatile Organic Compound
μg/L	micrograms per liter
UIC	Underground Injection Control
U.S. EPA	United States Environmental Protection Agency
VC	Vinyl chloride
VER	Vacuum-Enhanced Recovery
VOC	Volatile Organic Compound

#### LIST OF FIGURES

- Figure 1-1 Area Map of Hanscom AFB
- Figure 2-1 Anaerobic Transformations of Selected CAHs and their Daughter Products (after Vogel et al., 1987 and McCarty et al., 1993)
- Figure 2-2 Conceptual Design for an ERD System Layout
- Figure 2-3 Reagent Mixing and Injection System Schematic
- Figure 2-4 Reagent Mixing and Injection System
- Figure 3-1 Overview of Hanscom AFB Showing Extraction Wells and Modeled Top of Bedrock Contours
- Figure 3-2 Total VOC Concentrations in Lower Aquifer Near RAP1-6T, May 1998
- Figure 3-3 Cross Section of Injection Area, A A'
- Figure 3-4 Cross Section of Injection Area, B B'
- Figure 3-5 RAP1-6 Well Cluster, Viewed From Access Road Looking Southeast
- Figure 3-6 Drainage Channel Behind Fence and RAP1-6 Well Cluster, Viewed From Southeast Side Of Drainage Channel, Looking Northwest Toward Overrun For Runway 23. VER Unit At Site 1 Is Visible In Distant Background
- Figure 3-7 Drainage Channel Looking Northeast
- Figure 3-8 Potentiometric Surface in Lower Aquifer Near RAP1-6T, May 1998
- Figure 3-9 Hanscom AFB Site Layout, Pilot Test Area
- Figure 3-10 CAHs at Site 1 Pump Station (from Hanscom AFB, 2002)
- Figure 3-11 CAHs at Interceptor Well #3 (from Hanscom AFB, 2002)
- Figure 3-12 Vertical Correlation of Wells
- Figure 3-13 TCE Source Cell Concentrations in the Lower Aquifer
- Figure 3-14 Summary of VOC Results for Baseline Groundwater Sampling Event June 2000
- Figure 3-15 Rolling Average Weekly Molasses Loading
- Figure 4-1-(a) Dissolved Oxygen vs. Time, Showing All Data and Injection Events
- Figure 4-1-(b) Dissolved Oxygen vs. Time, "IRZ" Labeled Wells Only
- Figure 4-1-(c) Dissolved Oxygen vs. Time, "RAP and B" Labeled Wells Only
- Figure 4-2-(a) Oxidation Reduction Potential vs. Time, Showing All Data and Injection Events
- Figure 4-2-(b) Oxidation Reduction Potential vs. Time, "IRZ" Labeled Wells Only
- Figure 4-2-(c) Oxidation Reduction Potential vs. Time, "RAP and B" Labeled Wells Only
- Figure 4-3 TOC Trend in the Injection Well
- Figure 4-4-(a) pH vs. Time, All Data and Injection Events
- Figure 4-4-(b) pH vs. Time, "IRZ" Labeled Wells only
- Figure 4-4-(c) pH vs. Time, "RAP and B" Labeled Wells Only
- Figure 4-5 Monitoring Well TOC Data
- Figure 4-6 Monitoring Well DOC Data

- Figure 4-7 Precipitation for Bedford, MA, and Gradient Between Well B-239 and Selected Wells Over Time
- Figure 4-8 Precipitation for Bedford, MA, and Gradient Between Well IRZ-INJ and Selected Wells Over Time
- Figure 4-9 Potentiometric Surface Measured 6-15-00 and 6-16-00, Hanscom AFB
- Figure 4-10 Potentiometric Surface Measured 12-21-00
- Figure 4-11 Potentiometric Surface as of May 2001
- Figure 4-12 Potentiometric Surface as of 11-19-01
- Figure 4-13 Potentiometric Surface as of 1-11-01
- Figure 4-14 Potentiometric Surface as of 1-31-01
- Figure 4-15 Potentiometric Surface as of 3-19-01
- Figure 4-16 Potentiometric Surface as of 4-5-01
- Figure 4-17 Potentiometric Surface as of 6-18-01
- Figure 4-18 Potentiometric Surface as of 7-11-01
- Figure 4-19 Potentiometric Surface as of 10-12-01
- Figure 4-20 Potentiometric Surface as of 1-22-02
- Figure 4-21 Potentiometric Surface as of 2-22-02
- Figure 4-22 Potentiometric Surface as of 5-14-02
- Figure 4-23 Potentiometric Surface as of 9-16-02
- Figure 4-24 Potentiometric Surface as of 9-24-02
- Figure 4-25 Potentiometric Surface as of 10-1-02
- Figure 4-26 Potentiometric Surface as of 10-9-02
- Figure 4-27 Potentiometric Surface as of 10-15-02
- Figure 4-28 Extraction Well Flow Rates
- Figure 4-29 Monthly Precipitation
- Figure 4-30 Methane Trends
- Figure 4-31 IRZ-1 COC Response to Reagent Delivery
- Figure 4-32 Ethene Trends
- Figure 4-33 Historical VOC Trends for RAP1-6T Showing Both Laboratory and Field-Generated Data
- Figure 4-34 RAP1-6T COC Response to Reagent Delivery
- Figure 4-35 Recent VOC Trends for RAP1-6T Showing Both Laboratory and Field-Generated Data
- Figure 4-36 Dilution-Corrected IRZ-INJ VOC Trends (Lab Data Only)
- Figure 4-37 VOC Trends for IRZ-1 Showing Both Laboratory and Field-Generated Data
- Figure 4-38 VOC Trends for IRZ-1 (Lab)
- Figure 4-39 Dilution-Corrected IRZ-1 VOC Trends (Lab Data Only)

- Figure 4-40 Hydrogen Sulfide Trends
- Figure 4-41 (a) Conductance Trends vs. Time, Showing All Data and Injection Events
- Figure 4-41 (b) Conductance Trends vs. Time, "IRZ" Labeled Wells Only
- Figure 4-41 (c) Conductance Trends vs. Time, "RAP and B" Labeled Wells Only
- Figure 4-42 Hanscom AFB Ethene Trends RAP1-6T
- Figure 4-43 Recent VOC Trends for MW-RAP1-6T Lab Data Only
- Figure 4-44 Dilution-Corrected RAP1-6T VOC Trends (Lab Data Only)
- Figure 4-45 Qualitative Observation Ratings of Samples Taken from Demonstration Area Wells Over Time
- Figure 4-46 Methane Measurements from Demonstration Area Wells Over Time
- Figure 4-47 Ethene Measurements from Demonstration Area Wells Over Time
- Figure 4-48 VOC Trends for IRZ-2 Showing Both Laboratory and Field-Generated Data
- Figure 4-49 VOC Trends for IRZ-3 Showing Both Laboratory and Field-Generated Data
- Figure 4-50 VOC Trends for IRZ-4 Showing Both Laboratory and Field-Generated Data
- Figure 4-51 Trends for IRZ-5 Showing Both Laboratory and Field-Generated Data
- Figure 4-52 Historical VOC Trends for RAP1-6R Both Laboratory and Field-Generated Data
- Figure 4-53 Recent VOC Trends for RAP1-6S Showing Both Laboratory and Field-Generated Data
- Figure 4-54 TCE: cis-1,2-DCE Ratio (Concentration Basis) over Time for RAP1-6R
- Figure 4-55 Hanscom Bromide Trends Log Scale
- Figure 4-56 MCL Index at IRZ-1
- Figure 4-57 MCL Index at RAP1-6T (11-14-1984 through current)
- Figure 4-58 MCL Index at RAP1-6T (07-24-1998 through current)
- Figure 4-59 MCL Index at IRZ-INJ
- Figure 4-60 TCE vs. Conductance at IRZ-1 and RAP1-6T (June 2000 through October 2002)
- Figure 4-61 Biodegradation Rates at IRZ-INJ
- Figure 4-62 Biodegradation Rates at IRZ-1
- Figure 4-63 Biodegradation Rates at RAP1-6T
- Figure 4-64 Rates for Pre-Demonstration VOCs at RAP1-6T
- Figure 4-65 Literature Rate Constants for TCE
- Figure 4-66 Total Iron Lab
- Figure 4-67 Total Manganese Lab
- Figure 4-68 Total Arsenic in the Lower Aquifer from October 2002 Sampling Event
- Figure 4-69 Dissolved Arsenic in the Lower Aquifer from October 2002 Sampling Event
- Figure 4-70 Hydrogen Data
- Figure 4-71 Concentration of CAHs at MW-B239

- Figure 4-72 Concentration of CAHs at MW-B239, 10-28-1995 through 09-01-2002 with Field Data
- Figure 4-73 Concentration of CAHs at MW-B239, 05-01-1998 through 09-17-2002 with Field Data

#### LIST OF TABLES

Table 1-1 Summary of Physical Site Characteristics Table 2-1 **ERD Site Remediation Project Cost Elements** Table 2-2 Summary of Reagent Cost Ranges for Selected Soluble Carbohydrates Table 3-1 **Performance Objectives** Table 3-2 Suitability of Hanscom AFB Site Screening Characteristics for IRZ Implementation Pre-demonstration Project Concentrations of Volatile Organic Compounds in Table 3-3 Wells RAP1-6R and RAP1-6T Table 3-4 Soil Analysis Summary Table 3-5 Summary of Initial, Background Biogeochemical Data (min/max) Table 3-6 History Log of Demonstration Table 3-7 Parameters Included in Full and Abbreviated Groundwater Monitoring Events Table 3-8 Parameters Included in Soil Monitoring Events Table 4-1 Performance Criteria Table 4-2 **Expected Performance and Performance Confirmation Methods** Table 4-3 Bromide Tracer Data Summary Table 4-4 Biochemical Oxygen Demand (BOD) and Chemical Oxygen Demand (COD) Table 4-5 Data from Initial Groundwater Sampling Round at Hanscom AFB Table 4-6 Data from Midpoint Groundwater Sampling Round at Hanscom AFB, 05-03-01 to 05-08-01 Table 4-7 Data from Midpoint Groundwater Sampling Round at Hanscom AFB, October 2002 Table 4-8 **Reductive Dechlorination Processes** Table 4-9 Summary of Biodegradation Rates over Selected Time Intervals, Hanscom AFB Table 4-10 Total Dissolved Solids (TDS) in Groundwater Table 4-11 **VOCs of Secondary Concern** Table 4-12 Total Metals in Groundwater Table 4-13 Dissolved Metals in Groundwater Table 4-14 Molasses Analyses for Inorganics Table 4-15 Molasses – Water Analyses for Inorganics Table 4-16 Fatty Acids in the October 2002 Full Monitoring Round Table 4-17 **Comparison of Technology Alternatives** Table 5-1 Estimated IRZ Costs for a Hypothetical CAH Plume Table 5-2 Cost Savings for IRZ Technology Compared to Pump and Treat Systems Table 5-3 Economic Comparison of Probable Costs for Proposed ARCADIS CAH Site in South Carolina

- Table 5-4Results of DuPont Technology Evaluation
- Table 5-5
   Summary of IRZ Technology Application Costs

### ACKNOWLEDGEMENTS

We gratefully acknowledge the hosting of this demonstration Tom Best of Hanscom AFB. His ready assistance and thorough knowledge of the history of the Base's environmental remediation history were invaluable. We greatly appreciate the guidance and support of Environmental Security Technology Certification Program (ESTCP) (Cathy Vogel and Andrea Leeson) and Air Force Center for Environmental Excellence (AFCEE) for this work. The guidance of the AFCEE lead Jerry Hansen and of the ESTCP reviewers Hans Stroo and Dr Marvin Unger was vital. Numerous ARCADIS employees contributed to the work including D. Liles, J. Burdick, K. Saravanan, S. Darrigrand, A. Norton, B. Therriault, G. Gordon, T. Regan, D. Stake, B. Comstock, R. Hames and C. Castelluccio.

## PREFACE

ARCADIS is the owner of Contractor Patented Technology for the in-situ addition of carbohydrate substrate material to create reactive zones for the removal of chlorinated hydrocarbons from groundwater as set forth in U.S. Patent Nos. 6,143,177 and 6,322,700.

To discuss application of this technology at government sites please contact:

- Van Sands at ARCADIS in Denver CO 720-344-3792 regarding legal and contractual matters and
- Chris Lutes of ARCADIS in Durham, NC at 919-544-4535 or <u>clutes@arcadis-us.com</u> regarding technical information, or
- Jerry Hansen at AFCEE 210-536-4353 or jerry.hansen@brooks.af.mil.

For commercial application please contact ARCADIS only, at the above listed phone numbers.

## **EXECUTIVE SUMMARY**

The general purpose of this demonstration program was to evaluate the efficacy of the In-Situ Reactive Zone (IRZ)/Enhanced Reductive Dechlorination (ERD) technology to remove chlorinated aliphatic compounds (CAHs) from impacted groundwater in a range of geologic conditions and CAH concentrations. The demonstration also helped determine the rates of mass removal of CAHs in groundwater at the demonstration site at Hanscom Air Force Base.

Ultimately, the objectives of the demonstration were to demonstrate the ability to remediate contaminants in the subsurface over a relatively short time period (from one to five years in typical full-scale applications) and to gather information that can be used to estimate long term treatment effectiveness, life span and costs. The results of the demonstration were used to develop a protocol for the use of ERD technology for CAHs at DoD facilities (Suthersan, 2002). Also important in this demonstration was to show that the degradation of CAHs does not "deadend" at undesirable by-products such as cis-1,2-dichloroethene (cis-1,2-DCE) and/or vinyl chloride (VC). The primary benefits of ERD technology include ease of regulatory acceptance, in-situ nature and relatively low cost. Potential limitations to the application of the IRZ technology using soluble carbohydrates can include the following:

- Intermediate degradation products such as VC can be formed; however, proper system design can ensure their further degradation.
- Production of reduced gases or secondary water quality impacts from byproduct organic compounds, fermentation byproducts or mobilized metals is possible. The effects of these constituents usually do not extend beyond the reactive zone, but they should be monitored and addressed during implementation. At the demonstration site, secondary water quality impacts (including metals mobilization and high COD/BOD) were observed, but as expected were limited to the area of the reactive zone and did not appear to be significant downgradient. Ketones were generated as metabolic byproducts of molasses biodegradation, but did not appear to pose an appreciable risk.
- ERD's effectiveness on large pools of free-phase dense, non-aqueous phase liquid (DNAPL) has not been proven although it does appear to be applicable to sorbed or residual DNAPL.

#### **Scope and Location of Demonstration**

The demonstration area at Hanscom is downgradient of an area that has been used in the past for research and training exercises, including the dumping of drummed pyrokinetic waste for burning. Residual CAHs remain beneath this area of the site in sorbed and dissolved phases and evidence also suggests the presence of emulsified or pooled DNAPL. The semi-confined lower sandy till aquifer (the target zone for the demonstration) rests directly on bedrock at a depth of about 50 feet. Before treatment, trichloroethene (TCE) concentrations in the lower aquifer at the RAP1-6 well cluster averaged between 1,000 and 2,000 µg/L with elevated levels of both cis-1,2-DCE (2,000 to 5,000 µg/L) and VC (500 to 1,300 µg/L). The site was moderately reducing (DO <1.5 mg/l, ORP typically 0 to -50 mV) with pH ranging from 6.0 to 6.5 prior to treatment.

The active treatment phase of the demonstration was from October 2000 to October 2002. A total of forty-seven injections were conducted in a single injection well. Over this time, a total of 1,250 gallons of raw blackstrap molasses, 11,250 gallons of dilution water, 7,575 gallons of push water and 4,732 grams of potassium bromide were injected. Monitoring was conducted during the demonstration to gauge technology effectiveness, describe changes in biogeochemical conditions and gather process monitoring feedback.

#### **Summary of CAH Treatment Results**

Our discussion of CAH treatment focuses primarily on the two monitoring wells that received substantial doses of substrate TOC - RAP 1-6T and IRZ-1 - and secondarily on the injection well. The two monitoring wells were also the only wells screened in the target zone in which substantially increased levels of methane were observed.

The best treatment results were observed at IRZ-1 (approximately 45 feet downgradient from the injection well). At this well, highly effective treatment of TCE was observed five months after injections began and shortly after single injections with water pushes began (>95% reduction vs. pretest concentration). Substantial treatment of cis-1,2-DCE (eventually >85% reduction in pretest concentration) was not observed until a year later, during a second period of high TOC delivery. By that time, complete degradation was evidenced by the substantial increase in ethene production. The rate of ethene production continued to climb through the end of the demonstration in October 2002, indicating that treatment effectiveness continued to increase after two years of system operation. Ethene concentrations at this well increased to more than 20 times the pre-test value.

At RAP1-6T, a sharp decline in TCE was observed within six months of the first injection, coinciding with or slightly preceding the appearance of substrate. TCE levels for seven straight monitoring rounds were at 10% or less of the average of the preceding 10 years. This result exceeded the 80%-in-one-year treatment objective. Then, due to changes in the groundwater flow direction, the IRZ shifted away from RAP1-6T and IRZ-1. As a result of the shift, TCE, cis-1,2-DCE and VC concentrations rebounded at RAP1-6T. These concentrations dipped again when substrate levels increased, although it appears that the typical groundwater flow direction was somewhat more easterly rather than southeasterly as planned. Thus, this well was probably at the fringe of the effective reactive zone for much of the demonstration period.

Although the injection well is by definition a less accurate indicator of the overall effectiveness of the reactive zone than downgradient wells, it is useful to note that substantial evidence of effective treatment of all chlorinated species was seen at the injection well, even in data corrected for the dilution effect of the injected solution. Concentrations of TCE were reduced by more than 95% and VC by 85% over a long period from May 2001 through October 2002. DCE decreased substantially less during this period (at most about 75%). This suggests that although CAHs were being completely degraded, desorption from a localized source area upgradient from the injection well continued.

The wells that did not get substantial, consistent doses of substrate showed no evidence of treatment or at most only modest decreases in TCE (IRZ-2, IRZ-3, IRZ-4 and IRZ-5 in the

targeted lower aquifer, RAP1-6R in the bedrock aquifer and RAP1-6S in the upper aquifer). This strengthens the conclusions that substrate availability is linked with improved biodegradation and that contaminant removal was attributable to enhanced biodegradation rather than displacement.

#### Analysis of CAH Data: Conditions Required for Enhanced Biodegradation

Data for IRZ-1 and RAP1-6T demonstrate that TCE concentrations decrease when substrate concentration increases. However, this correlation between TCE degradation and substrate availability clearly does not hold for DCE degradation. This suggests that substrate is a necessary but not sufficient condition for complete enhanced biodegradation. Data suggests that it took almost a year of regular substrate injections and system adjustments to overcome the oxidative poise of more preferential electron acceptors and reach methanogenic conditions at or upgradient of IRZ-1. This result is also in keeping with our theoretical understanding that enhanced biodegradation, especially the biodegradation of cis-1,2-DCE, is more favored under methanogenic conditions than sulfate-reducing conditions (Suthersan 2002).

Taken together, this data set is consistent with a theoretical understanding in which both consistent substrate dosing and methanogenic conditions are required for optimal treatment (Suthersan, 2002). Therefore, the addition of sufficient carbohydrate substrate to drive redox values into the methanogenic or sulfate-reducing range in bacterial zones distant from the line of injection wells is required to achieve rapid, complete biodegradation.

#### **Comparison of Results with Primary Objectives**

During the two-year demonstration project, highly effective, complete TCE removal was observed in a source area that had a long history of fairly stable TCE concentrations before treatment. Evidence of complete treatment – a buildup of ethene, reduction in cis-1,2-DCE and no accumulation of vinyl chloride - was seen in the most effectively treated downgradient wells. TCE reductions in these wells (RAP1-6T, IRZ-1) exceeded the 80%-in-one-year treatment objective. Effective treatment was seen only where substantial substrate (molasses and its breakdown products) was observed in downgradient monitoring wells. The layout of the injection and monitoring well system was designed for a consistent southeasterly groundwater flow; however, during the demonstration period, the direction of flow varied, often to the east. Thus, it is suspected that a larger IRZ was formed than what was observed, but that the monitoring well network was not positioned to completely detect it.

The data from RAP1-6T, IRZ-1 and the injection well do not show evidence of "accumulation" or "dead-ending" at DCE or vinyl chloride. On the contrary, the data discussed above and especially the increases in ethene show that complete degradation was achieved in these wells. Therefore, the primary performance objectives were met for the wells within the reactive zone.

#### **Cost Analysis**

Cost comparisons provided in the report are based on a plume-wide application to a dissolved plume with residual, sorbed material in a source area. Applied under appropriate conditions, ERD provides significant cost savings over conventional pump and treat technology, and compares favorably with other more innovative technologies in a comparison including ex-situ substrate-enhanced bioremediation with recirculation, a zero valent iron barrier and natural

attenuation. Estimated unit costs will be provided at the conclusion of the project in the cost and performance report when complete financial information is available.

## 1. Introduction

#### 1.1 Background

Chlorinated solvent contamination of groundwater is a widespread problem at many military and civilian facilities. This class of compounds includes widely used industrial chlorinated aliphatic hydrocarbons (CAHs) such as carbon tetrachloride (CT), methylene chloride (MC), trichloroethane (TCA), trichloroethene (TCE) and tetrachloroethene (PCE). In addition to their roles in many industrial processes, CAHs have historically been used for cleaning and degreasing such diverse products as aircraft engines, automobile parts, electronic components and clothing. Contamination of groundwater by mobile metals (e.g., hexavalent chromium, lead, arsenic, nickel, mercury and cadmium) is also widespread at military facilities due to the use of these metals in ordnance, armament, armor, and as components of corrosion prevention coatings on vehicles. Because of the integral nature of CAHs and metals in efficient military operations, it is not surprising that the Unites States Armed Forces (USAF) are often faced with widespread, costly remediation problems related to these compounds.

The conventional remedy for CAH contamination in groundwater is groundwater extraction and ex-situ treatment, typically with air stripping or carbon adsorption, also known as pump and treat or in-situ air sparging. An alternative to these conventional technologies that has already been used at over 130 commercial sites is In-situ Reactive Zone (IRZ) technology for the remediation of CAHs and metals. The IRZ demonstration involves the addition of a food grade carbohydrate substrate, which serves as a supplemental energy source for microbiological processes in the subsurface. This substrate is typically molasses although these substrates can include high fructose corn syrup, whey, etc. (Suthersan, 2002). Through this subsurface molasses injection, the existing aerobic or mildly anoxic aquifers can be altered to highly anaerobic reactive zones. This creates suitable conditions for the biodegradation of CAHs and/or the precipitation of selected metals in insoluble forms. Thus this technology can be more specifically referred to as Enhanced Reductive Dechlorination (ERD) or Enhanced Anaerobic Reductive Precipitation (EARP). Other available innovative remedies for CAH contamination in groundwater include chemical oxidation, phytoremediation and vacuum enhanced recovery.

The primary benefits of this technology include ease of regulatory acceptance, in-situ nature and relatively low cost. The benefits of ERD technology include its record of successful application under the following conditions:

- At Various Constituent Concentrations Areas containing dissolved CAH concentrations in excess of 160 milligrams per liter (mg/L) have been successfully treated. Much more dilute plumes with concentrations of target constituents in the 10 – 100 micrograms per liter (µg/L) range have also been treated successfully.
- *In Varied Geologies* The ERD technology has been applied at sites with widely differing geologic and hydrogeologic settings, from low permeability silts and clays, to high permeability alluvial deposits, to bedrock settings and with groundwater velocities ranging

from a few feet per year to several feet per day. However, as discussed in Section 2, there are permeability and velocity limits beyond which the technology cannot be implied.

 Under Multiple Regulatory Programs – The ERD technology has been applied under multiple regulatory programs, including Comprehensive Environmental Response Compensation and Liability Act (CERCLA) and Resource Conservation and Recovery Act (RCRA), and under the jurisdiction of politically sensitive regulatory agencies such as the California Regional Water Quality Control Board. It has also been applied in several countries outside the U.S.

#### **1.2** Objectives of the Demonstration

#### 1.2.1 Objectives

The general purpose of this demonstration program was to evaluate the efficacy of the ERD technology to remove CAHs from the impacted groundwater in a range of geologic conditions and CAH concentrations. Ultimately, the objectives of the demonstration were to demonstrate the ability to remediate contaminants in the subsurface over a relatively short time period (from one to five years in typical full-scale applications) and also to gather information that can be used to estimate long term treatment effectiveness, life span and costs. The primary goal of this technology demonstration is to use the results to develop a protocol for use of ERD technology for CAHs at Department of Defense (DoD) facilities (this document, soon to be finalized, is formally titled "Technical Protocol for Using Soluble Carbohydrates to Enhance Reductive Dechlorination of Chlorinated Aliphatic Hydrocarbons" [Suthersan, 2000]). Performance objectives are discussed in more detail in Section 3.1. The demonstration helped determine the rates of mass removal of CAHs present in the groundwater at the demonstration sites. Also important in the demonstration was to show that the degradation of CAHs does not "deadend" at undesirable by-products such as *cis*-1,2-dichloroethene (*cis*-1,2-DCE) and/or vinyl chloride (VC).

To meet these objectives, a pilot test of the technology was conducted at Hanscom Air Force Base (AFB). Hanscom AFB is located in Middlesex County, Massachusetts, approximately 14 miles northwest of downtown Boston. The area selected for the demonstration at Hanscom is downgradient from Site 1, a former fire training area. Site 1 is located on Hartwell's Hill, northwest of the overrun for Runway 23 and southeast of Hartwell Road, as shown in Figure 1-1. A CAH plume in this area has exhibited TCE concentrations between 1,000-2,000  $\mu$ g/L, *cis*-1,2-DCE concentrations of 2,000-5,000  $\mu$ g/L, and VC concentrations of 500-1,300  $\mu$ g/L. Other physical characteristics of the subsurface at this site are listed in Table 1-1. The pilot test was conducted using molasses as a carbohydrate source, over a period of two years. Extensive monitoring data were collected before and during the test to address performance criteria. Three full monitoring events and five abbreviated monitoring events were conducted.

#### **1.3 Regulatory Drivers**

Groundwater impacts by CAHs at DoD sites are regulated under the RCRA and CERCLA programs. Hanscom AFB is a CERCLA (superfund) site, regulated by the EPA.

The overall remedial goals for this particular site are to achieve cleanup goals consistent with current and foreseeable future uses. For groundwater, the long-term cleanup goal is to achieve

drinking water standards, since this site lies within an area classified as GW-1 (a potential drinking water supply) by the Massachusetts Department of Environmental Protection (MADEP). For the CAHs detected at this Site, the MADEP's GW-1 standards are equivalent to the United States Environmental Protection Agency (U.S. EPA) Maximum Contaminant Levels (MCLs).

Under an Interim Record of Decision, a groundwater pump and treat system is operating in the demonstration area. Operation of the pump and treat system since 1991 has succeeded in substantially reducing TCE concentrations, but TCE is still well above the MCL and may have reached an asymptote. For this demonstration project, the objective was to determine if the natural attenuation processes could be enhanced to accelerate the progress towards the site-wide remedial goal.

The demonstration was successful in achieving the MCL for TCE. MCLs for TCE daughter products were not attained due to variable groundwater flow directions and thus inconsistent dispersal of reagent. However, generation of ethene was observed, indicating that TCE was being completely degraded without "dead-ending" at intermediate compounds, suggesting that remediation of TCE daughter products to MCLs could also be achieved using ERD technology.

#### 1.4 Stakeholder/End-User Issues

Stakeholders and end-users of ERD technology are concerned foremost with the issue of CAH cleanup. Under appropriate conditions, ERD offers significant advantages over conventional pump and treat technology, including lower cost and reduced treatment time. The advantages and limitations of the technology are discussed in detail in Section 2.4. The production of intermediate products is a potential concern to stakeholders and regulatory agencies. The ERD process converts more highly chlorinated CAHs to less chlorinated and eventually non-chlorinated end products. The cascading reactions can result in the production of VC. This product is more carcinogenic than the parent compound. Reductive dechlorination of VC should also occur with the ERD process, and it is also quickly biodegraded by aerobic microorganisms. For these reasons, the production of VC or other intermediate products is considered a temporary situation and does not represent a major impediment to the technology but should be monitored during application of the technology.

Another stakeholder/regulatory issue can be the production of gases such as methane, hydrogen sulfide, and carbon dioxide, and the migration and potential accumulation of these gases in the vadose zone. Concentrations of these gases can accumulate in the subsurface, when structures in the vicinity do not allow for passive diffusion of these gases. For this reason, vapor-phase concentrations of these compounds are monitored when a potential concern exists to ensure that safe conditions are maintained. If required, venting of subsurface gases or a modified donor injection routine will be used to protect against exposure or accumulation. This issue is not considered to be a major impediment to technology implementation, but must be considered.

Secondary water quality impacts from ERD can occur due to the by-products of substrate consumption as measured by parameters such as chemical oxygen demand (COD), biochemical oxygen demand (BOD), total dissolved solids (TDS), taste, odor, and sulfides. However these impacts are typically limited to the reactive zone itself. These byproducts, which are typical of

many natural processes in which bacteria consume a food source, are generally rapidly consumed when the conditions become more aerobic on the edges of the reactive zone. Secondary water quality impacts can also occur from mobilization of metals naturally occurring in the solid phase into the groundwater. Although enhanced anaerobic in-situ bioremediation processes will, in general, reduce the mobility of many metals (indeed it has been successfully used for the treatment of many), it will solubilize some other naturally occurring metals in the reactive zone (e.g., iron, manganese, and arsenic). However, even in solubilized form under anaerobic conditions, metals such as arsenic are substantially retarded by adsorption to the aquifer matrix. Furthermore, it is generally believed that they will be reprecipitated/immobilized downgradient of the reactive zone when the conditions return to their preexisting state (which, for the purposes of this discussion, is assumed to be aerobic). Similarly, reprecipitation/immobilization will occur within the IRZ area some time after system shutdown. These reducing conditions are by no means unique to IRZ systems – they occur, for example, at sites of total petroleum hydrocarbon (TPH) releases and landfills as well.

Molasses itself has been mentioned as a potential source of metals. Available analyses of metals in typical molasses-water mixtures used in IRZ applications have shown concentrations below regulatory standards. Injected metals did not produce secondary water quality issues in this demonstration (see Section 4.3.5). However, this is a potential issue that should be considered in the design phase for IRZ projects. The paucity of available data suggests that further work should be done to explore the metallic content of different sources of molasses.

Thus the potential for secondary water quality impacts needs to be fully identified and addressed during design and in consultation with all applicable regulatory agencies and the public.

As discussed in Section 4.3.5, secondary water quality impacts (including metals mobilization and high COD/BOD) were observed but as expected were limited to the area of the reactive zone and did not appear to be significant downgradient. Although ketones were generated as metabolic byproducts of molasses biodegradation they did not appear to pose an appreciable risk. Gas production was not an issue at this site since the demonstration was conducted in an open area, far from buildings or other structures where gases could accumulate.

## 2. Technology Description

#### 2.1 Technology Development and Application

#### 2.1.1 General Technology Description and History

ERD technology is intended to facilitate and expedite the biological reductive dechlorination of CAHs through the well-documented mechanisms pictured in Figure 2-1. The ERD technology stimulates indigenous microbiological organisms through the engineered addition of electron donors, which contain degradable organic carbon sources.

The general mechanism behind the application of ERD technology relies on enhancing or inducing the bioremediation of CAHs through periodic subsurface injection of a soluble electron donor solution (typically comprised of a carbohydrate such as molasses, whey, high fructose corn syrup, lactate, butyrate, or benzoate). Through periodic subsurface substrate injection, the ERD technology alters existing aerobic or mildly anoxic aquifers to anaerobic, microbiologically diverse, reactive treatment zones. Within such zones, conditions are conducive for the bioremediation of CAHs.

ERD technology facilitates and expedites the degradation of CAHs through biological reductive dechlorination. Chlorinated compound reduction can be a biologically mediated reaction that entails transferring electrons to the substrate of interest from various initial electron donors. The more oxidized the chlorinated compound is, the more susceptible it is to reduction.

Reductive dechlorination occurs when aquifer bacteria utilize chlorinated solvent molecules as electron acceptors in the oxidation of their carbonaceous food source (electron donors). The reduction of chlorinated solvent molecules that are used as electron acceptors cleaves one or two of their chlorine atoms, leading to the sequential dechlorination pattern observed in many contaminated aquifers. Several bacterially mediated anaerobic processes that lead to reductive dechlorination are discussed in detail in Section 1.3 of the protocol document (Suthersan, 2002). By injecting a degradable carbon source into the aquifer, the rate and extent of bacterial reductive dechlorination can be enhanced to levels that provide a cost-effective remedial method. These reductive dechlorination processes include dehalorespiration (in which reductive dechlorination is used for growth with CAHs serving as the electron acceptor) and cometabolic anaerobic biodegradation (in which the degradation does not yield a metabolic benefit to the bacteria). These cometabolic processes typically occur under either sulfate reducing or methanogenic conditions

In practice, ERD can be operated as an in-situ bioreactor that forms downgradient from a line of degradable substrate injection wells placed in a line perpendicular to groundwater flow. If sufficient carbon substrate is injected, oxygen and nitrate metabolism dominates near the injection line, while sulfate reduction, methanogenesis and reductive dechlorination zones form farther downgradient. The technology operates most effectively when groundwater is passing

through the sulfate-reducing zone, still bearing a degradable carbon load that will support methanogenesis and reductive dechlorination.

A conceptual design of this process has been provided as Figure 2-2. This technology can be implemented in a variety of ways, including fixed, automated systems and mobile, manually controlled systems (See also Sections 4 and 6 of the protocol document [Suthersan, 2002]). The particular system used in this demonstration was truck-mounted (see Figures 2-3 and 2-4 for a schematic and photograph of this system).

ERD technology developed from ex-situ biological reactor and precipitation technology, which has been routinely used for decades to treat a broad range of inorganic and organic compounds. However, some of these ex-situ processes involve addition of reagents, such as sulfide salts, which would be controversial to use in-situ. Efforts over the last 15-20 years have demonstrated that similar treatment approaches can be engineered in-situ. CAH biotransformation under anaerobic conditions has been studied for two decades at various scales (Vogel and McCarty, 1985; Parsons and Lage, 1985; Bouwer, 1993; and references cited therein). Researchers and remediation practitioners at ARCADIS recognized that biochemically-induced changes could be achieved without the need to inject potentially controversial reagents, and that naturally occurring mechanisms of attenuation could be enhanced.

In early 1994, when a commercial client requested an innovative remedial solution for chromateimpacted groundwater at a CERCLA site in Pennsylvania, ARCADIS chose molasses as a reagent to enhance these processes. In this case we avoided the technical, regulatory, safety and economic concerns associated with sulfide injection by using molasses to achieve reducing conditions. The Pennsylvania project clearly demonstrated that molasses IRZs could effectively produce controlled conditions required to remediate heavy metals. Subsequent projects have demonstrated the effectiveness of IRZs for remediation of CAHs and other organic and inorganic contaminants. Our experience has shown that molasses is not the only carbohydrate material that can be used for this purpose; other carbohydrates such as high fructose corn syrup and whey can also be effective. This approach has been accepted by regulators and has since been demonstrated in a wide variety of geological conditions with both high and low groundwater velocities. Enhancing CAH degradation using ERD has become an accepted practice in the last several years, but additional work remains to improve the design and optimize performance of ERD systems under varying conditions.

In addition to CAHs, ERD processes have a potential application to a wide spectrum of contaminants and co-contaminants such as:

- Chlorinated cyclic hydrocarbons, e.g., pentachlorophenol (PCP)
- Chlorinated pesticides, e.g., chlorinated propanes, lindane
- Metal precipitation, e.g., Cr<sup>+6</sup> to Cr<sup>+3</sup>; metal sulfide complexes of nickel and copper; metalhumic complexes of beryllium and other metals
- Other halogenated organic contaminants

### 2.1.2 Design Criteria

The key parameters that go into an IRZ/ERD system design include:

- Formation geochemistry (including the concentrations of electron acceptors such as dissolved oxygen (DO), nitrate, sulfate etc, pH and buffering capacity)
- Site-specific hydrogeology (including depth to water, saturated thickness, and hydraulic conductivity)
- Contaminant mass and form (dissolved, sorbed and free phase).

These parameters are discussed thoroughly in Sections 2 and 4.1 of the protocol document (Suthersan, 2002).

Ultimate design goals include contaminant removal rates and closure requirements (see Sections 1.2 and 3.1 of this report and Section 6.5 of the protocol document [Suthersan, 2002]). Interim design goals are set to ensure the creation of appropriate conditions for CAH biodegradation and may typically include these ranges for various field parameters (in this context, "monitoring wells" refers to those wells 1 to 3 months downgradient of the injection wells):

- pH > 4.0 s.u. in the injection wells; > 5.0 s.u. in the monitoring wells
- DO < 1.0 mg/L in both monitoring and injection wells
- Oxidation Reduction Potential (ORP) -> -400 mV and < -250 mV in the injection wells; < -100 mV in the monitoring wells. Note however that these ORP values should not be taken as absolutes since ORP is pH dependent. For sites where reducing environments are identified in the groundwater prior to initiation of reagent injections, a target goal of lowering the ORP by 200 mV in the injection wells and 100 mV in the monitoring wells should be employed.</li>
- Total organic carbon (TOC) >500 mg/L and < 9,000 mg/L in the injection wells and > 50 mg/L in the monitoring wells
- Specific conductance order of magnitude increase in the injection wells; 20 to 50% increase in monitoring wells

To achieve those goals parameters that must be specified during system design include:

- Substrate to be used and initial dose rate
- Intended radius of influence/injection well spacing
- Injection and monitoring well layout (which may be a barrier, source zone or plume treatment system)
- Injection system type (manual vs. automated, conventional well vs. direct push etc)
- Systems to handle byproducts (which may include the injection of buffers or the use of ventilation systems under structures).

These design considerations are discussed at length in Sections 4-6 of the protocol document (Suthersan, 2002). Pilot testing is usually required and adjustment or "tuning" of the system during operation is critical. These topics are discussed in Sections 5 and 6 of the protocol document.

#### 2.2 Previous Testing of the Technology

As of September 2002, ARCADIS has been involved with more than 130 IRZ sites, across five countries and 26 U.S. states. Thirty of these sites are full-scale implementations, five of which have achieved closure. The other sites are ongoing pilot applications, or Interim Remedial Measures, or they are completed pilot projects that are now in the full-scale design phase. The technology has successfully been applied to the following chlorinated compounds and metals:

- TCE, dichloroethene (DCE), VC, CT, chloroform (CF), chlorinated propanes, PCP, pesticides, trichlorofluoromethane, and perchlorate;
- Hexavalent chromium, nickel, lead, cadmium, mercury, and uranium.

Appendix A to the IRZ Protocol (Suthersan, 2002) includes a comprehensive table with information on ARCADIS IRZ sites for CAHs, as well as 15 case studies. Appendix A-1 of this document is an extensive bibliography of papers and book chapters published on this technology.

#### 2.3 Factors Affecting Cost and Performance

Although application of IRZ using soluble carbohydrates can occur in a variety of hydrogeologic settings, there are certain conditions that are better suited for cost effective use of the technology. Existing conditions that are anaerobic or borderline aerobic/anaerobic but with insufficient TOC can be most rapidly treated. Conditions that are anaerobic and already have sufficient degradable TOC may not be aided substantially by addition of soluble carbohydrates. One of the most important criteria is hydraulic conductivity. Generally, hydraulic conductivity of the aquifer needs to be greater than 1 ft/day and when coupled with hydraulic gradients, groundwater velocities on the order of 30 ft/year, or greater, are desirable. Another important criterion is the pH, which needs to be initially in the range from about 5 to 9 in order to have an active microbial population. Site screening criteria and methods are discussed more fully in Section 2 of the protocol document (Suthersan, 2002).

The implementation of an IRZ project is a dynamic process which requires a detailed understanding of the site geochemistry and hydrogeologic conditions before implementation and as it changes as a result of pilot or full-scale implementation. This technology is most likely to be successful when there is considerable process monitoring during the initial deployment of the pilot test that allows for adjustment of reagent deliverability (strength and frequency). Where ERD has failed, or has required longer than expected treatment periods, it is usually the result of improper monitoring (the wrong parameters or the wrong frequency) or data evaluation in the early stages of the pilot test. TOC loading and induced gradients must be reviewed early in the pilot process to allow delivery rates to be increased (for greater spreading and greater TOC levels within the treatment area) or reduced (or a buffering agent added), if pH levels drop too quickly.

Similarly, the effects of reagent injections must be reviewed in the context of how the addition of aqueous solutions affect hydraulic gradients (i.e., mounding) and flow directions. Groundwater flow directions and gradients should be viewed both in a macro and micro scale before and during the demonstrations.

Once a preliminary determination has been made that IRZ is an appropriate technology option to consider for the site, a more detailed data set needs to be gathered. Information required to fully review a site for IRZ includes:

- Site specific geology and hydrogeology, including: fraction of organic carbon (F<sub>oc</sub>) in the aquifer matrix, boring logs, predominant aquifer lithology, aquifer hydraulic characteristics, horizontal and vertical hydraulic gradient, groundwater velocity, and depth to water
- CAH concentrations and distribution, both current and historical, if available
- If available, data on general groundwater quality such as TDS, specific conductance, pH, dissolved organic carbon (DOC), carbonate and bicarbonate alkalinity, and general cation/anion scan [calcium, magnesium, sodium, potassium, chloride, sulfate, and nitrate]
- Any previously gathered biogeochemical data, including oxidation reduction potential (ORP), nitrate, sulfate/hydrogen sulfide, ferric/ferrous iron, dissolved oxygen, trace gases (including methane, hydrogen, carbon dioxide, ethane, and ethene)
- The site's situation (regulatory drivers, stage in the investigation/remediation process, cleanup goals and time frames, future plans for the site)
- Some brief historical information on the site (source of CAH, estimated date of release, and duration of release events)
- Maps showing the relationship of active operational areas (buildings, etc.) and impervious surfaces (roads and parking lots) to the contaminant plume(s).

Table 2-1 provides a comprehensive overview of cost elements associated with an ERD project.

In a general sense, with an IRZ system, the cost of the reagent material itself is relatively insignificant. When using reagents such as carbohydrates, the cost per pound of TOC delivered is as outlined on Table 2-2. The selection of a carbon substrate(s) will be primarily driven by overall reaction rates, which are, in turn, controlled by the site conditions. A goal should be to minimize overall project cost by minimizing the number of required injection points, the number of injection events, and reagent cost (Harkness, 2000). The physical characteristics of the substrate (i.e., phase and solubility) may also make certain substrates more suitable than others in particular applications.

The majority of the costs related to reagent injection include the labor associated with preparing the reagent mixture and injecting the material into the wells/points along with related costs (mobilization to the site, record keeping, preparation, etc.) Temporary equipment required for the injections includes a solution mixing/holding tank, a portable mixer, a transfer pump, and injection piping/hose. This equipment should be sized and consistent with use at the pilot test site and can be mobilized to each site in a conventional pick-up truck or by trailer. The mixer can be simple as a paddle, or agitation of the tank through truck movement. A nontoxic, non-reactive tracer, or pH buffers may also be included in the reagent solutions. Permanent equipment at the various injections wells includes a removable well seal for the injection wellhead, removable perforated diffuser tubing (to assure even reagent distribution along the screened interval of the well), and quick-disconnect fittings to allow easy attachment of the injection piping/hose to the diffuser tubes for the injection itself.

Budgetary limitations can often directly or indirectly affect design decisions such as source reduction versus plume-wide treatment. Appendix A of the IRZ Protocol (Suthersan, 2002) contains specific information regarding the technology application cost (capital and operation and maintenance) at a variety of sites in which ERD has been successfully applied. Based on our experience and analysis the two largest cost factors for ERD implementation are the injection well installation and the O&M associated with reagent injections. Three other factors that need to be given special consideration during design in order to develop the most cost effective approach for site remediation are:

- Plume Size to be Treated This is the primary factor driving the cost of the technology as the larger the plume area to be treated the more wells are needed (drilling costs) and the more time it takes for reagent delivery.
- Depth of Target Zone Drilling costs are the primary factor affecting overall technology cost. Therefore, deep contaminant settings and/or those requiring specialized drilling techniques (bedrock drilling, multiple conductor casings, etc.) can significantly increase costs. The depth to groundwater will define well design and contribute significantly to the capital cost of a full-scale system. The saturated thickness can also have an influence on cost, since there are practical limits on the maximum screened interval that can effectively be used in an injection well. Based on our experience, a 25-foot screened interval represents a practical limit for an injection point. Of course, this limit will be impacted by the heterogeneity of the subsurface lithology, hydraulic conductivity, and the resulting effects on groundwater flow characteristics. For example, if the lithology and resultant groundwater flow characteristics are such that there are variations in the flow characteristics within the target saturated interval, the use of multiple screened zones or multiple well points should be considered even if the interval is less than 25 feet.
- Groundwater Flux through Zone of Treatment Reagent injections also play a large role in overall technology costs. At sites in which there is a high groundwater flux, more substrate will be required, thereby increasing costs. In faster groundwater flow systems, the limited transverse dispersion in groundwater can limit the extent of the reactive zone created by an individual injection point. This is of particular importance in settings where drilling costs may be high, i.e., deep settings or complex geology. In such cases, an in-situ recirculation well can yield considerable cost savings over use of direct injection wells. This in-situ recirculation well concept aims primarily at delivering reagents in a cost effective manner while remediating larger, deeper contaminant plumes at sites with relatively high groundwater velocities.

#### 2.4 Advantages and Limitations of the Technology

As late as 1998, conventional remediation options for sites contaminated with CAHs were considered to be air stripping, granular activated carbon adsorption, and ultraviolet oxidation (Nyer 1998). Most of these technologies are the 'treat' portions of conventional pump-and-treat systems where impacted groundwater is removed to the surface for treatment and discharge. Pump-and-treat is known as a conventional technology with limitations due to long term operations and maintenance costs, which can be prohibitively expensive. These limitations stem from the fact that many contaminants partition preferentially to aquifer solids rather than the water carrier fluid. This results in moving vast quantities of groundwater while removing

increasingly smaller portions of contaminant mass with time. Established remediation methods for metals removal also employ groundwater extraction followed by ion exchange or chemical precipitation. Like pump-and-treat for CAHs, these remedial techniques are costly and require long periods of time to complete.

The primary advantages for IRZ using soluble carbohydrates can be summarized as follows:

- The *in-situ* process eliminates the need for transferring contaminant mass to other media (such as groundwater pumping and subsequent treatment with air stripping)
- IRZ processes have a potential application to a wide spectrum of contaminants and cocontaminants
- No ex-situ waste is generated
- The process usually uses electron donor sources that are typically easily accepted by regulators and the public
- The biologically mediated reactions involved can generally be driven by indigenous microflora
- The technology is flexible in application, yielding a spectrum of contaminant mass treatment options from passive/containment barrier applications to aggressive source area applications
- Promotes reduction of residual contaminant mass through desorption and disruption of the contaminant phase equilibrium
- Enhances natural attenuation processes
- Applicable to various geological settings and aquifer conditions
- Electron donor source is highly soluble and can move through both diffusive and advective processes into difficult lithologies such as fractured bedrock
- Systems can be designed with flexible operation approaches ranging from automated systems to manual bulk application
- Can be used in tandem with existing remediation systems to optimize performance
- Can be designed with minimal site and facility operation disturbance

All in-situ remediation technologies have an inherent limitation associated with subsurface conditions. The geology in which the technology is being applied will exert considerable control over remediation efficacy. Mass transfer and distribution rates in porous media are the primary factors influencing the efficiency of the IRZ technology using soluble carbohydrates. This can be compensated for to a great extent by a complete understanding of the geochemical and hydrological conditions of the aquifer system to be treated. A good conceptual model of the aquifer will produce a more effective IRZ design. Potential limitations to the application of the IRZ technology using soluble carbohydrates can be summarized as follows:

- Excessive depth of contamination tends to raise costs
- Low permeability aquifers require more injection points

- High permeability aquifers with high groundwater flows require an excessive amount of reactant to establish a reducing environment due to dilution and oxygen recharge
- Heterogeneous lithology, which incorporates preferential flow paths, can limit the distribution of the injected substrate
- Limited porosity of contaminated media such as fractured bedrock minimizes the propagation of treated area
- Biological fouling of injection wells or aquifer resulting from reagent injection is theoretically possible but is rarely observed in practice
- Systems with large amounts or influxes of electron acceptors such as oxygen, nitrate or soluble iron can require large doses of substrate, however, substrate cost is typically a small fraction of the total project cost
- Potential production of excessive quantities of reduced gases such as methane can be problematic in the vicinity of confined structures. Also, production of byproduct organic compounds containing reduced sulfur or nitrogen, including hydrogen sulfide is possible.
- Molasses in its pure form contains concentrations of several metals. In a dilute mixture, as is typically used in IRZ applications, the concentrations have been below regulatory standards. Injected metals did not produce secondary water quality issues in this demonstration (see Section 4.3.5). However, this is a potential issue that should be considered in the design phase.
- Longer lag times prior to effective treatment are noted in low concentration plumes
- Intermediate products such as VC can be formed, however proper system design can ensure their further degradation
- Highly brackish aquifers can pose problematic microbial ecology
- Effectiveness on large pools of free-phase dense, non-aqueous phase liquid (DNAPL) has not been proven although it does appear to be applicable to sorbed or residual DNAPL
- If not carefully controlled, fermentation effects of excessive molasses loading can create conditions conducive to formation of aldehydes, ketones and mercaptans, which, however, can then be further degraded biologically. Excessive fermentation can also decrease pH and potentially mobilize naturally occurring metals.

These potential limitations are general guidelines to be considered when evaluating potential sites for ERD treatment. Site-specific constraints should be considered for all remediation technology options.

Other innovative alternatives for the treatment of chlorinated aliphatic hydrocarbons in the saturated zone include chemical oxidation with permanganate or Fenton's reagent as well as various forms of reductive iron barriers.

## 3. Site/Facility Description

#### 3.1 Performance Objectives

This demonstration is the first of a series of Environmental Security Technology Certification Program (ESTCP)/Air Force Center for Environmental Excellence (AFCEE) funded demonstration projects that aim to evaluate the efficacy of the IRZ/ERD technology to remove CAHs from the impacted groundwater in a range of geologic conditions and CAH concentrations. A second demonstration project at Vandenberg AFB has been funded and will be reported in a separate document.

Primary and secondary performance objectives, as established and discussed in the demonstration plan, are presented in Table 3-1.

#### 3.2 Selecting Test Site

The Hanscom AFB site chosen for this demonstration was selected and proposed based on ARCADIS' review of obtainable site characterization data. Most existing DoD-wide databases of sites are limited in the depth of information available, i.e., (1) they treat bases as a whole but do not provide information specific to each site or operable unit or (2) types of contaminants are listed but not concentrations or closure standards. Thus, candidate sites were obtained in a non-systematic, networking-based approach. Information on candidate sites was solicited from ARCADIS, AFCEE, ESTCP and the Army Environmental Center staff. The qualifying criteria used during this initial site review included the following issues:

- Depth (size) of the contaminated aquifer requiring treatment generally, this is of little technical significance, however, there are cost implications as depth increases
- CAH concentrations preferably exceeding 10 times the treatment standard or 3 times the treatment standard AND 10 times the detection limit to allow easy detection of the effect of the treatment
- Site must exhibit at least moderate hydraulic conductivity (K>10<sup>-4</sup> cm/sec or 0.3 ft/day)
- Site should have completed an initial investigation, or be in the remedy selection process or have an operating pump-and-treat system in place
- Site should have no DNAPL present or DNAPL remedy selected/successfully implemented with ERD implementation as a polishing remedy. This was suggested just as a requirement for the initial demonstration site. The presence of DNAPL would represent a continuing source of dissolved CAHs that would complicate efforts to monitor the progress of the IRZ technology in a short-term demonstration. As discussed later, although the demonstration area was initially believed to be downgradient from the source, it was later determined to be a source zone, although no clear evidence of free phase DNAPL has been discovered.
- Available sulfate mass must correspond to the microbiology that is appropriate for the type of ERD desired. Aquifers that are high in sulfate may not be conducive to developing microbiology that is appropriate for CAH remediation

During the review, site hydrogeology and other data were examined. Existing data on CAH contaminant and intermediate breakdown product concentrations and the site's current regulatory status were considered during the initial site screening. Candidate sites were chosen from a grouping of approximately 25 DoD sites with CAH impacts. These 25 sites do not constitute an exhaustive list of DoD sites that could benefit from the implementation of the ERD technology. Sites were selected based on a combination of factors. One important factor was adherence to technical constraints of ERD technology discussed above. ARCADIS also considered the economic issues that impacted our ability to provide a cost effective demonstration program at a number of DoD sites. Thus, factors such as depth to the water table and geographical location (proximity to one of ARCADIS' offices) were important in site selection. Geographical factors and depth to the water table are not typically involved in choosing to implement the ERD technology from a technical prospective. Lastly, the sites were judged as to whether they were good "field laboratories" in which ARCADIS could implement the ERD technology and interpret the results in a manner consistent with the goals of an ESTCP/AFCEE demonstration project. Sites with extremely low groundwater velocities were eliminated as incompatible with a short-term field program although the technology can be applied at sites with low velocities as long as the aquifer hydraulic conductivity is reasonable.

Site selection criteria for ERD systems have been broadly described in Section 2 of the protocol document (Suthersan, 2002). Although application of ERD using soluble carbohydrates can occur in a variety of hydrogeologic settings, there are certain conditions that are better suited for cost effective use of the technology. One of the most important criteria is hydraulic conductivity. Generally, hydraulic conductivity of the aquifer needs to be greater than 1 ft/day and when coupled with hydraulic gradients, groundwater velocities on the order of 30 ft/year, or greater, are desirable. Another important criterion is the pH, which needs to be initially in the range of about 5 to 9 in order to have a microbial population suitable for microbial activity. Other general site selection criteria include:

- No large quantities of pooled DNAPL, or DNAPL remedy selected/implemented but a
  polishing step needed. (Note that the application of this technology for moderate amounts of,
  for example, emulsified or sorbed free product is an active area of technology development.
  This is possible, but not as rapid as applications for dissolved/sorbed CAH contamination).
  Elevated concentrations of solvents may act as toxic inhibitors to biodegradation as well,
  especially for sites where the release is relatively recent (i.e., within 1 to 3 years) and the insitu biological community has had little time to adapt and diversify.
- Sites that show some evidence of slow biodegradation, including those "stalled" at DCE and VC are desirable.
- The depth of the plume is also a factor in determining the cost effectiveness of an in-situ approach. The capital expense related to installing multiple injection wells in deep settings (greater than 50 feet below ground surface [bgs]), or in installing recirculation wells across thick homogenous settings needs to be compared to the costs associated with competing technologies.

Existing redox conditions that are anaerobic or borderline aerobic/anaerobic but with insufficient TOC can be most rapidly treated. Conditions that are anaerobic and already have sufficient degradable TOC may not be aided substantially by addition of soluble carbohydrates.

Table 3-2 summarizes evaluation criteria for implementing IRZ technology at Hanscom per established site screening parameters.

Based on the available geologic/hydrogeologic and groundwater chemistry data, the ability to successfully implement ERD technology at Hanscom Field appeared favorable. Calculated groundwater velocities and aquifer hydraulic conductivities (the ability to deliver reagents) at the site both appeared to be within acceptable ranges.

The long-term strategy for remediation of the CAH impacts at Hanscom Field includes natural attenuation. ERDs could be used in tandem with the current pump and treat approach, limiting the life span and high costs associated with pump and treat, enhancing natural attenuation of CAHs, and ultimately shortening the time needed to clean up the site. ARCADIS had already established a presence at Hanscom Field with the installation and startup of the vacuum enhanced recovery system for the bedrock aquifer near Site 1. Base representatives had expressed interest in implementing IRZ technology in this same area. Furthermore, the Base has an excellent relationship with the regulators at the Massachusetts Department of Environmental Protection, who had earlier expressed that they were "fully supportive" of a pilot demonstration of ERD technology at Hanscom field.

Field tasks for the pilot demonstration at Hanscom Field were handled by the ARCADIS office in Lowell, Massachusetts. The Lowell office is within 30 minutes from the site. Furthermore, the Lowell office had experience implementing ERD technology.

In summary, upon initial review, Hanscom AFB provided a fairly standard site for IRZ implementation. In retrospect, several factors complicated its use as a demonstration site, including variable gradient/potentiometric surface and the relative complexity of its subsurface lithology. However, a successful demonstration was conducted.

#### **3.3 Test Site Description**

#### 3.3.1 Site/Facility Description

Hanscom AFB is located in Middlesex County, Massachusetts, approximately 14 miles northwest of downtown Boston. The Base occupies about 800 acres in the Towns of Bedford, Concord, Lexington, and Lincoln. Hanscom AFB is home to the Electronics Systems Center (ESC), a dynamic nucleus of research and development. ESC is the Air Force's acquisition and development center for world-class command and control systems. Hanscom Field, located adjacent to and north of the Base, is a civilian airport operated by the Massachusetts Port Authority (Massport) and the Federal Aviation Authority. Prior to 1973, Hanscom AFB leased the runways and flight line of what is now Hanscom Field from the Commonwealth and the primary mission of Hanscom AFB was the operational maintenance of fighter aircraft and research and development (R&D) support. Historical operations at the Hanscom AFB/Hanscom Field complex involved the generation, use, and disposal of numerous hazardous substances,
such as petroleum products, paint, and chlorinated and aromatic solvents. As a result of waste disposal and resource management practices during the period that the Hanscom AFB had a flying mission, the groundwater and soil in areas of Hanscom AFB and Hanscom Field became contaminated. In response, the Air Force included Hanscom Field sites in its environmental restoration program. Initial field investigations of the airfield sites commenced in the summer of 1982. The site was listed on the National Priorities List in May 1994.

The area of interest for the demonstration of ERD technology at Hanscom AFB is downgradient from Site 1. The source area for Site 1 is located on Hartwell's Hill, northwest of the overrun for Runway 23 and southeast of Hartwell Road, as shown in Figure 3-1. There are no aboveground structures in this portion of the AFB. According to Haley and Aldrich (1988), the Site was known as Fire Training Area II, and was reportedly used from the late 1960s through 1973 by the Hanscom AFB Fire Department for training exercises and for research on pyrokinetic materials. Fire training exercises consisted of collecting drummed waste oils, solvents, paint thinners and degreasers from around the Base and transferring them to an aboveground tank on-site. These chemicals were dumped into a pit, ignited and then extinguished. CAHs have been detected in groundwater in a narrow plume which extends from the source area at Site 1, southeastward under the overrun for Runway 23 and through the area where the RAP1-6 well cluster is located (see Figure 3-2). Thus SIC codes 9711C and 4581 are applicable.

# 3.3.2 Site/Facility Characteristics

## 3.3.2.1 Geology and Hydrogeology

Hanscom Field is situated on the southern edge of a glacial lake bed known as Lake Concord, which was filled with sediment during the last phase of glaciation in the region. The overburden beneath the site consists of an upper glaciofluvial unit comprised primarily of fine sand; a middle lacustrine unit consisting primarily of interbedded silt, clay, and fine sand; and a lower glacial till unit comprised primarily of unstratified dense sand, silt, and clay. The boring logs for the RAP1-6 well cluster, within the focus area of this study, show the bedrock surface at approximately 50 feet bgs, with the upper and lacustrine units each approximately 15 feet thick, and the lower till unit approximately 20 feet thick. The underlying bedrock is comprised primarily of granite, with lesser amounts of diorite and gneiss. Cross-sections of the demonstration area are provided in Figures 3-3 and 3-4.

The area in the immediate vicinity of Site 1 is underlain by 18 to 25 feet of glacial till overburden that rests directly on granitic bedrock. The till typically consists of very dense, coarse to fine sand with variable amounts of silt, fine to coarse gravel, cobbles and boulders. Beneath the till, the bedrock surface slopes downward in an east-southeasterly direction from Site 1 towards the RAP1-6 well cluster. The bedrock surface drops an average of about 1 vertical foot for each 26 horizontal feet (Haley & Aldrich 1988). The glacial till layer that exists at ground surface at Site 1 grades into a more permeable, less dense, sandy till, interlayered with a denser, gray till near RAP1-6T. This lower sandy till comprises the "lower aquifer" described by Haley & Aldrich (1988). On top of the lower sandy till is a layer of stiff, laminated, glaciolacustrine silt with clay of varying thickness. Overlying the glaciolacustrine layer are glaciofluvial deposits of the "unconfined aquifer" described by Haley & Aldrich (1988). The unconfined aquifer material generally consists of brown, medium to fine-grained sand of medium to high density.

*In-situ* permeability tests had been completed for several wells in the lower (glacial till) aquifer near RAP1-6T. Hydraulic conductivity values for the lower aquifer range between 3 ft/day to 48 ft/day, and average 26 ft/day (CH2M Hill, 1997). Based on groundwater elevation contour maps, the hydraulic gradient of the lower aquifer in the vicinity of RAP1-6T was estimated at 0.006, and the effective porosity of the lower aquifer materials was estimated at 20 percent (CH2M Hill, 1997). Based on these data, the groundwater flow velocity in the lower aquifer was estimated at 0.8 ft/day, or approximately 290 ft/yr.

Groundwater at the site can be found in three distinct aquifers. The upper glaciofluvial unit comprises a shallow, unconfined aquifer. Groundwater flow velocity in this water-bearing zone is approximately 1,100 feet per year. The clays and silts in the middle lacustrine unit have a semi-confining effect on the aquifer in the underlying lower till unit. Groundwater flow velocity in the lower till unit is approximately 300 feet per year. The third aquifer beneath the Site consists of the fractured granitic bedrock. Estimating groundwater flow velocities in bedrock can be difficult due to preferential flow through individual fractures. By considering the bedrock beneath the site to be the equivalent of a porous medium, groundwater flow velocity is estimated at 120 feet per year. However, actual groundwater flow velocities in bedrock are expected to vary significantly, being higher near individual fractures or fracture zones, and lower in relatively unfractured zones (CH2M Hill, 1997).

The depth to groundwater has ranged from about 7 to 9 feet bgs in well RAP1-6S in the unconfined aquifer, from 4.5 to 8.5 feet bgs in RAP1-6T in the lower aquifer, and from 2 to 9.2 feet bgs in RAP1-6R. Vertical hydraulic gradients at this location would normally be upward from the lower and bedrock aquifers to the unconfined aquifer. However, due to pumping from nearby lower and bedrock aquifer interceptor wells, the gradients are reversed. Also, the groundwater elevations in all aquifers have generally been lower since mid-1991, apparently due to the groundwater collection and treatment system placed in operation in the spring of 1991.

Well cluster RAP1-6 is located approximately 36 feet from the access road that parallels the overrun for Runway 23. Approximately 10 feet beyond the wells is a chain-link fence, behind which is a drainage channel (see Figures 3-5 through 3-7). The channel has steep banks with approximately 20 feet of relief from the top of the slope to the bottom of the channel. The channel is approximately 40 feet wide from the top of the slope at either side. The channel is used to convey stormwater from the runways to a wetland area to the northeast. Beavers have built dams downstream across the drainage channel, which have backed up water through the length of the channel.

According to the Solute Transport Model Setup and Calibration Report completed by CH2M Hill in December 1997, natural regional groundwater flow direction is to the east/northeast. In the areas of Site 1 and Site 2, contaminant migration would have taken place under natural groundwater flow from some time in the early 1960s until 1991. This is manifested by the plume orientations, which may predominantly reflect historical groundwater flow patterns rather than current ones. This is significant because current groundwater flow patterns are complicated by the number of pumping influences at the site. These pumping influences create more radial flow patterns from the west to the east and from the southwest to the northeast. A potentiometric surface map from 1998 (Figure 3-8) shows predominant groundwater flow in the demonstration area from the northwest to southeast (bending eastward in the vicinity of RAP1-6T). Monitoring well locations were intended to be immediately downgradient of the injection well and potentiometric surface data was carefully monitored over the demonstration period to account for observed shifts in gradient.

## 3.3.2.2 Climatology

The climate of the greater Boston area is temperate, with an average monthly low temperature of 28.6 degrees Fahrenheit (°F) in January, and an average high monthly temperature of 73.5 °F in July. Average annual precipitation in the form of rain, hail and snow is 41.5 inches, with November the wettest month and July the driest.

# 3.3.2.3 Distribution of CAHs and Current Pumping Remedy

Residual CAHs remain beneath Site 1, in the adsorbed and dissolved phases, and also potentially as residual DNAPL. CAHs have been detected in groundwater in the unconfined, lower, and fractured bedrock aquifers.

Groundwater samples have been collected for laboratory analysis of CAHs from the RAP1-6 well cluster since 1986. A summary of the groundwater monitoring data from nearby monitoring wells in the lower and bedrock aquifers is provided in Table 3-3. The predominant CAHs detected have been TCE and 1,2-DCE. Other commonly detected CAHs include VC, tetrachloroethene (PCE), 1,1,1-trichloroethane (1,1,1-TCA), 1,2-dichloroethane (1,2-DCA), 1,1,1- dichloroethane (1,1,1-DCA), and 1,1- dichloroethene (1,1-DCE).

No biogeochemical analyses was conducted prior to selection of this site as a candidate, but a baseline round of analyses was conducted by ARCADIS prior to the decision to go forward with the demonstration.

## Lower Till Aquifer (Target Zone)

The semi-confined aquifer in the lower till unit was targeted for the pilot demonstration. This aquifer contains elevated total volatile organic compound (TVOC) concentrations ranging from a low of 3.2  $\mu$ g/L in RAP1-1T to a high of 5,400  $\mu$ g/L in RAP1-6T. Historic sampling data for the Site indicates that this water-bearing unit contains "source" CAHs such as TCE and 1,1,1-TCA, as well as biotic degradation compounds such as *cis*-1,2-DCE, 1,1- DCA, and VC. The presence of 1,1-DCE further suggests the abiotic transformation of 1,1,1-TCA via elimination reactions (1,1-DCE can then be reductively dechlorinated to VC and ethene). Historical trends in the concentration at RAP1-6T are presented and discussed in Section 4.3.3.3.2 of this report.

## Bedrock Aquifer

The bedrock aquifer typically has the highest TVOC concentrations, ranging from a low of 2.5  $\mu$ g/L in RAP1-1R to a high of 589,000  $\mu$ g/L in RAP1-3R. Well RAP1-3R is located within the area of influence of the Vacuum Enhanced Recovery (VER) system operating at Site 1, substantially upgradient from our demonstration zone. As with the lower lacustrine unit, historic sampling data indicates that wells installed in the bedrock contain source CAHs and degradation compounds such as TCE, 1,1,1-TCA, 1,2-DCE, 1,1-DCE, 1,1-DCA, and VC. Based on dissolved-phase concentrations and the configuration of the bedrock surface, it is believed that

there may be residual DNAPL along the bedrock/overburden interface under (and downgradient of) the original release locations at Site 1. Current pumping at IW-6 and operation of the VER system is intended to capture this source material so that the downgradient recovery wells (IW-2, IW-3, and IW-4) can contribute to cleanup of the plume. In bedrock well RAP1-6R, total CAH concentrations have ranged from a high of 8,100 µg/L in June 1996 to a low of 2,110 µg/L in February 1991, and concentrations of VC have increased substantially between 1986 and 1998.

## 3.3.2.3.1 Existing Groundwater Remediation

The current remedy at Hanscom Field is comprised of pump and treat collection trenches and recovery wells, coupled with a VER demonstration/pilot in the source zone. Both Site 1 and Site 2 have collection trenches which have essentially cleaned up the shallow unconfined aquifer, resulting in very limited areas of residual impact around the original source locations (TVOC concentrations in the shallow water-bearing unit identified during the May 1998 sampling event range from below laboratory detection limits to a maximum of 760  $\mu$ g/L). The lower till and bedrock aquifers contain four "interceptor" recovery wells installed along the Site boundary (wells IW-1 through IW-4), two additional recovery wells installed just downgradient of the source area at Site 1 (IW-6) and Site 2 (IW-5), and four VER wells installed just downgradient of the source area at Site 1.

A pilot test of permanganate oxidation was conducted in the VER area in 2001. According to Tom Best (personal communication, 2002) "We started injections in June 2001. 2 of the 3 injections (June & October) were in bedrock wells and the 3rd was in the surface/lower aquifer well (RAP1-3S) in August 01. However, by October 01 the water table (had) dropped into bedrock for a couple of months. When it got back up in(to) the till the permanganate was gone." Thus since most of these injections should not have affected the lower aquifer, and this area is two to three years' travel time upgradient of the demonstration zone, it was not expected to influence the demonstration results.

The primary objective of the collection trenches and interceptor pumping wells is containment of the plume, while the four VER wells are intended to focus on source removal. The groundwater remediation system at Site 1 is part of a pump-and-treat system that operates at three "sites" at Hanscom. Recovered groundwater from all three sites is piped to a large-capacity treatment system that uses two stripping towers in series to remove the CAHs and granular activated carbon to treat the off-gas. The average groundwater flow rate for the Site 1 collection system is 20 to 25 gallons per minute. Annual Base-wide groundwater volumes treated have ranged from 50 million gallons to 148 million gallons over the period 1991-2002 (Hanscom AFB, 2002).

## 3.3.3 Site/Facility Maps and Photographs

A drawing showing the features of the demonstration zone in the vicinity of RAP1-6 well cluster, including the overrun for Runway 23, the drainage channel, and nearby monitoring wells, is shown on Figure 3-9. Figure 3-8 also shows the groundwater elevation in the lower aquifer based on groundwater elevation measurements taken by Haley & Aldrich in May 1998. Figure 3-2 shows the total CAH distribution in nearby monitoring wells in the lower aquifer from the May 1998 groundwater monitoring event.

Site photographs as discussed above are provided as Figures 3-5 through 3-7. Figure 2-4 shows the injection well in-service.

### 3.3.4 Conceptual Site Model

The initial conceptual model for this site included groundwater transport of dissolved-phase CAHs from the source area at Site 1 to the area around the RAP1-6 well cluster and beyond to the southeast. The initial releases occurred when waste solvents and oils were placed into burn pits at Fire Training Area II. These contaminants infiltrated downward by gravity through fractures and pore spaces in the till until they reached the water table. Water table fluctuations at the source area ranged from 10 feet above, to just below the till/bedrock interface. Below the water table, the DNAPLs continued to move downward by gravity until they could no longer migrate through fractures due to decreasing fracture apertures. The DNAPLs pooled in low spots at the till rock interface where there was little fracturing, but also entered bedrock fractures and migrated down into the bedrock.

At the till/bedrock interface is a relatively high permeability zone consisting of basal gravel or highly weathered bedrock. The high permeability zone, combined with the sloping bedrock surface, allowed migration of DNAPL and dissolved-phase contaminants to the southeast towards the overrun for Runway 23 and well cluster RAP1-6. Groundwater transport modeling by CH2M Hill suggested that DNAPLs may have migrated along a bedrock trough from Site 1 towards well cluster RAP1-6. However, even though there are relatively high total dissolved CAH concentrations in monitoring wells downgradient from Site 1, there were no analytical data or visual observations to confirm that DNAPLs have migrated as far as the RAP1-6 well cluster prior to the start of this demonstration. Concentrations of 1,2-DCE and TCE in wells near RAP1-6 were less than 1% of the solubility limit of those compounds, and no DNAPLs have been observed in any of the monitoring wells screened in the lower or bedrock aquifers in this area.

Remedial pump-and-treat systems operating since 1991 have contained most of the migration of dissolved or non-aqueous phase CAHs, especially in the unconfined shallow aquifer, in which CAH concentrations are now in the low  $\mu$ g/L range. Between 1991 and August 1997, downgradient migration in the lower and bedrock aquifers continued and was enhanced by the interceptor wells of the treatment system, which recover groundwater from the lower and bedrock aquifers along the Hanscom Field boundary with the Town of Bedford to the north-northeast. In August 1997 a bedrock interceptor well in the immediate vicinity of the Site 1 source area was placed in operation to contain the CAHs near the source area. This effort has subsequently been augmented by a four-well VER system and four additional lower and bedrock aquifer interceptor wells, all in the immediate vicinity of the Site 1 source area on Hartwell's Hill (about 800 feet or nearly 3 years groundwater travel time upgradient of the IRZ demonstration zone).

Ongoing pump and treat actions and natural attenuation appear to be precluding further off-site migration of contaminants and reducing both on-site and off-site concentrations. The overall remedial goals for the site are to achieve cleanup goals consistent with current and foreseeable future uses. For groundwater, the long-term cleanup goal is to achieve drinking water standards, since this site lies within an area classified as GW-1 (i.e., potential drinking water supply) by the Massachusetts Department of Environmental Protection (MADEP). For the CAHs detected at

this Site, the MADEP's GW-1 Standards are equivalent to the U.S. EPA's Maximum Contaminant Levels (MCLs). Despite the fact that operation of a pump and treat system around the clock since 1991 has succeeded in substantially reducing TCE concentrations, TCE at Sites 1 and 2 is still well above the current MCL (see Figures 3-10 and 3-11 abstracted from Hanscom AFB, 2002) and may have reached an asymptote. For this demonstration project, the objective was to determine if natural attenuation processes could be enhanced to accelerate the progress towards the site-wide remedial goal.

## 3.4 Pre-Demonstration Testing and Analysis

As discussed in Section 3.3.2.1, the demonstration area is underlain by an unconfined upper aquifer and a semi-confined lower aquifer (the target zone for the demonstration), both consisting of glacial deposits, and separated by a stiff, laminated layer composed of glaciolacustrine silt with clay. The lower aquifer rests directly on bedrock at a depth of about 50 feet. We have prepared cross sections of the demonstration zone and vertical correlations of the installed wells (Figures 3-3, 3-4, and 3-12). These show that the injection well and most of the monitoring wells span the majority of the thickness of the lower aquifer – which is a tightly packed, poorly sorted till.

Pretreatment soil characterization in the lower aquifer during well installation (Table 3-4 and Appendix A-2) showed relatively low concentrations of soil TOC (<2%) and consequently low concentrations of sorbed TCE (<7 ppb), *cis*-1,2-DCE (<30 ppb) and VC (<11 ppb). The only other organic detected was a trace of acetone. The grain size analysis showed almost all the samples to be silty sand mixed with gravel which is consistent with the previous discussion of this zone as a till.

Prior to this demonstration project, a previous site consultant had conducted a modeling effort. In order to make the model reproduce the observed groundwater concentrations it was necessary for the modelers to postulate the presence of a source in the vicinity of well RAP1-6T (Figure 3-13), (CH2M Hill, 1997). According to Mr. Tom Best at Hanscom AFB, information from a former employee at the Base suggested that solvent disposal/fire training activities were conducted at one point on this southeast side of the runway. Previous attempts to locate a source zone in this area through vapor probing were unsuccessful. However, when ARCADIS installed five additional wells in June 2000, the existence of this source area was confirmed by a clear increasing trend in groundwater concentration in the lower aquifer in this area (Figure 3-14). As shown on that figure, the total volatile organic compound (VOC) increased dramatically from northeast to southwest in the lower aquifer in the immediate demonstration area before the demonstration. They were:

- 125 μg/L at B239-MW
- $2,700 \,\mu\text{g/L}$  at the well that would be used as the injection well later
- an average of 6,130 μg/L at the transect of monitoring wells that includes RAP1-6T
- an average of 8,060 µg/L at the transect that that includes IRZ-2 and IRZ-3 (note that IRZ-5 was not installed until after the demonstration began).

Other pretreatment physical characteristics of the subsurface at this site are listed in Table 1-1. Before treatment, the TCE in the lower aquifer at the RAP1-6 well cluster was fairly constant

over 15 years, averaging between 1,000-2,000  $\mu$ g/L with elevated levels of both *cis*-1,2-DCE (2,000-5,000  $\mu$ g/L range) and VC (500-1,300  $\mu$ g/L range) (see a more detailed discussion in Section 4.3.3). The site was moderately anoxic/reducing (DO <1.5 mg/L, ORP typically 0 to -50 mV) with a near neutral pH in the range of 6.0 to 6.5 prior to treatment. Note also that the ethene and methane level in IRZ-1 and these other neighboring wells was low (Figures 4-30, 4-32), and hydrogen sulfide was low (Figure 4-40). A summary of the initial baseline geochemical characteristics of the site is provided as Table 3-5. Notable are the relatively low concentrations of nitrate, sulfate and DO along with some evidence of dissolved methane.

The behavior of chlorinated solvent plumes with respect to reductive dechlorination has been categorized into three types (USEPA, 1998). Type 1 behavior occurs in the presence of anthropogenic carbon, and results in the rapid degradation of the more highly chlorinated solvents, provided that anaerobic conditions and an adequate supply of carbon substrate exist. Type 2 behavior is similar to Type 1, but is driven by a relatively high concentration of biologically available natural organic carbon rather than an anthropogenic carbon source. Biodegradation is generally slower with Type 2 than with Type 1 conditions. In both cases, the role of competing electron acceptors and the fate of VC are variables to be considered. Type 3 behavior dominates where little carbon is available and conditions are aerobic, or where the microbial community is incapable of chlorinated solvent biodegradation. Reductive dechlorination does not occur under Type 3 conditions, but VC may be oxidized in an aerobic environment. In this classification system Hanscom appears to be a type 1 site. Clear evidence of the first stage of degradation to DCE has been observed before treatment. This may be a site that is "stalled" at DCE although some VC production is probably also present under pretreatment conditions. Although relatively little TPH data has been collected at this site since they are not primary risk drivers, petroleum hydrocarbons were known to be released in substantial quantities in fire training activities at this site and were found at the site 1 source area on Hartwell's Hill (personal communication with Tom Best 2001). A recent National Academy of Sciences report (2000) points out that petroleum hydrocarbon can not always be relied upon as a long term electron donor for natural attenuation because they are often consumed over time before the CAH contamination is fully attenuated. TOC in the pre-injection baseline round ranged from 3 to 6 mg/L, suggesting that the amount of degradable carbon present was limited. Complete and efficient degradation would not be expected under these circumstances - that are probably not methanogenic or sulfate reducing but rather probably predominantly denitrifying or iron reducing. As shown in Table 1-1 of Suthersan, 2002 (reprinted from various ITRC sources), TCE degradation under these conditions has been observed on previous occasions but DCE and VC treatment has not. This suggests this site might indeed be "stalled" at DCE and VC. Thus although only one pretest round was available at IRZ-1 we have an internally consistent picture of the starting conditions that is also in agreement with our overall theoretical understanding of these processes developed based on experience at other sites and in the literature (Suthersan, 2002).

#### 3.5 Testing and Evaluation Plan

Operational and experimental methods for the ERD demonstration from system start-up through demobilization are discussed in this section.

### 3.5.1 Demonstration Installation and Startup

Physical setup for the Hanscom AFB system was minimal. Permanent equipment was limited to wells, with a removable well seal with fittings on the injection well to allow for connection. Utility requirements were limited to a source of potable water from mixing of the molasses solution. The demonstration area, including pre-existing wells and new sample locations, is shown in Figure 3-9.

The temporary equipment required for the injections included the following: a 210-gallon solution mixing/holding tank, a gasoline powered transfer pump, and an injection hose. A schematic of the injection system is presented in Section 2.1. Start-up testing of injection system only included filling tank with water to check for leaks. The tank and associated pumps generally functioned without difficulty. All temporary equipment, molasses, and reagents (bromide tracer), were stored in an existing site building. A conventional pick-up truck was used to transport the equipment to the injection well for each injection event.

Steps required for each injection event included testing of pH in the injection well, consulting guidance provided by the project manager as to what injection to make depending on the observed pH, manual mixing of the reagent solution, connection of the injection system to the injection well, and pumping of the solution into the injection well followed by an injection of clean water into the injection well. Typically, either a single batch (200 gallons) or a double batch (400 gallons) of solution was injected. A single batch of reagent solution consisted of 20 gallons of food-grade blackstrap molasses, 180 gallons of potable water, and 113 grams of potassium bromide. Mixing of the reagent solution was accomplished by partially filling the 210 gallon solution tank with water, adding 20 gallons of molasses and 113 grams of potassium bromide to the tank, stirring the tank manually for several minutes with a polyvinyl chloride (PVC) stir rod, and then filling the tank to the 200-gallon mark with clean water.

Prior to each injection event, a small volume of groundwater (one to three well volumes) was purged from the injection well. The groundwater pH was then measured using a pH meter or litmus paper. Well pH was used to determine the volume of reagent solution to be injected. No other analyses were performed, except for process monitoring and groundwater sampling for technology effectiveness verification described in Section 3.5.7.

Once mixed, the solution tank and injection equipment was transported in a pickup truck to the injection well. Hoses were connected between the solution tank, transfer pump, and injection well. The system valves were then opened, the pump started and run until all of the reagent solution had been transferred to the well. If a double injection was planned, a second batch of reagent solution was mixed and injected using the same procedure. Following the injection of reagent solution, if a water push was to be injected, the solution tank was filled with 200 gallons of clean water, which was injected using the same procedure.

During the initial injection events the reagent solution injection proceeded at a rate of approximately ten gallons per minute at observed well head pressures of approximately 10 to 15 pounds per square inch gauge pressure (psig). However, due to assumed biological fouling of the injection well (discussed in the "System Maintenance" section below), injection pressures

increased (due to higher flow resistance). In turn, the increased injection pressures apparently resulted in the degradation of the annular seal on the injection causing some leakage of reagent solution in the well vault during injections. As a corrective measure, injection pressures were controlled through a number of means to a lower pressure to prevent the leakage of the reagent solution.

During the final reagent injections, reagent and clean water injection flow rates were reduced to approximately one to two gallons per minute and pressures of two to three psig. Labor required for each injection at the initial injection rate, was approximately four to six hours for a single batch injection, with an additional one to two hours for a double injection. Due to the reduced flow rates toward the end of the demonstration, labor required for each event (exclusive of travel time to the site) had increased to approximately eight to ten hours for a single injection and twelve to sixteen hours for a double injection.

#### System Maintenance

Very little maintenance or repair work was required during the demonstration. A hose barb fitting on the injection system broke and was replaced on February 4, 2002. Additionally, the top of the injection well and its protective road box were damaged by a lawnmower as discovered on July 26, 2002. These items were promptly repaired and the top of well elevation was resurveyed as reported on October 16, 2002.

Given the enhanced biological nature of the in-situ remedy being used, there is some chance of biological fouling developing in and around the reagent injection wells (i.e., the well screen itself or possibly the well filter pack). However, given that the biological growth is anaerobic in nature, the actual mass of biological growth is typically minimal (as compared to biomass related to aerobic processes commonly observed in other remediation areas such as pumping wells or above-grade water treatment). In the case of reagent injection well fouling or plugging a typical remedy would be to surge the well using a well block to induce turbulence in the well and break up the biological mat.

During this demonstration, indirect evidence of biological growth in and around the injection well was observed as evidenced by a decrease in the maximum obtainable reagent injection rate after the first several injection events. This reduction in injection rate was also accompanied by an increase in injection pressure (i.e., resistance to flow created by the assumed fouling). In addition, the observation of increased injection pressures was coincident with the observation of a small amount of reagent solution leakage observed in the injection well vault. This leakage was observed to be coming from between the well casing and the surface seal.

Due to the nature of the injection equipment (e.g., a centrifugal pump without pressure regulation), the decreased flow rate caused a significant increase in injection pressure, which may have exceeded the soil fracture pressure. This in turn may have resulted in the creation of some fractures or voids in the annular seal for the injection well in turn resulting in the observed solution leakage at the surface.

Upon observation of the increased injection pressures and solution leakage, the corrective action employed by the field staff included either; a lowering of the injection pressure by reducing

pump speed, lowering the injection pressure by throttling the pump using a flow control valve or in some cases performing the injections solely under gravity flow. Each of these actions were intended to minimize surface leakage. This avoided the capital cost and disruption of repairing or replacing the well.

Safety issues were limited to those associated with handling equipment (vehicles, pumps, hoses, fittings) in the field, and working with contaminated groundwater from wells. No hazardous materials were used in the injection solution or generated during operation of the system, with the exception of purge water from the wells.

# **3.5.2** Period of Operation

Dates of major events relevant to the demonstration are summarized in the table below. Dates of intermediate injections and monitoring events are tabulated in Table 3-6. A final round of groundwater monitoring to test for rebound effects is planned for November 2003.

Event	Date						
Pre-Demonstration							
Initial investigations of Sites 1 and 2	Early 1980's						
Installation of monitoring wells pre-dating	1985 (RAP1-6T cluster); 1996(B239-MW through						
demonstration	B243-MW)						
Demonstration							
Well-installations and soil sampling	May 2000 (IRZ-INJ, IRZ-1 through 4); May 2001						
	(IRZ-5)						
Baseline groundwater sampling	June 2000						
First substrate injection	October 11, 2000						
Last substrate injection	October 9, 2002						
Demobilization	October 9, 2002						
Post-demonstration monitoring	Planned for November 2003						

# 3.5.3 Amount/Treatment Rate of Material Treated

A total of forty-seven injections were conducted at Hanscom AFB since October 11, 2000. The final injection was conducted on October 9, 2002, and final sampling occurred on October 14 – 16, 2002, for a total of almost exactly 24 months of demonstration operation. Over this time, a total of 1,250 gallons of raw blackstrap molasses, 11,250 gallons of dilution water, 7,575 gallons of push water and 4,732 grams of potassium bromide have been injected into IRZ-INJ. The injection rate is shown in Figure 3-15. The average injection rate over the period of treatment was 139 lbs molasses/week. The chloride concentration of the molasses injection solution was measured on March 12, 2002 to be 1,500 mg/L.

# 3.5.4 Residuals Handling

No hazardous waste was generated during the setup and operation of this demonstration, except for soil (drill cuttings) generated during injection well and monitoring well installation, and purge water generated during well development and sampling. Purge water was disposed of in the on-site wastewater treatment plant. Soil cuttings were stored in 55-gallon drums, characterized and disposed of at a licensed hazardous materials disposal facility.

# **3.5.5** Operating Parameters for the Technology

A description of operating procedures and labor requirements for the ERD demonstration is given in Section 3.5.1. Carbon dosing was variable during the demonstration, as was the use of water injections to disperse the substrate. These parameters were determined on the day of the injection event based primarily on the pH measurement in the injection well, but also on more detailed process monitoring conducted at regular intervals during the demonstration. Section 4.3.2.1 contains a discussion of process monitoring parameters and process control throughout the project.

Performance monitoring (to assess technology efficacy) for this demonstration was conducted using high quality assurance analysis during three full sampling rounds and five abbreviated sampling rounds. Analytical parameters, methods and analysis locations/ organizations are specified in Table 3-7. Furthermore, records were kept of the color, odor and other readily apparent characteristics of the sampled groundwater. Additionally, some groundwater samples at Hanscom were analyzed on-site by the Base's gas chromatograph (GC), operated by an independent contractor. The SOP for this procedure is given in Appendix A-3.

# 3.5.6 Experimental Design

The experimental design for the project was established in the demonstration plan (ARCADIS, March 2000). In brief, the types of measurements made are listed below. Discussions on the methods and outcomes of each type of data collection are provided elsewhere, as cited.

- Soil characterization Section 3.5.7.1.3 Soil samples were collected once, during well
  installation and analyzed for the parameters listed in Table 3-8.
- Process monitoring Sections 3.5.7.2, 4.3.2, 4.3.4 In 13 periodic events, measured DO, pH, ORP, specific conductance, temperature and water levels. In addition, used field test kits to analyze for hydrogen sulfide and ferrous iron, and laboratory analysis for bromide and TOC.
- Full and abbreviated groundwater monitoring Sections 3.5.7.1.2, 4.3.3, 4.3.5, 4.3.6.2, 4.3.6.3 three full and five abbreviated sampling rounds were conducted, with analysis for the parameters listed in Table 3-7.
- Process control Sections 4.3.2.1 Varied carbon dosing and water pushes in 47 injection events based on continuous evaluation of groundwater monitoring data.

## 3.5.7 Sampling Plan

The project sampling plan was developed on the basis of ARCADIS' previous experience at commercial ERD sites and existing site data for the Hanscom demonstration area. The sampling plan is detailed in the project's demonstration plan (ARCADIS, March 2000), which also contains the project QAPP.

# 3.5.7.1 Sample Collection

Field methods are described in this section for well installation and sampling of saturated soils and groundwater.

## 3.5.7.1.1 Well Installation Procedures

The general area studied during the IRZ demonstration laid within an approximately triangularshaped area between Monitoring Well B239-MW, Pumping Well No. 2 (BIW-2), and Pumping Well No. 4 (BIW-4) (Figure 3-2). The main focus area was around RAP1-6T, just west of the runway outrun, where one injection well and a series of new lower aquifer monitoring wells were installed (Figure 3-9). The target hydrogeologic unit was the basal sand, silt and clay glacial till unit that directly overlies bedrock at approximately 50 feet bgs in the demonstration area. The shape of the demonstration area and proposed well locations were determined based on the natural groundwater flow direction, the existence of the two downgradient pumping wells, as well as well placement limitations caused by the runway, a fence, and a drainage ditch located downgradient from the proposed injection well. Ten wells were used for data collection during the demonstration project: one (1) new injection well, five (5) new monitoring wells, and four (4) existing monitoring wells. Additional data was also collected from other nearby wells, when applicable.

The proposed injection well was installed immediately upgradient (relative to the originallyassumed southwest-to-southeast general groundwater gradient) from existing well cluster RAP1-6, which consists of wells RAP1-6S, RAP1-6T, and RAP1-6R, screened in the upper, lower, and bedrock aquifers, respectively. Existing monitoring well, B239-MW located approximately 400 feet upgradient from the proposed injection well was used as the primary background well. Downgradient wells used for demonstration monitoring included existing monitoring wells RAP1-6S, RAP1-6T, and RAP1-6R, and five new downgradient wells (IRZ-1 to IRZ-5) located as shown on Figure 3-9. Note that during the demonstration, the actual, measured direction of the groundwater gradient of the lower aquifer varied in response to various factors including pumpand-treat system operation (see discussion in Section 4.3).

The wells in the RAP1-6 cluster, including wells RAP1-6S (upper aquifer), RAP1-6T (lower aquifer), and RAP1-6R (bedrock aquifer), were installed in November 1985 for the US Army Corps of Engineers as part of a remedial investigation at the Base. Thus, these wells were present for use as groundwater monitoring wells during the demonstration project. RAP1-6T, which is of most importance to the demonstration project, is screened between about 30 and 45 feet below grade.

The installation of the injection well (IRZ-INJ) and four new monitoring wells (IRZ-1, IRZ-2, IRZ-3, and IRZ-4) was coordinated by ARCADIS and occurred in early-mid May 2000. The first of these wells installed was IRZ-2, located southeast of the drainage ditch, which was installed on May 3, 2000. The borehole for monitor well IRZ-2 was drilled using the hollow-stem auger techniques with 8-inch diameter augers. The boring was advanced to a total depth of 45 feet. The well screen was installed from 33.5 to 43.5 feet bgs (see Appendix A-4 for the relevant boring logs and well construction diagrams).

An attempt to drill IRZ-1 next was made on May 4, 2000, also using hollow stem augers. However, surging sand prevented the boring from progressing beyond 34 feet below grade. It was decided that a drive-and-wash drilling technique would be more appropriate for completing the remaining borings, and the borehole was abandoned. A replacement boring was installed on May 15 to 16, 2000, using the drive-and-wash technique and 6-inch diameter casing. The boring was advanced to refusal, and the well screen was set from 33.4 to 48.4 feet bgs. Wells IRZ-3, IRZ-4, and IRZ-INJ were installed during the period of May 5 through May 15, 2000, using the drive-and-wash technique. Borings for IRZ-3 and IRZ-INJ were advanced using 6-inch diameter casing, and well screens were set at 40 to 55 feet bgs and 34 to 49 feet bgs respectively.

Well IRZ-4 was installed initially using 6-inch diameter casing. However, a boulder was encountered at approximately 36 to 39 feet bgs, which could not be penetrated using the 6-inch casing. A 4-inch diameter casing was telescoped inside the 6-inch casing, and was successfully driven through the bolder. IRZ-4 was constructed with a 2-inch diameter well screen and riser, with the screen set at 36 to 51 feet bgs.

All of these wells were developed using a whacker pump and foot valve. 525 gallons of water were removed from IRZ-INJ with an improvement in the turbidity observed over the course of development. 250 gallons of water were removed from IRZ-1 and the water at the end of development was clear. 200 gallons of water were removed from IRZ-4, 125 gallons of water were removed from IRZ-3. The water was still slightly turbid at the end of development for wells IRZ-2, IRZ-3, and IRZ-4.

Well IRZ-5 was installed in May 2001, as a response to monitoring data suggesting that the flow direction from the injection well was toward IRZ-1 and missed monitoring wells IRZ-2, IRZ-3, and IRZ-4. A drive-and-wash drilling technique was used to advance a boring to a total depth of 53.5 feet bgs. The well screen was set at 38 to 53 feet bgs. Approximately 300 gallons of water were removed during subsequent well development.

Three sets of survey data were collected; after initial well installation, after well IRZ-5 was added in 2001 and after a well was damaged by a lawn mower in 2002. These data were used for potentiometric surfaces and mapping and are presented in Appendix A-5.

## 3.5.7.1.2 Groundwater Sampling

## Groundwater Sampling and Field Measurements

Two main types of field parameter monitoring were conducted during the Hanscom AFB demonstration, high QA/QC events and low QA/QC events. High QA/QC field monitoring events were conducted, typically associated with groundwater sampling events during abbreviated and full process monitoring events, and used low-flow sampling methodology. During low QA/QC events, a down-hole sonde was used to make similar field measurements for process monitoring when full sampling was not require (i.e., before or during injection events). These lower QA/QC events provided very valuable feedback data on the biogeochemical conditions, to assist in making decisions on the amount of substrate to inject thus controlling the reactive zone. These lower QA/QC events provided additional time points to help analyze changes in the characteristics and extent of the IRZ.

Groundwater sampling methods during high QA/QC performance monitoring rounds, utilized low-flow, or micropurge procedures, consistent with EPA and AFCEE published protocols. The basic tenet of the micropurge technique is to collect groundwater from a discrete portion of the well screen at a rate which most closely replicates the natural recharge of groundwater from the

formation into the well screen. This is accomplished by removing groundwater at low flow rates (typically between 100 and 500 ml/minute) while monitoring the water level within the well to ensure minimal (or preferably no) drawdown. While the well is being purged, field parameters are monitored at the well head using a flow through cell. DO, ORP, temperature, pH and specific conductance are monitored and recorded at ten minute intervals while the well is purged. When a minimum of three well volumes have been purged and these readings stabilize within 10% for three consecutive readings, the groundwater is considered to be representative of the aquifer (as opposed to stagnant water within the well) and groundwater samples for laboratory analysis are collected directly from the pump discharge at the surface. Depending on the depth to water and diameter of the existing wells at the site, different pumps may be utilized. At Hanscom, a low-flow, submersible, Redi-Flo pump was utilized.

During the demonstration period, a total of five high QA/QC abbreviated monitoring rounds, and three high QA/QC full monitoring rounds, including a baseline, midpoint and a final processmonitoring round, were conducted. The primary purpose of these events was to determine if the system was meeting its performance goals. The background full-monitoring round was conducted in June 2000, and the midpoint full-monitoring round was conducted in early May 2001. The final round was conducted October 14-16, 2002. Concerns about field equipment calibration during the mid-October monitoring round prompted a repeat of field parameter measurement "final" monitoring round, for field parameters only, on October 29, 2002. In addition, for comparison purposes a few samples were collected on October 29, 2002 for dissolved hydrogen using a different type of pump (see discussion below in this section). The final round of samples included secondary water quality parameters and will be followed approximately 12 months later by a "rebound" sampling round.

During the demonstration period, a total of thirteen low QA/QC process-monitoring rounds were conducted. The minimum parameter list for these events were pH, DO, specific conductance, and ORP. TOC and/or DOC were sampled when other indicators suggested the possibility of carbon overloading leading to fermentation. Additionally injection well pH was measured before every injection event and samples were obtained at intermediate time points for analysis by the Base's field GC to obtain additional information about TCE and DCE.

The times for these sampling and field monitoring events were selected during the demonstration by the ARCADIS project manager and team leaders, in consultation with ESTCP and AFCEE. As expected, the process monitoring events were more frequent near the beginning of the injection program, when the optimum injection dose was being established. These parameters provided information on the efficacy of carbon delivery to the reducing zone and the redox condition of the zone. From this information, carbon injection regimes were fine-tuned and more involved monitoring events could be effectively scheduled.

The first full round of high QA/QC sampling for performance evaluation (baseline sampling) occurred in June 2000 before the injection of any reagent and is known as the biogeochemical characterization. The biogeochemical characterization serves three purposes:

- a) confirms or refutes the applicability of the technology to the site
- b) establishes the baseline along with historical data

c) provides data necessary for system design

Determination of the baseline conditions of the appropriate biogeochemical parameters is a key element for the design of an ERD system. This evaluation gave a clear indication of the existing conditions and the necessary steps to be taken to optimize the environment to achieve the target reactions.

#### **Dissolved Gas Sampling**

ERD processes produce gases that can provide useful information about the process. Additionally, in some cases, the gasses produced may need to be managed for health and safety reasons. The evaluation of the potential for problems with gas generation is generally done as part of engineering design of a system. The depth to the zone of interest, likely paths for vapor migration, proximity of structures and other receptors and potential volumes of gasses produced are assessed in this context. For this demonstration program, it was determined that there were no causes for concern about gas generation hazards or nuisances at Hanscom AFB since no structures lay over the demonstration zone.

As previously described (ARCADIS, 2000; Suthersan, 2002), dissolved hydrogen was monitored in groundwater from IRZ monitoring wells at Hanscom AFB since its concentration can suggest which microbially-mediated redox processes are predominating in the reactive zone. Standard low-flow sampling techniques are used as the basis for sampling groundwater for dissolved hydrogen analysis as hydrogen is an extremely volatile gas that can easily be lost to the atmosphere if exposed to air particularly under turbulent conditions. The use of a flow-through cell increases the protection of samples/measurements against atmospheric loss or contamination. As such, Chapelle, et al. (1997) describe a gas stripping method (also known as the "bubble strip" method) for dissolved hydrogen sampling of groundwater monitoring wells which has become the accepted method given its relative simplicity and short sampling time. Simply stated, the gas stripping method involves creating equilibrium between the dissolved hydrogen in pumped groundwater (employing low-flow sampling techniques) and a small bubble of either air or nitrogen in a flow-through cell. Since hydrogen gas is extremely volatile, an equilibrium condition is reached fairly quickly. After equilibrium between the liquid and gas phases has been reached, a small amount of the gas phase is withdrawn and analyzed using a hydrogen detector (a typical GC technique which in most cases will be conducted by a subcontract analytical laboratory). The gas stripping sampling method and associated analysis (Wiedemeier, et al., 1996) has been incorporated by analytical laboratories performing dissolved hydrogen analysis (in this case Vaportech). They have developed relatively straightforward sampling kits and instructions which draw heavily on the gas stripping sampling method, and were employed here. These gas stripping methods incorporate the following steps:

- 1. Connect outlet tube from sampling pump to inlet tube of the pre-assembled sampling cell provided by the analytical laboratory.
- 2. Operate pump to flow at a rate of 100 to 700 ml/min (following laboratory's recommendation), purge any gas bubbles from sampling cell assembly.
- 3. Using a syringe (provided in kit), inject 20 to 30 ml (again following laboratory's recommendation) of air into the cell assembly.

- 4. Continue to pump through groundwater for a predetermined time between 10 and 30 minutes (depending on laboratory's recommendations which sometimes are dependent on pumping rate).
- 5. Via syringe, withdraw 1-2 ml of the gas bubble to purge, then withdraw a gas sample (typically 15-20 ml) and re-inject into a laboratory-provided sample vial, which is then shipped back to the analytical laboratory for analysis.

Other essential dissolved gas parameters monitored during the demonstration included light hydrocarbons, namely methane (suggesting methanogenesis), ethane, and ethene (which are by-products/end-products of biological-mediated chlorinated hydrocarbon degradation). These samples were collected in glass bottles provided by the analytical laboratory performing the analysis using low-flow methods. Two (2) 40 mL glass VOA vials with Teflon-faced septa and screw caps were filled for each sample. After collection, liquid samples were cooled and maintained at about 4°C until analyzed. Analytical methods for these light hydrocarbon gasses typically rely on GC techniques similar to those reported by Kampbell, et al., (1989), using SW3810 Modified, which is a static headspace technique for extracting volatile organic compounds from samples. Analytical methods for the light hydrocarbon gasses and  $CO_2$  typically rely on GC techniques similar to those reported by Kampbell et al., (1989), using SW3810 Modified, which is a static headspace technique for extracting volatile organic compounds from samples. Analytical methods for the light hydrocarbon gasses and  $CO_2$  typically rely on GC techniques similar to those reported by Kampbell et al., (1989), using SW3810 Modified, which is a static headspace technique for extracting volatile organic compounds from samples. Such methods are discussed further in subsequent report sections.

# 3.5.7.1.3 <u>Saturated Soil Sampling</u>

Soil samples were collected during well installation from within the saturated portion of the aquifer that was targeted for treatment. Samples were collected by driving a split spoon into the formation using a 140-pound weight. Because extensive lithologic information was available based on existing drilling logs, no soil samples were collected above the target depth, and borings were logged based on recovered cuttings. For each of the five new wells (one injection and four monitoring) initially installed, a soil sample from the suspected contaminated saturated zone was sampled and analyzed for the soil parameter list. One field blank and one field duplicate were also analyzed. Data collected was used to supplement existing geologic data and enhance the understanding of contaminant distribution, native soil TOC concentrations, and soil-groundwater contaminant partitioning.

# 3.5.7.1.4 <u>Sample Shipment and Labeling</u>

All sample sets were accompanied by a chain-of-custody form. Prior to shipment or transfer of custody, all samples were maintained in the custody of the field manager. Upon transfer of custody, the field manager verified the information on each sample label and assured that each container was intact and sealed using custody tape. He/she then signed and dated the chain-of-custody form. The individuals receiving the samples also signed, dated, and noted the time that they received the samples on the chain-of-custody form. This form documents transfer of custody of samples from the field investigator to another person, to the laboratories, or to other organizational elements.

Samples were properly packaged for shipment and delivered or shipped to the designated laboratory for analyses. Because common carriers (Federal Express, Airborne Express, etc.) will not sign chain-of-custody records, the original chain-of-custody form and one copy of the form

was placed in a plastic bag inside the secured shipping container when samples were shipped. One copy of the record was retained by the task field manager.

Shipping containers were secured shut using nylon strapping tape and custody seals. A custody seal was placed over the lid of the sample cooler to indicate whether the cooler had been opened during shipment prior to receipt by the laboratory. The original chain-of-custody form was transmitted to the project manager after samples were accepted by the laboratory. This copy then became a part of the project file.

A sample identification system was used to enable the field personnel to establish unique and appropriate identifications for each sample collected. This system incorporated identifiers for the site, sample matrix, the sample location, and the sample number. Field duplicates were designated by the sample code DUP, while equipment, field, and trip blanks were designated with the sample matrix codes EB, FB, and TB, respectively.

The site identification code for Hanscom was HAN. The matrix codes included:

- SS Soil
- GW Groundwater
- SG Shipped ground water sample for hydrogen analysis

Location codes were assigned in the field. Sample numbers were assigned in the field. Thus, for example, a groundwater sample at Hanscom might have been coded: HAN-GW-MW7-1. A duplicate of that sample would then have been coded: HAN-GW-MW7-DUP1. A trip blank for groundwater would be coded: HAN-GW-TB-1.

# 3.5.7.2 Sample Analysis

ESTCP demonstration sites are subjected to rigorous performance monitoring. Performance monitoring (to assess technology efficacy) for this demonstration was conducted using high quality assurance, low-flow groundwater sampling techniques and analysis during three full sampling rounds and five abbreviated sampling rounds. Analytical parameters, methods and analysis locations/organizations are specified in Table 3-7. Furthermore, records were kept of the color, odor and other readily apparent characteristics of the sampled groundwater. Additionally, some groundwater samples at Hanscom were analyzed on-site by the Base's GC (operated by an independent contractor). The SOP for this procedure is given in Appendix A-3.

The results of routine process monitoring were used to modify injection protocols and make other process control decisions, in an effort to maintain reducing conditions while avoiding overly depressing pH. Process monitoring was conducted using portable field instrumentation (e.g., Horiba U-22) and varies from relatively low QA (e.g., using down-the-well sondes) to relatively high QA (e.g., using flow-through cells) to measure DO, pH, ORP, specific conductance, and temperature. In addition, field test kits are used to analyze for hydrogen sulfide and ferrous iron, and samples are periodically submitted for laboratory analysis of bromide and TOC. Thirteen process-monitoring events were conducted at Hanscom. Table 3-6 is a week-byweek summary of the demonstration including injections and sampling events. The only groundwater sampling rounds not listed in this table are the initial full sampling round conducted in June 2000 and a "rebound" monitoring round planned for November 2003.

Finally, one soil sample per installed well was analyzed for the parameters listed in Table 3-8, which also summarizes the laboratory used, sample size, preservation, holding time, etc.

Further details of these methods are contained in the final demonstration plan.

# 3.5.7.3 Experimental Controls

Experimental controls included the use of background well B239-MW and other wells outside of the treatment zone in the groundwater monitoring program. Results for these control samples are discussed in Sections 4.3.3.3.4, 4.3.3.3.5, and Appendix A-6a.

# 3.5.7.4 Data Quality Parameters

Representativeness, completeness, comparability, accuracy and precision of the demonstration data are addressed in the data validation memoranda in Appendix A-6b. Deviations from the QA sampling plan, problems associated with QA samples, and their resolutions are recorded in Appendix A-6c. Relatively few data quality problems were identified, and most of these were judged inconsequential or were resolved by re-sampling or relying on alternate measurements of the same parameter.

CAH data in general were shown to be very reliable. Most of the data quality problems encountered during various sampling rounds occurred with the field data or the TOC/DOC data. In particular, the high levels of organic carbon in the injection wells caused problems with field measurements. However, these parameters are used more for process operation and interpretation than to assess the effectiveness of the technology. Therefore, it was judged that these problems did not materially affect the overall demonstration results.

# 3.5.7.5 Data Quality Indicators

Validation of the demonstration data were performed using the QA/QC criteria set forth in the "USEPA Contract laboratory (CLP) National Functional Guidelines for Organic Data Review," revised in June 2001. The data validation memoranda in Appendix A-6b describe the methods used to calculate data quality, and their results.

# 3.5.7.6 Calibration Procedures, Quality Control Checks, and Corrective Action

Calibrations of laboratory analytical equipment are addressed by the data validation memoranda in Appendix A-6b, which also describe the use of blanks, surrogates, matrix spikes and laboratory control spikes.

Procedures used to calibrate field equipment are described in the project's demonstration plan (ARCADIS, 2000). Specific instances of known problems with field equipment or measurements, and their resolutions, are recorded in Appendix A-6c. Relatively few problems were encountered, and many of these were able to be resolved by re-sampling or relying on alternate measurements of the same parameter.

## 3.5.8 Demobilization

Following the final injection event, reagent solution mixing and injection equipment was rinsed with potable water to remove residual molasses. No other above ground equipment or facilities were used during this demonstration, so no further demobilization was required. Underground equipment utilized during this demonstration was limited to the injection well and monitoring wells which were installed for this demonstration. At the request of the Air Force, these wells were left in place for use in the future for monitoring of the Site 1 pump and treat system. A written confirmation of the Base's wish to retain the wells is included in Appendix A-4.

### 3.6 Selection of Analytical/Testing Methods

Analytical methods used in the demonstration are listed in Table 3-7. Field instruments used in the program were identified in ARCADIS SOPs, in the project demonstration plan (ARCADIS, March 2000).

### 3.7 Selection of Analytical/Testing Laboratory

The locations or laboratories where analyses were performed are indicated in Tables 3-7 and 3-8. The majority of analyses were performed either by ARCADIS in the field or at Severn Trent Laboratories (STL) in Tampa, Florida (Mass DEP Certificate #M-FL224), or Savannah, Georgia (Mass DEP Certificate #M-GA006). Specialized analyses for dissolved gases in groundwater were performed by Vaportech Services, Inc. of Valencia, Pennsylvania.

In addition, although it was not planned in the demonstration plan, numerous additional rounds of TCE and DCE analyses by GC were provided courtesy of Tom Best of Hanscom AFB. These included samples collected on some occasions by the Base's contractor (Shaw/IT) and on other occasions by ARCADIS. The analyses were performed at Hanscom by IT. As discussed in Section 4.3 the data sets with and without the on-site data led to virtually identical conclusions, suggesting that the two laboratories are in rough agreement.

Grain size testing of soil samples was performed by Engineering Consulting Services, Inc. of Research Triangle Park, North Carolina.

# 4. Performance Assessment

### 4.1 Performance Criteria

Performance criteria are listed in Table 4-1. Criteria were based primarily on performance objectives agreed upon by ESTCP/AFCEE and ARCADIS in the planning stages of the project (see demonstration plan, ARCADIS, March 2000). Other criteria are included in an effort to conform with new reporting guidance issued during the preparation of this report (ESTCP, October 2002).

### 4.2 Performance Confirmation Methods

Performance confirmation methods and brief summaries of results are listed in Table 4-2. Because of the complexity of the demonstration (including the nature of the geology and the biology of CAH biodegradation), results are not easily presented in a table. Moreover, the temporal and spatial relationships of the multiple lines of evidence cannot be fully conveyed in this summary table. Therefore, references to relevant text are included as needed. A comparison of demonstration results with objectives is given in Section 4.3.7.

Data collection methods and data analysis procedures used in this demonstration, including the Quality Assurance Project Plan, were established in the project demonstration plan (ARCADIS G&M, March 2000). (See especially, Section 3.2, Pre-Demonstration Sampling and Analysis; Section 5.4, Sampling Plan; and Section 9.0, Quality Assurance Plan.)

## 4.3 Data Analysis, Interpretation and Evaluation

## 4.3.1 Performance Overview

The demonstration-scale system at Hanscom AFB was operated for two years between October 2000 and October 2002. During that time, the data discussed below demonstrated highly effective, complete TCE removal in a source area that had a long history of fairly stable TCE concentrations before treatment. Evidence of complete treatment – a buildup of ethene, reduction in *cis*-1,2-DCE and no accumulation of VC was also seen in the most effectively treated downgradient wells. Effective treatment was seen only where substantial substrate (molasses and its breakdown products) was observed in downgradient monitoring wells. The layout of the injection and monitoring well system was designed for southeasterly groundwater flow. During the demonstration period, the predominant direction of flow was eastward. Thus, it is suspected that a larger IRZ was formed than what was observed, but that the monitoring well network was not positioned to completely delineate it.

### 4.3.2 Process Monitoring Results and System Operational Adjustments

#### 4.3.2.1 Injection Rates and Field Parameter Observations

Injections of aqueous molasses solution that were begun in October 2000 (see Table 3-6 for a detailed list) were successful in quickly achieving favorable reducing conditions in the lower aquifer, as evidenced by depressed DO and ORP measurements in nearly all downgradient wells after a short period of time (Figures 4-1 and 4-2). Figure 4-3 shows that TOC in the injection well (typically measured immediately prior to the next injection event) remained steady in the desired range from October to May 2001. The pH of the injection well had also markedly decreased (Figure 4-4) which limited the rate of injection. See Sections 1.3.2, 4.1.2.1, 4.5, 5.3 and 6.2 of the protocol (Suthersan, 2002) for a detailed discussion of the causes of pH drop in EARP systems and its management. In brief, the pH of groundwater generally decreases during the injection of degradable organic substrates. The magnitude of the pH decrease depends on the dose of substrate and the natural buffering capacity of the system (both the groundwater and the aquifer solids). This site exhibited relatively low buffering capacity, and pH was controlled by careful carbon dose control and injection of a clean water "push" following reagent injection to disperse the dose away from the immediate vicinity of the well. As shown in Figure 3-15, the injection rate during this initial period (October 2000 – February 2001) was approximately 80 lbs of substrate per week. Thus the delivery of reagent (as indicated by only slightly elevated TOC and Bromide) to downgradient wells was inadequate after 2-3 months of such an injection regimen.

Thus, the injection frequency was increased to roughly a biweekly schedule and each molasses injection was followed by a water push (an injection of just water without diluted molasses) beginning February 2001. This revised dosing regime (averaging around 150 lbs/week as shown in Figure 3-15) greatly improved the distribution of reagent in roughly five weeks, delivering TOC and DOC to monitoring wells RAP1-6T (installed in the lower aquifer) and IRZ-1 starting in late March 2001 (Figures 4-5 and 4-6). As discussed in Section 4.3.4.2, a variety of methods were used to estimate velocity at this site, which suggested groundwater travel times on the order of 40 to 50 days to the initial transect of monitoring wells.

The reactive zone continued to be monitored and beginning in September 2001, the reagent dosage was doubled for most events, while still keeping a bi-weekly injection frequency and water push, in order to expand the size of the reactive zone and in response to increased groundwater flow (as shown in Figures 4-7 and 4-8, gradients increased in early summer 2001 and stayed relatively high throughout the demonstration) and slightly increased DO. As shown in Figure 3-15 the injection rate during this period (September 2001 through February 2002) was in the range of 150-250 lbs/week. It is important to note that despite the increased injection rate during this period the TOC at the most impacted pair of downgradient monitoring wells dipped (Figure 4-5). The reasons this occurred is discussed in Section 4.3.4.

The injection rate was reduced somewhat between March and September 2002 in response to diminished pH (frequently in the range of 3.9-4.0). As shown in Figure 3-15 the injection rate during this period was in the range of 100-200 lbs/week. However, the observed TOC in monitoring wells RAP1-6T and IRZ-1 (Figure 4-5) remained fairly high during this period.

Observations made between September 16 and 18, 2002, showed high DO in the monitoring wells (Figure 4-1) and TOC toward the low end of the desirable range (Figure 4-3). Moreover, as discussed in Section 4.3.4 potentiometric surfaces (Figures 4-23 through 4-26) indicated that the system was just recovering to a normal flow pattern following a series of outages in the pump and treat system that had likely altered the flow direction. Therefore the substrate injection rate was increased for a month to its highest level during the demonstration (Figure 3-15) in order to quickly restore the system to normal conditions.

At this site a fairly regular injection frequency has proven to be effective given its relatively porous geology and relatively high groundwater velocities, though injection well pH was monitored before each injection to determine if an injection could be done without excessive pH drop. The pH in the injection well held fairly steady between 3.9 and 5.5, near the low end of the acceptable range. However, the pH in all of the monitoring wells was basically unaffected by injection events typically holding between 6.0 and 7.5, regardless of whether they received significant substrate as indicated by TOC.

# 4.3.2.2 Changes in Groundwater Flow

Seasonal variations in the direction and magnitude of the groundwater gradient were observed (see Figures 3-8 and 4-9 through 4-27 for potentiometric surfaces as well as gradient plots in Figures 4-7 and 4-8). These appear to be caused at least in part by operational problems with the Base's pump and treat system (Figure 4-28) and variations in precipitation (see Figure 4-29).

These variations in magnitude and direction of gradient appear to have changed the size and shape of the reactive zone markedly. As shown in Figures 4-5 and 4-6, there were two periods where fairly strong doses of reagent were delivered to wells IRZ-1 and RAP1-6T, from late March 2001 through July 2001 and again from March 2002 through September 2002. These variations in hydrogeology and reactive zone shape are discussed more completely in Section 4.3.4.

# 4.3.2.3 TOC, DOC, and Other Measures of Reactive Zone Influence

In addition to TOC and DOC, several other measures can be used as indicators of reagent delivery and the consequent creation of the reactive zone: concentration of the bromide tracer, specific conductance and visual and visual/olfactory observations of groundwater. Bromide tracer shows a pattern essentially similar to that of TOC (primarily substrate), suggesting that the TOC was not significantly more sorbed to the soil matrix than the "nonreactive" bromide tracer at this site. As discussed above, the soil TOC is low at this site so this observation is reasonable. Specific conductance in IRZ-1 and RAP1-6T also shows two peak periods from March 2001 through early fall 2001 and again from later March 2002 through fall 2002. It is interesting to note that the peak in the conductance plot is less sustained for RAP1-6T than for IRZ-1 in both cases. This suggests that RAP1-6T was toward the fringe of the reactive zone formed. A similar trend is seen in the TOC/DOC data although the data points were less frequently obtained for those parameters.

Visual and olfactory observations were recorded periodically when groundwater samples were extracted from monitoring wells. Color, odor, turbidity, and other qualitative observations about the samples were documented. These observations were standardized during data entry into

numbers rating the observed level of reagent impact implied by the sample characteristics in an effort to present the qualitative data graphically. The numeric rating system utilized a scale from 0 to 5, with 0 (actually 0.1 so that the measurement would show up graphically) corresponding to no impact whatsoever and 5 corresponding to undiluted molasses injection solution. The numerical ratings were as follows:

0.1	No Influence	No color, odor, turbidity or other qualitative observations to suggest an impact
1	Slight Influence	One of the following: slight color, slight odor, or turbidity
2	Significant Influence	Two or more of the following: slight color, slight odor, or turbidity
3	High Influence	One or more of the following: moderate-to-strong color, odor, and/or turbidity
4	Very High Influence	Strong color and odor
5	Molasses Solution	Theoretical maximum. Of course, there will always be some dilution; thus this rating was never assigned to a sample.

Although not exactly in agreement this data set also suggests peaks in spring and summer 2001 and 2002 for reagent impact at IRZ-1 and RAP1-6T.

It is noted that the measured TOC values were not always higher than corresponding DOC values. Evaluations of field and laboratory data were undertaken in an attempt to explain this discrepancy (see Appendix A-6c). However, no universal explanation was found. Since these data were used only as process monitoring parameters, the discrepancy is not considered critical.

## 4.3.2.4 Bromide Tracer Data

The bromide tracer data (Figure 4-55 and Table 4-3) shows that:

- Bromide was undetected in all wells except the upper aquifer well in the background monitoring round
- Bromide was undetected in 9 straight monitoring rounds in the upgradient well B239
- High concentrations of bromide were detected only in RAP1-6T and IRZ-1 and generally coincide with other indicators of reactive zone influence such as TOC, DOC, BOD and COD (Figures 4-5, 4-6 and Table 4-4).
- Other than a brief period between late March 2001 and mid July 2001 all of the collected bromide data show that the injected fluid constituted less than 2% of the water sampled at the monitoring wells.
- Bromide concentrations increase gradually with time after the first two sampling rounds done on well IRZ-5 (recall that this well was installed after the demonstration began to better define the northeast edge of the reactive zone). TOC and DOC show a similar trend (Figures 4-5 and 4-6) suggesting that this well was receiving some influence at the far downgradient edge of the reactive zone by the end of the demonstration.

- Wells IRZ-2, IRZ-3, and IRZ-4 show erratic, low but often-detectable bromide concentrations. This when coupled with similar DOC, visual and olfactory data (Figures 4-6 and 4-45) may suggest that these wells were on the far fringes of the reactive zone and thus may have received occasional very dilute influences from the injection well.
- RAP1-6S in the upper aquifer begins to receive trace bromide starting in November 2001. This suggests some leakage or mixing into the upper aquifer and will be discussed more in Section 4.3.6.1.
- RAP1-6R in the bedrock aquifer was the only well to have detectable bromide before injection (perhaps due to differences in geochemistry). This well has no clear trend in bromide concentration during the demonstration.

# 4.3.3 CAH Treatment

CAH data for the three full groundwater monitoring rounds are summarized in Tables 4-5, 4-6, and 4-7.

# 4.3.3.1 Summary of CAH Treatment Results

Our discussion of CAH treatment will focus primarily on the two monitoring wells that received substantial doses of substrate TOC - RAP 1-6T and IRZ-1 (Figure 4-5) - and secondarily on the injection well. These two monitoring wells were also the only ones in the lower aquifer where substantially increased levels of methane were observed (Figure 4-30). Although reducing conditions as evidenced by reduced DO and ORP were observed at other wells, we would not expect from our theoretical understanding to observe treatment in the absence of delivered substrate (Suthersan, 2002).

The best treatment results were observed at IRZ-1 (approximately 45 feet downgradient) (Figure 4-31). At this well highly effective treatment of TCE was observed beginning in March 2001, 5 months after injections began and shortly after single injections with water pushes began (>95% reduction vs. pretest concentrations). Substantial treatment of *cis*-1,2-DCE (eventually >85% reduction in pretest concentration) was not observed until March 2002, a year later, during a second period of high TOC delivery. By March 2002, complete degradation was evidenced by the substantial increase in ethene production (Figure 4-32). The rate of ethene production continued to climb through the end of the demonstration in October 2002, indicating that treatment effectiveness continued to increase after two years of system operation. Ethene concentrations at this well increased to more than 20 times the pretest value.

The layout for the demonstration was designed to be centered around monitoring well RAP1-6T (approximately 40 feet downgradient), which had 14.5 years of relatively stable TCE, DCE and VC results before the demonstration began (Figure 4-33). A sharp decline in TCE was observed within 6 months of the beginning of injection, coinciding or slightly preceding the appearance of substrate as measured by increased TOC and conductance (Figure 4-34). TCE levels for seven straight monitoring rounds (between 3/26/01 and 9/7/01) were at 10% or less of the average of the proceeding 10 years (Figure 4-35). Then, for reasons discussed in detail below, the IRZ shifted away from this well as evidenced by the significant decline in TOC concentrations at this well. As a result of this shift, TCE, DCE and VC concentrations rebounded. These concentrations dipped again when substrate levels increased, although it appears that the typical

groundwater flow direction was somewhat more easterly rather than northeasterly as planned. Thus this well was probably at the fringe of the effective reactive zone for much of the demonstration period.

Due to fluctuations in the groundwater flow direction, rebounds were observed at IRZ-1 and RAP1-6T in the fall and winter of 2002. Concentrations declined again after substrate was restored in spring of 2002.

Although it is by definition a less accurate measure of the overall effectiveness of the reactive zone, it is useful to note that substantial evidence of effective treatment of all chlorinated species was seen at the injection well, even in data corrected for the dilution effect of the injected solution. Concentrations of TCE and VC were reduced by more than 95% for TCE and 85% for VC over a long period from May 2001 through the end of the demonstration in October 2002. DCE decreased substantially less (at most about 75%). This suggests that although CAHs were being completely degraded, desorption from a localized source area continued (Figure 4-36). This difference in temporal trends among the CAH constituents confirms that a degradation effect (and not merely dilution) is present.

The wells that did not get substantial, consistent doses of substrate showed no evidence of treatment or at most modest decreases in TCE only (IRZ-2, IRZ-3, IRZ-4 and IRZ-5 in the targeted lower aquifer, RAP1-6R in the bedrock aquifer and RAP1-6S in the upper aquifer). This strengthens the conclusions that substrate availability is linked with improved biodegradation and that contaminant removal was due to enhanced biodegradation rather than displacement.

## 4.3.3.2 Methodologies Used in CAH Data Analysis

We used several different approaches to understand the CAH data set from this demonstration. We had available CAH data from three primary sources:

- a) sampling conducted by ARCADIS using high QA/QC methods with off-site GC-MS analysis by STL (1 round pre-demonstration and 7 rounds during the demonstration)
- b) sampling conducted before and during the demonstration by ARCADIS or the Base's contractor (IT) with analysis conducted using an on-site GC by IT (as many as 12 additional rounds during the demonstration with many more prior to the demonstration)
- c) periodic compliance monitoring of some wells conducted before and during the demonstration by the Base's contractors using various off-site laboratories (15 or more rounds).

Between all these sources some of the wells had a wealth of data – as much as 65 samples over 16 years for RAP1-6T (of which 26 were during the treatment period)!

Various approaches were used to graphically review and interpret the data:

- a) plots of CAH concentration vs. time, with indications of the injection times
- b) plots of CAH concentration corrected for dilution as measured by the observed concentration of bromide tracer

- c) bar-line plots of CAH concentration compared to the concentration of indicators of substrate delivery such as TOC or conductance
- d) plots of CAH concentration on a molar rather than mass/volume basis
- e) plots of an "MCL index" (Payne, 2002) designed to measure overall progress toward reduced risk from CAHs. Note that this index should not be construed to imply that risks from CAHs are necessarily additive.

The significance of changes in concentration was determined primarily by comparisons to pretest concentrations, or in some cases, in comparisons to a level baseline that was observed before and for a period after injection before biodegradation evidently began. Although one well upgradient of the injection well (B239) was routinely sampled, it didn't turn out to be suitable for upgradient and downgradient concentrations for several reasons:

- a) pretest data and modeling by a previous consultant strongly suggested that the demonstration zone was in a source area
- b) concentrations at this well were relatively low and erratic for several years prior to and during the demonstration
- c) B239 is proximate to extraction well #6 which may exert a localized influence on it

It was judged impractical to install an additional upgradient well between IRZ-INJ and B239 because of the intervening active runway.

In most cases sufficient data was available that the presence or absence of trends in pre- and post- treatment concentrations could be readily discerned by inspection. In key cases, the students t-test was used to verify these conclusions. Multiple lines of evidence were also sought to confirm that biodegradation was being enhanced (NAS, 2000):

- a) evidence that biogeochemical conditions were appropriate for anaerobic biodegradation was obtained and reviewed
- b) the trends and time sequence of product production (i.e., DCE, VC, and ethene) were sought to verify that observed decreases in target compounds were not due to dilution and fit with the current theoretical/laboratory understanding of CAH biodegradation
- c) trends in historical data were compared inside and outside of the reactive zone
- d) tracer (bromide) corrections were used to verify that dilution was not the primary cause of observed trends
- e) downgradient and side-gradient well data were reviewed to rule out displacement effects

More details about particular methods of data interpretation are provided in the sections that follow.

## 4.3.3.2.1 <u>Dilution Calculations Using Tracer</u>

We have assumed that bromide tracer used is indeed conservative and does flow along with and at the same rate as the injected liquids. Thus the amount of dilution in a given well should be directly proportional to the amount of bromide that shows up in that well. For clarification, as an extreme example, if 100 mg/L of bromide were injected into the injection well and then 100 mg/L bromide were withdrawn a week later at a monitoring well located one-week downgradient of the injection well, we would surmise that the monitoring well contains only injection fluid. Likewise, if 0 mg/L bromide were measured in the monitoring well, none of the water from the injection well has made it to the monitoring well and there is thus no dilution.

Based on this logic, VOC measurements were corrected by calculating a dilution factor per the following equation:

$$DF = 1 - \frac{[Br]_{_{MW}}}{[Br]_{_{Inj}}}$$

Where,

- DF = dilution factor (that is, the percentage of monitoring well sample that is not dilution water),
- $[Br]_{MW}$  = the bromide concentration measured on a sample taken from a monitoring well on a given date, and
- [Br]<sub>Inj</sub> = Injection fluid bromide concentration (an approximate running average of injection fluid bromide concentrations made around the assumed travel time of the injection well to the monitoring well). This takes into account both the molasses solution and the water "push" fluids.

Then, the VOC concentration for a given monitoring well sample was corrected by calculating its actual concentration per the following equation:

$$VOC_{actual} = \frac{VOC_{measured}}{DF}$$

As an example, consider the Br concentration measured on a sample from RAP1-6T on 4/6/01. The measured Br in RAP1-6T was 21.4 mg/L, and the average injection concentration was determined to be 60 mg/L. Thus the dilution factor was: 1 - 21.4/60 = 0.643. The TCE concentration from this sample was measured to be 5.0 mg/L. Therefore, the actual, undiluted concentration should be 4.8 / 0.643 = 7.8 mg/L.

4.3.3.2.2 Bar-Line Plots of Substrate Dose vs. CAH concentration

We have prepared bar-line plots (Figures 4-31 and 4-34) to graphically show how the substrate dose (as measured by TOC and specific conductance and shown by the bars) affects the key VOCs (TCE and DCE shown by the lines).

# 4.3.3.2.3 <u>MCL Index</u>

The MCL index was developed to represent the cumulative effect of MCL exceedances for multiple compounds. Over a series of monitoring events, the index provides a sense of the overall progress of remediation relative to groundwater remediation goals (if based on MCLs) or drinking water toxicity (the basis for MCLs).

The index normalizes concentrations of all constituents to show the cumulative effect of exceedances of MCLs for multiple compounds. For instance:

MCL Index = 
$$\frac{PCE}{MCL_{PCE}} + \frac{TCE}{MCL_{TCE}} + \frac{cisDCE}{MCL_{cisDCE}} + \frac{VC}{MCL_{VC}}$$

For the Hanscom AFB site, only the primary detected CAHs were included in MCL Index calculations. Note that the use of this index should not be taken to imply that the risk from these compounds is necessarily additive.

4.3.3.2.4 <u>Rate Calculations</u>

The methodology for the rate calculations is discussed in Section 4.3.3.5.

### 4.3.3.3 Detailed Discussion of Observations at Individual Wells

### 4.3.3.3.1 CAH Data at Well IRZ-1

All of the available CAH data for this key well is plotted on a concentration basis in Figure 4-37. The initial concentrations of TCE, DCE and VC are reasonably consistent with those in neighboring wells measured in the same sampling round (Figure 3-14). Those neighboring wells include RAP1-6T, which has a long history of concentration stability. Thus although only one pretest round was available at IRZ-1, we can be reasonably certain that this range of CAH concentrations is representative.

TCE concentrations in IRZ-1 are erratic for the first several months after injection begins, swinging both substantially above and below the baseline. This may represent an interplay between not fully established enhanced biodegradation processes and enhanced desorption processes (see Section 4.3.1 of Suthersan, 2002 for a discussion of desorption effects in ERD systems). Then four sampling rounds from March 26, 2001 to July 11, 2001 showed dramatically lower TCE concentrations. These rounds included both on-site and off-site analyses. Essentially the same trend is shown in the high QA/QC offsite laboratory only data set (Figure 4-38) and in the dilution-corrected plot (Figure 4-39). As discussed previously, the bar-line plot (Figure 4-31) shows that this dip coincided with an increase in the presence of substrate, as indicated by TOC and specific conductance. During this March to July 2001 period DCE is stable, VC may be slowly declining (Figure 4-37), and methane (Figure 4-31) and ethene (Figure 4-32) are low and only a trace of hydrogen sulfide was observed (Figure 4-40). DO (Figure 4-1) and ORP (Figure 4-2) had substantially declined. These data taken together suggest enhanced degradation is occurring, but thermodynamically favorable electron acceptors have not been consumed to allow complete degradation to ethene.

Sometime between July and October 2001 the concentration of substrate at IRZ-1 as indicated by TOC and specific conductance dipped markedly which resulted in a rebound in the concentrations of most CAH analytes, especially TCE (see Figures 4-3, 4-31, and 4-37). This appears to have been due to changes in flow conditions, since the injection rate of substrate remained the same or even increased.

TOC and specific conductance began to increase, and TCE and VC decreased again between February and October 2002 (Figure 4-37). However, unlike the earlier period of high substrate concentration during this period, *cis*-1,2-DCE dramatically decreased! Concentrations of this analyte had been between 2600 and 4600 µg/L in 17 straight analyses between June 2000 and January 2002 were less than 1,000 µg/L in four of five monitoring rounds between February and October 2002. Ethene concentrations (regarded as the best indicator of complete CAH biodegradation) had not increased substantially by November 2001, but increased significantly by the March and October 2002 monitoring rounds (Figure 4-32). Ethene concentrations at IRZ-1 eventually increased to more than 20 times the pretest value. Methane concentrations that had begun to increase noticeably in November 2001 remained steady through the later portion of the test at a concentration of about 2 mg/L. Dissolved oxygen was typically less than 1 mg/L and ORP was typically between -150 mV and -200 mV during that period. As shown in Tables 4-5, 4-6, and 4-7, sulfate levels had substantially diminished in this well by May 2001 and were also low in October 2002. With the detection of trace sulfide, this suggests the conditions were at least in part, sulfate reducing. Although this is slightly above the range (below –240 mV) typically quoted for methanogenesis (Morin and Henry, 1998) the measurement of ORP is notoriously inaccurate in groundwater systems due to lack of equilibrium among other causes (Lindberg and Runnells, 1984). Another possible interpretation is that the methane was generated and the CAHs degraded somewhere upgradient of IRZ-1 where the ORP is lower. Thus, multiple lines of evidence suggest that complete degradation of CAHs was achieved upgradient of IRZ-1 under sulfate reducing/methanogenic conditions.

Analysis of the dataset using the off-site laboratory data only (Figure 4-38) and the dilution corrected data (Figure 4-39) supports the same conclusions.

#### 4.3.3.3.2 CAH Data at Well RAP1-6T

The RAP1-6 cluster was the only preexisting well in the immediate demonstration area. Conductance (Figure 4-41), TOC (Figure 4-5) and DOC (Figure 4-6) all suggest that the periods of elevated substrate concentration at this well were less sustained than at IRZ-1. As discussed more completely in Section 4.3.4, it appears that the typical groundwater flow direction was somewhat more easterly rather than southeasterly as planned. Thus this well was probably at the fringe of the effective reactive zone for much of the demonstration period. As shown in Figure 4-33, TCE and VC concentrations in monitoring well RAP1-6T immediately before the demonstration started in 2000 were essentially unchanged from the first measured values in 1986. Although a pump and treat system at the site had been operated for much of that time and other remedial measures were implemented which dramatically reduced concentrations in other areas of the site, they had little effect at RAP1-6T (see Figure 4-33, see also Hanscom AFB, 2002). The data shown on Figure 4-33 suggests that 1,2-DCE measured prior to 1998 is primarily *cis*-1,2-DCE. Based on this there is no clear trend in the DCE data except that it generally remained between 3,000 and 6,000 µg/L.

The concentration of TCE and DCE appeared to rise slightly in the fall of 2000 immediately after injection began on October 11<sup>th</sup>, which could possibly be a desorption effect. However, since the rise was slight and occurred so rapidly the existence of a desorption effect in this instance is questionable.

A sharp decline in TCE and a lesser decline in DCE was observed beginning in mid-November 2000 one month after injection and continued until September 2001, eleven months after the beginning of injection (Figure 4-35). This decline coincided or slightly preceded the appearance of substrate as measured by increased TOC and conductance (Figure 4-34). TCE levels for seven straight monitoring rounds (between 3/26/01 and 9/7/01) were at 10% or less of the average of the proceeding 10 years (Figure 4-35). Since VC was not measured by the on-site GC, there is less data for this compound, however it appeared to have a coincident, but slower decline during a series of four monitoring events from November 2000 to July 2001.

Then, for reasons discussed in detail below beginning in September 2001, substrate delivery to this well decreased markedly which led to a rebound in TCE, DCE and VC concentrations that peaked in November 2001. TCE concentrations dipped again from January to October 2002 in the presence of increased substrate levels although DCE and VC concentrations were too erratic to allow firm conclusions during that period. However, ethene at this well increased gradually and fairly steadily during the demonstration (Figure 4-42), with its most significant increase occurring in 2002. However, ethene never reached the high levels seen at IRZ-1.

The conclusions discussed above about trends in CAH data at this well are unchanged when the data is examined using the smaller off-site laboratory samples (Figure 4-43) or dilution-corrected (Figure 4-44) datasets.

ORP during most of the demonstration period ranged from -50 to -200 mV at this well (vs. 2 mV pretest) and DO was almost continuously held to less than 1 mg/L until October 2002. Modest increases in hydrogen sulfide (Figure 4-40) and decreased sulfate (Tables 4-5, 4-6, and 4-7) were observed in several monitoring rounds beginning in May 2001. Although methane concentrations increased in each monitoring round, increases to levels dramatically above those seen in any well in the background monitoring round did not occur until between November 2001 and March 2002. March 2002 was the peak for methane at this well and it had diminished some by October 2002 (although it was still substantially above background values). Taken together this data suggests that the predominant microbial processes in the vicinity of this well were probably iron reducing or denitrifying until May 2001, sulfate reducing in summer and fall 2001, and did not become methanogenic until the end of 2001, more than a year after injections began. Although very good TCE treatment was achieved at this well, more sustained delivery of substrate would probably be necessary for optimum rates of complete degradation to occur.

## 4.3.3.3.3 CAH Data at the Injection Well IRZ-INJ

Although it is by definition a less accurate measure of the overall effectiveness of the reactive zone since it is more prone to dilution effects and other artifacts, we discuss here the injection well data because:

- Since sampling was done before rather than immediately after injection events the water sampled from the injection well should represent the groundwater at the upgradient end of the reactive zone, where substrate has mixed at high concentration with the formation water
- Dilution correction using the bromide tracer can help estimate the effects of dilution
- The analysis of this data is similar to the conduct of field push-pull pilot tests which have been widely recommended.

Substantial evidence of effective treatment of all chlorinated species was seen at the injection well, even in data corrected for the dilution effect of the injected solution. Concentrations of TCE and VC were reduced by more than 95% for TCE and 85% for VC over a long period from May 2001 through the end of the demonstration in October 2002. DCE decreased substantially less (at most about 75%). The sequence of observed substantial concentration decreases was TCE first, then VC, followed by DCE. The continued presence of DCE suggests that although CAHs were being completely degraded, desorption from a localized source area continued (Figure 4-36) through the end of the demonstration (since the low concentration at B239 discussed below suggests that upgradient inputs were small).

As shown in Figure 4-3, TOC levels in the injection wells were maintained at a high level fairly consistently throughout most of the demonstration, which is logical since the injection well area is much less subject to variations in flow direction than the downgradient wells. The one major dip in this concentration on July 11/12, 2001 coincided with peaks in the DCE and VC concentration trends (Figure 4-36). Visual and olfactory observations (Figure 4-45) also suggested that this well was surrounded by a consistent reactive zone. However hydrogen sulfide levels were generally only modestly elevated (Figure 4-40), methane concentration proceeded (Figure 4-46) and ethene levels actually decreased as the demonstration proceeded (Figure 4-47). Data for the very light gasses may be unreliable because this well was described at some times as "frothy" due the effects of substrate. It is also possible that ethene decreased as most of the available CAHs in the immediate area was consumed by biodegradation and rates were limited by mass transfer from a source.

# 4.3.3.3.4 Upgradient Monitoring Well B239 CAH Data

When the demonstration zone was laid out, this well (which was present before the demonstration) was intended to serve as an upgradient control well since it lies between the primary Site 1 source and the demonstration zone. CAH data from this well is shown in Figures 4-71 through 4-73. Though this well is indeed hydraulically upgradient of the demonstration zone, and it lies just opposite the runway from the injection well, the data shows it has much lower and more erratic levels of CAHs than wells within the demonstration zone. Potentiometric surfaces (i.e., Figures 3-8, 4-11, 4-12) suggest that this well, though downgradient of extraction well BIW-6, may at some times be within that well's zone of influence and is certainly benefiting from the remediation influence of that well. Data acquired during this demonstration suggest the presence of another CAH source between B239 and RAP1-6T (see Section 3.4). While classified as a lower aquifer well the screen in this well is somewhat shallower than the other lower aquifer wells used in this demonstration (see Figure 3-12). These data taken together suggest that pump and treat has been successful in significantly reducing observed concentrations immediately to the northwest of the runway but has not yet reached MCLs and may have reached an asymptote.

# 4.3.3.3.5 CAH Data at all Other Monitoring Wells

With only a few exceptions CAH concentrations at the other wells monitored for this demonstration showed no evidence of treatment or at most modest decreases in TCE only (IRZ-2, IRZ-3, IRZ-4 and IRZ-5 in the targeted lower aquifer, RAP1-6R in the bedrock aquifer and RAP1-6S in the upper aquifer). These data have been reported in:

- Figure 4-48 for IRZ-2,
- Figure 4-49 for IRZ-3,
- Figure 4-50 for IRZ-4,
- Figure 4-51 for IRZ-5,
- Figure 4-52 for RAP1-6R in the bedrock aquifer, and
- Figure 4-53 for RAP1-6S in the upper aquifer.

These wells were generally outside of the IRZ. Some exceptions to this are that IRZ-5 appeared to be on the very downgradient edge of the IRZ near the end of the demonstration. Similarly, RAP1-6S shows some influence late in the demonstration. This is discussed in Section 4.3.2.4. The following observations were made for completeness:

- VC in well IRZ-2 which appears to show a consistent although modest downward trend until the last data point in November 2002 when it rebounded to initial conditions
- TCE in IRZ-3 declines about 50% in the spring and summer of 2001, then rebounds in the fall of 2001, and remains fairly steady throughout the remainder of the demonstration period
- Ethene production in IRZ-5 appears to steadily increase after this well was installed, after the demonstration had already begun (Figure 4-47)
- TCE in RAP1-6R declined slowly during the demonstration. However, an examination of the ratio of TCE to DCE (on a concentration basis) both before and after injections begin suggests that a trend that was ongoing was little changed by the demonstration (Figure 4-54).

As seen in the TOC (Figure 4-5), DOC (Figure 4-6), conductance (Figure 4-41), bromide (Figure 4-55) and visual/olfactory observations (Figure 4-45) plots none of these wells had consistent evidence of substantial substrate delivery. However DO and ORP did appear to decrease in many of these wells (Figures 4-1 and 4-2). This strengthens the conclusions that substrate availability is linked with improved biodegradation. Since these wells were above, below, upgradient and side gradient from the wells that showed clear evidence of biodegradation (IRZ-1 and RAP1-6T) this strengthens the conclusion that contaminant removal at those three wells was due to enhanced biodegradation rather than displacement or dilution.

# 4.3.3.3.6 <u>MCL Index</u>

The MCL index calculation (Figures 4-56 through 4-59) of course mirrors the VOC concentration plots – showing risk reductions were achieved for TCE in all wells and for DCE in IRZ-1 and IRZ-INJ. But they emphasize the importance of VC. Although VC did not increase during the demonstration and actually decreased somewhat at IRZ-1 and substantially IRZ-INJ it is typically the last of the chlorinated ethenes to degrade under anaerobic conditions. It is also important to note that VC is the most readily degraded under aerobic conditions of the chlorinated ethenes so would be expected to degrade outside the reactive zone.

# 4.3.3.3.7 <u>Statistical Analysis of CAH Data</u>

A one-sided Student's t-test was conducted to compare concentrations of CAHs before the demonstration to those during the demonstration. The t distribution is assumed to be normal, and

the variances of the two samples are assumed not to differ significantly. Using data for RAP1-6T, all available data before October 11, 2000 were compared to all data collected after October 11, 2000. For IRZ-1, post-demonstration data starting in late 2001 through October 2002 were used instead of all post-October 2001 data, to reflect the daughter product response to methanogenic conditions. Where multiple analyses were conducted on one sample on the same day (e.g., when both on-site and traditional laboratory testing were conducted on the sample), results were averaged for a single data point.

Well	САН	Data Set	Mean Conc. (µg/L)	Std. Dev.	Sample Size	Т	Probability
RAP1-6T	TCE	Before	1289	615	24	4.75	1.2E-05
	TCE	After	444	528	19		
	Cis-DCE	Before	3763	1363	24	1.45	0.0769
	Cis-DCE	After	3230	937	19		
	VC	Before	955	373	14	1.92	0.0346
	VC	After	679	272	9		
IRZ-1	TCE	Before	492	646	2	0.89	0.2049
	TCE	After	238	252	6		
	Cis-DCE	Before	3219	54	2	1.11	0.1540
	Cis-DCE	After	1922	1562	6		
	VC	Before	1100		1	0.72	0.2732
	VC	After	831	323	3		

Statistics for each data set were as follows:

The last column indicates the maximum probability that the observed change in concentration was due to random variation. We note that a simple t-test is probably not the best statistical approach to this problem, because the conditions that impact treatment efficiency (i.e., groundwater flow direction and thus substrate delivery) are frequently changing.

#### 4.3.3.4 Analysis of CAH Data: Conditions Required for Enhanced Biodegradation

The bar-line plots for IRZ-1 and RAP1-6T (Figures 4-31 and 4-34) demonstrate that the VOC concentrations decrease when substrate concentration increases as evidenced by TOC and specific conductance. Attempts to plot VOC concentrations as a function of TOC (not shown for brevity) did not yield a linear correlation. But a very clear correlation was observed between specific conductance (another measure of substrate influence) and TCE degradation in these two monitoring wells (see Figure 4-60).

This correlation between TCE degradation and substrate availability clearly does not hold for DCE degradation. This suggests that substrate is a necessary but not sufficient condition for enhanced biodegradation. For example from inspection of the bar-line plot for IRZ-1 (Figure 4-31) it is notable that good TCE treatment occurred in the spring and summer of 2001 but good treatment for both TCE and DCE did not occur until the spring and summer of 2002, despite similar TOC and specific conductance levels. Moreover there are no significant differences in pH, DO and ORP for well IRZ-1 between those time periods - but there is a significant differences in a suggests that it took almost a year of regular substrate injections and system adjustments to overcome the oxidative poise of more preferential electron acceptors (Suthersan,

2002) and reach methanogenic conditions at or upgradient of IRZ-1. This result is also in keeping with our theoretical understanding that enhanced biodegradation, especially the biodegradation of DCE is more favored under methanogenic conditions than sulfate-reducing conditions (Suthersan, 2002).

At RAP1-6T elevated methane levels were observed during the last two sampling periods; however this well never showed irrefutable evidence of DCE treatment (although there is some evidence). However as previously discussed RAP1-6T is on the fringe of the IRZ and did not experience the same sustained substrate loading IRZ-1 received.

Taken together this data set is consistent with a theoretical understanding in which both consistent substrate dosing and methanogenic conditions are required for optimal treatment (Suthersan, 2002). Bioremediation of CAHs can proceed at pilot- and full-scale via the formation of distinct subsurface zones characterized by a predominant terminal electron accepting process. Electron donor injected into the subsurface travels at a site specific rate downgradient while being utilized by differing bacterial communities which develop in response to the frequently varying, microbiologically processed electron donor(s) passing through them. Thus, starting with ERD implementation, what was likely a relatively uniform indigenous bacterial community diverges into "banded" zones characterized by differing dominant terminal electron acceptor processes. For example, near the line of injection wells, the bacterial community in an anoxic aquifer might be dominated by iron-reducing or nitrate-reducing bacteria that can utilize the sugars present in carbohydrate substrates immediately and are continually provided with an acceptable terminal electron acceptor by upgradient groundwater re-supply. Sulfate reducing and methanogenic and potentially dehalorespiring zones form further downgradient. Even further downgradient once all the substrate is consumed conditions gradually return to the ambient redox state. Dojka et al. documented this zonation concept on a vertical basis rather than a lateral basis at a site where a fuel hydrocarbon plume and a CAH plume commingled (Dojka et al., 1998).

As upgradient electron donors are depleted, downgradient zones are selected for and sized based on the presence of the next most energetically favorable electron acceptor available in the groundwater (the immediate area around the injection well itself is usually characterized by a very low redox condition). Downgradient zones can vary in their redox condition with the tendency being for increasingly negative redox values as distance from the injection well line increases. Even further downgradient all of the substrate is consumed, and thus the redox potential begins to increase again with distance. Redox is known to influence the degradation of lesser-chlorinated CAH daughter products whose accumulation would likely lead to a conclusion that the bioremediation effort is stalled at *cis*-1,2-DCE. One interpretation that can be drawn from the recent ITRC literature summary in Table 4-8 is that PCE and TCE can be reductively dechlorinated at higher redox values than are required for the successful reductive dechlorination of *cis*-1,2-DCE and VC. Therefore, the addition of sufficient carbohydrate substrate to drive redox values into the methanogenic or sulfate-reducing range in bacterial zones distant from the line of injection wells is required to achieve complete biodegradation.

#### 4.3.3.5 Calculated Biodegradation Rates

In order to quantify the rate of decrease of constituents of concern (COC) during the demonstration, first-order attenuation rates were calculated for TCE, *cis*-DCE and VC using

exponential regression methods. The first-order attenuation rate is described by the following relationship:

$$y = y_o e^{-kx}$$

where

 $y_o$  is the initial constituent concentration y is the constituent concentration at time x and k is the first-order degradation rate constant.

To account for decreases in concentration caused by dilution and dispersion, the data were normalized using bromide tracer data (see Section 4.3.3.2.1 for an explanation of dilution calculations). The use of a normalized data set results in a more conservative estimate of attenuation than would the raw data, and the resulting k can be attributed principally to biodegradation.

Rates were calculated at three well locations: IRZ-1 and RAP1-6T, the nearest downgradient monitoring wells screened within the injection interval, and at IRZ-INJ, the injection well. Bromide tracer data allowed for estimates of dilution even at the injection well, making it a reasonable monitoring point for evaluation of biodegradation rates. Initial concentrations were represented by an average of data collected within one year prior to the first molasses injection.

For each well, normalized concentration data were plotted versus time. An exponential regression was then fitted to a selected time interval, yielding an estimate of k and a correlation coefficient ( $R^2$ ) for that interval.  $R^2$  measures how well the regression equation represents the trend in the data.  $R^2$  values range from 0 to 1; the closer to 1, the more accurate the representation. From the first-order rate constant, the attenuation half-life was calculated. Time intervals over which rates were calculated were selected to best represent periods of continuous and supportable biodegradation as follows:

- For IRZ-INJ, rates were calculated for the entire demonstration period, starting with the first injection in October 2000 (Figure 4-61). This approach is conservative because it incorporates the initial equilibration and desorption phases. A more detailed breakdown of the injection well data was not attempted because the available VOC data with associated bromide data (depicted in Figure 4-36) does not clearly define the equilibration period and desorption peak.
- For IRZ-1, TCE concentration decline attributable to biodegradation occurred primarily in the early part of the demonstration following the initial desorption peak, and ending in July 2001 (Figures 4-31 and 4-37 show the data and 4-62 the rate calculation). After this date, an apparent change in the flow field caused a disruption of the substrate delivery to IRZ-1 (see Figure 4-31) and a discontinuity in the VOC trend data. As discussed above *cis*-DCE biodegradation appears to require elevated methane in addition to TOC, which occurs at IRZ-1 starting in late 2001 (see Figure 4-31). The *cis*-DCE rate calculation is therefore based on data from October 2001 through October 2002 (see Figure 4-62). A VC biodegradation rate was not calculated at IRZ-1.

At RAP1-6T, concentration declines in TCE and *cis*-DCE occurred early in demonstration starting at the initial desorption peak and ending in September 2001 (see Figures 4-34 and 4-35 for the data and 4-63 for the rate calculations). After this date, an apparent change in the flow field caused a disruption of the substrate delivery (see decline in TOC levels at RAP1-6T in Figure 4-34) and a discontinuity in the VOC trend data. A VC biodegradation rate was not calculated at RAP1-6T.

Degradation rates were also calculated for the pre-demonstration period using historical data from RAP1-6T (Figure 4-33 shows the data and Figure 4-64 the rate calculations). These calculations generally suggest that biodegradation was very slow or nonexistent before the demonstration. Clearly, biodegradation was occurring in this location prior to the demonstration, as evidenced by the presence of substantial concentrations of TCE daughter products. However, the fluctuations in the historical data also represent continuing influx from a source and changes in water levels, influenced in part by the groundwater pumping system. *Thus this calculation is conservative (underestimates biodegradation) as are all of the calculations in this section because they assume that a fixed amount of CAH is present and degrades. We know from the discussions presented above that a source is existent in this area and thus that CAHs continue to be introduced from a solid phase or free phase source to the dissolved phase. The rate calculation however cannot take this into account because these interphase transfer rates are unknown*.

Results of the rate calculations are summarized in Table 4-9. Rates of degradation during treatment were substantially higher than in the pretreatment period for each of the three compounds considered.

Rates for all three of the compounds were higher at the downgradient wells than at the injection well. This is attributable partly to the time intervals over which the rates were calculated, and also to the repetition of injections. With repeated introductions of fresh substrate at the injection well, some biochemical conditions (e.g., highly reducing conditions including methanogenic) that are created with time (or distance) from the injection well are enhanced in downgradient locations relative to the substrate injection area.

Rates for other ERD sites where similar substrates have been used are provided for comparison. The sites for which data are listed are TCE sites, one with PCE as a parent compound, with degradation products including *cis*-DCE and VC. Total CAH concentrations at the sites ranged from 1.2 to 22 mg/L, under a variety of hydrogeologic conditions. The Hanscom site produced TCE degradation rates that fall within the range given for other sites. *Cis*-DCE rates for Hanscom were slower than for other sites; this is attributed in part to the inconsistent delivery of substrate to downgradient wells.

The rates in Table 4-9 are compared to rates given in Howard et al. (1991) for anaerobic, aqueous biodegradation, or natural attenuation. Calculated half-lives for TCE in the Hanscom demonstration ranged from 28 to 80 days, improving on the published range of 98 to 1653 days given by Howard et al. (1991). Rates for *cis*-DCE and VC were consistent with the published ranges.
Another comparison can be made with TCE attenuation rate constants compiled by EPA (1998) from Aronson and Howard (1997). Figure 4-65 shows the distribution of the rate constants, which represent natural attenuation of TCE as cited in published reports. EPA (1998) notes a likely bias toward high attenuation rates in this data. The Hanscom rates for TCE biodegradation in Table 4-9 are three to nine times the average published TCE natural attenuation rate constant of around 1/year.

## 4.3.4 Hydrogeology Discussion

## 4.3.4.1 Changes in Groundwater Flow Direction due to Changes in Pumping Rates

The ERD at Hanscom exhibited changes in groundwater flow that were not expected and led to complex patterns in substrate delivery and CAH concentration as discussed above. The most probable explanation for these changes is a combination of multiple sources and changing pumping patterns imprinted on top of a complex geology with relatively thin aquifer zones. A thorough discussion of the effect of groundwater extraction on flow direction in the demonstration area is provided in Appendix A-7 and is briefly presented in this section.

Under natural conditions, the flow direction in the three conductive geologic units is generally to the east-northeast (CH2M Hill, 1997). In the design of the pilot test, the influence of the recovery wells was taken into account, and a consistent southeasterly flow direction was assumed. However, flow patterns varied during the course of the demonstration. The variability of flow patterns is attributed to nearby pumping wells and the following hydrogeologic features of the lower sandy till, the unit in which the demonstration was conducted:

- The sandy till, the most productive of the three units, is semi-confined and continuous over the demonstration area as well as a larger area encompassing five production wells (BIW-1 through 4, BIW-6); and
- It is at most about 20 feet thick.

The confinement means that changes in withdrawals are expected to show effects much more quickly than in unconfined conditions. The quickness of reaction over larger areas is the result of the fact that in confined aquifers, unlike unconfined aquifers, the removal of large amounts of water is not needed to affect changes in head over relatively large areas because what is being transmitted is head, not water. In addition, the limited thickness of the till would result in transmittal of head changes to a larger area than in a thicker aquifer.

The locations of the recovery wells relative to the IRZ/ERD area are shown in Figure 3-1. Extraction histories for four of the five recovery wells are graphed in Figure 4-28. The most extreme variations in pumping rates since mid-2000 have occurred in wells BIW-3 and BIW-4. A detailed evaluation of well production and groundwater flow direction in the demonstration shows a correlation between the two, as discussed in Appendix A-7. In brief, changes in flow can be seen are corresponding to the following events:

- The BIW-3 shutdown in September and October 2000, and again in May 2001
- Increased pumping of BIW-4 starting in October 2001
- Decreases in BIW-3 and BIW-4 pumping rates in July through September 2002.

These changes in flow direction no doubt affected substrate delivery and thus treatment efficiency. The basic assumption made in setting up the injection and monitoring wells is that the direction of groundwater flow and hence contaminant and substrate transport is relatively stable in a southeasterly direction. While the reduction in the parent compound, TCE, has generally been favorable in the two monitoring wells that fell within the reactive zone, excursions from that trend occurred in late 2000 and in fall and winter 2001. The late 2000 changes, when a precipitous drop in TCE concentration was followed by a sharp rise (Figure 4-37), occurred when BIW-3 was first shut off, then increased to very high rate of withdrawal. The second occurred primarily in fall and winter 2001 and coincided with the increase in pumpage from BIW-4.

In summary, the changes in pumpage from wells BIW-3 and BIW-4 have induced changes in the flow regime at the IRZ/ERD area in the basal unit that help explain the variations in substrate delivery and treatment efficiency.

## 4.3.4.2 Groundwater Velocity Estimation

Based on preexisting data we estimated the groundwater flow velocity at the site to be 0.8 ft/day (ARCADIS Geraghty & Miller, 2000). Groundwater velocity observed in the demonstration area was calculated using the following three methods. A detailed explanation of each method can be found in Appendix A-7.

- a) Based on average bromide concentrations observed at the first line of observation wells and the mass bromide loading rate at the injection well, a volumetric flow rate through the lower aquifer was estimated and used to derive a flow velocity. A velocity of 0.64 feet/day was calculated using this method.
- b) The bromide arrival time at the first line of observation wells was used to calculate flow velocity. Using this method, the velocity was estimated to be between 0.30 feet/day and 1.57 feet/day.
- c) Measured or estimated values of hydrogeologic characteristics were used to calculate flow velocity based on a variation of Darcy's Law. Using an average hydraulic conductivity of 19 feet/day, the flow velocity was calculated by this method to be 0.76 feet/day.

A discussion of the merits and limitations of each method is included in Appendix A-7. The values derived from the three methods are reasonably close, thus an average velocity during the demonstration period of approximately 0.80 ft/day is assumed.

#### 4.3.5 Secondary Water Quality Issues

We recognize that while the substrate injected (molasses) and its breakdown products are generally nontoxic, it may elevate certain parameters in the water within the reactive zone. For example, by definition, any substrate used to enhance anaerobic bioremediation will elevate the BOD, a traditional measure of water quality. Furthermore, since we are intentionally creating reducing conditions within the reactive zone, this will necessarily alter the geochemistry of the reactive zone. This will make some soil mineral metals more mobile (more dissolved) and others less mobile (more inclined to the solid phase). Further information about these matters can be found in Sections 1.2, 2.1, 3.1 and 7 of the protocol document (Suthersan, 2002). An additional

potential water quality issue, i.e., the metals content of the injected molasses solution, is addressed in Section 4.3.5.2. Although the injectate is not expected to introduce metals in harmful concentrations, the metals evaluations in this section also indirectly address this potential problem.

In general, it is believed that enhanced anaerobic in-situ bioremediation processes will reduce the mobility of many metals (indeed it has been successfully used for the treatment of many) but it will solubilize some other naturally occurring metals in the reactive zone (for example iron, manganese, and arsenic). However, even in solubilized form under anaerobic conditions, metals such as arsenic are substantially retarded by adsorption to the aquifer matrix. Furthermore, it is generally believed that they will be reprecipitated/immobilized downgradient of the reactive zone when the conditions return to their preexisting state (which for the purposes of this discussion is assumed to be aerobic). Similarly, reprecipitation/immobilization will occur within the IRZ area some time after system shutdown. Finally, we note that these reducing conditions are by no means unique to IRZ systems – they occur for example at sites of TPH releases and landfills as well.

Therefore, we agreed with ESTCP on a multi-step process to evaluate the following issues in the context of the Hanscom demonstration. Additional detail regarding each item is provided in Appendix A-8.

- 1. <u>Review existing base monitoring data:</u> Only very limited data were identified, and were deemed to be of little use to this demonstration.
- 2. <u>Review mineralogy</u>: This provides guidance for parameter selection and data interpretation but cannot be definitive.
- 3. Review data gathered to date on iron, manganese, COD and BOD
- 4. <u>Expand the final monitoring round for this project</u> to include total and dissolved metals plus TDS.
- 5. <u>Incorporate analysis of data gathered on a related project:</u> A large pilot scale study and a related simultaneous bench scale column study for Fort Devens, Massachusetts. Fort Devens and Hanscom are approximately 25 miles apart but their mineralogy may be somewhat different. The primary focus of the bench scale column study is the mobilization of arsenic during treatment of CAHs.
- 6. <u>Expand the rebound monitoring</u> to include total and dissolved metals plus TDS in wells in the heart of the treatment zone.

The results of these evaluations are discussed in detail in Appendix A-8. Key findings are briefly summarized below. It is noted that mentions of metals in this section and throughout this document also refer to the metalloids arsenic and selenium, as appropriate.

A number of parameters were found to be present at their highest concentrations at the injection well or in the reactive zone at the first line of downgradient monitoring wells (IRZ-1, RAP1-6T). Further downgradient, concentrations decreased to background levels at the next line of wells (IRZ-3, IRZ-4, IRZ-5, B242-GW). This trend was found for the following parameters:

- BOD and COD (Table 4-4) as expected, elevated BOD and COD occurred at the injection well after injections began (BOD and COD are measures of injected molasses and its metabolic products just as TOC is).
- Hydrogen sulfide (Figure 4-40) is substantially elevated in the injection well and RAP1-6T as would be expected under anaerobic (sulfate reducing) conditions.
- TDS (Table 4-10) like many of the other secondary water quality parameters, TDS was elevated in the reactive zone but decreased dramatically downgradient.
- Chloroform and carbon tetrachloride (Table 4-11) were likely introduced with the tap water used for injections (California Department of Health Services, Howard 1990). They were rarely detected and were never above MCLs.
- Ketones (acetone, 2-butanone, 2-hexanone, MIBK; see Table 4-11) were probably byproducts of molasses biodegradation, but all detections were below regulatory standards. The risks posed by these expected metabolic byproducts of the degradation of food grade carbon sources are very low in comparison to the risks posed by the chlorinated constituents that are targeted for remediation.
- Metals including iron, manganese, arsenic, beryllium, chromium, copper, lead, nickel, selenium, and zinc (Figures 4-66 through 4-69, Tables 4-12 and 4-13). The metals results indicate that secondary water quality impacts may occur within the reactive zone during implementation of ERD as a result of increased mobilization/solubilization of some metals. However, the effect appears to be limited to the injection area and to the extent of the strongly reducing zone. The metals data supports the concept that the affected metals, including arsenic, iron and manganese, are reprecipitated/immobilized downgradient of the reactive zone when conditions return to preexisting (less reducing) state. Similarly, it is expected that reprecipitation/ immobilization will occur within the IRZ area some time after system shutdown. Furthermore, the data supports the assertion that metals concentrations in the injectate do not create secondary water quality impacts.

#### 4.3.5.1 Arsenic – Field and Bench Scale Observations from Another DoD IRZ Site

Similar issues have been discussed during the implementation of pilot tests performed for DoD at another Massachusetts site. At the Devens Reserve Forces Training Area, ARCADIS conducted bench-scale and field-scale ERD pilot tests that, while the primary objective was degradation of PCE, were designed in part to test for arsenic mobilization. The following discussion is summarized from the Devens ERD Pilot Test Evaluation Report (ARCADIS, November 2002).

There appear to be three primary triggers that can cause the release/solubilization of geogenic arsenic, including development of high pH (greater than 8.5), the presence of high concentrations of competing anions (such as phosphate, bicarbonate, or silicate), and development of reducing conditions at circumneutral pH.

Within the anaerobic and reducing IRZ created by ERD technology, there is evidence that some control on arsenic solubility can be realized through the formation of low-solubility arsenic sulfide compounds. However, it is expected that the primary control on arsenic solubility will be provided by adsorption to and co-precipitation with hydrous ferric (iron) oxides under ambient oxidizing conditions.

Under the Devens site's normal aerobic groundwater conditions, both dissolved-phase arsenic and iron concentrations were below laboratory detection limits. In the field pilot, arsenic was solubilized in the pilot study area at levels greater than both the current and proposed MCLs for arsenic. However, field tests supported the expectation that the presence of soluble arsenic will be limited to the boundaries of reducing zones created by the ERD technology. Once the original aerobic and oxidizing poise of those reducing zones is restored, it is expected that dissolved arsenic will decrease to non-detectable levels. ERD application was therefore considered appropriate for treatment of CAHs provided the temporary presence of arsenic was appropriately monitored and managed.

In the bench-scale treatability study (flow-through column study), the initial aerobic poise of each of three soil columns was overcome by passing reduced groundwater containing dissolved concentrations of arsenic, iron, and manganese through the columns. Measurements of the three metals/metalloids and DO and ORP were recorded at intervals as the water was applied to the columns. After reducing conditions had been achieved, the aerobic poise of two columns was restored using two different oxidation techniques (air injection and hydrogen peroxide injection). Based on the treatability study results, the following observations were made:

- Even under reducing conditions, the aquifer materials provided a significant level of control on arsenic solubility
- The injection of air or hydrogen peroxide in the field can create an aerobic environment (most suitable for controlling arsenic solubility)

Thus, both empirical data from the Devens site and published research indicate that arsenic solubility as it relates to the use of ERD can be controlled, mitigating concerns associated with use of the technology.

#### 4.3.5.2 Metals in Molasses

Molasses in its pure form contains concentrations of several metals that may exceed water quality criteria. Published analyses of blackstrap molasses (US Sugar, 2001) and analyses of metals in molasses/water mixtures by ARCADIS are presented in Tables 4-14 and 4-15. The ARCADIS metals sample was from a commercial remediation site in Ohio where a different molasses source was used than at Hanscom, but the results should be similar for Hanscom. Also note that the water-to-molasses mixture used at the Ohio site was slightly more dilute than the Hanscom mixture. None of the metals detected exceeded available Federal MCLs, and would not if adjusted to match the more concentrated Hanscom mixture. In addition, the site metals groundwater data discussed above encompass any solute quality issues. In other words, the groundwater metals data in Section 4.3.5 show that metals, whether solubilized from the formation or introduced as a trace component of molasses (or both) were not problematic in this demonstration outside the reactive zone. On the basis of this evidence, we would not typically expect to see water quality impacts from the molasses injectate. However, this is a potential issue that should be briefly considered in the design phase of IRZ projects.

## 4.3.6 Other Observations

#### 4.3.6.1 Reactive Zone Size and Residence Time

As discussed earlier, since the groundwater flow direction was on average more easterly than the planned southeasterly, the reactive zone was formed on the north side of the monitoring well array and it is highly probable that a large portion of the reactive zone formed to the north of well IRZ-1. We can estimate the maximum and minimum sizes of the reactive zone as follows:

<u>Width perpendicular to average groundwater flow direction:</u> The zone width was most clearly defined by RAP1-6T and IRZ-1. IRZ-1 received an adequate dose of substrate during most periods. RAP1-6T received an adequate dose of substrate for somewhat briefer times indicating that it was on the edge of the reactive zone, which is consistent with the potentiometric surface observations. Thus the minimum zone width was 20 feet and an estimated zone width, including the portion passing north of IRZ-1, was 40 feet.

<u>Length parallel to average flow direction</u>: The injection well, IRZ-1 and RAP1-6T were all clearly within the zone. The zone extended slightly upgradient of the injection well as evidenced by the decline in CAH concentrations. Well IRZ-5 was showing increased TOC and DOC by the end of the demonstration, suggesting that it was on the far edge of the zone. Thus the zone length is estimated as between 45 and 160 feet and likely was less than 100 feet.

<u>Depth (vertical)</u>: The screened interval for the monitoring and injection wells was 10-15 feet. Near the end of the demonstration, some evidence was seen of influence in the upper aquifer and bedrock aquifer wells of the RAP1-6 cluster (see Figure 4-6 and Table 4-3). This suggests either further vertical mixing due to vertical gradients, seal leakage as discussed in Section 3.5.1, or diffusion. Thus the zone depth can be estimated as between 15 and 50 feet, but for all practical purposes the depth of the zone was equivalent to the thickness of the lower aquifer (18 to 25 feet).

These zone size definitions have been based on delivery of measurable substrate, which as discussed above is necessary for effective treatment. However, the effect on DO and ORP appears to have propagated much more widely (Figures 4-1 and 4-2).

To determine the residence time required to reduce CAHs in the demonstration area, it is necessary to separate out the travel time of the substrate between the injection well and the monitoring point, and the acclimation time required for microbial growth. The Hanscom results were also complicated by substrate distribution issues. At this site, there was a lag of 3 to 5 months (90 to 150 days) between the time of first injection and the beginning of enhanced TCE treatment at IRZ-1 and RAP1-6T. Travel time for the substrate to reach these wells is estimated at 53 days (see Section 4.3.4.2. and Appendix A-7), leaving approximately 40 to 100 days after substrate distribution issues (see Figures 4-31 and 4-34) rather than biological acclimation. It is reasonable that no appreciable acclimation time was required at this site for enhanced TCE degradation because TCE biodegradation to DCE was already occurring prior to treatment.

Changes in DCE concentration took longer to occur. Although it is difficult to separate out the rates of production of DCE from TCE and of DCE degradation, the data suggests that at IRZ-1 DCE removal was not enhanced and ethene production increased until about 15 months after the

first injection and was correlated with increases in methane concentration. This suggests that there was up to a year's lag beyond the time required to distribute substrate that may be due either to the successive consumption of electron acceptors required to reach methanogenic conditions, bacterial acclimation or both.

In summary, enhanced TCE treatment was thought to occur almost immediately upon sufficient substrate delivery. The residence time required for cis-1,2-DCE treatment, independent of travel and acclimation time, was not definitive, but is expected to be on the order of a few months for this site. The time required to achieve similar reductions in VC would be longer, but was not determined in the demonstration. However, the strong performance of the system under imperfect coverage suggests that a full-scale system with more homogeneous substrate distribution could achieve stringent cleanup goals within five years. Moreover, the demonstration area was apparently within a source zone, and would be expected to perform more efficiently in lower-concentration portions of the plume. A longer treatment zone parallel to groundwater flow in a plume-wide or multiple barrier application would also enhance full CAH treatment, and would likely be required at Hanscom to achieve MCLs.

An estimate was made of the number of injection points that would be required to treat the full extent of the lower aquifer plume in Figure 3-2. In hypothetically scaling up the system, it was assumed that a plume-wide application would be appropriate for the site, rather than another configuration such as a barrier or a limited source area application. A plume-wide application was also the basis of the cost analysis in Section 5, allowing comparison to common uses of pump-and-treat and other groundwater remediation technologies, as well as to most other ERD sites. Other geometries would likely require fewer injection points. In addition to the configuration, the following assumptions were made:

- A full-scale system would be run in lieu of pump and treat, or pump and treat system operation would be more steady, eliminating most of the variability in groundwater flow direction.
- The reactive zone size and residence time established for the demonstration would be representative of a system performing in a less variable flow field. This is conservative, since the amount of substrate delivered to the test plot would presumably treat the same CAH concentrations more effectively if it were dispersed in a narrower area.
- The targeted aquifer is of a fairly uniform thickness throughout the extent of the plume, similar to the 18 to 25 feet encountered in the demonstration area, and a single injection depth is adequate to treat the affected zone.
- The reactive zone size and residence time determined above are applicable for lower concentrations of CAHs found in the majority of the plume.

Based on a "treatment cell" emanating from an injection point measuring 40 feet wide by 100 feet long, or 4,000 square feet, and a total plume area of approximately 1,140,000 square feet (based on the plume map in Figure 3-2), approximately 285 injection points would be required to treat the full extent of the plume. This injection well spacing is consistent with recommendations made in Section 4.2.1 of the protocol document (Suthersan, 2000), in which typical spacing is cited at 20 to 50 feet transverse to flow direction and approximately 100 days travel time parallel to flow direction. Downgradient well spacing of 100 feet at the Hanscom site, divided by the

groundwater velocity of 0.8 feet per day, represents a 125-day travel time. As discussed in the protocol document, well spacing parallel to groundwater flow is primarily a budgetary concern.

## 4.3.6.2 Utility of Hydrogen Data

Molasses is not directly consumed but rather goes to form a number of organic monomers, alcohols and organic acids, which in turn break down to form acetate and hydrogen, which serve as energy sources for methanogens involved in reductive dechlorination (Suthersan, 2002). It has been suggested that dissolved hydrogen can be a diagnostic parameter to monitor in groundwater from ERD monitoring wells, as it can suggest which microbially-mediated redox processes are predominating in the reactive zone. Though sampling and measurement of dissolved hydrogen from monitoring well groundwater is feasible, the results are subject to several potential problems. We believe that the cost of acquiring reliable hydrogen data is generally not justified at routine sites since the predominant redox processes in various zones can normally be delineated from other chemical measurements.

According to Chapelle 1999 "Significantly, each of these terminal electron-accepting processes (TEAPs) has a different affinity for H<sub>2</sub> uptake. CO<sub>2</sub> reduction (methanogenesis) has the lowest H<sub>2</sub> affinity, and observed steady-state H<sub>2</sub> concentrations in methanogenic aquifers are relatively high (5-30 nmol/L (nM)). Sulfate reduction has a slightly greater affinity for H<sub>2</sub> than methanogenesis and is observed to have lower characteristic H<sub>2</sub> concentrations (1-4 nM). Fe(III) reduction (0.2-0.8 nM) and nitrate reduction (<0.1 nM) have even greater affinities for H<sub>2</sub> and are observed to have progressively lower steady-state H<sub>2</sub> concentrations. Thus, concentrations of H<sub>2</sub> are a useful indicator of TEAPs in groundwater systems, particularly when interpreted in the context of electron acceptor (nitrate, Fe(III), and sulfate) availability and the presence of final products (Fe(II), sulfide, and methane) of microbial metabolism."

The measurement of dissolved hydrogen in groundwater is a two-step process: sampling and analysis. Both steps are critical and special care must be taken by the field crew during sampling to avoid potential contamination/interference of the sample. The most relevant sources of sample contamination/interference include:

- Contamination with atmospheric air
- Generation of hydrogen gas from well construction materials and techniques
- Generation of hydrogen gas resulting from choice of sampling pump employed

Standard low-flow sampling techniques are used as the basis for sampling groundwater for dissolved hydrogen analysis as hydrogen is an extremely volatile gas whose concentration in groundwater can be biased if exposed to the atmosphere. The use of a flow-through cell increases the protection of samples/measurements against atmospheric loss or contamination. Methodologies for hydrogen sampling used in this project were discussed in Section 3.5.7.1.2. A general discussion of hydrogen sampling and analysis methodologies can be found in Appendix C of the protocol document (Suthersan, 2002).

In addition to sampling technique, well construction is also critical. Wells with casings and/or screens constructed of iron-containing metals (e.g., cast iron, galvanized steel) have been shown to artificially generate dissolved hydrogen via the reduction of water by metallic iron under

anoxic conditions (Chapelle et al., 1997). Wells constructed of PVC are recommended, although stainless steel and Teflon may also be acceptable (Microseeps, 2002; Bjerg, et al., 1997). In addition to well construction materials, the process of well installation itself has been shown to generate hydrogen gas, presumably from the disturbance of soil (Bjerg, 1997). For this reason, one to three months is recommended between well installation and the first hydrogen-monitoring event. Furthermore, the installation of wells or other site disturbances in the vicinity of a monitoring well may generate hydrogen from the soil, which can show up in the monitoring well. The radius of influence of such site disturbances as they relate to dissolved hydrogen measurements is most likely highly dependent on the specific characteristics of the site geology and is therefore difficult to generalize. However, noting nearby site disturbance activities is particularly salient if unusually/unexpectedly high hydrogen results are observed in an adjacent monitoring well.

The selection of sampling pump is also important and requires careful consideration of several factors including the pump operating characteristics (flow, head) and means of operation. Peristaltic pumps (which draw water under negative pressure) are generally preferred as they are low-flow, minimum turbulence/disturbance, and have been shown to enhance the gas-stripping process required during sampling (Chapelle 1997), though there may be times when conditions dictate that an alternative pump be used (e.g., the required head to be overcome may necessitate the use of a higher head pump). Positive pressure pumps such as piston or bladder pumps may be effectively used, although the surging operation of a bladder pump may cause the instantaneous flow rate to exceed those required to achieve liquid-gas hydrogen equilibrium during sampling. This limitation may be overcome by decreasing the duty-cycle of the bladder pump and/or by pumping into a surge tank with a peristaltic pump to feed the bubble strip sampling apparatus (Microseeps). Direct current submersible pumps should be avoided as the electrical current passing through the monitoring well may produce hydrogen gas from water via electrolysis (Chapelle 1997).

Hydrogen data from this demonstration is reported in Figure 4-70 and Tables 4-5 and 4-7. Many of the values from the June 2000 and March 2002 sampling rounds were reported as saturating the analytical process (> 50 nM/L). These values were above those conventionally associated with steady state concentrations in presence of various microbial communities. However in each of these rounds several samples, often ones that would be expected to be unimpacted by the reactive zone, such as RAP1-6R and B-239 showed substantially below saturation concentrations. Several possible explanations were considered for this trend in the data:

- Well construction methods were reviewed. Metals in the construction material were eliminated as a possible cause. However it is possible that since the June 2000 data was collected one month after well installation that metal drilling implements could have caused artifacts in that data, but this would not explain similar results in March 2002.
- Pumps used were reviewed and initially eliminated as a possible cause of artificially high values since the values occurred in some wells and not others and the pumps used were not of the type most suspected in hydrogen generation.
- Metal objects in the ground could not be eliminated as a possible cause since groundwater is relatively shallow and it is possible that pipes, fence posts or debris could be in contact with groundwater at some times.

We conferred with Dr. Francis Chapelle of USGS about the first two hydrogen data sets. He offered the opinion that the one round of post injection data may suggest that we are producing hydrogen faster than the microorganisms can use if and therefore it is building up to higher than expected levels. In general he believes that fermentation processes that produce hydrogen are more robust than the sulfate reducing or methanogenic processes that use hydrogen. He feels something (perhaps a co-contaminant) may be inhibiting hydrogen utilization here. He also noted that high hydrogen values had previously been observed in the presence of Aqueous Fire fighting foam (AFFF). Based on site history the presence of AFFF is possible.

We then performed a final round of hydrogen sampling in October 2002 sampling with an ACpowered Grundfos variable speed 2-inch submersible (centrifugal) pump. We requested that the laboratory dilute these samples as necessary to avoid detector saturation and thus extend the analytical range. In this instance, all of the results reported were above the saturation limit of the previous analyses. They ranged from 134 nM/L at RAP1-6T to 1452 nM/L at IRZ-1 and 1514 nM/L at IRZ-INJ. Since these values were judged to be unreasonably high, we resampled. Thus we repeated the sampling on October 29, 2002 in three selected wells using a peristaltic pump: B239, RAP1-6T and IRZ-1. The values obtained were 2.2 nM/L at B-239, 7.9 nM/L at RAP1-6T, and 9.7 nM/L at IRZ-1. Thus the values given in the tables and figures for this round should be considered very questionable. It appears that though it is DC current-driven submersible pumps that are associated with the most problems with hydrogen generation in the literature (Chapelle, 1997, p. 2876), that an AC-powered pump generated an artifact in this instance.

We conclude from this experience that the sampling of hydrogen during ERD projects may not be justified in most circumstances since:

- More reliable diagnostic information from which the predominant metabolic processes ongoing at the site can be inferred is available (measurements of alternate electron acceptors and the products of their use.
- Hydrogen sampling and analysis is relative complex and costly, subject to numerous potential artifacts and constrains pump selection substantially.

## 4.3.6.3 Fatty Acids

In the last sampling round in October 2002, fatty acids, which are metabolic byproducts of molasses degradation were sampled and analyzed in selected wells (Table 4-16). Acetic, propionic and butyric acids were found within the reactive zone but not upgradient. More information on the breakdown of molasses and other carbohydrates is presented in Section 1.3 of the protocol (Suthersan, 2002).

## 4.3.6.4 Microbial Population Characterization

Since there was strong evidence for natural attenuation at this site, no microbial characterization was performed prior to the pilot scale test. However, at the request of ESTCP, a microbial characterization was performed of the reactive zone during the final sampling round in October 2002 using groundwater samples. Phospholipid Fatty Acid Analysis, Denaturing Gradient Gel Electrophoresis and Targeted Gene Detection for Dehalococcoides ethenogenes (DHE) were performed. Samples were drawn from the generally upgradient well B239 and from the wells in

the reactive zone or on its fringe - RAP1-6T and IRZ-1. The methods and results are presented and discussed in Appendix A-10. The main conclusions from this work were that:

- Good population numbers (for a groundwater sample) and good diversity were found in all samples.
- The samples from within the reactive zone showed less evidence of starvation then the upgradient sample.
- The samples within the reactive zone appeared to have more anaerobic populations.
- DHE was detected in the upgradient and RAP1-6T samples. Results from IRZ-1 were inconclusive due to interferences in the sample.

## 4.3.7 Comparison of Results with Objectives

Objectives for this demonstration were discussed in Section 4.1 and were grouped into primary and secondary objectives. A brief comparison of results with objectives is given in Table 4-2; longer discussions are provided in this section.

## 4.3.7.1 Primary Objectives

During the two-year demonstration, the data discussed below demonstrated highly effective, complete TCE removal in a source area that had a long history of fairly stable TCE concentrations before treatment. Multiple lines of evidence of complete treatment – a buildup of ethene, reduction in *cis*-DCE and no accumulation of VC was seen in the most effectively treated downgradient wells. Effective treatment was seen only where substantial substrate (molasses and its breakdown products) was observed in downgradient monitoring wells. The layout of the injection and monitoring well system was designed for a consistent southeasterly groundwater flow; however, during the demonstration period, the direction of flow varied with most observations suggesting an eastward flow direction. Thus, it is suspected that a larger IRZ was formed than what was observed, but that the monitoring well network was not positioned to completely delineate it.

This summary discussion of CAH treatment will focus primarily on the two monitoring wells that received substantial doses of substrate TOC, RAP 1-6T and IRZ-1 (Figure 4-5), and secondarily on the injection well, INZ-INJ. The two primary monitoring wells were also the only ones in the lower aquifer where substantially increased levels of methane were observed (Figure 4-30).

The best treatment results were observed at IRZ-1 (Figure 4-31). At this well, highly effective treatment of TCE was observed beginning in March 2001, 5 months after injections began and shortly after single injections with water pushes began (>95% reduction vs. pretest concentrations which greatly exceeds the objective of 80% reduction within one year). Substantial treatment of *cis*-1,2-DCE (eventually >85% reduction in pretest concentration) was not observed until March 2002, a year later, during a second period of high TOC delivery. Complete degradation was also shown by the substantial increase in ethene production in March 2002 (Figure 4-32), which continued to climb through the end of the demonstration, indicating

that treatment effectiveness continued to increase after two years of system operation. Ethene concentrations at this well increased to more than 20 times the pretest value.

Well RAP1-6T had 14.5 years of relatively stable TCE, DCE and VC concentrations before the demonstration (Figure 4-33). A sharp decline in TCE was observed within 6 months of the beginning of injection, coinciding or slightly preceding the appearance of substrate (Figure 4-34). TCE levels for seven straight monitoring rounds were at 10% or less of the average of the proceeding 10 years (Figure 4-35), exceeding the 80% in one year treatment objective. Then the IRZ shifted away from this well as evidenced by the significant decline in TOC. As a result of this shift, TCE, DCE and VC concentrations rebounded. These concentrations dipped again when substrate levels increased, although it appears that the typical groundwater flow direction was somewhat more easterly rather than southeasterly as planned. Thus this well was probably at the fringe of the effective reactive zone for much of the demonstration period.

Although it is by definition a less accurate measure of the overall effectiveness of the reactive zone, it is useful to note that substantial evidence of effective treatment of all chlorinated species was seen at the injection well, even in data corrected for the dilution effect of the injected solution. Concentrations of TCE and VC were reduced by more than 95% for TCE and 85% for VC from May 2001 through the end of the demonstration. DCE decreased substantially less (at most about 75%). This suggests that although CAHs were being completely degraded, desorption from a localized source area (Figure 4-36) upgradient continued to impact the injection well.

The wells that did not get substantial, consistent doses of substrate showed no evidence of treatment or at most modest decreases in TCE only. This strengthens the conclusions that substrate availability is linked with improved biodegradation and that contaminant removal was due to enhanced biodegradation rather than displacement. At full scale a greater number of injection wells would provide overlap and would be less subject to changes in flow direction.

As discussed in Section 4.3.6.1, the residence time required to treat the source area CAH concentrations encountered in the demonstration area to Federal MCLs was not directly determined in the demonstration. However, the strong performance of the system under imperfect coverage suggests that a full-scale system with more homogeneous substrate distribution could achieve stringent cleanup goals within five years.

The data from RAP1-6T, IRZ-1 and the injection well do not show evidence of "accumulation" or "dead-ending" at DCE or VC (Figures 4-35 through 4-37). On the contrary, the data discussed above and especially the increases in ethene (Figures 4-32 and 4-42) show that complete degradation was achieved in these wells. Therefore the primary performance objectives were met for the wells within the reactive zone.

#### 4.3.7.2 Secondary Objectives

In addition to the primary treatment effectiveness objectives discussed above, certain other objectives were established in the demonstration plan to improve understanding of the process (see Section 3.1 for more details):

<u>Objective</u>: Demonstrate the ability of the ERD to induce an anaerobic and reducing environment in groundwater.

<u>Result:</u> As shown by the DO, ORP, hydrogen sulfide and methane data (Figures 4-1, 4-2, 4-30 and 4-40), an anaerobic environment was created within the reactive zone.

<u>Objective</u>: Evaluate the ability of the ERD to desorb contaminants that are most likely sorbed to the aquifer materials.

<u>Result:</u> There is some evidence of a TCE desorption "hump" early in the demonstration (see Figure 4-37 for example). The fact that levels of CAHs at all of the wells in the demonstration zone before the test greatly exceeded the levels in upgradient well B239, and that pretest VOC concentrations increased from northwest to south east across the zone, suggests that desorption was ongoing within the zone even before treatment (see a more in-depth discussion in Section 3.4). Note also that the levels of CAHs at side-gradient or fringe wells such as IRZ-2 (Figure 4-48) were much higher than those in B239 (Figures 4-71 through 4-73) throughout the demonstration, also demonstrating that desorption was continuing through the demonstration. However, the rate of desorption is difficult to quantify and the increased desorption due to the IRZ is thus difficult to quantify (since the desorption and enhanced biodegradation processes occur simultaneously).

Objective: Evaluate degradation rates for CAHs in groundwater.

<u>Result:</u> Degradation rates calculated, as discussed in Section 4.3.3.5, were substantially enhanced over both the pretreatment rates and typical natural attenuation rates previously observed in the field at other sites. TCE degradation rates were shown to be in the range expected in enhanced in-situ bioremediation systems.

Objective: Determine the optimal strengths and frequency of reagent delivery for the site.

<u>Result:</u> Strengths and frequency of injection are discussed in Section 4.3.2.1. The first period of effective treatment occurred when the injection rate was approximately 125 lbs molasses per week in the single injection well. Injection rates ranged from 100 to 250 lbs molasses per week during the second period of highly effective treatment. However, due to changes in groundwater flow direction and flow rate that occurred during the demonstration, no single optimal strength and frequency can be defined. Rather, the optimal injection rate for full-scale system operation is better defined as that rate necessary to maintain the optimal substrate loading (shown by TOC and specific conductance) and microbial community (methanogenic) at the monitoring wells. These optimal conditions are defined and discussed in Section 4.3.3.4.

The demonstration plan discussed that the following target concentrations or ranges for various field parameters would be used as a guide to system operations:

- pH > 4.5 s.u. in the injection wells; > 5.0 s.u. in the monitoring wells;
- DO < 1.0 mg/L in both monitoring and injection wells;
- ORP > -400 mV and < -250 mV in the injection wells; < -200 mV in the monitoring wells. For demonstration sites where reducing environments are identified in the groundwater prior to initiation of reagent injections, a target goal of lowering the ORP by 200 mV in the injection wells and 100 mV in the monitoring wells will be employed.
- TOC -> 500 mg/L and < 5,000 mg/L in the injection wells and > 50 mg/L in the monitoring wells. TOC in monitoring wells should be expected to decline with distance from the injection well.

Specific conductance – order of magnitude increase in the injection wells; 20 to 50 % increase in monitoring wells.

The actual observed values for these parameters are presented in Section 4.3.2 and will only be briefly summarized here. The desired pH was achieved without difficulty in the monitoring wells within the demonstration zone, therefore somewhat lower pHs in the injection wells were tolerated to achieve acceptable TOC loadings.

DO was generally held to less than 1 mg/L in the monitoring wells; data for the injection well was believed to be unreliable because of "frothing" in that well. Occasional upward trends in DO were noted, which served as early warnings that the injection strategy needed to be adjusted.

ORP (Figure 4-2) was lower than -400 mV in the injection well for several months. ORP in the monitoring wells within the reactive zone held fairly steady between -100 and -200 mV during most of the demonstration. Although these were somewhat less reducing than the stated objective, as noted in Section 4.3.3.5, ORP measurement is inherently unreliable and methanogenic conditions and complete treatment were achieved at the observed ORP levels.

TOC >50 mg/L was achieved at times at the monitoring wells (see Figure 4-5). As discussed in Section 4.3, TOC levels above 10 mg/L at the monitoring wells were associated with effective treatment. TOC in the injection well was maintained over 500 mg/L throughout the demonstration. Although we stated a goal of less than 5,000 mg/L in the injection well in the demonstration plan, later experience at other sites caused us to modify that guidance to state that the upper limit is site-specific but generally around 9,000 mg/L (see Section 4.5 of the protocol [Suthersan, 2002]). Injection well TOC was generally but not always below 9,000 mg/L during this demonstration. However, as noted above, little or no pH change was noted at the monitoring wells and TOC at the monitoring wells was not excessive. Therefore, since this was a high flow site with only one injection well, high loadings at the injection well were used.

Specific conductance increased an order of magnitude in the injection well and more than 50% in the monitoring wells within the reactive zone. As shown in Figure 4-60, specific conductance increases were closely correlated to effective TCE treatment.

Thus, in summary, the reactive zone was generally operated within the expected ranges for the process monitoring parameters and all of the secondary objectives were met.

#### 4.3.8 Technology Comparison

Based on the results of the demonstration as outlined in this document, the use of ERD to treat CAH impacts in groundwater via transformation to innocuous end-products has been demonstrated to be successful. In addition, as outlined in the work performed during the demonstration, the technology has provided many advantages over other conventional and emerging remediation techniques including the following:

- Ease of deployment including very limited 'hard' design,
- Limited permitting & approvals,
- Ease of operations & maintenance,

- Flexibility,
- Limited health and safety risk directly related to the technology and reagent handling,
- Implementation with little impact to ongoing facility operations and/or future development activities.

These advantages as well as the competitive cost of application of the technology provide a convincing case for the applicability/desirability of the technology in a variety of application scenarios. However, the results of the demonstration illustrate some limitations of ERD application in comparison to other technologies. These limitations include the following:

- Speed at which desired reactions/treatment results can be expected to occur,
- Possible incomplete treatment of parent CAHs, and
- Possible solubilization of inorganics as a result of the reducing conditions.

Overall, these limitations are likely to be limited to a small percentage of sites and/or applications if the technology is implemented properly. However, they need to be carefully considered during both the technology selection and remedial design phases of the project to assure success.

Attached Table 4-17 contains a general comparison of ERD to several other common remediation technologies used for the treatment of CAHs, specifically, groundwater extraction & treatment, aquifer sparging, and chemical oxidation. This general comparison considers the relative effectiveness, reliability, speed and ease of use of each technique for comparison purposes.

Evaluation and comparison of ERD to other remedial techniques, specific to conditions at the Site can also be made. This comparison is especially valid at the Site given that the Base has undertaken numerous remediation projects including the use of groundwater extraction and treatment, vacuum-enhanced recovery (a modified version of groundwater extraction and treatment) and in-situ chemical oxidation. A discussion of these technologies as they relate to actual or potential application at the site is outlined below.

## 4.3.8.1 Groundwater Extraction and Treatment (Pump & Treat)

The use of groundwater extraction and treatment has been ongoing at the Site for many years and has provided valuable remediation progress for the overall restoration program at the Base. The dissolved CAH plumes have been stabilized and off-site migration and thus risks have been controlled.

However, it is clear that due to the portions of the Site underlain by low permeability geologic materials as well as the expansive size of the CAHs plumes at the Base, restoration of the site using pump and treat will require a very long time to achieve. This is clearly illustrated by the several locations in which high concentrations of CAHs are still present even after the lengthy pumping program. It is likely that a more cost effective approach would be to utilize ERD on source areas, and perhaps additional IRZs between source areas and the existing extraction well

system. The goal would be to terminate use of the pump and treat system, and reduce constituent concentrations to levels suitable for application of monitored natural attenuation.

## 4.3.8.2 Aquifer Sparging

As outlined in the general analysis above, the use of sparging is often an effective means to remediate CAH impacts. However, at Hanscom this technology is technically unfeasible for the semi-confined aquifer since the confining unit would prevent recovery of the vapors resulting in the uncontrolled migration of gas-phase CAHs.

## 4.3.8.3 Chemical Oxidation

Given the in-situ nature of the technology, chemical oxidation would be expected to be a successful means of treating residual dissolved and adsorbed phase CAH impacts at the Site. Currently, the Air Force is in the process of evaluating this technology at the Base using permanganate as the oxidant. At the time of this report, performance data from these tests are unavailable.

It is expected that a chemical oxidant could be delivered to the impacted areas. Assuming this is the case, the suitability of chemical oxidation at the Site versus IRZ is likely an economic decision. Given the cost of the chemical reagents needed to not only oxidize the target compounds, but to overcome the natural reductive poise in the formation, the cost of chemical oxidation is expected to be high, if used for a full-scale plume treatment approach. More likely, chemical oxidation would be selected to play a limited or 'surgical' role in the overall restoration strategy using it in a program where it would be reserved for treatment of higher concentration areas or areas where rapid clean-up time periods outweighed cost concerns.

## 4.3.8.4 ERD

The results of the ERD demonstration at the Site indicate that the technology can be successfully applied and, if properly operated, can result in complete degradation of the CAHs present in the dissolved phase as well as the enhancement of desorption of adsorbed phase CAHs. Given the scope and limitation of the demonstration treatment to MCL's was not demonstrated, although substantial degradation was. However, given ARCADIS' experience at numerous other sites, as well as taking into account the lessons learned on this project, we are confident that the ERD technology can achieve those objectives.

In comparing the use of ERD to other technologies, the chief advantage of ERD is likely cost. The limited infrastructure required to deploy the technology as well as the low reagent costs will likely make ERD the least expensive means to address the residual impacts when implemented at full-scale.

## 4.3.9 Lessons Learned

ERD in an IRZ application has been applied at a broad range of sites since 1995. These sites have included a variety of constituents to be treated - including PCE, TCE, 1,1,1-TCA, carbon tetrachloride, pentachlorophenol, and chlorinated pesticides; various groundwater concentration ranges; and numerous hydrogeologic settings (including shale and karstic limestone bedrock, low permeability glacial tills and saprolite, and high permeability alluvium and glacial outwash environments).

As with all groundwater remediation activities both in-situ and ex-situ, the successful application of ERD relies mainly on sufficient and accurate hydrogeological information for the given site. The application of ERD to treat CAHs in groundwater at many sites located in varied in-situ hydrogeologic settings under different concentrations has provided a valuable knowledge base that has taught many lessons for future applications of the technology both at the pilot-and full-scale. These lessons learned are also applicable to applying other in-situ remedial techniques. Some specific lessons learned from the Hanscom demonstration are included below.

## 4.3.9.1 Substrate Dosing Required for Successful Treatment

As discussed in depth in Section 4.3, this demonstration clearly showed that a considerable concentration of substrate is required for successful treatment. As depicted in Figure 4-60, a specific conductance of 600 mS/cm (300 mS/cm above background) at the downgradient monitoring wells appears to be the threshold for successful treatment. Figures 4-31 and 4-34 suggest that successful treatment is usually associated with TOC values between 10 and 200 mg/L. This is in basic agreement with the guidance in the protocol document (Suthersan, 2002, Section 4.5) based on observations at many sites that 50-200 mg/L TOC in monitoring wells is ideal. They further suggest that methanogenic conditions as indicated by methane concentrations in excess of 1000  $\mu$ g/L are associated with rapid, complete treatment.

## 4.3.9.2 Microbial Acclimation / The Role of Bioaugmentation

Following the addition of an electron donor, the microbiological community present at the site is required to acclimate to the changing aquifer conditions. During this interval, some originally prominent members of the community may decline in numbers or cease to exist entirely within the community at large. Other microorganisms that were previously present in relatively insignificant concentrations may find the changing conditions more suitable to their metabolic needs and expand in number. Once substrate has been delivered to an area within the aquifer, a period of several additional months is often required for the successive consumption of various electron acceptors, which in turn requires successive changes in the microbial community. It is ARCADIS' experience that the implementation of the ERD technology typically results in a bacterial succession as described above where bacteria that are important for the biodegradation of CAHs become a functional part of the new bacterial community that is given rise by ERD implementation. The bacterial community present in the aquifer prior to carbohydrate injection is shifted towards species better adapted to a more reduced environment. In rare instances, ARCADIS has implemented bioaugmentation at ERD sites. Bioaugmentation is the introduction of a specific bacteria or mixture of bacteria to a site where it is felt that the community already in existence is lacking the capability to biodegrade a given contaminant.

At this site although there was a lag of 3-5 months between the time of first injection and the beginning of enhanced TCE treatment, this appeared to be due almost entirely to substrate distribution issues (see Figures 4-31 and 4-34) rather than biological acclimation. It is reasonable that no appreciable acclimation time was required at this site for enhanced TCE degradation because TCE biodegradation to DCE was already occurring prior to treatment.

Changes in DCE concentration took longer to occur. Although it is difficult to separate out the rates of production of DCE from TCE and of DCE degradation, the data suggests that at IRZ-1

DCE removal was not enhanced and ethene production increased until about 15 months after the first injection and was correlated with increases in methane concentration. This suggests that there was up to a year's lag beyond the time required to distribute substrate that may be due either to the successive consumption of electron acceptors required to reach methanogenic conditions, bacterial acclimation or both.

## 4.3.9.3 Fermentation and Byproduct Formation

During the application of ERD, a highly reducing biogeochemical environment is generally created throughout the treatment zone. In addition, this zone will contain a large excess of organic carbon. During the application of ERD, most commonly when the contaminated aquifer possesses lower hydraulic conductivity (10-5 cm/sec [2.8 x 10-2 ft/day] or less), these conditions can result in the formation of organic acids in the groundwater as part of the degradation process. As a result of the formation of these acids, the ambient pH in the treatment zone can be lowered and in turn conditions conducive to fermentation-based reactions are then created. This environment can create low pH conditions that are detrimental to methanogenic bacteria.

The formation of undesirable byproducts including acetone and 2-butanone has been observed at sites where reagent dosing has commenced without careful monitoring of groundwater conditions near the injection wells. The occurrences of these byproducts are generally limited in extent and often sporadic in nature. It is expected that these ketones are also utilized by microbes in the IRZ. Almost all of these products are readily aerobically degradable as well and so are degraded on the downgradient edge of the ERD zone. Furthermore, almost all have higher risk-based limits (i.e., MCLs) than the target compounds of the ERD system. However, the possibility of production of these byproducts needs to be accounted for in the project planning stage. Therefore, the lessons learned regarding these potential occurrences are as follows:

- Careful and regular monitoring of groundwater within the treatment zone should be provided in order to ensure that pH levels are not depressed (pH < 5 at monitoring wells, pH < 4 in injection wells) and TOC levels are not excessive (site specific, but generally above 5,000 to 10,000 mg/L in injection wells).
- The remedial plan for application of ERD should be flexible enough to allow for modification of both the delivery frequency and mass of organic carbon delivered, preventing the build-up of organic carbon and creation of conditions amenable to creation of excessive amounts of these byproducts. Modifications in reagent delivery should be tied to regular pH and TOC monitoring in the treatment zone.

At Hanscom, as discussed in Section 4.3, pH changes in the monitoring wells were never significant but pH decreases in the injection well limited the substrate injection rate. The use of a clean water push was beneficial to disperse the injected TOC further beyond the injection well right after injection. This allowed the molasses injection rate to be substantially increased without further pH drops at the injection well. As discussed in Section 4.3.5, although ketones were generated as metabolic byproducts of molasses degradation, they did not pose an appreciable risk. At other sites, it may be desirable to add a buffer rather than using a water push (see Protocol Sections 4.1.2.1 and 6.2 [Suthersan, 2002]).

## 4.3.9.4 Pilot Test Design

This demonstration used one injection well with six monitoring wells in the downgradient target zone (arranged as two transects of three wells each). At this site substantial variability was observed in the groundwater flow direction vector. This resulted in marked changes over time in the substrate concentration at the most affected downgradient monitoring wells (RAP1-6T and IRZ-1). It also resulted in several downgradient monitoring wells not being impacted by the introduced substrate. Although it is desirable to have some downgradient monitoring wells outside of the reactive zone so that the radius of influence of a single injection well can be determined, installing too many uninfluenced pilot zone monitoring wells is not cost effective. Thus consideration should be given to an alternate test design with three injection wells in a transect and a smaller number of downgradient monitoring wells. This approach has the following advantages:

- It is less subject to changes in the groundwater flow direction vector once the demonstration begins, because the created zone is wider and thus the likelihood of a given monitoring well seeing wide variations in substrate concentration is lessened.
- It is less subject to errors in estimation of the exact direction of the average groundwater flow for similar reasons.
- The center of such a reactive zone would be more likely to maintain strongly reducing conditions because the wider zone would be less subject to diffusive mixing of electron acceptors from the sides.

However, the disadvantage of such a system is that it would be substantially more costly to estimate the radius of influence of each injection well (because a larger total number of wells would be needed if radius-of-influence data was vital).

#### 4.3.9.5 Application in Areas of High Constituent Concentration/DNAPL

Given the inherent problems with the use of conventional remediation techniques in areas where the constituent concentrations are very high and/or where free phase constituent (DNAPL) may be present, ERD has been an attractive potential alternative for these settings (Yang and McCarty, 2000; Sharma and McCarty, 1996; Nielsen and Keasling, 1999; Cope and Hughes, 2001; Hughes, 2001). One benefit of applying ERD in high concentration regimes is related to the natural surfactant effect that usually accompanies this technology. When the groundwater equilibrium is altered, the transfer of more constituent mass from the free or adsorbed phase into the dissolved phase should be expected. An increase in the levels of dissolved constituents in groundwater results in an increase in a more treatable portion of the total CAH mass. This effect can be used by itself or in conjunction with other ongoing technologies (such as pump and treat) to reduce treatment life span and costs. Care needs to be taken that desorption does not result in the vertical or horizontal migration of elevated dissolved concentrations away from the treatment area (i.e., expansion of the constituent plume).

The possibility of enhancing migration to off-site areas or sensitive receptors is even more pronounced when applying ERD in a potential DNAPL environment. Therefore, prior to ERD application in these settings, a clear plan to address these possibilities must be available. This could include application of the technology on an "outside-in" approach in which the lower concentration areas downgradient of the areas of higher concentration are treated initially, or an

approach in which the high concentration area is avoided altogether and the technology is applied in a containment role.

However, if properly accounted for, the possibility of concentration increases and/or migration of the impacts can be overcome and ERD can successfully be applied in these settings, especially when very high biological treatment rates can be established. The application of ERD in these areas will increase the levels of mass reduction in the subsurface, and once the initial disruption in phase equilibrium is overcome it can be expected that the technology will provide greater control of constituent migration from the source area.

Although it was not originally designed as a study of source zone treatment, evidence presented in Section 4.3 suggests that this demonstration was successfully operated in a source area.

#### 4.3.9.6 Secondary Water Quality Impacts

As discussed in Section 4.3.5, secondary water quality impacts (including metals mobilization and high COD/BOD) were observed but as expected were limited to the area of the reactive zone and did not appear to be significant downgradient. Although ketones were generated as metabolic byproducts of molasses biodegradation, they did not appear to pose an appreciable risk.

## 5. Cost Assessment

## 5.1 Cost Reporting

An evaluation of costs specific to the demonstration and an estimate of unit costs will be provided in the Cost and Performance Report at the completion of the project, when final project financial information is available. At that time, we will also discuss costs associated with the demonstration that would be expected to differ at full-scale. A cost breakdown for a hypothetical case is provided in Section 5.2.4 and Table 5-1, using the Level 2 and 3 work breakdown structure given in U.S. Army Corps of Engineers, 2001 and elements of the cost tracking guidance provided in ESTCP, October 2002. Information is also presented in this section regarding cost comparisons between ERD and other technologies, based on our experience and that of others as presented in the literature.

#### 5.2 Cost Analysis

In general, CAH plumes in groundwater may take one of three forms:

- Pure dissolved phase contamination
- Sorbed or emulsified source material with a dissolved phase plume
- Free phase (pumpable) DNAPL source with a dissolved phase plume.

The second case is by far the most common and is the condition assumed to exist at Hanscom. Cost analyses presented in this section are therefore geared toward this type of plume.

Additionally, ERD can be applied in at least three configurations – as a barrier, as a plume-wide treatment, and as a spot treatment of a source area. The choice of configuration for a given site depends on a variety of technical, economic, regulatory and risk factors. However, a common ERD approach is to treat the whole plume (above specified concentrations, leaving low-concentration fringes to attenuate naturally). This approach is assumed. Those analyses presented in this section are based on a dissolved phase plume with a sorbed, emulsified source area.

#### 5.2.1 Cost Comparison

#### 5.2.1.1 Cost Comparisons at Commercial Sites vs. Pump & Treat

The best way to estimate the potential benefit of an innovative remediation technology is to evaluate its cost at sites where it has been demonstrated alongside more conventional technologies. ARCADIS has extensive experience in replacement of pump and treat systems with IRZ technology. Some examples of actual and projected savings associated with these sites are listed in Table 5-2. The geometries of the listed CAH sites are inter-comparable, being generally plume-wide or multiple-transect applications (as opposed to single linear containment barriers) and not solely source area hot spot treatments. The CAH sites also generally fall into the category of dissolved phase plumes with sorbed source material. The same type of application would likely apply at the Hanscom plume.

#### 5.2.1.2 Cost Comparisons at Commercial Sites vs. Other Innovative Technologies

Cost comparisons with other, more innovative technologies are available as well. For a South Carolina site, ARCADIS performed a cost comparison of several potentially applicable technologies (Table 5-3). The site contained a dissolved PCE/TCE plume in low-permeability, saprolitic soils. The comparison favorably portrays the application of an IRZ as a cost-competitive way of treating the contamination in the shortest predicted remedial interval.

DuPont has developed and published a computerized, controlled methodology to compare the costs of remediation for a standardized hypothetical site contaminated with PCE (Quinton et. al., 1997). The site was hypothetically established as being 1,000 ft long and 400 ft wide with free product. The DuPont study considered remediation duration, estimated engineering and flow/transport modeling costs, equipment costs, operation and maintenance, and monitoring costs when designing the controlled methodology. Following development of the comparison methodology, DuPont considered these treatment options: natural attenuation, substrate-enhanced anaerobic bioremediation (recirculating contaminated groundwater through the source area of the plume while injecting sodium benzoate as a carbon source), a biological substrate-enhanced anaerobic barrier (comparable to ARCADIS' IRZ technology), an *in-situ* permeable reactive barrier incorporating zero valent iron, and a pump and treat system with air stripping and carbon adsorption.

Natural attenuation, biological substrate-enhanced anaerobic barrier, *in-situ* permeable zerovalence iron reactive barrier, and pump and treat were evaluated as plume containment to be implemented 1,000 ft from the hypothetical spill zone. The scenario assumed that no free product removal technology would be implemented at the source area for containment technologies. Substrate-enhanced anaerobic bioremediation was evaluated as a technology that directly attacked the contamination in the spill zone.

To accurately determine and compare the costs of the listed technologies, DuPont included unit cost measure, cost elements making up the overall cost and period of time over which the cost is incurred in the actual remediation of the evaluation. The results of the evaluation from Quinton et al. are summarized in Table 5-4.

With the assumptions made during the DuPont evaluation, substrate-enhanced biobarrier (comparable to ARCADIS' IRZ technology) ranks third on cost. However, ARCADIS does not typically implement this technology as a containment technology in remedial situations where there is known to be free product in the source zone. In combination with a free product removal technology and a good knowledge of the subsurface hydrogeology, our company has found that it can more cost-effectively remove the free product and remediate the dissolved plume with our IRZ technology. It is our belief that, if DuPont's approach took this change in assumption into account, the substrate enhanced biobarrier evaluation would exchange places in the table with the recirculating source zone remedial approach to become the most cost-effective technology except natural attenuation.

Cost will certainly depend on scale, and generally the cost of the IRZ technology expressed per unit of CAH mass or gallon of water treated, decreases with increasing scale. This decrease occurs since transportation, mobilization, design and reporting costs are nearly fixed and can thus be spread over more units. This effect is generally similar for most remediation technologies, conventional or innovative.

## 5.2.2 Cost Basis

Three cost comparisons were made in the preceding section. In the first, ARCADIS compared projected costs for remediation of several sites based on estimated capital and O&M costs and the number of years required for each technology option to reach remedial goals, adjusted to present worth. The second (DuPont) cost comparison was constructed on a similar basis, but since it was based on a hypothetical site, was also extended to unit costs per volume of water and mass of contaminant treated. For a real site, the mass or volume of water treated in-situ is difficult to estimate with acceptable accuracy.

## 5.2.3 Cost Drivers

Section 2.3 provides a general discussion of cost factors associated with ERD. An even more extensive discussion of ERD cost drivers has recently been published as Sections 4 through 6 of the protocol document (Suthersan, 2002). Although a project-specific cost analysis has yet to be conducted (but will be provided in the cost and performance report), anticipated cost drivers are briefly summarized as follows:

- The majority of the costs related to reagent injection are related to labor (including preparation and support), temporary and permanent equipment, type of application (source reduction vs. plume-wide treatment), etc. The cost of the reagent material is relatively insignificant. The typical cost per pound of TOC delivered is as outlined on Table 2-2.
- Based on our experience and analysis, the two largest cost factors for ERD implementation are the injection well installation and the O&M associated with reagent injections. Three other factors that need to be given special consideration during design in order to develop the most cost effective approach for site remediation are:
  - Plume size
  - Depth of target zone
  - Magnitude of groundwater flux

## 5.2.4 Life-Cycle Costs

Extensive information about cost experience in actual practice with this technology has been provided in Appendix A of the protocol document (Suthersan, 2002). These costs are broken down into capital and O&M only.

A more detailed breakdown of life-cycle costs for a hypothetical, typical site is provided in Table 5-1, using the Level 2 and 3 work breakdown structure given in U.S. Army Corps of Engineers, 2001. The hypothetical example assumes the following conditions:

The hypothetical site is a commercial property with a TCE plume. The TCE exists largely in the dissolved phase, but residual source material remains in a sorbed, emulsified state; no pooled DNAPL remains. A combination of ERD and monitored natural attenuation (MNA) will be used to achieve cleanup standards. The ERD is targeted for the portion of the plume where TCE concentrations range from 1 to 5 mg/L. This area is approximately 400 feet in length, 100 feet in width, 20 feet in thickness, and extends to a depth of 50 feet. Groundwater velocity is 0.5 feet/day.

The portion of the plume targeted for ERD is to be treated with a 10% molasses solution, injected through 25 injection wells. Injections are performed monthly for the first two years of treatment, using mobile, trailer-mounted injection equipment. The rate of injection is then reduced to bimonthly for three additional years. The project duration is five years from the initiation of the ERD program, including three years of MNA. MNA costs are not presented here.

Table 5-1 includes capital, operating and regulatory (permitting and reporting) costs. Since costs are based on complete destruction of CAHs, no future liability costs are included.

The hypothetical site represents a whole-plume application where no DNAPL is present in the source area. This approach is potentially appropriate for the Hanscom plume, where no separate phase CAH is thought to exist. Similar applications at real sites are represented in Table 5-5.

The duration of ERD injections and MNA are of course different for each site and dependent on site conditions. The example of five years of injections, followed by three years of MNA, is typical, based on ARCADIS' experience. Treatment at many sites is much faster. At least six ARCADIS ERD sites have succeeded in reaching MCLs for target CAHs or even obtaining closure certifications within 18 months to 2.5 years after the initiation of ERD (see Appendix sections A-2.4 and A-2.8 of the protocol document [Suthersan, 2002]; Panhorst et al. [2002]; and Payne et al. [2001]).

Unit costs for remediation of the hypothetical plume using ERD were estimated on a basis of gallons of water treated. Two estimation methods were used: first, assuming that the initial volume of contaminated water is the total volume to be treated, and second, considering the flux of water through the plume area for the five-year duration of treatment. This approach provides a conservative range of costs.

- For a an aquifer with total porosity of 0.3, the initial volume of groundwater in the hypothetical plume is 240,000 cubic feet, or 1,795,200 gallons. With a total project cost of \$680,298 (from Table 5-1), the cost per initial gallon of water treated is \$0.39. This cost is overestimated, since desorption and other effects require treatment of several pore volumes of water, thereby substantially increasing the volume of water that must be treated.
- The cross-section of the plume perpendicular to groundwater flow is 100 feet wide and 20 feet thick, for a cross-sectional area of 2,000 square feet. At a velocity of 0.5 ft/day, the flux through the cross-section is 300 cubic ft/day. Over the 5 years (1,825 days) of treatment, 547,500 cubic feet, or 4,095,300 gallons of water will flow through the cross sectional area. With a total project cost of \$680,298 (from Table 5-1), the cost per gallon of water treated is \$0.17. This estimate provides a lower bound on the potential range of unit costs.

Conditions at the Hanscom site differed from the hypothetical cost example in having a variable flow direction, thereby affecting the distribution of substrate and the volume of substrate used. Under these circumstances, the relative substrate cost was likely higher than at the average ERD site, or at the hypothetical site discussed below. In particular, the hypothetical example uses 20 gallons of molasses per well per injection (see Table 5-1), with injections occurring at one-month

intervals, eventually decreasing to bimonthly intervals. In contrast, the overall molasses use at Hanscom averaged just under two injections per month, at 27 gallons of molasses per injection. Without the variability in groundwater flow direction, less molasses would have been required. Hanscom's flow velocity was also higher than at the hypothetical site (0.8 ft/day vs. 0.5 ft/day), which increases groundwater flux and the required molasses feed rate. In a full-scale application at Hanscom, it is expected that the flow field could be controlled (e.g., by shutdown or more uniform operation of the pump and treat system) and/or that substrate distribution would be less problematic due to broader coverage of the area to be remediated by multiple injection points. Thus, the injection parameters used in the hypothetical example are considered typical based on ARCADIS' experience, and are not solely based on the Hanscom example.

Based on ARCADIS' experience, actual project costs have ranged from approximately \$75,000 for a small-scale application and/or pilot study or demonstration-scale project to \$2,000,000 for a large plume treatment with a fully automated reagent injection system. Table 5-5 presents a selection of cost examples with concentration and size information. The full-scale system for the automated site included installation of over 100 reagent injection wells to provide aggressive plume-wide treatment.

Operating costs (including reagent injection, monitoring and reporting) are generally on the order of \$50,000 to \$100,000 per year. The percentage of the total costs associated with the reagent injections is typically greater than 50%. On the other hand, the actual cost of the reagent itself typically represents less than 10% of the total project cost.

The cost data presented in Table 5-5 clearly illustrate the effective nature of the ERD technology in addressing CAH contamination in groundwater. For example, two sites have been completed with "no further action" notifications from the regulatory agencies, for less than \$500,000 each.

## 6. Implementation Issues

#### 6.1 Environmental Checklist

Potential regulations that affect the ERD demonstration are limited to those addressing *in-situ* remediation technologies. These regulations include underground injection control (UIC) permit issues and the products of the ERD treatment process. There are no unusual issues involving the transport, storage or disposal of wastes and treatment residuals. The standard issues of drill cuttings produced during injection well installation and purge water produced during well sampling may apply.

The amount of interaction with regulatory agencies required to execute the ERD projects is sometimes substantially greater than with traditional technologies, until a particular regulatory agency becomes familiar and comfortable with these technologies. However, the technology has been successfully permitted in numerous jurisdictions and the regulatory community's experience base is growing. ARCADIS currently has more than 130 IRZ projects in 26 states that are complete or underway. Of these, 30 are full-scale implementations, and five have achieved closure status. Reagents approved for use at various ERD sites include molasses, corn syrup and whey.

Many states regulate the injection of materials into the subsurface and may require a Safe Drinking Water Act-mandated UIC permit prior to implementing the technology. The UIC permit includes information regarding the chemical nature of the substrate solution, and addresses potential concerns with water quality resulting from the injection process. Typically, the carbohydrate reagents recommended are food-grade, contributing to the rapid acceptance of the technology. UIC permitting for injection of carbohydrates is generally waived or is implemented with minimal paperwork (for example, permitting by rule). This issue is not considered to be a major impediment to ERD implementation.

Previous experience with state regulatory agencies where ERD technology has been performed indicates that an initial meeting to establish the proposed course of action for the project is the most effective process. The concerns of the UIC permit staff at state regulatory agencies must be addressed at the onset of the project to avoid delays. Usually, the information required to satisfy the requirements of the UIC permits is readily available, and should not represent a major regulatory hurdle. Continued close communications with the regulatory agencies during the planning and execution of ERD greatly increase the potential for a successful demonstration. A teaming relationship with the local environmental regulatory agencies is important to technology success.

Public participation during the technology process should be addressed on a site-specific basis. Inquiries on behalf of public entities should be addressed in a timely manner by the project management. The ERD technology is a relatively straightforward and non-threatening process, and thus it is anticipated that any public communications will be favorably received.

#### 6.2 Other Regulatory Issues

Hanscom AFB is a CERCLA (Superfund) Site, regulated by the United States Environmental Protection Agency (USEPA). Remedial Investigations and Remedial Actions have been ongoing at the facility since 1985. A pre-NPL "Remedial Action Plan" for what is now NPL Operable Unit 1 (which includes Site 1) was approved and implemented in 1987. Subsequently, in October 2000, an Interim Record of Decision was issued for NPL Operable Unit 1 to continue operation of the existing dynamic groundwater collection and treatment system. MADEP concurred with the Interim Record of Decision.

Interactions with USEPA and MADEP are handled by Hanscom AFB personnel. Correspondence, reports, and work plans are routed through Hanscom AFB personnel to the regulatory agencies. Approval of the project work plan was granted by the USEPA in a January 4, 2000 letter (Michael Barry of USEPA to Tom Best of Hanscom AFB), with MADEP deferral. Because of the demonstration status of the project, EPA approved its implementation without a UIC permit. The addition of a potassium bromide tracer to the reagent was also approved by the USEPA in a separate communication.

A formal briefing on the progress of the project has not yet been conducted with regulators. However, summary information was recently provided to the USEPA in Hanscom's *Five Year Review Report* (August 2002), with which USEPA has indicated its concurrence. The report recommended continuance of ongoing efforts in the demonstration area to reduce source area contamination and to support a final groundwater remedy.

Since this demonstration was entirely within the Hanscom Field/Hanscom AFB boundaries and not in a residential area there was no formal public participation requirement. Hanscom AFB personnel are responsible for attending and presenting at regularly scheduled public meetings the overall remediation problem at the Base. Hanscom AFB communicates environmental information to the public through its website <u>http://www.hanscom.af.mil/</u> and through the availability of documents at the Bedford Town Library. Hanscom also has a Restoration Advisory Board (RAB) and works with officials in the four surrounding communities on environmental issues.

#### 6.3 End-User Issues

The IRZ/ERD technology is being implemented at over 130 commercial and government-owned facilities and, within the limits recommended in Table 3-2, may potentially be applied to any groundwater CAH plume. CAH contamination is a common remediation concern at DoD installations. EPA has estimated that more than 3,000 DoD sites in the US alone are contaminated with chlorinated hydrocarbons. The information in the FY 2001 DERP Annual Report to congress, which is organized by site type, was used to estimate that DoD has 6,800 total solvent sites of which 2,300 are sites "in progress". ARCADIS' applications of ERD to Federal sites are detailed in Section 6.3.3.

#### 6.3.1 Secondary Water Quality and Gas Production Issues

Secondary water quality and gas production issues as they relate to stakeholders and end-users are discussed in detail in Section 1.4. Briefly, the following issues were identified:

- The production of intermediate products of CAHs as the ERD process converts more highly chlorinated CAHs to less chlorinated and eventually non-chlorinated end products. The cascading reactions can result in the production of vinyl chloride. The production of vinyl chloride or other intermediate products is considered a temporary situation and does not represent a major impediment to the technology but should be monitored during application of the technology.
- Gases such as methane, hydrogen sulfide, and carbon dioxide may be produced and may
  potentially migrate and/or accumulate in the vadose zone. Since engineering solutions are
  fairly easily implemented, this issue is not considered to be a major impediment to
  technology implementation.
- The by-products of substrate consumption may include metals mobilized from the solid phase, COD, BOD, TDS, taste, odor, and sulfides. However, these impacts are typically limited to the reactive zone itself. Furthermore, it is generally believed that metals will be reprecipitated or immobilized downgradient of the reactive zone when the conditions return to their preexisting state and within the IRZ area some time after system shutdown.
- Molasses itself is a potential source of metals. The molasses-water mixture did not produce secondary water quality issues in this demonstration (see Section 4.3.5). However, this is a potential issue that should be briefly considered in the design phase for IRZ projects. The paucity of available data suggests that further work should be done to explore the metallic content of different sources of molasses.

Thus the potential for secondary water quality impacts needs to be fully identified and addressed during design and in consultation with all applicable regulatory agencies and the public.

## 6.3.2 Procurement and Implementation Issues

Equipment required for technology implementation as applied at Hanscom, described in Sections 2.3 and 3.5.1, is non-specialized and readily available. System design must be customized for each application to account for regulatory and site conditions, hydrogeological and geochemical characteristics, but the elements of a batch-fed IRZ/ERD are available commercially-off-the-shelf (COTS) and through subcontract with laboratories, drilling contractors, etc. As summarized in Table 4-2, ERD technology is relatively easy to implement and beyond the design phase and should generally only require environmental technicians for field implementation and maintenance. Automated systems and those involving extraction/ reinjection systems require custom design, and the ease of implementation of such systems is design-dependent.

The primary scale-up issue is the addition of injection wells to expand the IRZ, based on the geometry of the IRZ as determined during the field pilot test. If the number of injection wells required is excessive, or if drilling costs are prohibitive due to depth or difficult geological conditions, scaling up could pose significant hurdles. However, such barriers are usually foreseen before a pilot test is implemented.

Licensing is required to apply the technology. ARCADIS is the owner of Contractor Patented Technology for the in-situ addition of carbohydrate substrate material to create reactive zones for the removal of chlorinated hydrocarbons from groundwater as set forth in U.S. Patent Nos. 6,143,177 and 6,322,700. ARCADIS and the U.S. Government have agreed for ARCADIS to grant to the Government a paid up, non-exclusive limited license for government owned facilities only (this agreement is related to the demonstration effort "In situ Substrate Addition to Create Reactive Zones for Treatment of Chlorinated Aliphatic Hydrocarbons" under Contract No. F41624-99-C-8032). No rights to assign, sublicense or other ownership interests are to be conveyed therein, nor shall the License apply to any other patented technology that is owned or licensed by ARCADIS. To discuss application of this technology at government sites please contact "Government contact name and #" and Chris Lutes of ARCADIS in Durham, NC at 919-544-4535 or clutes@arcadis-us.com. For commercial application please contact ARCADIS only.

The technology was customized for the demonstration only in the sense that all sites require a customized system design and adjustment of operations as monitoring data is gathered. As implied by the widespread existing use of the technology (see Section 2.2), it has already been successfully commercialized and transferred.

#### 6.3.3 Transition

The in situ reactive zone (IRZ) bioremediation technology discussed herein was developed primarily in the private and sector and has been applied (at more than 130 sites) to treat metals and CAHs. These sites involved regulators and a variety of site conditions in several different geographic areas of the country. The technology is mature as a plume remediation strategy or barrier strategy and ready to transition to commercial application in the DoD. It has been used successfully in "source" type zones up to 150 mg/L initial TCE (see Section A.2.13 of the protocol document [Suthersan, 2002]). It is applicable to a wide range of contaminants and geological conditions (see Sections 1 and 2 of Suthersan, 2002). It is not, however, a "silver bullet" applicable to all sites and all contaminants and/or mixtures of contaminants. Additional demonstrations are not necessary for treatment of groundwater plumes but would provide useful data to further elucidate applicability to varying conditions and/or contaminants. Additional demonstrations are desirable to evaluate potential to various source zone architectures (for example sites with DNAPL entrapped in soil pores or present in fractured bedrock) and to evaluate different delivery mechanisms, such as recirculation wells, for deeper/thicker plumes). Additional demonstrations are also desirable to extend the technique to additional contaminants such as explosives and perchlorates. ARCADIS has demonstrated that the technique is applicable to a wide variety of subsurface conditions. However, experienced personnel familiar with IRZ must carefully evaluate each site in order to identify conditions, including adverse geochemistry, that could impact remedial design.

Finalization of the protocol "Technical Protocol for Using Soluble Carbohydrates to Enhance Reductive Dechlorination of Chlorinated Aliphatic Hydrocarbons" occurred in December 2002 and will be a major technology transfer step. While not a demonstration need, collection of case histories will provide a useful guidance tool. This process has begun as Appendix A of the protocol (Suthersan, 2002) and is expected to continue during DoD's effort to prepare the "Principles and Practices of Enhanced Anaerobic Bioremediation" document now ongoing with ARCADIS participation.

As inventor of the technique (and current patent holder), ARCADIS is the most experienced firm to apply soluble carbohydrates for the enhancement of anaerobic biodegradation. ARCADIS is already aggressively and successfully seeking to roll out the IRZ technique at other DoD and DOE facilities. ARCADIS has the following IRZ projects underway at Federal facilities:

- A pilot scale application underway this year at Fort Devens, Massachusetts (see Section 4.3.5.1) under a guaranteed fixed price contract
- A recently completed pilot study for the Navy at Naval Weapons Industrial Reserve Plant, Dallas, Texas, (see Enhanced In-Situ Biotransformation Pilot Study Report, to NFESC March 29, 2002)
- Two sites at Lompoc Federal Penitentiary where an IRZ pilot study that has been submitted for regulatory approval at a guaranteed performance contracting site (this is a DoD/FORSCOM project it was a disciplinary barracks and got transferred to Bureau Of Prisons during BRAC)
- A bench scale study of IRZ for Uranium underway under a contract with DOE NETL using samples from Fernald
- Under a recently signed guarantied fixed price contract at Fort Leavenworth KS we are planning application at two sites the work plans for these are under EPA and state review
- A planned application under a recently signed guaranteed fixed price AFCEE task order for Charleston AFB in South Carolina
- A demonstration for Bechtel Jacobs LLC at the DOE/Piketon facility for both TCE and radionuclides
- A demonstration for energetics that has been contracted through AEC/Plexus for Milan Army Ammunition Plant in Tennessee.

Perhaps one of the most likely sources of funding will be redirection of funding as IRZ approaches are substituted, with regulatory approval, for marginal or ineffective pump and treat systems. The Army's Groundwater Effectiveness Technical Evaluation Review (GWETER) program managed by the Army Environmental Center (AEC) and supported by ARCADIS demonstrates how this process works. Under this contract, ARCADIS has performed life-cycle analyses and expert technology reviews of existing groundwater pump and treat systems at ten active and inactive (Army Base Realignment and Closure [BRAC]) sites to evaluate whether or not existing remediation systems are appropriate, efficient and cost effective in achieving sitespecific goals. Success include Former Fort Ord (Monterey, CA) where plans are under way for remediating a large chlorinated plume using in-situ technologies, and an in-situ pilot test at Milan Army Depot (Milan, TN) on explosives (RDX, TNT, HMX) in groundwater. The Milan project will demonstrate extension of IRZ to destroy explosives. As noted previously, there are many potential opportunities under Army, Navy and Air Force jurisdiction to substitute IRZ for existing pump and treat systems and realize substantial savings and a shorter path to closure. The action remains for DoD contracting groups and installation restoration program managers to seek more effective solutions.

As noted previously, IRZ is a bioremediation technique utilizing an electron donor to effect conditions needed to achieve treatment objectives in situ. While IRZ is applicable to a wide variety of sites and contaminants, expert knowledge is necessary to choose those situations with greatest potential for performance success and treatment cost reduction. In the last few years the DoD has extended performance based contracts (PBCs) to include remediation projects. PBCs are nearly ideal approaches for transfer of IRZ and other similar technologies to the user. A well-written PBC scope of work describes treatment objectives but leaves remedy choice to potential remediation service providers. This paradigm shift away from "cost plus" approaches leaves DoD officials in control as performance is monitored against various milestones set by the firm providing remediation services while allowing the private sector firm flexibility to work with regulators to choose remedies that best meet site specific requirements and optimize profitability of the job. The Army, Navy and Air Force are all working PBC remediation strategies. Additional procurement guidance will be needed as lessons learned are applied. Potential for cost savings are substantial. The need to transition to PBC approaches and realize maximum benefit from PBC approaches is urgent.

This demonstration project was performed by ARCADIS. ARCADIS is a private sector firm providing remediation services to a wide variety of industrial and government clients. The work was funded in part by ESTCP and AFCEE and managed by AFCEE. ARCADIS is already aggressively marketing IRZ to industry and government sectors and seeking other private sector organizations interesting in adding this technology to their tool kit. To date ARCADIS has implemented IRZ solutions at more than 130 sites in the U. S. and abroad and is working with other private sector partners to facilitate broader application of the technique.

ARCADIS will continue ongoing IRZ marketing efforts to both private sector and government clients. The firm's objective has never been to be the low cost provider but to provide best and most cost effective solutions. IRZ is an integral part of ARCADIS' GRiP<sup>®</sup> fixed price remediation contracting approach. Ongoing efforts at AFCEE, US Army Corps of Engineers (USACE) Omaha District, Army FORSCOM and elsewhere all support IRZ technology transfer efforts by emphasizing performance and price vs. low unit cost plus fee. Contracts are already in place with these agencies that encourage adoption of innovative technologies like IRZ. Increased utilization of these contracts should be encouraged. In addition ARCADIS is very willing to cooperate with other firms in implementing this technology under other government contracts at sites where they are the lead consultant.

## 7. References

ARCADIS Geraghty & Miller. 2000. *Final Demonstration Plan and Work Plan for In-situ Substrate Addition*. Prepared for AFCEE and ESTCP, March 2, 2000, under contract F41624-99-C-8032.

Aronson, D. and Howard, P. 1997. *Anaerobic Biodegradation of Organic Chemicals in Groundwater: A Summary of Field and Laboratory Studies*. Environmental Science Center, Syracuse Research Corporation, 6225 Running Ridge Road, North Syracuse, NY 13212-2509. SRC TR-97-0223F.

Best, Tom, 2001. Personal communication with Chris Lutes ARCADIS December 18, 2001.

Bjerg, P.L., Jakobsen, R., Bay, H., Rasmussen, M., Albrechtsen, H., and Christensen, T.H. 1997. *Effects of sampling well construction on*  $H_2$  *measurements made for characterization of redox conditions in a contaminated aquifer*. Environ. Sci. Tech. 31(10).

California Department of Health Services. 2001. *Drinking Water: Overview of Monitoring Results 1984-2000*. <u>http://www.dhs.cahwnet.gov/ps/ddwem/chemicals/monitoring/results84-00.htm</u>. Last updated September 19, 2001.

CH2M Hill. *Draft Solute Transport Model Setup and Calibration Report, Operable Unit 1, Hanscom AFB.* 1997. Prepared for 66 SPTG/CEV under AFCEE Contract F41624-94-D-8053, Delivery Order 0031, December 1997 and "Draft Groundwater Flow Model Report, Operable Unit 1, Hanscom Air Force Base, July 1997, same contract.

Chapelle, F.H., Vroblesky, D.A., Woodward, J.C., and Lovley, D.R. 1997. *Practical considerations for measuring hydrogen concentrations in groundwater*. Environ. Sci. Technol. *31*(10).

FINAL REPORT, SERDP/ESTCP Expert Panel, Workshop on Research and Development Needs for Cleanup of Chlorinated Solvent Sites. 2002. http://www.estcp.org/documents/techdocs/chlorsolvcleanup.pdf.

*Fiscal Year 2001 Defense Environmental Restoration Program (DERP) Report to Congress.* 2001. <u>http://63.88.245.60/derparc\_fy01/derp/indexTen.htm</u>.

Friedman and Erdman. 1982. *Techniques of Water-Resources Investigations of the United States Geological Survey, Chapter A6.* 

Haley and Aldrich. Architect-Engineer Field Investigation Report, Sampling Round No 11, May 1998, Long Term Sampling Program, Hanscom Air Force Base Bedford Massachusetts. 1998. USA COE September 1998 File No 05833-065, September 1998.

Hanscom AFB, Environmental Flight, 66<sup>th</sup> Civil Engineering Squadron. 2002. Second Five Year Review Report for Hanscom Field/Hanscom Air Force Base Superfund Site. August 2002.

Horst, J. F., Beil, K. A., Burdick, J. S., and Suthersan, S. S. 2000. *Comparison of Natural and Enhanced Attenuation Rates through Substrate Amendments*. 2<sup>nd</sup> International Conference on Remediation of Chlorinated and Recalcitrant Compounds. Proceedings available at <a href="http://www.battelle.org/bclscrpt/Bookstore99/list\_all.cfm?OrderVar=Subject">http://www.battelle.org/bclscrpt/Bookstore99/list\_all.cfm?OrderVar=Subject</a>. May 2000.

Howard, P. et al. 1991. Handbook of Environmental Degradation Rates. Lewis Publishers, Inc.

Howard, Phillip H. 1990. *Handbook of environmental fate and exposure data for organic chemicals*. Lewis Publishers, Inc., Chelsea, Michigan,

Interstate Technology and Regulatory Cooperation (ITRC) Work Group. 2002. *DNAPL source reduction: Facing the challenge*. <u>http://www.itrcweb.org</u>. April 2002.

Interstate Technology and Regulatory Cooperation (ITRC) Work Group. 1999. *Natural attenuation of chlorinated solvents in groundwater: Principals and practices*. http://www.itrcweb.org. May 1999.

Interstate Technology Regulatory Cooperation (ITRC) Work Group. 1998. *Technical and Regulatory Requirements for Enhanced In Situ Bioremediation of Chlorinated Solvents in Groundwater*. December 23, 1998.

Lindberg, R.D. and D.D. Runnells. 1984. *Ground Water Redox Reactions: An Analysis of Equilibrium State Applied to Eh Measurements and Geochemical Modeling.* Science, p 925-927. August 31, 1984.

Microseeps. *Marketing literature and sampling instructions*. Pittsburgh, PA: <u>www.microseeps.com</u>.

Morin, T.C. and S.M. Henry "Using Intrinsic Processes to Manage Site Remediation", Environmental Technology, January/February 1998, P18-24.

National Academy of Sciences (NAS). 2000. *Natural Attenuation for Groundwater Remediation*. Prepublication Copy, <u>http:///nas.nap.edu/openbook/0309069327/html/R1.html</u>. Copyright 2000.

Panhorst, E.M., G.B. Page, J.B. Stephenson, and D. Glenn. 2002. *Expedited Closure of a PCE/TCA Site Using IRZ<sup>TM</sup> Technology*. Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA. May 20-23, 2002.

Payne, Fredrick. 2002. Personal communication with Chris Lutes (ARCADIS). 2002.

Payne, F.C., S.S. Suthersan, F.C. Lenzo and J.S. Burdick. 2001. *Mobilization Of Sorbed-Phase Chlorinated Alkenes In Enhanced Reductive Dechlorination*. In: Anaerobic Degradation of Chlorinated Solvents. Proc. Internat. In Situ and On-Site Bioremediation Symp. 6(2):53-60.

Quinton, G.E. et al. 1997. 1997. A method to compare groundwater cleanup technologies. *Remediation*, Vol. 7, No. 4, pp 7-16. *See Also* Quinton, Gary E. et al. 1997. *Cost evaluation of* 

anaerobic bioremediation vs. other in situ technologies; In situ and on-site remediation. Bioremediation, Vol. 4. p.315. From Battelle Press, Fourth Int'l symposium on in-situ and on-site bioremediation, New Orleans, LA, April 28-May 1, 1997.

Suthersan, Suthan. 2002. FINAL: Technical Protocol for Using Soluble Carbohydrates to Enhanced Reductive Dechlorination of Chlorinated Aliphatic Hydrocarbons.

U.S. Army Corps of Engineers, Washington, D.C. 2001. *A Guide to Preparing and reviewing Remedial Action Reports of Cost and Performance*. Engineering Pamphlet: EP 1110-1-19, 30. June 2001.

USEPA. 1998. Technical Protocol for Evaluating Natural Attenuation of Chlorinated Solvents in Ground Water. EPA/600/R-98/128. September 1998.

US Sugar. 2001. http://www.suga-lik.com/molasses/molasses\_frame.html.

## 8. **Points of Contact**

POINT OF	ORGANIZATION	Phone/Fax/Email	Role in Project
CONTACT			
Dr. Andrea Leeson	ESTCP	703-696-2118 (ph)	ESTCP Lead
Program Manager,	901 North Stuart Street	703-696-2114 (fax)	
Cleanup	Suite 303	andrea.leeson@osd.mil	
	Arlington, VA 22203		
Jerry Hansen	AFCEE	210-536-4353,	AFCEE Lead
Project Manager	Technology Transfer Div.	DSN 240-4353 (ph)	
	3207 Sydney Brooks	210-536-4330,	
	Brooks AFB, TX 78235-5344	DSN 240-4330 (fax)	
		Jerry.Hansen@brooks.af.mil	
Pete Palmer, P.E.,	ARCADIS G&M, Inc.	813-961-1921 (ph)	Principal Investigator
P.G.	14497 North Dale Mabry Hwy,	813-961-2599 (fax)	
	Suite 115	ppalmer@arcadis-us.com	
	Tampa, FL 33618		
Christopher C. Lutes	ARCADIS G&M, Inc.	919-544-4535 (ph)	Project Manager/Lead
1	4915 Prospectus Drive	919-544-5690 (fax)	3 6
	Suite F	clutes@arcadis-us.com	
	Durham, NC 27713		
Thomas W. Best	Hanscom AFB	781-377-4495,	Hanscom AFB Site
Environmental Engr.	66 CES/CEVR	DSN 478-4495 (ph)	Contact
_	120 Grenier Street	781-377-8545,	
	Hanscom AFB, MA 01731	DSN 478-8545 (fax)	
		Thomas.Best@hanscom.al.mil	
Michael Berry	Federal Facilities Superfund	617-918-1344 (ph)	Regulatory Contact
Remedial Project	Section	barry.michael@epa.gov	
Manager	U.S. EPA, Region 1		
-	1 Congress Street, Suite 1100		
	(HBT)		
	Boston, MA 02114-2023		
Glen Gordon, P.E.	ARCADIS G&M, Inc.	918-937-9999	ARCADIS G&M Base
	Wannalancit Office & Tech.	ggordon@arcadis-us.com	Liaison
	Ctr		
	175 Cabot Street, Suite 5400		
	Lowell, MA 01854		

Project Lead Signature:

Christopher C. Lutes, ARCADIS G&M, Inc.

Date

# Figures


Figure 1-1. Area Map of Hanscom AFB







Figure 2-2. Conceptual Design for an ERD System Layout



Figure 2-3. Reagent Mixing and Injection System Schematic



Figure 2-4. Reagent Mixing and Injection System



Figure 3-1. Overview of Hanscom AFB Showing Extraction Wells and Modeled Top of Bedrock Contours



Figure 3-2. Total VOC Concentrations in Lower Aquifer Near RAP1-6T, May 1998



Figure 3-3. Cross Section of Injection Area, A – A'



Figure 3-4. Cross Section of Injection Area, B – B'



Figure 3-5. RAP1-6 Well Cluster, Viewed From Access Road Looking Southeast



Figure 3-6. Drainage Channel behind Fence and RAP1-6 Well Cluster Viewed from Southeast Side of Drainage Channel, Looking Northwest toward Overrun for Runway 23. VER Unit at Site 1 is Visible in Distant Background.



Figure 3-7. Drainage Channel Looking Northeast



Figure 3-8. Potentiometric Surface in Lower Aquifer Near RAP1-6T, May 1998



Figure 3-9. Hanscom AFB Site Layout, Pilot Test Area



Figure 3-10. CAHs at Site 1 Pump Station (from Hanscom AFB, 2002)

## **OU-1 Boundary Interceptor Well #3**



Figure 3-11. CAHs at Interceptor Well #3 (from Hanscom AFB, 2002)



Figure 3-12. Vertical Correlation of Wells



Figure 3-13. TCE Source Cell Concentrations in the Lower Aquifer



Figure 3-14. Summary of VOC Results for Baseline Groundwater Sampling Event – June 2000

## Rolling Average Weekly Molasses Loading



Figure 3-15. Rolling Average Weekly Molasses Loading



Hanscom AFB - DO trends

4-1-(a) Dissolved Oxygen vs. Time, All Data and Injection Events



Hanscom AFB - DO trends

4-1-(b) Dissolved Oxygen vs. Time, "IRZ" Labeled Wells Only

Hanscom AFB - DO trends



4-1-(c) Dissolved Oxygen vs. Time, "RAP and B" Labeled Wells Only

Hanscom AFB - ORP trends



4-2-(a) Oxidation Reduction Potential vs. Time, All Data and Injection Events

Hanscom AFB - ORP trends



4-2-(b) Oxidation Reduction Potential vs. Time, "IRZ" Labeled Wells Only

Hanscom AFB - ORP trends



4-2-(c) Oxidation Reduction Potential vs. Time, "RAP and B" Labeled Wells Only

Hanscom AFB IRZ-INJ TOC Trend



Figure 4-3. TOC Trend in the Injection Well

Hanscom AFB - pH trends



4-4-(a) pH vs. Time, All Data and Injection Events

10 9 8 7 Нd 6 Ж 5 ж ж Ж 4 3 29-Feb-00 29-May-00 27-Aug-00 25-Nov-00 23-Feb-01 24-May-01 22-Aug-01 20-Nov-01 18-Feb-02 19-May-02 17-Aug-02 15-Nov-02 IRZ-1 ▲ IRZ-2 First Injection IRZ-3 -IRZ-4 IRZ-5 .....  $\diamond$ 

Hanscom AFB - pH trends

4-4-(b) pH vs. Time, "IRZ" Labeled Wells Only

Hanscom AFB - pH trends



4-4-(c) pH vs. Time, "RAP and B" Labeled Wells Only

1000 100 TOC (mg/l) 10 1 -IRZ-1 IRZ-4 B-239 IRZ-2 IRZ-3 IRZ-5 6S 6R 6T Monitoring Well/Sample Date ■ 6/16/2000 ■ 4/5/2001 ■ 6/6/2002 ■ 11/1/2000 ■ 5/4/2001 ■ 9/16/2002 □ 11/7/2000 □ 7/11/2001 □ 10/15/2002 □ 12/21/2000 □ 10/12/2001 □ 1/4/2001 □ 11/19/2001 ■ 1/11/2001 ■ 1/23/2002 ■ 1/31/2001 ■ 2/22/2002 □ 3/26/2001 □ 3/27/2002

Monitoring Well TOC Data - LOG SCALE

Figure 4-5. Monitoring Well TOC Data

Monitoring Well DOC Data - LOG SCALE



Figure 4-6. Monitoring Well DOC Data

## **B-239 Gradients**



Figure 4-7. Precipitation for Bedford, MA, and Gradient Between Well B-239 and Selected Wells Over Time

## **IRZ-INJ Gradients**



Figure 4-8. Precipitation for Bedford, MA, and Gradient Between Well IRZ-INJ and Selected Wells Over Time



Figure 4-9. Potentiometric Surface Measured 6-15-00 and 6-16-00



Figure 4-10. Potentiometric Surface Measured 12-21-00


Figure 4-11. Potentiometric Surface as of May 2001



Figure 4-12. Potentiometric Surface as of 11-19-01



Figure 4-13. Potentiometric Surface as of 1-11-01



Figure 4-14. Potentiometric Surface as of 1-31-01



Figure 4-15. Potentiometric Surface as of 3-19-01



Figure 4-16. Potentiometric Surface as of 4-5-01



Figure 4-17. Potentiometric Surface as of 6-18-01



Figure 4-18. Potentiometric Surface as of 7-11-01



Figure 4-19. Potentiometric Surface as of 10-12-01



Figure 4-20. Potentiometric Surface as of 1-22-02



Figure 4-21. Potentiometric Surface as of 2-22-02



Figure 4-22. Potentiometric Surface as of 5-14-02



Figure 4-23. Potentiometric Surface as of 9-16-02



Figure 4-24. Potentiometric Surface as of 9-24-02



Figure 4-25. Potentiometric Surface as of 10-1-02



Figure 4-26. Potentiometric Surface as of 10-9-02



Figure 4-27. Potentiometric Surface as of 10-15-02

## HANSCOM AFB EXTRACTION WELL FLOW RATES



Figure 4-28. Extraction Well Flow Rates

**Demonstration Period Monthly Precipitation** 



Figure 4-29. Monthly Precipitation

Hanscom AFB - Methane trends



Figure 4-30. Methane Trends



**IRZ-1 COC Response to Reagent Delivery** 

Figure 4-31. IRZ-1 COC Response to Reagent Delivery

Hanscom AFB - Ethene trends



Figure 4-32. Ethene Trends

## MW-RAP1-6T with field data



Figure 4-33. Historical VOC Trends for RAP1-6T Showing Both Laboratory and Field-Generated Data



**RAP1-6T COC Response to Reagent Delivery** 

Figure 4-34. RAP1-6T COC Response to Reagent Delivery

MW-RAP1-6T with field data



Figure 4-35. Recent VOC Trends for RAP1-6T Showing Both Laboratory and Field-Generated Data



Dilution-Corrected IRZ-INJ VOC Trends (lab data only)

Figure 4-36. Dilution-Corrected IRZ-INJ VOC Trends (Lab Data Only)



Note: On dates when both field and laboratory measurements were taken, data points for TCE and cis-1,2-DCE are averages of the field and lab results.

Figure 4-37. VOC Trends for IRZ-1 Showing Both Laboratory and Field-Generated Data

IRZ-1



Figure 4-38. VOC Trends for IRZ-1 (Lab)



Dilution-Corrected IRZ-1 VOC Trends (lab data only)

Figure 4-39. Dilution-Corrected IRZ-1 VOC Trends (Lab Data Only)



## Hydrogen Sulfide trends

Well ID

Figure 4-40. Hydrogen Sulfide Trends

H2S (mg/l)

Hanscom AFB - Conductance trends



Figure 4-41-(a) Conductance Trends vs. Time, Showing All Data and Injection Events

Hanscom AFB - Conductance trends



Figure 4-41-(b) Conductance Trends vs. Time, "IRZ" Labeled Wells Only

Hanscom AFB - Conductance trends



Figure 4-41-(c) Conductance Trends vs. Time, "RAP and B" Labeled Wells Only

Hanscom AFB - Ethene trends RAP1-6T



Figure 4-42. Ethene Trends RAP1-6T

MW-RAP1-6T - Lab Data Only



Figure 4-43. Recent VOC Trends for MW-RAP1-6T – Lab Data Only



Dilution-Corrected RAP1-6T VOC Trends (lab data only)

Figure 4-44. Dilution-Corrected RAP1-6T VOC Trends (Lab Data Only)


### Visual Observations by Well

Figure 4-45. Qualitative Observation Ratings of Samples Taken from Demonstration Area Wells Over Time

Methane Data - Log Scale



Figure 4-46. Methane Measurements from Demonstration Area Wells Over Time

Ethene Data - Log Scale



Figure 4-47. Ethene Measurements from Demonstration Area Wells Over Time



Figure 4-48. VOC Trends for IRZ-2 Showing Both Laboratory and Field-Generated Data



Figure 4-49. VOC Trends for IRZ-3 Showing Both Laboratory and Field-Generated Data



Figure 4-50. VOC Trends for IRZ-4 Showing Both Laboratory and Field-Generated Data



Figure 4-51. Trends for IRZ-5 Showing Both Laboratory and Field-Generated Data

MW-RAP1-6R with field data



Figure 4-52. Historical VOC Trends for RAP1-6R, Laboratory and Field-Generated Data

## MW-RAP1-6S with field data



Figure 4-53. Recent VOC Trends for RAP1-6S Showing Both Laboratory and Field-Generated Data



TCE:cis-1,2-DCE Ratio (concentration basis) over Time for RAP1-6R

Figure 4-54. TCE:cis-1,2-DCE Ratio (Concentration Basis) over Time for RAP1-6R



Hanscom Bromide Trends - LOG SCALE

Figure 4-55. Hanscom Bromide Trends – Log Scale





Figure 4-56. MCL Index at IRZ-1

MCL Index at RAP1-6T



Figure 4-57. MCL Index at RAP1-6T (11-14-1984 through current)

## MCL Index at RAP1-6T



Figure 4-58. MCL Index at RAP1-6T (07-24-1998 through current)

MCL Index at IRZ-Inj



Figure 4-59. MCL Index at IRZ-INJ

TCE vs. Conductance



Figure 4-60. TCE vs. Conductance at IRZ-1 and RAP1-6T (June 2000 through October 2002)

**Biodegradation Rates at IRZ-INJ** 



Figure 4-61. Biodegradation Rates at IRZ-INJ



**Biodegradation Rates at IRZ-1** 

Figure 4-62. Biodegradation Rates at IRZ-1



**Biodegradation Rates at RAP1-6T** 

Figure 4-63. Biodegradation Rates at RAP1-6T



Rates for Pre-Demonstration VOCs at RAP1-6T

Figure 4-64. Rates for Pre-Demonstration VOCs at RAP1-6T



Figure 4-65. Literature Rate Constants for TCE





Figure 4-66. Total Iron –Lab





Figure 4-67. Total Manganese – Lab



Figure 4-68. Total Arsenic in the Lower Aquifer from October 2002 Sampling Event



Figure 4-69. Dissolved Arsenic in the Lower Aquifer from October 2002 Sampling Event

10000 1000 100 Hydrogen (nM) 10 1 0.1 0.01 B-239 IRZ-1 IRZ-2 IRZ-3 IRZ-4 IRZ-5 IRZ-INJ RAP1-6R RAP1-6S RAP1-6T Well 6/14/2000 3/27/2002 **10/15/2002** 

Hydrogen Data - Log Scale

Note: 50 nM = saturated in first two monitoring rounds

Figure 4-70. Hydrogen Data



MW-B239

Figure 4-71. Concentration of CAHs at MW-B239

## MW-B239 with field data



Figure 4-72. Concentration of CAHs at MW-B239, 10-28-1995 through 09-01-2002 with Field Data

#### 500 450 400 cis-1,2-DCE þ 350 Concentration (ug/I) 300 -0 250 200 150 Ò þ **♦** $\diamond$ Ò 100 \$ ♦ $\diamond$ $\diamond$ 50 $\circ$ $\diamond$ $\sim$ Â $\diamond$ Ş Å $\diamond$ C Ø 0 5/1/1998 11/17/1998 12/22/1999 7/9/2000 1/25/2001 8/13/2001 3/1/2002 9/17/2002 6/5/1999 Date

## MW-B239 with field data

Figure 4-73. Concentration of CAHs at MW-B239, 05-01-1998 through 09-17-2002 with Field Data

# Tables

SITE	DEPTH TO GW	HYDRAULIC CONDUCTIVITY	VELOCITY	GRADIENT	TREATMENT DEPTH
Hanscom	1.2 - 2.4 m BLS	~790 cm/d	24 cm/d	0.006 m/m or	15 m BLS
	4 – 8 ft BLS	26 ft/d	0.8 ft/d	ft/ft	50 ft BLS

Table 1-1. Summary of Physical Site Characteristics

COST ELEMENT	COMMENTS		
Costs incurred before treatment			
Engineering design	See text.		
Work Plan preparation	Also include submittal and editing required for regulatory acceptance; permit preparation. Include costs for bid solicitation for subcontractors.		
Mobilization and preparatory work	Includes mobilization for injection well installation and system construction. May also include installation of additional monitoring wells if the preexisting network is inadequate.		
Well installation	Surface and subsurface structures at the facility may interfere with well design and placement. Proper planning and design can minimize these costs.		
Treatment costs			
Field supervision	Oversight of subcontractors for drilling, laboratory analyses, etc.		
Injection system	Manual batch loading of molasses into the injection well array can be performed using relatively low cost injection systems that may be truck or trailer mounted (see section 2.1 of this report and sections 4 and 6 of the protocol document, Suthersan 2002). If a permanent injection set-up is required, additional capital costs will be incurred, including a system enclosure, permanent mixing tank/equipment, automated injection pumps and valving, and controls. Additional costs for this type of system may include below grade piping to transfer the solution from the enclosure to the wells and provision of utilities (water and electric). However, a portion of the costs associated with the more permanent installation will be off-set by the lower labor and field expense costs associated with the manual batch injections.		
Substrate (food grade carbohydrate)	As mentioned in the text and in the protocol, these costs are relatively low on a per pound basis but can become substantial if a site requires high doses due to high flow or electron acceptor load. Feed rates are discussed in detail in protocol sections 4.3,4.5 and 5.3 (Suthersan 2002)		
Labor, O&M	Automated loading of molasses into the injection well array will require more control equipment, but will reduce operations and maintenance costs.		
Sampling and sample analysis	Labor required to collect groundwater samples from the treatment area. as well as costs for shipping, analysis and data interpretation should be included.		
Utility costs	The main requirement should be a readily available source of potable water, preferably with a large flow rate near the site. Fuel for vehicles and electrical power or gasoline for pumps is also likely to be required but in small quantity.		
Other costs	These include disposal of drill cuttings and purge water. During application of ERD, process waste is limited to disposal of contaminated groundwater generated during well purging.		
Interim reporting	Technical performance and financial interim reports are normally required		
After treatment costs			
Final reporting	Reports documenting system performance must be prepared for site closure.		
Demobilization (equipment, material, and personnel.	Must include labor and subcontractor costs required to remove any equipment or surface facilities associated with the demonstrations. It must be assessed if injection and monitoring wells need to be removed/abandoned. Some site restoration can typically be anticipated.		

Reagent	Range of Costs (Per Pound of TOC)			
	Low		High	
Molasses (Food Grade)	\$	0.25	\$	0.60
Corn Syrup	\$	0.25	\$	0.44
Whey (Powder)	\$	1.17	\$	1.33
Sodium Lactate	\$	1.25	\$	1.46

 Table 2-2.
 Summary of Reagent Cost Ranges for Selected Soluble Carbohydrates

## Table 3-1. Performance Objectives

Type of Performance Objective	Primary Performance Criteria	Expected Performance (Metric)	Actual Performance Objective Met?
Qualitative	1. <u>Technology Evaluation</u> - Gather information (for estimation of long-term treatment effectiveness, life span and costs) to use in a protocol for use of ERD technology for CAHs at DoD facilities	Collection of extensive performance data	Yes
Quantitative	2. <u>Reduce Time to Remediate</u> - Demonstrate the ability of ERD to remediate contaminants in the subsurface over a relatively short time period	1 to 5 years in typical full-scale applications	Time was limited but results support this metric
Quantitative	<ul> <li>3. <u>Contaminant Reduction (%)</u> - Reduce total CAH concentrations from baseline levels of</li> <li>a) &gt;200 ppb</li> <li>b) 50 to 200 ppb</li> <li>c) &lt;50 ppb</li> </ul>	a) 80% in 1 year b) 75% in 1 year c) 50% in 1 year	Yes for TCE; Qualified Yes for cis- DCE; in a limited area
Qualitative	4. <u>Prevent "Stalling"</u> - Demonstrate that degradation of CAHs by ERD does not stall at undesirable by-products (cis-DCE and/or VC)	Reduction of cis-DCE, VC after initial production, production of ethene	Yes in limited area
Type of Performance Objective	Secondary Performance Criteria	Expected Performance (Metric)	Actual Performance Objective Met?
Quantitative	5. <u>Geochemistry Manipulation</u> - Demonstrate the ability of ERD to enhance the anaerobic and reducing environment in groundwater where anaerobic conditions prevail	DO to <1 mg/L ORP <50 mV	Generally yes; anaerobic environment created within reactive zone
Quantitative	6. <u>Contaminant Mobility</u> - Evaluate the ability of ERD to desorb CAHs from aquifer materials	Presence of "spike" in concentration after initial injections	Yes in limited area
Quantitative	7. <u>Contaminant Reduction (Rate)</u> - Evaluate degradation rates before & after treatment	Calculate k	Yes

Type of Performance Objective	Secondary Performance Criteria	Expected Performance (Metric)	Actual Performance Objective Met?
Qualitative	8. <u>System Performance Optimization</u> - Determine optimal strengths and frequency of reagent delivery for the site	$eq:linear_line$	Generally yes; continuously "tuned" system to metrics, determined required strength, frequency of injections (see Sections 4.3.2.1, 4.3.7.2)
Quantitative	9. <u>Hazardous Materials</u>	Potentially hazardous materials limited to soil cuttings from well drilling and purge water	Yes; no other haz. materials generated
Qualitative	10. <u>Reliability</u>	No significant reliability issues anticipated	
Qualitative	11. <u>Ease of Use</u>	Field implementation (substrate delivery) requires an environmental technician with 40 Hour HAZWOPER training, and office support from degreed scientists or engineers	Yes
Qualitative	12. <u>Versatility</u>	ERD can be used for other applications (e.g., metals, perchlorate) and under variable site conditions	N/A
Qualitative	13. <u>Maintenance</u>	Maintenance limited to occasional well development, normal equipment maintenance by technician	Yes
Qualitative	14. Scale-Up Constraints	Scale-Up Constraints Scale-up hasn't occurred at this site	

Site Characteristic	Suitable for IRZ	Unsuitable for IRZ	Hanscom AFB
Aquifer hydraulic conductivity	> 1 ft/day	<0.01 ft/day	26 ft/day
Groundwater velocity	30 ft/year - 5 ft/day	< 30 ft/year, $> 5$ ft/day	0.8 ft/day
pH	6.0-8.0	< 5.0, > 9.0	5.7-7.1
Natural attenuation of CAHs	Slow, complete degradation, or stalled degradation	No degradation	Slow
DNAPL presence	None, or emulsified, sorbed, or residuals	IRZ not appropriate for targeting pooled DNAPL at this point in technology development	Although the demonstration site was believed to be well downgradient of the primary source area, and the initial dissolved phase concentrations did not indicate DNAPL according to the conventional definition (1-2% of solubility), later results suggested the presence of a source in the demonstration area.
Sulfate	< 700 ppm		39 ppm, max
Redox	Aerobic or borderline	Anaerobic with sufficient TOC	Borderline: DO of 0.5 to 1 mg/l, ORP of -60 to 200 mv
Depth of Target Zone		>50 feet can become expensive	50 feet
CAH concentration	Non-toxic	Toxic	Non-Toxic

# Table 3-2. Suitability of Hanscom AFB Site Screening Characteristics for IRZ Implementation
		Max 00	0.1.07	0 00	NI		4	1.1.04	NI04	1	Max 07	NA- 00
KAP1-6R	Feb-86	Iviar-86	Oct-87	Sep-88	NOV-90	Feb-91	Aug-91	Jul-94	NOV-94	Jun-96	May-97	May-98
PCE	11	NS	1	1	5.4	9.6	14	1	1	1	1	1
TCE	820	NS	730	1400	570	670	760	1100	1400	1800	1600	1800
1,1,1-TCA	70	NS	36	1	22	50	32	1	1	1	1	1
1,2-DCA	8.5	NS	1	1	14	36	10	1	1	1	1	1
1,2-DCE	2600	NS	2900	5000	2600	680	1800	3500	3700	4932	4800	4800
1,1-DCE	48	NS	42	60	37	64	95	69	110	130	130	1
1,1-DCA	60	NS	77	140	84	140	130	130	190	240	230	1
VC	<u>67</u>	NS	220	340	<u>170</u>	460	300	<u>600</u>	<u>710</u>	<u>990</u>	850	<u>1300</u>
Total VOCs	3685		4007	6943	3502	2110	3141	5402	6113	8095	7613	7905
Concentration	Data/Lower A	quifer										
RAP1-6T	Feb-86	Mar-86	Oct-87	Sep-88	Nov-90	Feb-91	Sep-91	Jul-94	Nov-94	Jun-96	May-97	May-98
PCE	6.6	7.1	1	1	3.8	1	13	1	1	1	1	1
TCE	650	550	340	430	450	600	1400	1800	2500	2200	2000	1600
1,1,1-TCA	300	290	130	120	130	160	62	1	1	30	1	1
1,2-DCA	17	18	1	1	9.7	58	11	1	1	1	1	1
1,2-DCE	5000	6000	3200	3000	2700	1000	2900	4200	5000	6200	6600	3800
1,1-DCE	96	84	42	1	61	68	140	1	1	210	220	1
1,1-DCA	190	160	110	98	120	130	220	1	270	350	350	1
Vic	780	760	510	560	570	660	1200	950	1200	1600	1600	1
VC												

Table 3-3. Pre-demonstration Project Concentrations of Volatile Organic Compounds in Wells RAP1-6R and RAP1-6T

PCE	tetrachloroethene
TCE	trichloroethene
1,1,1-TCA	1,1,1-trichloroethane
1,2-DCA	1,2-dichloroethane
1,2-DCE	1,2-dichloroethene
1,1-DCE	1,1-dichloroethene
1,1-DCA	1,1-dichloroethane
VC	vinyl chloride
NS	not sampled

Notes: All sample results are reported in micrograms per liter (ug/L) Sample results shown as 1, represent concentrations less than the detection limit of 1 ug/L.

# Table 3-4. Soil Analysis Summary

	IRZ-Inj	HAN-SS-IRZ-1	IRZ-1 Dup	IRZ-2	IRZ-3	IRZ-3 Dup	IRZ-4	TB-01
Analysis	(44-46)	(44-45)	(44-45)	(39-41)	(44-46)	(44-46)	(44-46)	
Volatile Organic Compounds (ug/kg)								
1,2-Dichloroethene (total)	19	28	NA	18	18	22	8.5	1 U
Vinyl chloride	10 U	2.4 J	NA	1.1 J	0.83 J	2.4 J	11 U	1 U
Acetone	12 J	21 J	NA	27 U	16 J	19 J	18 J	10 U
Carbon disulfide	5.1 U	5.6 U	NA	5.5 U	1.4 J	1.1 J	5.5 U	1 U
1,1-Dichloroethane	2.6 J	2.2 J	NA	2.2 J	4.5 J	5.3 J	3.4 J	1 U
cis-1,2-Dichloroethene	19	28	NA	18	18	22	8.5	1 U
Trichloroethene	6.9	6.7	NA	5 J	6.8	6.2	3.3 J	1 U
Total Organic Carbon (mg/kg)	2000 U	1900 B	1570 B	2000 U	770 B	NA	2000 U	NA
	Gray silty gravel	Gray silty sand	Gray poorly graded gravel with silt and	Gray silty sand	Gray silty sand		Gray silty sand	
Grain Size Description	with sand	with gravel	sand	with gravel	with gravel		with gravel	

#### Notes:

Only detected compounds listed

VOCs analyzed by method SW846 8260B, by Severn Trent Laboratories, Inc. of Tampa, FL

TOC analyzed by method SW846 9060, by Severn Trent Laboratories, Inc. of Tampa FL

Grain size analysis performed by Engineering Consulting Services, Ltd. Of Research Triangle Park, NC

J = Estimated result

B = Estimated result

NA = Not analyzed

	DO	ORP	PH	NITRATE	SULFATE	SULFIDE	CO <sub>2</sub>	METHANE
SITE	(mg/l)	(mv)	(s.u.)	(mg/l)	(mg/l)	(mg/l)	(mg/l)	(µg/l)
Hanscom	0.35	-57.5	5.73	ND	21.5	ND	9.4	15.0
	1.48	200	7.10		38.9	0.1	86.2	138.8

Table 3-5. Summary of Initial, Background Biogeochemical Data (min/max)

Week	Dates (from -	to)	Event
1	10/9/2000	10/15/2000	Initial injection 10-11-00
2	10/16/2000	10/22/2000	Process monitoring #1 w/TOC 10-18-00
3	10/23/2000	10/29/2000	Process monitoring #2 w/TOC 10-25-00
4	10/30/2000	11/5/2000	Process monitoring #3 w/TOC 11-1-00
5	11/6/2000	11/12/2000	Abbreviated monitoring #1 w/TOC, sulfide, Br, CAHs 11-6/7-00, injection #2 11-8-00
6	11/13/2000	11/19/2000	
7	11/20/2000	11/26/2000	Injection #3 11-21-01
8	11/27/2000	12/3/2000	Process monitoring #4 w/TOC 11-30-00
9	12/4/2000	12/10/2000	
10	12/11/2000	12/17/2000	
11	12/18/2000	12/24/2000	Process monitoring #5 w/TOC, injection #4 12-21-00
12	12/25/2000	12/31/2000	
13	1/1/2001	1/7/2001	Process monitoring #6 w/TOC Br- 1-4-01
14	1/8/2001	1/14/2001	Process monitoring #7 w/TOC Br- 1-11-01
15	1/15/2001	1/21/2001	Injection #5 1-15-01
16	1/22/2001	1/28/2001	
17	1/29/2001	2/4/2001	Process monitoring #8 w/TOC, Br-, injection #6 1-31-01
18	2/5/2001	2/11/2001	
19	2/12/2001	2/18/2001	
20	2/19/2001	2/25/2001	Injection #7 with clean water push 2-20-01
21	2/26/2001	3/4/2001	
22	3/5/2001	3/11/2001	
23	3/12/2001	3/18/2001	
24	3/19/2001	3/25/2001	Process monitoring #9, injection #8 with clean water push 3-19-01
25	3/26/2001	4/1/2001	Injection #9 with clean water push 3-26-01 with sampling.
26	4/2/2001	4/8/2001	
27	4/9/2001	4/15/2001	Injection #10 with clean water push, abbreviated monitoring #2 4-9-01
28	4/16/2001	4/22/2001	
29	4/23/2001	4/29/2001	Injection #11 with clean water push 4-23-01
30	4/30/2001	5/6/2001	5/3-5/8 midpoint full monitoring round and installation of IRZ-5
31	5/7/2001	5/13/2001	IRZ-5 developed 5-8-01, injection #12 with clean water push 5-9-01
32	5/14/2001	5/20/2001	
33	5/21/2001	5/27/2001	Injection #13 with clean water push 5-22-01.
34	5/28/2001	6/3/2001	
35	6/4/2001	6/10/2001	Injection #14 with clean water push 6-4-01.
36	6/11/2001	6/17/2001	
37	6/18/2001	6/24/2001	Injection #15 with clean water push 6-18-01 along with water surface measurements.
38	6/25/2001	7/1/2001	
39	7/2/2001	7/8/2001	
40	7/9/2001	7/15/2001	Injection #16 with water push 7-12-01 with abbreviated monitoring #3

# Table 3-6. History Log of Demonstration

Week	Dates (from - to	)	Event
41	7/16/2001	7/22/2001	
42	7/23/2001	7/29/2001	Injection #17 with water push 7-25-01.
43	7/30/2001	8/5/2001	
44	8/6/2001	8/12/2001	Injection #18 with water push 8-7-01.
45	8/13/2001	8/19/2001	
46	8/20/2001	8/26/2001	Injection #19 with water push 8-22-01.
47	8/27/2001	9/2/2001	
48	9/3/2001	9/9/2001	Double injection #1 (Injection #20) with standard water push 9-7-01.
49	9/10/2001	9/16/2001	
50	9/17/2001	9/23/2001	Double injection #2 (Injection #21) with standard water push 9-19-01.
51	9/24/2001	9/30/2001	
52	10/1/2001	10/7/2001	Double injection #3 (Injection #22) with standard water push 10-3-01.
53	10/8/2001	10/14/2001	Process monitoring #10 10/12/01
54	10/15/2001	10/21/2001	Water push 10-16-01.
55	10/22/2001	10/28/2001	
56	10/29/2001	11/4/2001	Double injection #4 (Injection #23) with standard water push 10-30-01.
57	11/5/2001	11/11/2001	
58	11/12/2001	11/18/2001	11-13-01 Injection #24 (single)
59	11/19/2001	11/25/2001	Abbreviated monitoring #4 11-19-01
60	11/26/2001	12/2/2001	Double injection #5 (Injection #25) with double water push 11-26-01.
61	12/3/2001	12/9/2001	
62	12/10/2001	12/16/2001	Double injection #6 (injection #26) with double water push 12-11-01.
63	12/17/2001	12/23/2001	
64	12/24/2001	12/30/2001	Double injection #7 (injection #27) with a little more than a single push 12- 26-01.
65	12/31/2001	1/6/2002	
66	1/7/2002	1/13/2002	Double injection #8 (injection #28) with single push 1-8 and 1-9-02.
67	1/14/2002	1/20/2002	
68	1/21/2002	1/27/2002	Process monitoring #11, 1-22-02. Double injection #9 (injection #29) with single push 1-23-02.
69	1/28/2002	2/3/2002	
70	2/4/2002	2/10/2002	Single injection #25 (injection #30) with single push 2-5-02.
71	2/11/2002	2/17/2002	
72	2/18/2002	2/24/2002	Double injection #10 (injection #31) with half push 2-19-02. TOC/DOC/Br sampling 2-22-02.
73	2/25/2002	3/3/2002	
74	3/4/2002	3/10/2002	Water push only (#2) 3-5-02. Limited process monitoring (#12).
75	3/11/2002	3/17/2002	pH on IRZ-INJ and single molasses injection/single push #26 (injection #32) 3-12-02.
76	3/18/2002	3/24/2002	Half injection with water push #1 (injection #33) 3-19-02.
77	3/25/2002	3/31/2002	3-27 and 3-28-02: abbreviated monitoring #5 with gas sampling
78	4/1/2002	4/7/2002	Single injection #27 (Injection #34) with single push 4-2-02
79	4/8/2002	4/14/2002	
80	4/15/2002	4/21/2002	Double injection #11 (Injection #35) with single push and pH on IRZ-INJ 4- 16-02

Week	Dates (from -	to)	Event
81	4/22/2002	4/28/2002	
82	4/29/2002	5/5/2002	Single injection #28 (Injection #36) with single push and pH on IRZ-INJ 4-30- 02.
83	5/6/2002	5/12/2002	
84	5/13/2002	5/19/2002	Water level measurements on 5-14-02.
85	5/20/2002	5/26/2002	Double injection #12 (Injection #37) with water push and pH on IRZ-INJ 5- 20-02.
86	5/27/2002	6/2/2002	
87	6/3/2002	6/9/2002	Single injection #29 (Injection #38) with water push and limited process monitoring #13 6-6-02.
88	6/10/2002	6/16/2002	
89	6/17/2002	6/23/2002	Single injection #30 (Injection #39) with water push and pH on IRZ-INJ 6-18- 02.
90	6/24/2002	6/30/2002	
91	7/1/2002	7/7/2002	Single injection #31 (Injection #40) with water push and pH on IRZ-INJ 7-2- 02.
92	7/8/2002	7/14/2002	
93	7/15/2002	7/21/2002	Single injection #32 (Injection #41) with water push and pH on IRZ-INJ 7-16-02.
94	7/22/2002	7/28/2002	
95	7/29/2002	8/4/2002	Lawnmower damage discovered 7-29-02. 7-30-02 single injection #33 (inj. #42), with pH. Single push on 8-1-02
96	8/5/2002	8/11/2002	
97	8/12/2002	8/18/2002	8-12-02 200 gallon water push.
98	8/19/2002	8/25/2002	
99	8/26/2002	9/1/2002	Double injection #13 (Injection #43) with water push and pH in IRZ-INJ 8-28- 02.
100	9/2/2002	9/8/2002	
101	9/9/2002	9/15/2002	
102	9/16/2002	9/22/2002	Water level and TOC/Br samples 9-16-02. Double injection #14 (Inj. #44) w/water push and IRZ-INJ pH 9-18-02.
103	9/23/2002	9/29/2002	Double injection #15 (Injection #45) with water push and pH in IRZ-INJ 9-24- 02.
104	9/30/2002	10/6/2002	Double injection #16 (Injection #46) with water push 10-2-02, pH in IRZ-INJ and water levels on 10-1-02.
105	10/7/2002	10/13/2002	Single injection #33 (Injection #47) with water push 10-9-02, pH in IRZ-INJ and water levels.
106	10/14/2002	10/20/2002	10-14 through 10-16-02 Final sampling round.
107	10/21/2002	10/27/2002	
108	10/28/2002	11/3/2002	10-29-02 Remeasurement of final round field parameters.

End of Table 3-6. History Log of Demonstration

Parameter	Analytical Method	Concentration Units Reported In	Volume, Container, Preservative & Storage Requirements	Hold Time	Parameter Included also In Abbreviated Monitoring Events?	Location of test/ Firm
Temperature	ARCADIS SOP D1 (based on EPA 170.1)	Degrees C	NA	Analyze immediately	Y	ARCADIS in the field
ORP	See appendix 'field procedures' & 'instrument calibration procedures'	mV	NA	Analyze immediately	Y	ARCADIS in the field
Dissolved Oxygen	ARCADIS SOP D5 (Based on EPA 360.1)	mg/L	NA	Analyze immediately	Y	ARCADIS in the field
РН	ARCADIS SOP D2 (based on EPA 150.1)	S.U.	NA	Analyze immediately	Y	ARCADIS in the field
Specific Conductance	ARCADIS SOP D3 based on standard methods for examination of water & wastewater, 15 <sup>th</sup> edition method 205 & USEDA method 120.1	microsiemens/cm	NA	Analyze immediately	Y	ARCADIS in the field
Alkalinity	310.1	mg/L	250 mL Glass or plastic Cool to 4 °C	14 days	Ν	STL

# Table 3-7. Parameters Included in Full and Abbreviated Groundwater Monitoring Events

Parameter	Analytical Method	Concentration Units Reported In	Volume, Container, Preservative & Storage Requirements	Hold Time	Parameter Included also In Abbreviated Monitoring Events?	Location of test/ Firm
Nitrate	300.0A	mg/L	250 mL Glass or plastic Cool to 4 °C	48 hours	N	STL
Nitrite	300.0A	mg/L	250 mL Glass or plastic Cool to 4 °C	48 hours	N	STL
Sulfate	300.0A	mg/L	100 mL Glass or plastic Cool to 4 °C	28 days	N	STL
Chloride	300.0A	mg/L	250 mL Glass or plastic	28 days	N	STL
Methane, Ethane, Ethene	Modified RSK-175, WA 1.02	ug/l	Glass VOA vials	7 days	N	Vaportech
Carbon Dioxide	WA 2.01 modified	mg/l	Glass VOA vials	7 days	Ν	Vaportech
Chemical Oxygen Demand	410.4 or 410.1	mg/L	250 mL Glass or Plastic Cool to 4 °C $H_2SO_4$ to pH<2	28 days	N	STL
Biochemical Oxygen Demand	405.1	mg/L	100 mL Glass or plastic Cool to 4 °C	48 hours	N	STL

Parameter	Analytical Method	Concentration Units Reported In	Volume, Container, Preservative & Storage Requirements	Hold Time	Parameter Included also In Abbreviated Monitoring Events?	Location of test/ Firm
Total Organic Carbon (TOC)	415.1	mg/L	100 mL Glass or Plastic Cool to 4 °C $H_2SO_4$ to pH<2	28 days	Y	STL
Dissolved Total Organic Carbon	415.1	mg/L	100 mL Glass or Plastic Cool to 4 °C $H_2SO_4$ to pH<2	28 days	Y	STL
Ammonia	350.1	mg/L	500 mL Glass or Plastic Cool to 4 °C $H_2SO_4$ to pH<2	28 days	N	STL
Sulfide	Color Chart/ Effervescence of H <sub>2</sub> S (Hach Kit 25378- 00)	mg/L	500 mL Glass or Plastic Cool to 4 °C $H_2SO_4$ to pH<2	7 days	Y	ARCADIS in the field
Total Iron	6010B and CHEMetrics kit in field	ug/L	1 L Glass or plastic HNO <sub>3</sub> to pH<2	6 months	N	STL, Also in field by ARCADIS
Total Manganese	6010B and CHEMetrics kit in field based on APHA 314C and CHEMetrics kit in field	ug/L	1 L Glass or plastic $HNO_3$ to pH<2	6 months	N	STL, Also in field by ARCADIS

Parameter	Analytical Method	Concentration Units Reported In	Volume, Container, Preservative & Storage Requirements	Hold Time	Parameter Included also In Abbreviated Monitoring Events?	Location of test/ Firm
Dissolved Iron	6010B and CHEMetrics kit in field	ug/L	1 L Glass or plastic HNO <sub>3</sub> to pH<2	6 months	N	STL, Also in field by ARCADIS
Dissolved Manganese	6010B and CHEMetrics kit in field (APHA 314C)	ug/L	1 L Glass or plastic HNO <sub>3</sub> to pH<2	6 months	N	STL, Also in field by ARCADIS
CAHs	8260	ug/L	VOA vials, no headspace HCl to pH<2; Cool to 4 °C	14 days	Y	STL
Hydrogen	RSK-196	nM/L	Special; see text Re: dissolved gas sampling	28 days	N	Vaportech
Explosives (Badger AAP only)	8330	ug/L	Glass or teflon store @ 4 °C	14 days/extract analyzed within 40 days	Y	STL
Bromide	300.0	mg/l	250 ml plastic or glass unpreserved	28 days	Y	STL

 Table 3-8. Parameters Included in Soil Monitoring Events

Parameter	Analytical Method	Concentration Units Reported In	Container & Preservative Requirements	Hold Time	Parameter Included also In Abbreviated Monitoring Events?	Location of test
Total Organic Carbon (TOC)	9060	mg/kg	None specified	28 days	Y	STL
CAHs	8260	ug/kg	4 oz. Glass with teflon lined septa; store @ 4 °C	14 days	Y	STL
Explosives	8330	ug/kg	250 mL glass w/teflon lined septa; cool to 4 °C	14 days/extract in analyzed within 40 days	Y	STL
Grain Size	ASTM D-422	% passing	500 mL wide mouth glass or plastic (purchased by field crew)	None	Y	ECS

### Table 4-1. Performance Criteria

Performance Criteria	Description	Primary or Secondary
Technology Evaluation	Gather information to use in a protocol for use of IRZ technology for CAHs at DoD facilities	Primary
Reduce Time to Remediate	Demonstrate the ability of ERD to remediate contaminants in the subsurface over a relatively short time period	Primary
Contaminant Reduction	Reduction of baseline levels of CAHs, primarily TCE, cis-1,2-DCE, and VC at Hanscom AFB	Primary
	Enhancement of CAH degradation rates	Secondary
Prevent "Stalling"	Demonstrate that degradation of CAHs by ERD does not stall at undesirable by- products (cis-DCE and/or VC)	Primary
Geochemistry Manipulation	Demonstrate the ability of ERD to enhance the anaerobic and reducing environment where anaerobic conditions prevail	Secondary
Contaminant Mobility	a. Evaluate the ability of ERD to desorb CAHs from aquifer materials	Secondary
	b. Evaluate the propensity of ERD to mobilize metals	Secondary
System Performance Optimization	Determine optimal strengths and frequency of reagent delivery for the site	Secondary
Hazardous Materials	Identify any hazardous materials introduced or generated by ERD technology	Secondary
Reliability	Identify potential problems that may cause system shutdowns	Secondary
Ease of Use	Describe the number of people, skill level(s) and safety training required to perform injections and monitoring	Secondary
Versatility	Describe whether ERD can be used for other applications and under other site conditions	Secondary
Maintenance	Identify operations and maintenance requirements and level of training required to implement O&M	Secondary
Scale-Up Constraints	Identify engineering constraints associated with scaling up an ERD system	Secondary

### Table 4-2. Expected Performance and Performance Confirmation Methods

Performance Criteria	Expected Performance Metric (Pre-Demonstration)	Performance Confirmation Method	Actual (Post-Demonstration)							
PRIMARY CRITERIA (Performance Objectives) (Qualitative)										
Technology Evaluation	Collection of extensive performance data	Body of data from 11 monitoring wells conforms to demonstration plan	Performance data collection plan was met with few exceptions							
Prevent "Stalling"	Reduction of cis-DCE, VC after initial production, production of ethene	CAH and ethene data from wells IRZ-1 and RAP1-6T in the reactive zone	Ethene concentrations rose to more than 20 times pre-test value at IRZ- 1, 5 times pre-test value at RAP1- 6T							
PRIMARY CRITERIA (Performance Objectives) (Quantitative)										
Reduce Time to Remediate	1 to 5 years in typical full-scale applications	Evidence of contaminant reductions (% and rates) and ethene production	In the 2-year pilot, observed significant contaminant reductions and ethene production (see Sections 4.3.3.5 and 4.3.7.1), suggesting that remediation time of 5 years or less is realistic for a full- scale system							
Contaminant Reduction (%)	Total CAH concentrations reduced by at least 80% in 1 year	CAH data from IRZ-1 and RAP1- 6T, from baseline sampling through October 2002	IRZ-1: TCE reduced >95% in 5 months. Cis-DCE reductions >85% in 17 months. VC reductions 41% in 17 months. RAP1-6T: TCE reduced >80% in 1 year, cis-DCE and VC increased due to inadequate substrate delivery (see Section 4.3.7.1)							

Performance Criteria	Expected Performance Metric (Pre-Demonstration)	Performance Confirmation Method	Actual (Post-Demonstration)						
SECONDARY CRITERIA (Performance Objectives) (Qualitative)									
System Performance Optimization	$eq:linear_line$	Performance monitoring data evaluated before each injection event to determine optimal strengths and frequency of reagent delivery for the site	An anaerobic environment was created within the reactive zone with few exceptions to performance criteria (see Section 4.3.7.2). Strength and frequency of injection discussed in Sections 4.3.2.1 and 4.3.7.2						
Reliability	No significant reliability issues anticipated	Field records	Met performance metric; minor corrective actions needed for well fouling and seal leakage (Section 3.5.1)						
Ease of Use	Field implementation (substrate delivery) requires an environmental technician with 40-hr HAZWOPER training, and office support from degreed scientists or engineers	Experience from demonstration operation and other site applications	Met performance metric for substrate delivery. Geologist required for permanent well installations.						
Versatility	ERD can be used for other applications (e.g., metals, perchlorate) and under variable site conditions	Experience from other site applications	Versatility discussed in Sections 1.1, 2.1.1						

Performance Criteria	Expected Performance Metric (Pre-Demonstration)	Performance Confirmation Method	Actual (Post-Demonstration)
Maintenance	Maintenance limited to occasional well development, normal equipment maintenance by technician	Field records	Met performance metric; maintenance issues discussed in Section 3.5.1
Scale-Up Constraints	Primary scale-up issues anticipated to be efficacy of manual batch injection mode and area of influence determination	Experience from demonstration operation and other site applications	Scale-up hasn't occurred at this site, but batch injection successful, area of influence determined in Section 4.3.6.1. Scale-up issues and cost implications are discussed in Section 6.3 and in Section 5.7 of the protocol document (Suthersan, 2002)
	SECONDARY CRITERIA	(Performance Objectives)	
	(Quant	itative)	
Geochemistry Manipulation	DO to <1 mg/L ORP <50 mV	Performance monitoring data evaluated before each injection event	An anaerobic environment was created within the reactive zone (see Section 4.3.7.2)
Contaminant Mobility	Presence of "spike" in concentration after initial injections	CAH data for wells IRZ-1, RAP1- 6T	Spikes observed in TCE and cis- DCE concentrations shortly after first injection (see Section 4.3.7.2)
Contaminant Reduction (Rate)	Calculate k	K determined from long-term pre- demonstration data at RAP1-6T and from data trends at IRZ-1 and RAP1-6T	Calculated k (see Section 4.3.3.5)
Hazardous Materials	Potentially hazardous materials limited to soil cuttings from well drilling and purge water	Field records, analyses of soil cuttings	Purge water disposed of in on-site wastewater treatment system, cuttings from soil borings characterized and disposed of off- site

 Table 4-3. Bromide Tracer Data Summary

	20	00						2001						2002			
Well ID	6/15	11/7	1/4	1/11	1/31	3/26	4/7	5/4	7/11	10/12	10/30	11/19	1/23	2/22	3/27	9/16	10/15
IRZ-1	<0.027	0.14	NS	0.17	0.21	0.85	13.7	4.1	1.2	0.54	NS	0.22	0.2	<0.027	0.28	0.54	0.72
IRZ-2	<0.027	0.16	NS	NS	<0.027	NS	0.15	0.14	0.12	<0.027	0.082	0.08	<0.027	<0.027	0.14	NS	<0.027
IRZ-3	<0.027	0.15	NS	NS	<0.027	NS	0.16	<0.027	<0.027	<0.027	NS	0.095	0.11	<0.027	<0.027	NS	<0.027
IRZ-4	<0.027	0.16	NS	0.14	<0.027	NS	<0.027	<0.027	<0.027	<0.027	NS	0.091	0.16	NS	0.071	<0.027	0.096
IRZ-5	NS	NS	NS	NS	NS	NS	NS	<0.027	<0.027	0.092	NS	0.09	0.1	<0.027	0.12	0.12	0.23
RAP1-6T	<0.027	0.14	0.17	0.18	0.3	41.3	21.4	5.1	1.4	0.12	NS	0.086	0.13	<0.027	0.34	0.21	0.19
RAP1-6R	0.11	0.12	NS	NS	NS	NS	<0.027	0.13	<0.027	NS	NS	0.077	NS	NS	<0.027	NS	0.1
RAP1-6S	<0.027	<0.027	NS	NS	NS	NS	<0.027	<0.027	<0.027	NS	NS	0.1	NS	NS	0.12	NS	1.6
B239-MW	<0.027	<0.027	NS	NS	<0.027	NS	<0.027	<0.027	<0.027	NS	NS	<0.027	NS	NS	<0.027	NS	<0.027

NS = Not Sampled

Results in units of mg/L

Constituent	Date	B239-MW	RAP1-6R	RAP1-6S	IRZ-INJ	RAP1-6T	IRZ-1	IRZ-4	IRZ-2	IRZ-3	IRZ-5
BOD (mg/l)	06/16/00	3 U	3 U	3 U	3 U	3 U	3 U	3 U	3 U	3 U	NA
	05/07/01	3 U	3 U	3 U	10800	280	320	3 U	3 U	3 U	3.6
	10/15/02	2 U	24	160	14000	58	140	2 U	2 U	2 U	10
COD (mg/l)	06/16/00	106	20 U	20 U	26.9	52.1	23.3	55.7	120	70.1	NA
	05/07/01	20 U	20 U	20 U	12900	308	367	20 U	20 U	20 U	20 U
	10/15/02	20 U	52	360	51000	115	250	20 U	20 U	20 U	40

Table 4-4. Biochemical Oxygen Demand (BOD) and Chemical Oxygen Demand (COD)

Notes:

Analytical methods: BOD by SW846 Method 405.1/5210B, COD by SW846 Method 410.4 NA = Not analyzed U = Undetected at the listed detection limit

Analyte	Units	B239	IRZ-INJ	IRZ-INJ-DUP	RAP1-6R	RAP-6T	IRZ-1	IRZ-4	IRZ-3	IRZ-2	RAP1-6S
Trichloroethene	ug/L	29	560	530	1400	810	1100	1500	1300	1900	1.4
cis-1,2-Dichloroethene	ug/L	70	1600	1600	4300	2100	3500	5300	4400	5300	3.8
trans-1,2-Dichloroethene	ug/L	0.88	16	16	22	21	23	ND	27	27	ND
1,1-Dichloroethene	ug/L	1.7	48	47	130	72	130	ND	140	170	ND
Vinyl chloride	ug/L	16	360	370	690	660	1100	1100	860	1300	ND
1,1-Dichloroethane	ug/L	4.8	110	98	240	130	210	350	280	330	ND
Ethane	ug/L	0.02	1.31	1.18	0.36	0.5	2.13	10.15	3.37	3.02	ND
Ethene	ug/L	0.61	24.73	22.91	30.57	25.85	45.24	96.41	71.32	66.83	0.02
Dissolved Oxygen	mg/L	1.38	1.19		1.48	0.43	0.47	0.5	0.61	0.47	0.35
Oxygen - Lab	mg/L	0.19	0.72	0.85	2.72	2.96	1.13	0.86	1.12	0.16	2.47
Oxidation Reduction Potential	mv	-32.1	200		-57.5	2.9	14	-30	-16	-38	-21.4
рH	SU	6	5.79		7.1	5.9	6.03	6.28	6.17	6.3	5.73
Conductance	uS/cm	33	356		16	37	445	512	501	472	48
Dissolved Organic Carbon	mg/l	3.7	1.8	2	2.2	2.5	2.8	3.5	3.4	3.3	1.4
Total Organic Carbon	mg/l	2	1.9	1.9	1.8	6.2	2.8	3.5	3.3	3.3	1.8
Chemical Oxygen Demand	mg/l	106	30.5	23.3	ND	52.1	23.3	55.7	70.1	120	ND
Biochemical Oxygen Demand	mg/l	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND
,,,	5										
Nitrate as N	mg/L	ND	0.04	ND	ND	ND	ND	ND	ND	ND	2.1
Nitrite as N	mg/L	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND
Ammonia as N	mg/L	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND
Total Manganese-Field	mg/L	0	0		0.2	0.3	0.3	0.6	0	0.3	0
Dissolved Manganese-Field	5										
(mg/l)	mg/L	0	0		0.3	0.3	0	0.3	0	0.3	0
-	-										
Ferrous Iron - Field	mg/L	>10	0.4		0.4	>10	5	>10	>10	>10	>10
Unfiltered Iron - Lab	mg/L	24.6	1.4	1.4	0.71	9.1	9.2	16.6	24.3	20	3.8
Dissolved Iron - Field	mg/L	>10	0.1		0.4	4	5	>10	>10	>10	0.3
Dissolved Iron - Lab	mg/L	23.4	0.31	0.31	0.15	5.5	8.6	12.3	14	12.3	0.74
Bromide	mg/L	ND	ND	ND	0.11	ND	ND	ND	ND	ND	ND
Chloride	mg/L	9.2	14.7	14.8	17.1	17	18.6	23.6	21.8	21.1	1.6
Sulfate	mg/L	38.9	32.3	32.2	22.8	29.4	28.3	22.7	24.2	21.5	21.6
Sulfide - Field	mg/L	0	0		0	0	0	0	0	0	0.1
Carbon Dioxide	mg/L	86.2	78	80.9	9.4	74.9	62.2	47.3	67.6	38.5	50.2
Methane	ug/l	15	63.5	58.1	72.2	51.5	84	138.8	130.1	122.2	1.4
Hydrogen	nM/L	1.6	>50	>50	5.2	>50		>50	>50	>50	>50
Nitrogen	mg/L	11.1	13.1	12.5	14.7	17.5	14.8	14.5	17.3	14.1	10.8

Table 4-5. Data from Initial Groundwater Sampling Round at Hanscom AFB

ND - Not Detected

Analyte	Units	B239	IRZ-INJ	RAP1-6R	RAP1-6T	IRZ-1	IRZ-4	IRZ-5	IRZ-3	IRZ-2	RAP1-6S	RAP1-6S-Dup
Trichloroethene	ug/L	22	15	1300	4.7	0.7	1200	1200	320	1300	1.7	1.7
cis-1,2-Dichloroethene	ug/L	54	640	4600	3000	3600	4900	4600	5000	5700	6.4	6.4
trans-1,2-Dichloroethene	ug/L	0.68	3.1	34	23	39	44	36	38	34	ND	ND
1,1-Dichloroethene	ug/L	1.3	9	140	110	91	160	170	140	160	0.12	0.12
Vinyl chloride	ug/L	9	67	690	640	630	990	970	820	870	0.45	0.47
1,1-Dichloroethane	ug/L	3.3	24	250	140	170	270	280	240	290	0.18	0.18
Ethane	ug/L	0.02	0.35	0.25	0.47	0.62	0.63	0.79	0.72	0.77	ND	ND
Ethene	ug/L	0.36	2.3	22	22	34	41	36	51	46	0.03	0.02
Dissolved Oxygen-Field	mg/L	0	1.35	0	0	0	0	0	0	0	0.58	NA
Oxygen-Lab	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Oxidation Reduction Potential	mv	22	-63	32	-190	-175	-66	-9	-87	-47	270	NA
рН	SU	5.7	4.72	6.46	6.83	6.84	6.59	5.95	6.44	6.31	4.51	NA
Conductance	uS/cm	204	4000	224	1040	1130	269	258	306	256	65	NA
Alkalinity	mg/L	70.7	673	99	299	368	120	86.9	156	104	7.1	7.1
Dissolved Organic Carbon	mg/L	2.2	4680	2.6	131	143	3.6	4.2	3.6	3.2	1.5	1.7
Total Organic Carbon	mg/L	2.1	5840	2.1	186	201	3.5	3.5	3.6	3.2	1.5	1.5
Chemical Oxygen Demand	mg/L	ND	12900	ND	308	367	ND	ND	ND	ND	ND	ND
Biochemical Oxygen Demand	mg/L	ND	10800	ND	280	320	ND	3.6	ND	ND	ND	ND
Nitrate as N	mg/L	ND	ND	ND	0.2	0.25	ND	ND	ND	ND	1.5	1.5
Nitrite as N	mg/L	ND	17.4	ND	ND	0.27	ND	ND	ND	ND	ND	ND
Ammonia as N	mg/L	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND	ND
Soluble Manganese-Field	mg/L	0.3	NA	1	>2	>2	0.3	0.3	0	0	0	0
Dissolved Manganese-Lab	mg/L	1.3	21.2	1.4	6.9	5.9	0.93	0.77	0.93	0.79	0.076	0.078
Total Managanese-Lab	mg/L	1.3	21.1	1.4	6.9	6.1	0.97	0.8	0.99	0.91	0.078	0.077
Total Iron - Field	mg/L	>10	>10	0.2	>10	>10	>10	>10	>10	>10	0.7	0.7
Total Iron - Lab	mg/L	26.4	730	0.14	141	141	14.4	11	24.8	28.9	0.51	0.5
Dissolved Iron-Field	mg/L	>10	>10	0	>10	>10	>10	>10	>10	8	0.4	0.4
Dissolved Iron-Lab	mg/L	25.9	729	ND	142	135	13	9.7	21.8	9.2	0.39	0.41
Bromide	ma/l	ND	ND	0.13	5 1	<i>A</i> 1	ND	ND	ND	0 14	ND	ND
Chloride	mg/L	6.8	253	19.7	23.3	26.9	21.8	22.7	19.7	19.8	1.8	1.8
Total Dissolved Solids	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Sulfate	mg/L	43.7	30.1	21.7	3.6	0.22	23.2	27.4	10.9	21.8	18.9	19.7
Hydrogen Sulfide - Field	mg/L	0	5	0	1	0	0.1	0.1	0	03	0	0
Carbon Dioxide	ma/l	106.8	1149.7	14.8	99.1	86	60.3	57.6	54.1	50.5	74.2	79.7
Methane	ug/l	3.6	5.8	51.1	106.5	ND	74.2	72.8	92.8	86.3	3.3	3.9
Hvdrogen	nM/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Nitrogen	ma/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
	g, L					/ \		/ 1				

Table 4-6. Data from Midpoint Groundwater Sampling Round at Hanscom AFB, 05-03-01 to 05-08-01

NA = NOT ANALYZED ND = NON DETECT

Analyte	Units	B239	IRZ-INJ	RAP1-6R	RAP-6T	RAP1-6T-DUP	IRZ-1	IRZ-4	IRZ-5	IRZ-3	IRZ-2	RAP1-6S
Trichloroethene	ug/L	39	ND	1400	510	530	68	1400	1100	1400	1800	0.4
cis-1,2-Dichloroethene	ug/L	88	110	4900	3300	3300	980	5000	4400	4800	5100	1.6
trans-1,2-Dichloroethene	ug/L	1	ND	ND	34	30	26	ND	33	40	36	ND
1,1-Dichloroethene	ug/L	2.1	ND	130	87	84	23	160	140	ND	180	ND
Vinyl chloride	ug/L	16	ND	980	1000	980	650	1100	1100	1100	1200	0.35
1,1-Dichloroethane	ug/L	5.4	ND	240	180	190	190	250	240	250	290	ND
Ethane	ug/L	0.02	0.01	0.36	0.24	NA	0.38	0.57	0.83	0.61	0.8	0.12
Ethene	ug/L	0.52	0.16	37.21	133.09	NA	1107.96	41.4	83.94	45.9	45.63	0.09
Dissolved Oxygen-Field	mg/L	0.34	0.36	0.44	1.65	NA	0.58	7.67	0.93	0.37	1.04	2.59
Oxygen-Lab	mg/L	0.56	0.21	0.29	0.2	NA	ND	0.52	0.87	0.17	0.19	0.41
Oxidation Reduction Potential	mv	-15	-147	-226	-169	NA	-175	-75	-86	-96	-81	-49
рН	SU	6.26	4	6.92	6.6	6.6	7.11	6.7	6.7	6.73	6.6	6.2
Conductance	uS/cm	216	5910	310	401	NA	616	286	308	317	271	385
Alkalinity	mg/L	54	ND	120	160	170	210	120	140	110	110	200
Dissolved Organic Carbon	mg/L	2	13000	13	36	39	74	2.5	12	2.4	2.3	130
Total Organic Carbon	mg/L	1.3	13000	15	33	38	77	2.6	10	2.4	2.3	110
Chemical Oxygen Demand	mg/L	ND	51000	52	100	130	250	ND	40	ND	ND	360
Biochemical Oxygen Demand	mg/L	ND	14000	24	56	59	140	ND	10	ND	ND	160
Nitrate as N	mg/L	ND	ND	0.018	ND	ND	ND	ND	ND	ND	ND	0.11
Nitrite as N	mg/L	ND	ND	0.05	ND	ND	ND	ND	ND	ND	ND	ND
Ammonia as N	mg/L	0.72	2.7	3.5	0.3	0.3	0.44	0.2	0.2	0.23	0.25	0.63
Soluble Manganese-Field	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved Manganese-Lab	mg/L	NA	1.7	1.9	1.7	1.7	1.8	0.86	0.91	0.68	0.73	1.1
Total Managanese-Lab	mg/L	NA	1.7	1.7	1.6	1.6	1.8	0.96	0.91	0.73	0.75	1.2
Total Iron - Field	mg/L	1	50	<25	25	NA	50	NA	NA	<25	NA	50
Total Iron - Lab	mg/L	NA	220	0.33	54	56	91	19	16	15	17	160
Dissolved Iron-Field	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved Iron-Lab	mg/L	NA	220	0.32	59	59	88	13	16	23	13	140
Bromide	mg/L	ND	58	0.1	0.2	0.18	0.72	0.096	0.23	0.027	ND	1.6
Chloride	mg/L	2.7	120	16	24	24	33	3.1	2.6	22	15	3.4
Total Dissolved Solids	mg/L	130	15000	190	260	280	360	210	220	210	190	400
Sulfate	mg/L	36	890	20	13	13	2.3	23	22	23	22	17
Hydrogen Sulfide - Field	mg/L	0	0.7	0.3	1	NA	0	0.5	NA	0	NA	0.5
Carbon Dioxide	mg/L	64.3	1327.5	19.8	41.1	NA	19.7	58.2	45.4	57.7	46.5	175.6
Methane	ug/L	16.1	31.7	68.4	983.7	NA	2057.7	80.6	117.1	95.8	94.2	1845.3
Hydrogen	nŴ/L	495.6	1514	157.7	134.1	NA	1452.4	620.9	179.6	423.5	667.8	367.4
Nitrogen	mg/L	9.8	1.2	13.9	10.3	NA	8.7	14.2	15.9	15	15.6	13.6

 Table 4-7. Data from Midpoint Groundwater Sampling Round at Hanscom AFB, October 2002

NA = NOT ANALYZED

ND = NON DETECT

PROCESS	<u>PCE</u>	<u>tce</u>	<u>c-DCE</u>	<u>vc</u>	<u>tca</u>	DCA	<u>ст</u>	<u>CF</u>	<u>DCM</u>
Direct Aerobic	N	Ν	Y&N	Y	N	Ν	Ν	N	Y
Cometabolic w/ CH4	Ν	Y	Y	Y	Y&N	N *	N	Υ	NR
Cometabolic w/ toluene	Ν	Y	Y	Y	Ν	N *	N	Y&N	NR
Cometabolic w/ NH4	Ν	Y	Y	Y	Υ	N*	Ν	Y	NR
Direct Anaerobic	Ν	Ν	Ν	Υ	Ν	Ν	Ν	Ν	Y
Anaerobic/ Denitrification	Y&N	Y&N	N *	N*	N*	N *	Y	Y&N	NR
Anaerobic/Sulfate reduction	Y	Y	Y	Y	Y	Y	Y	Y	NR
Anaerobic/ Methanogenic	Y	Y	Y	Y	Y	Y	Y	Y	NR

N: Not documented in the literature

Y: Documented in the literature many times; concensus opinion

Y&N: Documented in the literature more than once of both occurrence and absence N\*: Not documented in the literature to date, but not investigated significantly

NR: Process may occur but Not Relevant since competing process occurs more rapidly

Source: ITRC 2002, ITRC 1999

Key	
Carbon Tetrachloride	CT
Chloroform	CF
Dichloromethane	DCM
1,1,1-Trichloroethane	TCA
Dichloroethane	DCA
Tetrachloroethene	PCE
Trichloroethene	TCE
cis-1,2-Dichloroethene	<i>c</i> -DCE
Vinyl chloride	VC

		P	re-Treatme	nt	Du	ring Treatn	nent	Other EF	RD Sites*	Published	NA Rates
Constituent	Well	k	$R^2$	Half Life	k	$R^2$	Half Life	k	Half Life	k	Half Life
		(1/yr)		(days)	(1/yr)		(days)	(1/yr)	(days)	(1/yr)	(days)
TCE	IRZ-Inj				3.16	0.7093	80	0.98	257	0.15-2.58	98-1653
	IRZ-1				8.98	0.6632	28	3.95	64		
	RAP1-6T	-0.07	0.3638	(Gain)	5.12	0.4726	49	3.10	82		
								2.33	108		
								1.31-3.20	79-193		
								1.83-8.40	30-139		
								15.33	17		
cis-DCE	IRZ-Inj				0.59	0.4035	428	2.45	103	0.35-2.26	112-720
	IRZ-1				0.92	0.0590	275	3.18	80		
	RAP1-6T	3E-03	0.0010	97287	1.14	0.7348	223	2.15	117		
								1.26	200		
								1.46-6.21	41-173		
								15.33	17		
Vinyl Chloride	IRZ-Inj				2.33	0.8910	109	2.92	87	0.35-2.26	112-720
	IRZ-1							0.95	267		
RAP1-6T		-0.04	0.3050	(Gain)				0.69	365		
								1.10-5.48	46-231		

Table 4-9. Summary of Biodegradation Rates over Selected Time Intervals

Notes:

See Figures 4-62 through 4-64 for illustration of selected intervals over which rates were calculated

Rates calculated for other ARCADIS ERD sites, as published in Horst et al. (2000) and Suthersan et al. (2002)

Published data are anaerobic, aqueous biodegradation half-lives from Howard et al. (1991), assumed to represent natural attenuation

Table 4-10. Total Dissolved Solids (TDS) in Groundwater

Well	TDS (mg/l)
HAN-GW-B239	130
HAN-GW-RAP1-6S	400
HAN-GW-RAP1-6R	190
HAN-GW-IRZ-INJ	15000
HAN-GW-RAP1-6T	260
HAN-GW-RAP1-6T DUP	280
HAN-GW-IRZ-1	360
HAN-GW-IRZ-4	210
HAN-GW-IRZ-2	190
HAN-GW-IRZ-3	210
HAN-GW-IRZ-5	220
HAN-GW-B242	150

### Notes:

Samples collected October 14-16, 2002

Analytical method: SW846 Method 160.1

Shading indicates exceedance of the Federal Secondary

Drinking Water Standard of 500 mg/l for TDS

# Table 4-11. VOCs of Secondary Concern

Constituent	Stan	dard	Well	Date	Result (ug/L)
Acetone	610	PRG	HAN-GW-IRZ-1	05/07/01	38
			HAN-GW-RAP1-6S	10/15/02	47
2-Butanone (MEK)	1900	PRG	HAN-GW-IRZ-1	05/07/01	240 E
			HAN-GW-RAP1-6S	10/15/02	34
			HAN-GW-RAP1-6T	05/07/01	180 E
Carbon tetrachloride	5	MCL	HAN-GW-RAP1-6S	10/15/02	0.11 J
Chloroform	80	MCL	HAN-GW-IRZ-INJ	05/08/01	8.6
				10/15/02	18 J
1,2-Dibromoethane (EDB)	0.05	MCL	HAN-GW-IRZ-5	10/16/02	6.1 J
Dichlorodifluoromethane	390	PRG	HAN-GW-IRZ-1	10/14/02	9.4 J
2-Hexanone	1500	RBC	HAN-GW-IRZ-5	10/16/02	79 J
Methylene chloride	5	MCL	HAN-GW-IRZ-3	06/15/00	0.53 J
(Dichloromethane)				05/03/01	0.21 J,B
				10/15/02	8.4 J,B
			HAN-GW-IRZ-4	05/04/01	0.36 J,B
			HAN-GW-IRZ-INJ	06/16/00	0.55 J
			HAN-GW-RAP1-6R	06/15/00	0.57 J
				05/07/01	0.3 J,B
				10/15/02	8.1 J,B
4-Methyl-2-pentanone (MIBK)	160	PRG	HAN-GW-IRZ-5	10/16/02	70 J
Methyl t-butyl ether (MTBE)	13	PRG	HAN-GW-RAP1-6S	10/15/02	1.3 J
Styrene	100	MCL	HAN-GW-IRZ-INJ	05/08/01	0.51 J

(continued on next page)

#### Table 4-11. (continued)

Constituent	Stand	lard	Well	Date	Result (ug/L)
Toluene	1000	MCL	HAN-GW-B239	05/03/01	0.21
			HAN-GW-IRZ-1	06/16/00	0.59 J
				05/07/01	0.34 J
			HAN-GW-IRZ-2	06/15/00	2.1 J
				05/04/01	2.3
			HAN-GW-IRZ-3	06/15/00	0.92 J
				05/03/01	0.97 J
			HAN-GW-IRZ-4	06/15/00	1.2
				05/04/01	1.4
			HAN-GW-IRZ-5	05/08/01	2.8
			HAN-GW-IRZ-INJ	06/16/00	0.54 J
				05/08/01	0.66 J
			HAN-GW-IRZ-INJ-DUP	06/16/00	0.7 J
			HAN-GW-RAP1-6R	05/07/01	3
			HAN-GW-RAP1-6S	10/15/02	0.45 J
			HAN-GW-RAP1-6T	05/07/01	1
			HAN-GW-RAP1-6T DUP	10/14/02	12 J
1,2,3-Trichloropropane	0.0056	PRG	HAN-GW-IRZ-5	10/16/02	26
Xylenes, Total	10000	MCL	HAN-GW-IRZ-1	05/07/01	0.4 J
			HAN-GW-IRZ-2	06/15/00	3.9
				05/04/01	2.6
			HAN-GW-IRZ-3	06/15/00	1.4
				05/03/01	0.98 J
			HAN-GW-IRZ-4	06/15/00	4.4
				05/04/01	2.2
			HAN-GW-IRZ-5	05/08/01	1.5
			HAN-GW-RAP1-6S	10/15/02	0.57 J
			HAN-GW-RAP1-6T	05/07/01	0.56
			HAN-GW-RAP1-6T DUP	10/14/02	18 J,B

Notes:

Analytical method - SW846 Method 8260

B = Detected in blank

E = Estimated

J = Estimated

MCL - Federal Maximum Contaminant Level for drinking water

PRG - US EPA Region 9 Preliminary Remedial Goal for tap water (provided where no MCL exists)

RBC - US EPA Region 3 Risk-Based Concentration for tap water (provided where no MCL or PRG exists) Shading indicates exceedance of listed standard

Only detected compounds listed; chlorinated ethenes and ethanes not included

#### Table 4-12. Total Metals in Groundwater

-												RAP1-	6Т						_		_		_	_	_		
Constituent (mg/l)	MCL or other Std.	B-239		IRZ-IN	IJ	RAP1-	6R	RAP1-6	SS	RAP1-	6T	DUP		IRZ-	1	IRZ-4	1	IRZ-2	2	IRZ-	3	IRZ-	5	B-242	2	FB	
Antimony	0.006	0.02	U	0.0086	В	0.02	U	0.02	U	0.02	U	0.02	U	0.02	U	0.02	U	0.02	U	0.02	U	0.02	U	0.02	U	0.02	U
Arsenic	0.045 (PRG)	0.012		0.049		0.01	J	0.092		0.023		0.02		0.042		0.0083	В	0.011		0.031		0.01		0.01	U	0.01	U
Beryllium	0.004	0.004	С	0.0076		0.004	U	0.004	U	0.004	U	0.004	U	0.004	U	0.004	U	0.004	U	0.004	U	0.004	U	0.004	U	0.004	U
Cadmium	0.005	0.005	U	0.0019	В	0.005	U	0.005	U	0.005	U	0.005	U	0.005	U	0.005	U	0.005	U	0.005	U	0.005	U	0.005	U	0.005	U
Chromium	0.1	0.0033	В	0.2		0.0037	В	0.003	В	0.0022	В	0.0065	В	0.01	U	0.072		0.01	U	0.0049	U	0.01	U	0.015		0.01	U
Copper	1.3	0.006	В	0.22		0.0016	В	0.002	В	0.0015	В	0.0016	В	0.0017	В	0.0093	В	0.0009	В	0.0026	В	0.02	U	0.0017	В	0.0016	В
Iron	11 (PRG)	<mark>19</mark>		220		0.33		<mark>160</mark>		54		56		91		19		17		23		16		4.1		0.05	U
Lead	0.015	0.005	U	0.028		0.005	U	0.005	U	0.005	U	0.005	U	0.005	U	0.0041	В	0.005	U	0.005	U	0.005	U	0.0019	В	0.005	U
Manganese	0.88 (PRG)	<mark>1.0</mark>		1.7		1.7		<mark>1.2</mark>		1.6		1.6		1.8		0.96		0.75		0.73		0.91		0.12		0.01	U
Mercury	0.002	0.0002	U	0.0012	U	0.0002	U	0.0002	U	0.0002	U	0.0002	U	0.0002	U	0.0002	U	0.0002	U	0.0002	U	0.0002	U	0.0002	U	0.0002	U
Nickel	0.73 (PRG)	0.0082	В	0.25		0.04	U	0.04	U	0.04	U	0.0071	В	0.04	U	0.049		0.04	U	0.0059	U	0.04	U	0.026	В	0.04	U
Selenium	0.05	0.01	U	0.011		0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U
Silver	0.18 (PRG)	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U
Thallium	0.002	0.01	U	0.02	U	0.01	U	0.012		0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U	0.01	U
Zinc	11 (PRG)	0.11		0.7		0.0063	В	0.0072	В	0.02	U	0.02	U	0.02	U	0.016	В	0.02	U	0.0077	U	0.02	U	0.088		0.02	U

Notes:

Samples collected October 14-16, 2002

Analytical methods - SW846 Methods 6010B and 7470

B = Detected in blank

U = Undetected at the detection limit listed

MCL - Federal Maximum Contaminant Level for drinking water

PRG - US EPA Region 9 Preliminary Remedial Goal for tap water (provided where no MCL exists) Shading indicates exceedance of listed standard

#### Table 4-13. Dissolved Metals in Groundwater

Constituent (mg/l)	MCL or other Std.	B-239	IRZ-INJ	RAP1-6R	RAP1-6S	RAP1-6T	RAP1-6T	DUP	IRZ-1	IRZ-4	IRZ-2	IRZ-3	IRZ-5	B-242	FB
Antimony	0.006	0.02 U	0.0056 <mark>B</mark>	0.02 U	0.02 U	0.02 U	0.02	U	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
Arsenic	0.045 (PRG)	0.0041 B	0.051	0.01 U	0.079	0.022	0.024		0.04	0.0062 B	0.0051 B	0.0077 B	0.0082 B	0.01 U	0.01 U
Beryllium	0.004	0.004 U	0.0077	0.004 U	0.004 U	0.004 U	0.004	U	0.004 U	0.004 U	0.004 U	0.004 U	0.004 U	0.004 U	0.004 U
Cadmium	0.005	0.005 U	0.0023 B	0.005 U	0.005 U	0.005 U	0.005	U	0.005 U	0.005 U	0.005 U	0.005 U	0.005 U	0.005 U	0.005 U
Chromium	0.1	0.01 U	0.19	0.0023 B	0.01 U	0.01 U	0.01	U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U
Copper	1.3	0.0018 B	0.17	0.0013 B	0.0011 B	0.0012 B	0.0013	В	0.0012 B	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U	0.0016 B
Iron	11 (PRG)	<mark>19</mark>	220	0.32	<mark>140</mark>	<mark>59</mark>	<b>5</b> 9		<mark>88</mark>	<mark>13</mark>	<mark>13</mark>	15 15	<mark>16</mark>	4.4	0.05 U
Lead	0.015	0.005 U	0.027	0.005 U	0.005 U	0.005 U	0.005	U	0.005 U	0.005 U	0.005 U	0.005 U	0.005 U	0.0017 B	0.005 U
Manganese	0.88 (PRG)	1.0	1.7	<u>1.9</u>	1.1	1.7	1.7		<mark>1.8</mark>	0.86	0.73	0.68	0.91	0.13	0.01 U
Mercury	0.002	0.0002 U	0.0012 U	0.0002 U	0.0002 U	0.0002 U	0.0002	U	0.0002 U	0.0002 U	0.0002 U	0.0002 U	0.0002 U	0.0002 U	0.0002 U
Nickel	0.73 (PRG)	0.0058 B	0.25	0.04 U	0.04 U	0.04 U	0.04	U	0.04 U	0.04 U	0.04 U	0.04 U	0.04 U	0.016 B	0.04 U
Selenium	0.05	0.01 U	0.017	0.01 U	0.01 U	0.01 U	0.01	U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U
Silver	0.18 (PRG)	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U	0.01	U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U
Thallium	0.002	0.01 U	0.01 U	0.01 U	0.014	0.01 U	0.01	U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U	0.01 U
Zinc	11 (PRG)	0.043	0.71	0.02 U	0.0066 B	0.02 U	0.02	U	0.02 U	0.02 U	0.02 U	0.02 U	0.0069 B	0.032	0.02 U

Notes:

Samples collected October 14-16, 2002

Analytical methods - SW846 Methods 6010B and 7470

B = Detected in blank

U = Undetected at the detection limit listed

MCL - Federal Maximum Contaminant Level for drinking water PRG - US EPA Region 9 Preliminary Remedial Goal for tap water (provided where no MCL exists)

Shading indicates exceedance of listed standard

Table 4-14. Molasses Analysis for Inorganics

Blackstrap Molasses Analysis (US Sugar Corp., 2000)									
Weight/gallon	12.0 lbs								
Calcium	0.80%								
Chloride	2.10%								
Cobalt	negligible								
Copper	14 ppm								
Iron	130 ppm								
Magnesium	0.27%								
Manganese	5 ppm								
Nitrogen	1.01%								
Phosphorus	negligible								
Potassium	4.20%								
Selenium	negligible								
Sodium	0.09%								
Sulfur	0.78%								
Zinc	8 ppm								

from http://www.suga-lik.com/molasses/molasses\_frame.html

#### Table 4-15. Molasses-Water Analysis for Inorganics

		Laboratory	Laboratory	Analysis
Constituent	SDWA MCL	Detection Lmt	Mixture	Qualifier
Molasses u	sed at a commerci	al site in Ohio; 10:1	water:molasses mix	ature
Arsenic	0.05	0.01	0.0088	В
Barium	2	0.2	0.031	В
Cadmium	0.005	0.005	0.00053	BJ
Lead		0.003	0.005	
Chromium	0.1	0.01	0.0078	В
Selenium	0.05	0.005	0.028	В
Silver	0.1*	0.01	<0.01	
Mercury	0.002	0.0002	0.000077	В
Molasses u	used at Hanscom;	9:1 water:molasses i	mixture (October 20	02)
Chloride	250*		1500	

Concentrations reported in milligrams per liter

Metals analysis conducted by USEPA Method 6010B

Laboratory

Qualifiers:

"B" - Estimated result below laboratory method detection limit

"J" - Method blank contamination, associated method blank contains the target analyte

at a

reportable level

Federal Standards are SDWA MCLs or \*secondary drinking water regulations

Arsenic MCL is currently 0.05 mg/L and will change to 0.01 mg/L In 2006

Well	Date	Pyruvic	Lactic	Formic	Acetic	Propionic	Butyric
IRZ-1	10/14/2002	<4	<1	<1	111.4	17.5	14.1
RAP1-6T	10/14/2002	<4	<1	<1	57.3	5.9	6.6
MW-B239	10/14/2002	<4	<1	<1	<1	<1	<1

Table 4-16. Fatty Acids in the October 2002 Full Monitoring Round

Results reported in mg/L

# Table 4-17. Comparison of Technology Alternatives

	Groundwater Pump & Treat		Aquifer Sparging
Effe	ctiveness	Effectiveness	
	Rapid results (containment & mass removal) once system is deployed. Effective at mass removal of contaminants.		Rapid results (containment & mass removal) once system is deployed. Effective at mass removal of contaminants.
	Very effective for hydraulic containment & easily demonstrated.		complex to demonstrate in short-term. In-situ treatment allows for more effective treatment of organics suchs as VOCs. However, overall effectivness limited to compounds with high Henry's Law constant or those
Reli	Not effective in meeting all but the least stringent clean-up goals. <b>ability</b>	Reliability	that can degrade aerobically.
	Moderate reliability - number of fixed/engineered components increase likelhood of operational problems/failures.		In-situ nature and limited fixed components make technology very reliable.
	Fixed, engineered nature of systems severely limit flexibility and adaptability.		Fixed, engineered nature of systems severely limit flexibility and adaptability.
	Operational experience suggests systems can be plauged by reliability problems associated with non-target contaminants (ie, fouling).		More reliable than ex-situ treatment techniques given no need to handle extracted groundwater.
Spe	Can address wide range of contaminants (VOCs, SVOCs, metals, other inorganics, etc.). ed	Speed	above).
	Short-term - Slow speed. Fairly complex design, approval & permitting process needed for implementation.		Short-term - Moderate speed. Reasonable design & approval, limited permitting process needed for implementation.
Eas	Long-term - Poor speed. Nature of technology requires very long time to reach closure. e of Use	Ease of Use	Long-term - Moderate speed. Nature of technology requires some time to reach closure - especially if goals are low.
	Technology is very complex due to water handling, energy requirments, manpower requirements, and residuals management.		Technology is moderately complex due to energy requirments, manpower requirements. Limited residuals management.
	Health & safety concerns are moderate. Technology can cause additional routes of exposure to media.		Health & safety concerns are low. Technology does not provide additional routes of exposure to media.
	Abive grade nature of treatment system can impact Site activities and/or development potential.		Abive grade nature of treatment system can impact Site activities and/or development potential.
			(continued)

### Table 4-17. (concluded)

Effectiv	Chemical Oxidation reness	Enhanced Reductive Dechlorination Effectiveness Technology will provide effective mass removal upon acclimitization of
Ver Effe mol	y rapid results (mass removal) upon application of technology. active at mass removal of contaminants. activeness for simple to demonstrate in short-term. Long-term nitoring required to evaluate 'rebound'	reactive zone. Effective at mass removal of contaminants. Effectiveness for containment and/or plume treatment is more complex to demonstrate in short-term.
In-s as <sup>v</sup> In a <b>Reliabil</b>	situ treatment allows for more effective treatment of organics suchs VOCs. However, overall effectivness limited to organic compounds. addition, mixed organic plumes may require multiple oxidants. <b>ity</b>	In-situ treatment allows for more effective treatment of organics suchs as VOCs and others. Technology can also be used to treat other compounds including metals. <b>Reliability</b>
In-s Lac ada	situ nature and no fixed components make technology very reliable. k of fixed, engineered systems make technology flexible & aptable.	In-situ nature and no fixed components make technology very reliable. Lack of fixed, engineered systems make technology flexible & adaptable.
Mo extr	re reliable than ex-situ treatment techniques given no need to handle racted groundwater.	More reliable than ex-situ treatment techniques given no need to handle extracted groundwater.
Lim Speed	ited suite of compounds that can be reliable treated (see above).	Larger suite of compounds that can be reliably treated (see above).
Sho pro	ort-term - Fast speed. Limited design, approval, & permitting cess needed for implementation.	Short-term - Fast speed. Limited design, approval, & permitting process needed for implementation.
trea sup Ease of	atment of constituents assuming sufficient oxidant chemical is oplied.	Long-term - Moderate speed. Nature of technology requires some time for reactive zone to fully acclimatize. Ease of Use
Tec pote resi Hea tem	chnology is moderately complex due handling of chemicals and ential for aquifer preparation prior to treatment. However, no iduals management is required. alth & safety concerns are high. Technology can create high operature reactions and/or high levels of oxygen in the subsurface	Technology is very simple to implement. Limited manpower requirements, no residuals management, and no chemical handling concerns.
that	t need to be addressed.	No appreciable health & safety concerns.

Below grade nature of technology and lack of fixed systems limitBelow grade nature of technology and lack of fixed systems limit impactsimpacts to Site activities and/or development potential.Site activities and/or development potential.

WBS I	Numb	ber	DESCRIPTION	QTY	UOM	UNIT COST	COST \$
33XXX			HTRW CONSTRUCTION ACTIVITIES				
331XX			HTRW REMEDIAL ACTION (Capital and Operating)				
	01		MOBILIZATION AND PREPARATORY WORK				
	01	01	Mobilization of Construction Equipment (Drilling Rig)	1	EA	\$3,000	\$3,000
	01	03	Submittals/Implementation Plans/Permits	1	EA	\$30,000	\$30,000
	01		Pilot Testing	1	EA	\$75,000	\$75,000
	_						
	02		MONITORING, SAMPLING, TESTING, AND ANALYSIS				
	02	04	Monitoring Wells - Installation	4	EA	\$3,000	\$12,000
	02		Injection Wells - Installation	25	EA	\$3,000	\$75,000
	02		Well Development	1	EA	\$4,000	\$4,000
	02		IDW Disposal (soil cuttings)	1	EA	\$1,000	\$1,000
	02	05	Sampling Groundwater				
	02		Quarterly (10 wells)	4	EA	\$5,300	\$21,200
	02		Semi-Annual (10 wells)	8	EA	\$5,300	\$42,400
	02	06	Sampling Soil	1	EA	\$3,000	\$3,000
	02	09	Laboratory Chemical Analysis	12	EA	\$3,600	\$43,200
	_						
	11		BIOLOGICAL TREATMENT				
	11	04	In-Situ Biodegradation/Bioreclamation				
	11		Trailer-Mounted Molasses Injection System	1	EA	\$10,000	\$10,000
	11		Monthly Molasses Injections - Labor	24	EA	\$2,250	\$54,000
	11		Bi-Monthly Molasses Injections - Labor	18	EA	\$2,250	\$40,500
	11		Field Process Monitoring - Equipment	42	EA	\$200	\$8,400
	11		Field Process Monitoring - Analytical	42	EA	\$200	\$8,400
	11		Laboratory Chemical Analysis (TOC)	100	EA	\$30	\$3,000
	11		Molasses (20 gallons per well per injection)	21000	GAL	\$3	\$63,000
	11		Water for Injection (180 gallons per well per injection)	189000	GAL	\$0.0029	\$548
	11		Well Rehabilitation	63	EA	\$300	\$18,900
	11		Progress Reporting	5	YR	\$25,000	\$125,000
	11		Completion Report	1	EA	\$20,000	\$20,000
	21			05	<b>F A</b>	<b>*</b> =00	\$17 FOO
	21			35	EA	\$500	\$17,500
			INDIRECT ENVIRONMENTAL ACTIVITY COSTS				
			Environmental and Safety Training	2	FTE	\$250	\$500
			OSHA Ambient Environment Sampling	1	EA	\$250	\$250
			Waste Manifesting	1	EA	\$500	\$500
	1					,	
			TOTAL AMOUNT FOR HYPOTHETICAL SITE				\$680,298

# Table 5-1. Estimated IRZ Costs for a Hypothetical CAH Plume

Location	Description	Target COCs	Actual/Projected Savings
Rogersville, Tennessee	Parts manufacturing for trucks	PCE, TCA	\$200,000
Eastern Tennessee	Fuel facility	PCE, radionuclides	\$1,500,000
Chattanooga, Tennessee	Former manufacturing facility	PCE	\$500,000 (50%)
Northeastern New Jersey	Pharmaceutical	PCE	\$6,000,000
Williamsport, Pennsylvania	Textron/manufacturing	Cr <sup>+6</sup> , TCE, DCE, VC	\$2,250,000 (75%)
Reading, Pennsylvania	Textile equipment	TCE, Cr <sup>+6</sup> , Pb, Cd	\$700,000 (70%)
Emeryville, California	Metal plating manufacturer	TCE, DCE, Cr <sup>+6</sup>	\$1,600,000 (80%)
Hampton, Iowa	Metal plating	Cr <sup>+6</sup>	\$500,000 (66%)
Dallas, Texas	Graphics	Cr <sup>+6</sup>	\$1,500,000 (75%)
Pennsylvania	Lord Corporation	CAHs	\$6,400,000 (74%)
East Coast	Metal plating	CAHs, Cr <sup>6+</sup>	\$6,000,000

# Table 5-2. Cost Savings for IRZ Technology Compared to Pump and Treat Systems

Economic Category	Natural Attenuation	Vacuum- Enhanced Recovery	In-Situ Air Sparging	Iron Reactive Wall	IRZ
Capital					
Best	\$25,000	\$350,000	\$200,000	\$600,000	\$150,000
Worst	\$30,000	\$500,000	\$250,000	\$700,000	\$160,000
Annual O & M					
Best	\$25,000	\$60,000	\$45,000	\$25,000	\$30,000
Worst	\$35,000	\$75,000	\$60,000	\$35,000	\$40,000
Present Worth of Total					
n (years) =	30	20	20	30	15
Best	\$429,000	\$1,135,000	\$789,000	\$1,004,000	\$477,000
Worst	\$595,000	\$1,481,000	\$1,035,000	\$1,265,000	\$596,000
Total Opinion of Probable Costs					
Best Case	\$400,000	\$1,200,000	\$750,000	\$900,000	\$500,000
Worst Case	\$600,000	\$1,500,000	\$1,100,000	\$1,300,000	\$800,000

# Table 5-3. Economic Comparison of Probable Costs for Proposed ARCADIS CAH Site in South Carolina

Table 5-4. Results of DuPont Technology Evaluation	
--	--

Metric	Pump and Treat	Zero-Valent Iron PRB	Substrate Enhanced Biobarrier	Recirculating Source Zone	Natural Attenuation
Present Cost, (\$1000s)	\$9,800	\$3,900	\$3,100	\$1,300	\$890
\$/1,000 gallons treated	\$8.90	\$5.30	\$4.20	\$1.80	\$1.20
\$/lb PCE Removed	\$1,600	\$640	\$520	\$220	\$150
### Table 5-5. Summary of IRZ Technology Application Costs

	Estimated	Estimated Annual	Actual or Predicted	Initial	Dimensions
Site	Capital Costs	O&M Costs	Costs to Closure	Concentration	
Industrial Laundry/Dry Cleaning Facility, Eastern PA	\$75,000	\$45,000	\$250,000	46,000 ug/l PCE	10,000 ft <sup>2</sup> x 20 ft deep
				5 - 14,000 ug/l PCE	19.3 acres or 1200 x
Uranium Processing Facility, Eastern US	\$480,000	\$65,000	\$760,000	(plus U)	700 ft
				24,000 ug/l TCE (plus	< 2 acres or <87,000 ft <sup>2</sup>
Former Metal Pating Site, Western US <sup>1</sup>	\$100,000	\$150,000	\$250,000	Cr)	x 10 feet deep
				800 ug/l CT,	3.25 acres or 141,600 ft <sup>2</sup>
Industrial Manufacturing Site, South Carolina	\$1,400,000	\$75,000	\$2,000,000	chloroform, TCE	x 10 ft deep
					3000 ft long in bedrock ·
Industrial Site, Northeastern US	\$150,000	\$80,000	\$750,000	120 ug/L PCE	depth varies
Former Dry Cleaner, Wisconsin <sup>2</sup>	\$200,000	\$100,000	\$400,000	1,500-4,000 ug/L PCE	$30,000 \text{ ft}^2 \text{ x 5 ft deep}$
					1000 x 400 ft x 20 ft
Former Automotive Manufacturing Site, Midwestern, US	\$75,000	\$60,000	\$375,000	800 ug/l TCE	deep
					3000 x 400 ft x 40 ft
AOC 50, Ft. Devens, Ayer, Massachusetts	\$150,000	\$150,000	NA <sup>3</sup>	4,000 ug/L PCE	deep

Note:

All costs presented in current dollars.

1 - Site has received regulatory closure.

2 - Site has received regulatory closure.

3 - No Predicted Costs to Closure Available. Pilot study ongoing.

## List of Appendices

Appendix A-1	IRZ/ERD Bibliography
Appendix A-2	Soil Sieve Analyses
Appendix A-3	Standard Operating Procedure: Field Screening of Water & Vapor Samples for Volatile Organic Compounds by Gas Chromatography
Appendix A-4a	Boring Logs and Well Construction Diagrams
Appendix A-4b	Authorization to Leave Monitoring Wells in Place
Appendix A-5	Survey Data
Appendix A-6a	Additional Discussion of Data Quality
Appendix A-6b	Validation Memos
Appendix A-6c	QC Notes
Appendix A-7	Additional Discussion of Hydrogeologic Evaluations
Appendix A-8	Additional Discussion of Secondary Water Quality Issues
Appendix A-9	Data Archiving and Demonstration Plans
Appendix A-10	Microbial Analysis Report
Appendix C	Quality Assurance Project Plan (QAPP)

# Appendix A-1

IRZ/ERD Bibliography

#### Appendix A-1: IRZ/ERD Bibliography

"Engineered *in situ* Anaerobic Reactive Zones", S.S. Suthersan, US Patent 6,143,177, 2000. http://www.epa.gov/epaoswer/hazwaste/ca/success/r4s\_nucl.pdf

*"In situ* anaerobic reactive zone for *in situ* metals precipitation and to achieve microbial denitrification" S.S. Suthersan, US Patent 5,554,290, 1996.

Burdick, J., "Enhanced Natural Attenuation of Chlorinated VOCs," Natural Attenuation December 1998 Conference, IBC, Pasadena, California.

Burdick, J., and Jacobs, D., "Field Scale Applications to Demonstrate Enhanced Transformation of Chlorinated Aliphatic Hydrocarbons," National Ground Water Association Northeast Regional Conference, Burlington, VT, October 1998.

Burdick, J., Bent, T., and Suthersan, S., "Field Scale Applications to Demonstrate Enhanced Transformation of Chlorinated Aliphatic Hydrocarbons." Proceedings from the First International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Natural Attenuation -Chlorinated and Recalcitrant Compounds, pp 81-87, May 1998.

Burdick, J., Lenzo, F., and Jacobs, D., "Enhanced Reductive Dechlorination of PCE in Bedrock," 2<sup>nd</sup> International Conference on Remediation of Chlorinated and Recalcitrant Compounds, May 2000. Proceedings available at <u>http://www.battelle.org/bclscrpt/</u> Bookstore99/list all.cfm?OrderVar=Subject

C.S. Morie, J. Greene, M. Ledford, G.B. Page, C.T. Hughes, and B.D. Ilgner, "Tetrachloroethylene and Uranium Remediation Using IRZ®," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

D.L. Jacobs, F. Lenzo, and J. Burdick, "PCE Remediation in Bedrock Using Full-Scale Enhanced Reductive Dechlorination," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

Dennis, E. and Suthersan, S., "Microbial Precipitation of Dissolved Metals Using Molasses," EPA 542-N-98-010, No. 30, December 1998.

E.M. Panhorst, G.B. Page, J.B. Stephenson, and D. Glenn, "Expedited Closure of a PCE/TCA Site Using IRZ<sup>TM</sup> Technology," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

F. Lenzo, S.S. Suthersan, M. Hansen, and J.Burdick, "Enhanced Reductive Dechlorination Using Food-Grade Substrates: State-of-the-Practice," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

Hansen, M. A., Lenzo, F., Burdick, J., and Suthersan, S., "Enhanced Reductive Dechlorination: Lessons Learned at over Twenty Sites," 2<sup>nd</sup> International Conference on Remediation of Chlorinated and Recalcitrant Compounds, May 2000. Proceedings available at http://www.battelle.org/bclscrpt/Bookstore99/list\_all.cfm?OrderVar=Subject Horst, J. F., Beil, K. A., Burdick, J. S., and Suthersan, S. S., "Comparison of Natural and Enhanced Attenuation Rates through Substrate Amendments," 2<sup>nd</sup> International Conference on Remediation of Chlorinated and Recalcitrant Compounds, May 2000. Proceedings available at http://www.battelle.org/bclscrpt/Bookstore99/list\_all.cfm?OrderVar=Subject

J. Burdick, D.L. Jacobs, P. Rowland, and G. Lock, "Enhanced Reductive Dechlorination — Treatment and Vapor Control Below an Active Building," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

J.S. Burdick, K.A. McGuinnes, E. Rodriguez, and D.J. Wanamaker, "Pilot Test of Enhanced Reductive Dechlorination of TCE and Full-Scale Design," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

K.A. Beil, E. Vanyo, J.F. Horst, and M.A. Hansen, "Remediation of TCE and Nickel Via an In Situ Reactive Zone," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

Lenzo, F., "*In Situ* Treatment Technology (Chapter 8)," 2<sup>nd</sup> Ed., Lewis Publishers, CRC Press, Inc., Boca Raton, FL, 2000.

Lenzo, F., Burdick, J., and Jacobs, D., "Enhanced Reductive Dechlorination of PCE in Bedrock: Pilot to Full-Scale," WEFTEC 2000, October 2000.

Lenzo, F.C., "Metals Precipitation Using Microbial *In situ* Reactive Zones," Abiotic *In situ* Technologies for Groundwater Remediation, August 31-September 2, 1999.

Liles, D.S., Lutes, C.C., G.B. Page and S.S. Suthersan, "*In Situ* Chemical Stabilization of Metals and Radionuclides Through Enhanced Anaerobic Reductive Precipitation: Application of a Commercial Technology to DOE Needs," Poster presented at DOE's Industry Partnership for Environmental Science and Technology Conference - Morgantown, West Virginia (10/30-11/1/01), paper submitted for publication in proceedings, October 2001.

Lutes, C. and D. Liles (ARCADIS G&M, Durham, North Carolina, USA), M. Hansen, J. Burdick, S. Suthersan (ARCADIS G&M, Langhorne, Pennsylvania, USA), J. Hansen (AFCEE, Brooks AFB, Texas, USA), D. Kampbell (USEPA, ORD, Ada, Oklahoma, USA), D. McInnes (East Central University, Ada, Oklahoma, USA), Utilization of Treatability and Pilot Tests to Predict CAH Bioremediation," Poster accepted for presentation at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, May 20-23, 2002. Monterey, California, Paper submitted for proceedings.

Lutes, C. and V. D'Amato (ARCADIS G&M, Durham, North Carolina, USA), M. Hansen, J. Burdick, S. Suthersan (ARCADIS G&M, Langhorne, Pennsylvania, USA), J. Hansen (AFCEE, Brooks AFB, Texas, USA), Development of an AFCEE/ESTCP Substrate Enhanced CAH Bioremediation Protocol," Poster accepted for presentation at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, May 20-23, 2002. Monterey, California, Paper submitted for proceedings.

Lutes, C., M. Hansen, J. Burdick, S. Suthersan and J. Hansen, "Development of an AFCEE/ESTCP Substrate Enhanced CAH Bioremediation Protocol," Presented at Partners in Environmental Technology Environmental Symposium and Workshop, Washington D.C., November 2001.

Lutes, C., M. Hansen, J. Burdick, S. Suthersan, D. Liles, J. Hansen, D. Kampbell and D. McInnes, "Utilization of Treatability and Pilot Tests to Predict CAH Bioremediation," Presented at Partners in Environmental Technology Environmental Symposium and Workshop, Washington D.C., November 2001.

Lutes, C., V. D'Amato and D. Liles (ARCADIS G&M, Durham, NC, USA), M. Hansen, J. Burdick, and Dr. S. Suthersan, (ARCADIS G&M, Langhorne, PA, USA), J. Hansen, (AFCEE, Brooks AFB, TX, USA), "Advances in Understanding and Predicting Enhanced Anaerobic CAH Bioremediation," Poster accepted for presentation at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, May 20-23, 2002, Monterey, California, Paper submitted for proceedings.

Lutes, C.C., "Demonstration of *In Situ* Substrate Addition to Create Reactive Zones for Treatment of Chlorinated Aliphatic Hydrocarbons," Presented at Cape Fear Post, Society of Military Engineers, Wilmington, NC, December 14, 2000.

Lutes, C.C., J. Burdick, P. Palmer, S.S. Suthersan, D.H. Kampbell, and J.E. Hansen, "Demonstration of *In Situ* Substrate Addition to Create Reactive Zones for Treatment of Chlorinated Aliphatic Hydrocarbons - Initial Results," Poster Presented at SERDP/ESTCP sponsored *Partners in Environmental Technology*, Arlington, VA, November 2000

M.A. Hansen, K.A. Beil, F.C. Lenzo, and S. Suthersan, "Engineering Details for Enhanced Reductive Dechlorination (ERD) Technology," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

M.R. Pahr, S.S. Suthersan and E. Rhine, "In Situ Remediation of Carbon Tetrachloride and Chloroform in Groundwater," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

Nyer, E., Lenzo, F., and Burdick, J., "*In Situ* Reactive Zones: Dehalogenation of Chlorinated Hydrocarbons," Groundwater Monitoring Review, Spring 1998.

Payne, F.C., S.S. Suthersan, F.C.Lenzo and J.S. Burdick. 2001. Mobilization Of Sorbed-Phase Chlorinated Alkenes In Enhanced Reductive Dechlorination. In: Anaerobic Degradation of Chlorinated Solvents. Proc. Internat. In Situ and On-Site Bioremediation Symp. 6(2):53-60.

R. Fitzpatrick, D.L. Jacobs, and J.S. Burdick, "Enhanced Reductive Dechlorination of Pentachlorophenol and Precipitation of Chromium in Group Groundwater," Presented at The Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

S.T. Potter, F. Lenzo, S.S. Suthersan, and M.P. Kladias, "Alternative Substrate Delivery Systems for In Situ Reactive Zone Perchlorate Treatment," Presented at The Third International

Conference on Remediation of Chlorinated and Recalcitrant Compounds, Monterey, CA, May 20-23, 2002.

Shetty, N. and Liles, D., "Application of *In Situ* Reactive Zone Concept to Enhance the Remediation of Chlorinated Aliphatic Hydrocarbons," Presented at American Institute of Chemical Engineers Spring Meeting, Biological Treatment of Wastewater and Groundwater, March 15-18, 1999. Unpublished.

Suthersan, S., "Natural and Enhanced Remediation Systems," Lewis Publishers, CRC Press, Inc., Boca Raton, FL, 2001.

Suthersan, S., "Remediation Engineering: Design Concepts (Chapter 8)," CRC Press, Inc., Boca Raton, FL, 1997.

Suthersan, S., F. Lenzo, M. Hansen, J. Burdick and C. Lutes, "Short Course: Enhanced Reductive Dechlorination Using In-Situ Reactive Zone (IRZ) Technology: Practical Applications," to be presented at the Third International Conference on Remediation of Chlorinated and Recalcitrant Compounds, May 22, 2002; Monterey, California.

Suthersan, S., Maierle, M., and Palmer, P., "Utilization of In Situ Reactive Zones for Microbial Precipitation of Heavy Metals and Denitrification," Superfund XVI Conference, Washington, DC, November 6-8, 1995.

## Appendix A-2

Soil Sieve Analyses



ENGINEERING CONSULTING SERVICES, LTD.

Geotechnical • Construction Materials • Environmental

June 8, 2000

Mr. Chris Lutes

Arcadis Geraghty and Miller 4915 Prospectus Dr., Suite F Durham, NC 27713

RE: Subject: Laboratory Testing Hanscom AFB Geotechnical Samples ECS, Ltd. Project Number T7597-A

Dear Mr. Lutes:

This letter of transmittal accompanies the results of the laboratory work you requested for the above referenced project. Five (5) soil samples were tested for particle size distribution in accordance with the American Society of Testing and Materials (ASTM) designations D422. In addition, one sample was selected to split for duplicate testing, bringing the total number of tests run to six (6). All of the samples were classified either as SM, GM, or GP-GM and all contained some material in the gravel as well as silt size range. For purposes of classification, all minus #200 material was assumed to be ML.

Sample HAN-SS-IRZ-1was selected for duplicate testing, and was split for this purpose using a sample splitter, however the presence of gravel sized particles made the resulting quantity of material available marginally small. As a result of this limitation, the classification of this material is different for each of the tests. This is due to the fact that small changes in the number of gravel sized grains has a large effect on the relative grain size distribution.

Customarily, we will hold the remaining portions of these samples for 60 days unless otherwise instructed. Please contact us if you would like any additional testing or if you have any questions.

We appreciate this opportunity to provide our laboratory testing services to you and look forward to serving you again in the near future.

Respectfully,

William L. Stone

William L. Stone Laboratory Manager

P.O. Box 12015, Research Triangle Park, North Carolina 27709 • (919) 544-1735 • Fax (919) 544-0810 • 1-800-327-5832 • www.ecslimited.com

Aberdeen, MD · Atlanta, GA · Austin, TX · Baltimore, MD · Chantilly, VA · Charlotte, NC · Chicago, IL · Danville, VA \* · Frederick, MD · Fredericksburg, VA · Greensboro, NC Greenville, SC · Norfolk, VA · Research Triangle Park, NC · Richmond, VA · Roanoke, VA · Williamsburg, VA · Wilmington, NC\* · Winchester, VA \* \*Testing Services Only

ARCADIS GERAGHTY & MILLER	Laboratory Task (	Order No./P.O. No		_ CHAII	N-OF-C	USTOD	Y RECORD Pa	age	of
Project Number/Name RN00990	0,0003,MA00-	3 [	<u></u>	ANALYSIS		) / SIZE			
Project Location Hanson	AFB		/	/	/	/	7 7		
_aboratory					/	/			
Project Manager Chris Lutes	(919-544-455								
Sampler(s)/Affiliation J13/60	-ARCHIE	- 2 /							
Sample ID/Location Matrix	Date/Time Time Sampled Lab D	/0/					Remarks	;	Tota
IRZ-4/44-46 5	51100 9:50	1	[	Í		(	(		
IRZ-INJ 5	51200 -	1							1
JRZ-2 (31.41) 5	5300 1540	Ì							1
IRZ-3/44-46) 5	58,00 1100	1			···· · · · · · · · · · · · · · · · · ·				1
MAN-55=FRZ-1/44-15/5	51600 12:00								1
								*	
ample Matrix: L = Liquid; S = S	olid; A = Air						Total No. (	of Bottles/ Containers	5
Relinquished by:	Organiza	tion: ARCANE	-	Da <sup>.</sup>	te 5 /1	,00	Time 2:007	Seal I	ntact?
Received by:	Organiza	tion: <u>605, 1</u> -	d	Da <sup>-</sup>	te <u>5</u> //	7100	Time 11:25	14 (Yes) N	io N//
Relinquished by:	Organiza	tion:		Da	te/	1	Time	Seal I	ntact?
Received by:	Organiza	tion:		Da	te/		Time	_ Yes N	o N//
pecial Instructions/Remarks:		<u> </u>							
			60			•			



P.O. Box 12015 RTP, NC 27709 (800) 327-5832 (919) 544-1735 Fax

#### **Grain Size Analysis**

Job Name Arcadis, Geraghty & Miller Job No. 7597-A

Boring No.IRZ-INJSample No.IRZ-INJSample Depth (ft.)Not AvailableSample DescriptionGray silty gravel with sandU. S. C. S. Classification:GM

Dry Weight of Sample 788.39 g

Sieve Number	Diameter [mm]	Percent Passing	Percent Retained
1 1/2"	37.5	100.0	0.0
1"	25.4	96.7	3.3
3/4"	19.10	81.0	19.0
3/8"	9.52	67.4	32.6
#4	4.76	56.2	43.8
#10	2.00	46.1	53.9
#20	0.84	38.2	61.8
#40	0.42	31.4	68.6
#60	0.25	26.0	74.0
#140	0.106	16.7	83.3
#200	0.074	13.2	86.8



.

\_\_\_\_\_



P.O. Box 12015 RTP, NC 27709 (800) 327-5832 (919) 544-1735 Fax

#### **Grain Size Analysis**

Job NameArcadis, Geraghty & MillerJob No.7597-A

Boring No. HAN-SS-IRZ-1Sample No.HAN-SS-IRZ-1 (44-45) ISample Depth (ft.)44 - 45Sample DescriptionGray silty sand with gravelU. S. C. S. Classification:SM

Dry Weight of Sample 406.55 g

Sieve Number	Diameter [mm]	Percent Passing	Percent Retained
	[]	1 4551116	noumou
1 1/2"	37.5	100.0	0.0
1"	25.4	100.0	0.0
3/4"	19.10	93.9	6.1
3/8"	9.52	70.4	29.6
#4	4.76	57.6	42.4
#10	2.00	47.1	52.9
#20	0.84	38.4	61.6
#40	0.42	31.5	68.5
#60	0.25	26.3	73.7
#140	0.106	17.1	82.9
#200	0.074	13.2	86.8





P.O. Box 12015 RTP, NC 27709 (800) 327-5832 (919) 544-1735 Fax

#### **Grain Size Analysis**

Job NameArcadis, Geraghty & MillerJob No.7597-A

Boring No. HAN-SS-IRZ-1Sample No.HAN-SS-IRZ-1 (44-45) IISample Depth (ft.)44 - 45Sample DescriptionGray poorly graded gravel with silt and sandU. S. C. S. Classification:GP-GM

Dry Weight of Sample 344.22 g

Sieve	Diameter	Percent	Percent
Number	[mm]	Passing	Retained
1 1/2"	37.5	100.0	0.0
1"	25.4	100.0	0.0
3/4"	19.10	89.1	10.9
3/8"	9.52	64.5	35.5
<b>#4</b>	4.76	52.4	47.6
#10	2.00	42.2	57.8
#20	0.84	34.2	65.8
#40	0.42	27.9	72.1
#60	0.25	23.3	76.7
#140	0.106	15.2	84.8
#200	0.074	12.0	88.0





P.O. Box 12015 RTP, NC 27709 (800) 327-5832 (919) 544-1735 Fax

#### **Grain Size Analysis**

Job Name Arcadis, Geraghty & Miller Job No. 7597-A

Boring No.IRZ-2Sample No.IRZ-2 (39-41)Sample Depth (ft.)39-41Sample DescriptionGray silty sand with gravelU. S. C. S. Classification:SM

Dry Weight of Sample 777.12 g

Sieve Diameter Percent Percent Number [mm] Passing Retained 1 1/2" 37.5 100.00 0.00 1" 25.4 93.09 6.91 3/4" 19.10 90.58 9.42 3/8" 9.52 84.78 15.22 #4 4.76 79.83 20.17 #10 2.00 69.98 30.02 #20 0.84 58.41 41.59 #40 0.42 49.20 50.80 #60 0.25 40.78 59.22 #140 0.106 20.52 79.48 #200 0.074 15.58 84.42





P.O. Box 12015 RTP, NC 27709 (800) 327-5832 (919) 544-1735 Fax

#### **Grain Size Analysis**

785.74 g

Job NameArcadis, Geraghty & MillerJob No.7597-A

Boring No. IRZ-3Sample No.IRZ-3 (44-46)Sample Depth (ft.)44 - 46Sample DescriptionGray silty sand with gravelU. S. C. S. Classification:SM

Dry Weight of Sample

Sieve Diameter Percent Percent Number [mm] Passing Retained 1 1/2" 37.5 100.00 0.00 1" 25.4 93.22 6.78 3/4" 19.10 85.52 14.48 3/8" 9.52 72.21 27.79 #4 4.76 62.01 37.99 #10 2.00 51.38 48.62 #20 0.84 42.84 57.16 #40 35.68 0.42 64.32 #60 0.25 29.84 70.16 #140 0.106 18.23 81.77 #200 0.074 14.16 85.84





P.O. Box 12015 RTP, NC 27709 (800) 327-5832 (919) 544-1735 Fax

#### **Grain Size Analysis**

Job Name Arcadis, Geraghty & Miller Job No. 7597-A

Boring No. IRZ-4Sample No.IRZ-4 (44-46)Sample Depth (ft.)44 - 46Sample DescriptionGray silty sand with gravelU. S. C. S. Classification:SM

Dry Weight of Sample

725.56 g

Diameter [mm]	Percent Passing	Percent Retained
37.5	100.00	0.00
25.4	100.00	0.00
19.10	93.30	6.70
9.52	75.93	24.07
4.76	65.26	34.74
2.00	54.75	45.25
0.84	45.66	54.34
0.42	39.38	60.62
0.25	29.28	70.72
0.106	20.02	79.98
0.074	15.81	84.19
	Diameter [mm] 37.5 25.4 19.10 9.52 4.76 2.00 0.84 0.42 0.25 0.106 0.074	Diameter [mm]Percent Passing37.5100.0025.4100.0019.1093.309.5275.934.7665.262.0054.750.8445.660.4239.380.2529.280.10620.020.07415.81



## Appendix A-3

Standard Operating Procedure: Field Screening of Water & Vapor Samples for Volatile Organic Compounds by Gas Chromatography

## Standard Operating Procedure Field Screening of Water & Vapor Samples for Volatile Organic Compounds by Gas Chromatography

#### 1.0 Scope and Application

This gas chromatograph/photoionization detector (GC/PID) method is applicable to the detection of trichloroethylene (TCE) and cis-1,2-dichloroethene (cis-DCE) in water and vapor samples at the Hanscom AFB project site.

#### 2.0 Summary

Samples are analyzed on site using a portable GC/PID, using direct injection of . vapor samples from off-gas treatment systems or injection of the headspace of water samples. The identification of target analytes is determined by retention time comparison to standard materials. The concentration of target analytes is calculated using the external standard technique.

#### 3.0 Interferences and Limitations

This method is useful for providing reliable screening data quickly and cost effectively, and should not be considered definitive data. As with any GC method, any compounds co-eluting with cis-DCE or TCE can produce an erroneously high concentration or a misidentification of the target analyte.

#### 4.0 Sample Collection

Standard volatile sample collection procedures should be followed for collecting a representative sample. Water samples should be collected in 40 ml VOA vials, with no observable headspace.

#### 5.0 Sample Preservation

- 5.1 Water samples should be stored at 4°C until analyzed. Unpreserved samples will be analyzed within 7 days of collection and acid preserved samples will be analyzed within 14 days.
- 5.2 No preservation of vapor samples is required. Vapor samples will be analyzed within eight hours of collection if using Tedlar Bags. The holding time using the Microseeps technique is 14 days.

#### 6.0 Apparatus

6.1 Photovac Model 10s50 portable GC with PID, strip chart recorder, and electronic integrator.

1

777221GC-SOP

- 6.2 Capillary column 1 meter CPSIL 5 pre-column with a 9 meter CPSIL analytical column.
- 6.3 Photovac low power isothermal oven powered by a globe/cell rechargeable battery pack.
- 6.4 Dilution, standard preparation, and sample injection syringes (with syringe cleaner) in varying sizes allowing for accurate injection volumes from 10 to 1000  $\mu$ L. Syringes used for injection of sample into the GC should be gas tight.
- 6.5 Dual flow meter (range 0 to 55cc/mL).
- 6.6 Zero grade air will be used as the carrier gas.
- 6.7 Tedlar Bags and 40 mL VOA vials.
- 6.8 Calibration gas mixture for vapor analysis.
- 6.9 Liquid standards, methanol and distilled water for preparation of water standards.

### 7.0 GC/PID Operating Parameters

- 7.1 Set gain at 10<sup>°</sup>
- 7.2 Set isothermal oven at 30°C
- 7.3 Set flow at 6 mL
- 7.4 Set backflush at 30 seconds
- 7.5 Set runtime at 360 seconds

## 8.0 Standard Preparation

- 8.1 All standard preparation shall be documented in the Standard Preparation Logbook. At minimum, the following will be recorded for each standard solution; Date, initials, source of standard, composition of standard, and lot #'s for reference solutions.
- 8.2 All standards must be labelled with the standard name and date prepared, such that it can be traced back to the Standard Preparation Logbook.

777221GC-SOP

66 SPTG CEV HAFB

**GC-SOP** 

#### 8.3 Gas Standard

- 8.3.1 A specialty gas mixture cylinder with a certified concentration of 1 ppm (v/v) of cis-1,2-DCE and TCE will be purchased for standardizing the GC for vapor samples.
- 8.3.2 The cylinder will be used to create a working standard by filling a Tedlar bag dedicated to the standard. This working standard will be valid for eight hours.
- 8.4 Liquid Stock Standard
  - 8.4.1 Certified reference mixtures of individual standards for cis-1,2-DCE and TCE at a concentration of 5,000 μg/ml in methanol will be purchased for the preparation of the working water standards.
  - 8.4.2 The reference standards should be tightly sealed and stored in the refrigerator and allowed to warm to room temperature before use.
  - 8.4.3 Observe the meniscus of a previously used standard to ensure that the vial has remained tightly sealed during storage. If the meniscus does not match the marking on the vial from its previous use, the standard must be discarded and a new reference mix should be used.
  - 8.4.4 After the reference has been used, mark the meniscus of remaining standard before returning the standard to the refrigerator for storage.
- 8.5 Liquid Working Standard
  - 8.5.1 The working standard, at a concentration of 1,700  $\mu$ g/ml of each target analyte is prepared by combining 300  $\mu$ l of each reference standard and 420  $\mu$ l of reagent grade methanol.
  - 8.5.2 The working standard should be prepared in a screw top vial equipped with a Mininert valve to reduce evaporation during use, and must be refrigerated between uses.
  - 8.5.3 A new working standard will be prepared when calibration standards indicate degradation of the standard or a significant change in standard concentration due to evaporation. See Section 11.0.

777221GC-SOP

March 31, 1999

3

#### **GC-SOP**

- 8.6 Liquid Calibration Standard
  - 8.6.1 Add 34 ml of distilled water to a 40 mL VOA (40mL is the nominal volume, but the actual capacity is 44 mL). This will leave a headspace volume of 10 mL.
  - 8.6.2 Allow the working standard to warm up to room temperature.
  - 8.6.3 Open the Mininert valve, insert a 10 μl syringe, and fill the syringe with 10 μl of the standard.
  - 8.6.4 Inject the working standard into the water in the vial prepared in step 8.6.1.
  - 8.6.5 This calibration standard, equivalent to 0.5 ppm, or 500 ppb, will be used to calibrate the GC.
  - 8.6.6 The calibration standard must be used within eight hours.
  - 8.6.7 Additional concentration standards can be prepared by adjusting the volume of working standard added to the 40 mL vial.

#### 9.0 Sample Preparation

- 9.1 Samples will be allowed to set undisturbed at room conditions in order to reach thermal equilibrium prior to analysis. (A minimum of 30 minutes for vapor samples and 60 minutes for liquid samples.)
- 9.2 10 mL of water will be removed from the vial to create a headspace prior to analysis. Insert a syringe needle through the vial septa then use another syringe to withdraw and discard 10 mL of sample. Water samples will then be shaken vigorously for 20 seconds prior to extracting the headspace for analysis.

#### 10.0 Initial Calibration

- 10.1 A three-point calibration curve will be conducted to establish the linear range of the instrument. A zero grade air blank will be analyzed before and following the three standards.
- 10.2 Three different volumes of the specialty gas standard will be used to generate the curve for vapor analysis.
- 10.3 Three different concentrations water standards will be prepared by injecting three different volumes of the working standard into vials filled with 34 mL of water.

777221GC-SOP

**GC-SOP** 

10.4 The linear regression for the three points will be calculated. The curve will be acceptable if the correlation coefficient for the curve is greater than or equal to 0.98.

#### 11.0 Daily Calibration

- 11.1 Daily calibration shall be performed using a working standard within the calibration range established with the three-point curve. A blank will be analyzed before and after the standard.
- 11.2 The response factor will be calculated by:

$$R_f = (A/B)/C$$

Where:

- A = Total instrument response (sum of the area under the peak in the retention time window).
- B = Injection volume (µl) of the standard
- C = concentration of standard
- 11.3 If the daily response factor is >150% the mean response factor for the initial calibration curve, then the appropriate check and corrective action will be taken (i.e. reinjection of standard, re-make standard, clean detector etc...).

#### 12.0 Sample Analysis

- 12.1 Samples will be injected into the instrument in the same manner as standards. Injection volumes may vary based on sample concentration.
- 12.2 If no peaks of interest are observed or the peak heights are below the calibration range, then reanalysis may be performed at a higher injection volume (up to 1000µl).
- 12.3 If a sample peak exceeds the linear range established with the three point calibration curve, then reanalysis may be performed using smaller injections (down to 10  $\mu$ l). A dilution may be conducted if the instrument response still exceeds the linear range.
- 12.4 The concentration of a compound can be determined either manually using Rf's or by setting up the Photovac's integrator.

12.4.1 Concentrations by Rf's are calculated by the formula:

 $C = A/(R_f x B)$ C = Concentration of compound ppm

777221GC-SOP

5

GC-SOP

- A = VS of the peak as printed out by the integrator
- $R_f$  = Daily response factor (Reference Section 10.2)
- $B = Injection volume (\mu l)$
- 12.4.2 Following the calibration of the 10s50, enter the required data into the integrator memory per the instruments operation manual. Correct for variation in injection volume.

C = A/(B/D)

C = Actual concentration of the compound in ppm

A = Concentration in ppm calculated and printed out by the 10s50 integrator

B = Injection used for standard that the integrator was set up on D = injection volume of sample

#### 13.0 Quality Control

- 13.1 Instrument blanks will be run at the start of each day.
- 13.2 A zero grade air blank (syringe blank) will be analyzed at the start and end of each day, after every 20 samples and after grossly contaminated samples.
- 13.3 A duplicate sample will be analyzed at a frequency of once per 50 samples analyzed.
- 13.4 Calibration standards will be analyzed at the start and end of each day and at a minimum frequency of once per every 20 samples.
- 13.5 A calibration standard will be analyzed whenever system maintenance (such as septa change, carrier gas recharge) is performed.

#### 14.0 Method Detection Limit

- 14.1 The method detection limit (MDL) will be calculated at least once per year.
- 14.2 Prepare a calibration standard at three to five times the estimated MDL.
- 14.3 Analyze this standard seven times.
- 14.4 Calculate the standard deviation of the seven results.
- 14.5 The MDL is 3.14 times the calculated standard deviation.

777221GC-SOP

## Appendix A-4a

Boring Logs and Well Construction Diagrams

ARCADIS GERAGHTY & MILLER DAILY LOG A X Weil(s) IRZ-2 Project/No. KN 2099111. 10033 MAMAR \_\_\_\_Page \_\_\_\_ Hanscom AFB Site Location 5 Presared By ÷ Description of Activities Date/Time  $\infty$ Slann 82 arriver 500WI A n.C C.VPD PA nl M. MARR COM NRO container? The sample nJamanan containels a short with 11/5 Savaran have Th 1170 p/PA ロの 1330 'snr 0 n MANGANO AUDPA 411 440 221  $\tau_1$ 30 39-41 - dallad Sommer MURNER intill 520 5.200 1. eel:12 +0 TA 41 de in 2 حبرد 15 Ani 1.201 milaria SANY that w n, XX al- in this to do to so 335 sand to 31 42 + to 18 43\$ < 1700 : it stop

## ARCADIS GERAGHTY & MILLER DAILY LOG

۱

ζ  $\gamma$ oage \_\_\_\_ Project/No. Weil(s)\_\_\_\_ Site Location Prepared By Description of Activities Date/Time 781-377-4455 Tom bos SIIMN 830 Dec  $\sim$ 411992 Tem otor 915 Tom (umm TIMPERC 0)5 SILAPA aucris because halo -750 avaran Bnsite 340 lacations surging up queers talked 34-50nd 1430 ochina-dillers GLAPFING DD BURRAN Th n washing-Tomo SugarsTy ang P

## ARCADIS GERAGHTY'S MILLER DAILY LOG

RN009900.0003. MADO3 Project/No. Page \_ Weil(s) 77 Site Location Han 4000 Prepared By An . . Date/Time Description of Activities 5/5/00 1. NO WA 740 aroup said ai 8 X I द्र २७ 11/ 9 h -UNC PSIM 5A 0 える

ARCADIS GERAGHTY & MILLER SAILY LOG D.003. MAM3 Page \_\_\_\_ Weil(s)\_\_\_ Project/No. ì Site Location IAN Precared Sy Date/Time **Description** of Activities - 0 20 74 VPI 7 🗸 Y R ?/) on1  $\mathcal{A}$ nc<u>२</u>.१ 2 -. 71 د درسد E -7 5915 5.7m 113 212 2011 (mni) '?]X T.15 AM :700 < 11: . ? .?/ ٢L ຈ. -17 (nmm) *....* , 15.30 5 20 212 520 11  $\rho$ 62 Ĉ < 1 · , SOU - 41 ノイショ 25 شرار 1600 -.

### ARCADIS GERAGHTY & MILLER

## DAILY LOG

Weil(s)	Project/No. 21299222223, MAV23 Page
	15055
Site Location	15. 1
Prepared By	5 Arrigiand
Date/Time	Description of Activities
stam	harry number 80
5.75	anied angite drillers setting re!
	well sot at 54- seven to 39- sand to 37
845	state with Tam R- wants to make ripetion
	1. F. 11 1. 7 -
IPIA	drivers arounting as with
F	The second of the second of a second the second
11/103	drillars decision - anomant
(1:07)	softing to will - 2- 11
122)	
1530	down to 24' grev silty clay
1630	down to 29'-packed up - leftste
C	(95)
24	
	1 . 29
	idan
	La <sup>2</sup>

## ARCADIS GERAGHTY & MILLER DAILY LOG

Weil(s)\_\_\_\_\_ Project/No.\_\_\_\_ R 11 0099003, 0003, M 400 3 Dage . Site Location 101 Prepared By **Description** of Activities Date/Time nn 745 ALTAP ョッイi . RM TP Plan 10 Э 100 0 05 1005  $21^{T}$ 0 10 20 DCOCAI SUSTP Th in Ð 1200 ackaray nd 737 De hist Ah/ Si (CSII 1 ~ < 'n 15
Project/No. RN0099BO.0003. MADO3 Page \_\_\_\_ Well(s) 1FCEE Site Location Variara no Prepared By Description of Activities Date/Time 50 Im laun arrived onsite 800 sample T.!! TP7-4 lor in SCODP ึกเ 1.10 24" 10 mapin 714 . 830 201 1.1 415 < t.i!44-45-T ina  $\mathcal{Z}_{f}$ DDM SIMI un to hedrack at 51.2" 17/3 Tar 1200PL SP Ting W? stersupered to 36 1390 in'

Weil(s)	Project/No. R.1009900,0003Page	
Site Location_	AFCEE	
Prepared By	5 Tair arand	
Date/Time	Description of Activities	
5/12/00		
740	prived onsite	
	coordinated with Rich From IT for airport	
	clearance and power down of BKV line	
	moved sul an 3 + dd	
	Legan dy. Ing	
<u> </u>	Jawn to the cra- 1- the the	
15 77	1 At to I take	
	PH I DI LINK ON SKIPNI	
161.5	That Valv & On condensation line	
/620	lines from wells are 14" piping	
	check valves are l'-did not install	•
	-	

Weil(s) The factor	Project/No. 12NCC9900.0003
Site Location	AFCEF
Prepared By _	J. Bonsteel
Date/Time	Description of Activities
51500	
9.30H	Arrive in Lexington. Gill Sam who meets we and
	brings me to site, Meet drillers (Hike Hike)
	Rich Fem IT Grap. Ec. arr wort w/san,
	Sampling requirements = te-
6.45	Sam de arts site Well Cotath (513' bis)
	Installing well JasiAcy.
11.15	Well screen 3655, faid to 34', bratante to 32'
	apart to artain
	Call Tom Best to intratice might joine him my
	cell number and what an et procurss
11:40	AFTER BRING DTR MESSurements, Mike determines
	Well bottome 49', sand to 30' (areshot by 2').
	Talk Wism and have drilled they to which at ?'
	of sand. Attempt + include it is hell resing
	(can't get like turn)
12:20	Finish apartman injection well
13:10	Mareria + Get-up on IRZ-1
13:40	Brgin Miz-1
14.15	30' Gasing in around. Tom Best stops out and
	Says me can leave rich have down on hole overnight.
15:45	Drive 35' casing into around.
1/2:30	itie Un staining inter a raining

7

Weil(s)	_ Project/No R. NOO9900, 0003 Page
Site Location_	MANECUM AFF3
Prepared By _	J. Barsteel
Date/Time	Description of Activities
51600	
7:00A	Arrive ansite, Set-up sample bottles. Get
	FOD PFD. The difficulty fouring at contest.
	Will call form to confirm sample bottle
	quantity - FAD/PAD,ct.
7:45	Geologic anive Anaiting clearance to cretion AFB.
0.30	Access quarted Prive to 522-1 Taction
9:20	Drive 35-40' rapiroy.
10;00	Ruled OFF site by MASSPORT due to environcy
	landing.
10.30	Told emergency landing would occur in 2-3 his
	the to feel spoly of plane. Asked For
	permission to continue work with requested to
	leave again - ramission denied. Will have
	to wait with landing with report work.
	On start-by.
11:15	Notified the plane will make landing at different
	aigort. Changed to continue work
11.45	Attempt 44-46' 5 Doon - very clase ister
	blass after getting to 45'
12:00	Collect MAN -55- JAZ-1 (44-45')

:

ARCADIS GER	AGHTY & MILLER
DAILY LOG	mar 919-Stylen
	insurs 1-035
Well(s) FFF-	Project/No
Site Location	Bidford (Hansen AFB)
Prepared By	J. Parstel
Date/Time	1400 Description of Activities
12:50	Batters exanter a droth. Unde to drive
	asing Fither. Call Rich Ravell) and confirm
	it is ok to set well in open burchde
	@ 48.5' (Protrack).
13:05	Tak while + ap aver progress.
13:55	Nal Lottom e 42,4
	Screen 424 to 33.4
	Sand to 30.51 IRZ-1
	Bartonite to 28.5' Well Construction
15:05	Grat to grade
	Talk w Christites (NC) about well details + lab
	For arain, size analysis
	Filk w/ Chuck (Conell) about development water storage
	disposal. Will great w/ Sam to see if he knows
	, but agreement was readed w/ Tom 135t.
16:00	Frich FRZ-T. Patring up rig + equipment
16:45	Dogo off Sampes & Fel the
	• ·
	State Black

CAME TONOL THE

ARCADIS GERAGHTY & MILLER

DAILY LOG

Weil(s)	Project/No. RN009900.0003. 14003 23ge 1
Site Location	Manscen AFB
Prenared By	J_Bonsteel
Date/Time	Description of Activities
517/00	
7:001	Arrive ousite
9:00A	Geologic armus ousit. Unlead empty chuns, Call
	Em Best (leave message) to see if we can bring
	ra(bam up) onto poperty.
9:30	Set up on TR2. This to decolop w/ "whater sump
	<u>JZ2-2 to JZ2-3 ~ 30</u>
	$\frac{67}{11} \frac{10}{12} 10$
	61 to 122-200 x 16'
	05 10 +12-1 -10
10.7.0	Deducing TOZATATE Witherer and + Foot value
0.50	Ring on 1.5 her - 2359 still sitty timbed
12400	$\frac{1}{1} = \frac{1}{2} $
	10v = 200g
1:1008	TRZ-IRT dudoped for 3 his. 505g remarch
	Turbidity -> slightly turbid (sitty (better than, star))
1:30P	Set up on IPZ-1 and start dalapment.
J:3012	200g at of FRZ-1, daring up, will pump 250y They stop
3:157	250 g remard IRZ-1 Jean
1	

Weil(s)	Project/No
Site Location_	Housean AF13
Prepared By	J. Bonsteel
Date/Time	Description of Activities
5100	
7:00A	Arrive ansite. Gologic arrives ansite
3'.15A	Moving Soil drams.
8'.50	Set-up for development of IPRZ-4
9:101	Start IRZ-4 Development (whecher pungs + Fact value)
	1 1v= 7g 3v=21g 10v= 70g
10:55A	TRZ-4 2000 removed - Slightly Inted (sitty)
11:45	Set-up for delegoment on Fizz-2
12:00	Call Chris Lits + leave message. Then call
	Rich (Lavel) and discuss development. Foot
	the Grains method append to be contempodeding
	as violation of whater surplused will
	constantly stirup File sitt. Curetly using
	rialwards on IRZ-2 and very little volume
	of water bang remarket. Discuss possibility
	of using alternative method,
19:30	At Geologic is they have while pumps with
	them, they do, but no tubing.
1.00P	Speak in berf Fundick about development. Agree
	That best way to develop will have to be
	using "whether" and to remare more terbid
	water than use cantrifical to Dick up any
	istimat of the mal bottom.

:

Weil(s)	Project/No
Site Location_	Honstan ATTZ
Prepared By	J. Barsteel
Date/Time	Description of Activities
1,202	Speck willich (lande) + determine that thest
	method to douter will be whater Dunp (time
	15. quelity of using centrificgel doesn't add ys)
	Will purge last a wals with "super-turbed"
	water raised.
	Developma JRZ-21
2:05	Water From TRZ-2 cleaning to sightly turbid (100g)
	Will Jump 125g (TOTAC) at them start IRZ-3
2:20P	132-2 125 removed Slightly turbid
9:295	Set up on IRZ-3
2:35P	Dadoping JRZ-3(125 15g clar)
3:30P	TRZ-3 25g remard slightly tubid
<u> </u>	· .

## ARCADIS GERAGHTY & MILLER Sample/Core Log

Boring/Wel	IRZ	-	_Project/No.	RNOC990	0,0003 MAC	03		Page	of	_
Site Location		Bedf	And , MA		Drilling Started 5	shu	Drilling Completed			
Total Depth			Feet	Hole Diameter	inches	Type of Coring I	- Sample/ Device	<u>Solt</u>	5000 (44 41	- =)
Length and of Coring D	Diameter evice		Duethi	th Cosing		_	Sampling Inte	erval	feet	
Land-Surfac	ce Elev.		feet	Surveyed	Estimated	Datum				-
Drilling Fluid	d Used		Potebk	Water			Drilling Metho	bd	Drive Twash	L
Drilling Contractor		Geol	egic			Driller	Hike	_Helper	Mike	
Prepared By	-	J.Ro	nstrl			Hamme Weight.	r 	Hamme Drop	erins.	
Sample/Core (feet below lar	Depth nd surface)	Core Recovery	Time/Hycraulic Pressure or Blows per 6							
From	То	(feet)	Inches	Sample/Core Descriptio	nc A f					٦
0			ļ	- KOFOI ( 5	and-cky-cy	05.5	<u>}</u>			
1	10			DK BRUM	GRY Fm	<u>SA</u>	17,00	5 <b>C</b>		
10	20			Orgy F	AND STET	+CU	M'			4
20	33			Gray F.S	AND - SUT					1
33	35			DRIGET STAT	E GR. HVR t	small	COFREE	5		
35	44	1		The- to	nsp. silt flow	GRU	Small	an	4	
				OVENUE	17	$\overline{OT}$	<i>,</i> ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	$\overline{}$		
411	45		10,210	Nort dans	TUI Con	an de	and Lann	-	d).	
	- 10			L'M W	hist Di			~~~~~	<u> </u>	
					por (FH		- pp	$\rightarrow$		
				-	01106'					
				op of red	- <u>e</u> 400					
				• 						
				*****						
1										

02/10/2002 10:38 FAX 978 937 7555 ARCADIS GAM INC. Ø 003 919-544-535 RN009900.0023. HADO3 Page 3 Zelfod Harson AFB Barte 5/1/00 Description of Activities evanter @ Battas the Order to drive adin Confirm and Ś 421 RA orende 6 48.5' Provat 13:05 Idk 13:55 Va 3 TA 0 33,4 305 TRZ 1 6 28.5 Wal Costration 10 5:05 ande Gat 0 Talkw NC) about well details + lab aran gze analisis Fil velopment water strenge : will grat 42 F an b the know aarranget was round m Rest. 6:00 Paking up • • 3:45 States the

## ARCADIS GERAGHTY & MILLER Sample/Core Log

Boring/We	IRA	7-)	Project/No.	MARO 7	RNDD99AA	<u>199</u> .	?	Page	of
Site Location	Ha	SCOM	AFB S	itel	Drilling Started	3/10	Drilling Completed		
Total Depti	n Drilled	45	Feet	Hole Diameter	inches	Type of Coring [	Sample/ Device	<u>-5</u> PI	1/1 span
Length and of Coring D	Diameter evice		long	2" diant	ter		Sampling Inte	7 erval	feet
Land-Surfa	ce Elev.		_feet	Surveyed	Estimated	Datum			
Drilling Flui	d Used						Drilling Metho	bd	angers
Drilling Contractor		Geo	Logic			Driller	Mike	Helper	Todd
Prepared By		Day	rigran	<u>nc/</u>		Hammer Weight	140	Hamme Drop	r ins.
Sample/Core (feet below ian	Depth nd surface)	Core Recovery	Time/Hydraulic Pressure or Blows per 6						
From			incres				- Hanna	:12	daire
				Lite 1	11 14(1 Gra	SEV		114.0~	AT A DAGE
0	0.5			Oppnic,	ich tansoil	1210	nts		
0.5	10			sand tar	misth	DSC	fine-	mpd	/
	-{\			Same al	avelsized ~	notes'a	1 Coun	ded	
10				ster-	sand satinin	tpd	,, ,, ,, ,, ,, ,, ,, ,, ,, ,, ,, ,, ,,	/	
10	29			some fi	ne-med so	nd	situra	ted	
19	25			sand - ale	V. saturatod	Fre	-med		
25	31			sandesili	Emastly Fil	NP Gr	ev 59	twat	ed
.31	39			suspected	till lavera	vaers	PRCOU	inte	rina
				Jarger ra	cks				
37	41	Same	led	VELY DOO	rly sorted	lava	ers of	Sar	d
			48-57-	and silt.	arev. tight	y pa	ckel		
			5-77			//			
					*****				

## ARCADIS GERAGHTY & MILLER

### Well Construction Log (Uncolsolidated)

	T the LAND SURFACE	Project <u>AFCER</u> Town/City Le photop	
	drilled hole	County Permit No Land-Surface Elevation and Datum:	StateA
	- Well casing, <u>4</u> inch diameter, Backfill Grout <u>Por Thand</u> <u>28</u> ft <sup>*</sup>	feet Installation Date(s) <u>5/3/00</u> Drilling Method <u>60/16w S</u> Drilling Contractor <u>Geolog</u> Drilling Fluid <u>Woter</u> Development Technique(s) and Date(s)	Surveyed Estimated
	Bentonite Surry		
	<u>33.5</u> n	Fluid Loss During Dnilling Water Removed During Development Static Depth to Water	gallons
	Well Screen. inch diameter <u>Q.01</u> slot	Pumping Depth to Water	feet below M.P.
	Gravei Pack	Yieldgpm Specific Capacitygpm/ft	Date
		Well Purpose	
	<u>43.5</u> n° <u>45</u> n°	Remarks	
44	Measuring Point is Top of Well Casing Unless Otherwise Noted. * Depth Below Land Surface	Prepared by	

.

$\langle \langle \langle \rangle$	$\sim$	5		Jupa may	
	DIS GERAG	HTY & MILL	ER	• • • • • • • • • • • • • • • • • • • •	
Samp	le/Core	Log		11	
Boring/We	" <u>I.R.Z</u>	1-3.	Project/No.	R/079903.003	
Site	ITC	イビ		Drilling _/_	Drilling 5/0/30
Location	_At(	EE		Started 5/5	// 1 _ Completed / 2/0//
Total Depti	h Drilled	54.5	Feet	Hole Diameter inches	Type of Sample/ Coring Device <u>SP/17</u> SP2337
Length and	I Diameter	_	;		
of Coring D	)evice		spoon_		- Sampling Interval
Land-Surfa	ce Elev.		feet	Surveyed	Datum
Drilling Flui	id Used	- <u></u>	wher	·	Drilling Method
Drilling Contractor		col a	gíc_		Driller Mike Helper Tann
Prepared By		5 7	, , <u>rrigra</u> i	20	Hammer Hammer Weight <u>14</u> Drop ins.
Sampie/Core	Deoth		Time/Hydraulic		
(feet below la	ind surface)	Core	Pressure or		377 44-2-
5 mm	To	Recovery	Blows per 6	Sample/Core Description	
				sond tor figs man	famined lase moist
10	1			heraning let to	of instar table
10	21			same sand	
	21	· ·		Ease a cand acercin	calar
	20			silt sout acard	fine sandasilt
-25				coturts of	
72	32			Siturday arey 5	ativated law plastic
23	4-1				initation is the second
<u></u>	12			- In-Treduc	<u> </u>
6.17	LI LI			koulder Hock re	11- Dassihla
	11			bedrach materia	
44	40	14	68-73-44	to coody ested ant	Erial subsourced
	-13	<u> </u>	10 01	College a cord for	
		H m m			
45	11 J	23	62-58.	mostly argular cot	bles and tightly
				Departed sand	· V /
545				hedioch	
1					

### ARCADIS GERAGHTY & MILLER

# Well Construction Log (Uncolsolidated)

•

$frt \\ \psi  und surface$ $frt \\ drilled hole \\ Well casing, \\ Q inch diameter, \\ schedule 40 \\ Backfill \\ Grout Portland$	Project <u>RMOMMONO3</u> well <u>IRZ-3</u> Town/City <u>Bedford</u> County <u>Middlesex</u> State <u>MA</u> Permit No. Land-Surface Elevation and Datum: feet Surveyed Estimated Installation Date(s) <u>5/8/00</u> Drilling Method <u>driveg wash/casing</u> Dnlling Contractor <u>Geologic</u>
Bentonite sturry 38 ft <sup>•</sup> Speilets	Drilling Field
<u>4Ω</u> π <sup>•</sup> Well Screen. <u>4</u> inch diameter <u>2</u> Stot	Fluid Loss Dunng Drilling      gallons         Water Removed During Development      gallons         Static Depth to Water      feet below M.P.         Pumping Depth to Water      feet below M.P.         Pumping Duration      hours         Yield      gpm       Date
Gravel Pack Sand Pack Formation Collaspse	Specific Capacity gpm/ft Well Purpose
Measuring Point is Top of Well Casing Unless Otherwise Noted. • Depth Below Land Surface	Prepared by

•

4

## ARCADIS GERAGHTY & MILLER Sample/Core Log

Poring M/ell	127	-4	Project/No.	RN09910.000.3.	MAD3Pageof
Site	-AA	- 1+		Drilling	-// Drilling -////mO
Location	AF	CEE	-	Started	<u>/9/00</u> Completed <u>5///////</u>
Total Depth	Drilled	.51.2	Feet	Hole Diameter inches	Type of Sample/ Coring Device <u>Split Synon</u>
Length and	Diameter		<u>~</u>		
of Coring D	evice		2'<2001	2	Sampling Intervalfeet
Land-Surfac	æ Elev.		_feet	Surveyed Estimated	Datum
<b>Drilling Fluid</b>	Used	h/i	Ter		Drilling Method <u>UT: VP9-W454</u>
Drilling Contractor		Geo	Laic		Driller Mike Helper Mike
Prepared By		5	Thering.	and	Hammer Hammer Weight <u>40</u> Drop ins.
Sample/Core (feet below lar	Depth nd surface)	Core Recovery	Time/Hydraulic Pressure or Blows per 6		
From	То	(feet)	Inches		C'and aning lass
<u> </u>	22			Sand - land aley, 7	Fing-med granter, 1200
10				saturated	
12	24			sanda silt grey -f	ine saturated
24	33			silty chy - grey WE	t. low-med plastic
32	355			till tightly maken	I sond a captles porty
				SATEN	
756	4B			larce boulder very	dense material
2.2	10			The card and the	
710		ia		till and tightly	when wet
74	46 PCD-C	$\frac{1/8}{4}$		H N I F	and the
	FID-1	r. 5 6		SIT, GM TOCKT	
	FLYK	/.s			
51				Ledrock	
	······································				
			<u> </u>		
					·

### ARCADIS GERAGHTY & MILLER

### Well Construction Log

(Uncolsolidated)

		RI	nnagan nn	2 TR7-4
本 ft y LAND SURFACE		Project 7 ///	101111 au	D Well <u>L/K</u> /
44		Town/City <u>Be</u>	ratoro	44
			galesex	State ////
dnillet hole	nch olameter	Permit No.	tion and Datum	
		Lano-Sunace Eleva	luon and Datum:	<b>—</b> .
			feet	
	ich diameter,	' Installation Date(s)	5/11/00	
scheq	ule 40	Drilling Method	drive del	inch
Backfill	,	Durand Medica	CAT WCGW	<u> 4577</u>
Grout Part	land	Drilling Contractor	Geologia	2
		Drilling Fluid	vater	
32 "				
		Development Techni	ique(s) and Date(s)	
Bentonite	slurry			
<u>34</u> n A	pellets			
	-			
20		Fluid Loss Dunng Dn	illing	gallons
<u> </u>   <u>-&gt;b</u> <sup>n</sup>		Water Removed Duri	ing Development	gallons
		Static Depth to Water	r	feet below M.P.
Well Screen.	h dha an dana		· · · · ·	
$\frac{1}{PVC}$	<u>), //3</u> slot	Pumping Depth to Wa	ater	
	•	Pumping Duration	hours	
		Yield	gpm	Date
		Specific Capacity	gpm/fl	t
	2050 ·	Moll Purpose		
	9434	wei ruipose		
	-			
目 <u>·31</u> fr	-	Remarks		
<u>51.2</u> #		-		
	-			
Measuring Point is Top of Well Casing	-			
Unless Otherwise Note	rd			
<ul> <li>Depth Below Land St</li> </ul>	urtace F	Prepared by		
Unless Otherwise Note  Depth Below Land S	ed urface -	Prepared by		

## Well Construction Log

4

(Uncolsolidated)

.

	Project RN0099010012 Well IRZ-5
	Town/City Bedford
ИИ	CountyState <u>µA</u>
dirilled hole inch diameter	Permit No
	Land-Surface Elevation and Datum:
	feet 📈 Surveyed
Well casing,	Estimated
PVC	Installation Date(s) <u>5/4/() - 5/8/() - 1</u>
	Drilling Method <u>dy IVE 9 Wash</u>
Grout	Drilling Contractor
N N	Drilling Fluid
<u>34</u> nº	
	Development Technique(s) and Date(s)
	Surger PUMP - Check Value
	5/8/0)
20	Fluid Loss Dunng Dnlling
<u>- 50</u> f*	Water Removed During Development 300 gallons
	Static Depth to Waterfeet below M.P.
Well Screen.	Pumping Depth to Water
<u><u><u><u></u></u><u><u></u><u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u></u></u></u>	Pumping Duration 2.5 hours
	Yield ~ Opris Date
	Specific Capacity com/ft
	Well Purpose
<u>5.3</u> n-	Remarks
<u></u>	
Measuring Point is Top of Well Casing	
Uniess Otherwise Noted.   • Depth Below Land Surface	

5 Darrigrang

Wellcstr.xls 3/16/98

Prepared by

i

@004

r

02/1	5/2002 15:40 F	AX 978 937 7555	ARCADIS GRM	INC.		@ 004
	• •	4444 - 4544 - 444 4444 - 444 444		•	• •	
• •	DAILY LO	G				
		_	· ·	_		
	Well(s)	_ Project/No	(37992) (3	G12 M+	<u>703</u>	'age
	Site Location_	Hanso	om AFR			
F	Prepared By_	5 Parr	igrand_			
	Date/Time		Descript	ion of Activities	-	-
	5/8/01			t		
م المرد المراجع المراجع محمد المراجع الم	800	dillers 1	posite o	extarmit	na finis	hing !!
		work an	well 1	Calibrate	al Hariba	11-22
	100	benn Da	in Dina ar	TPZ-1	ni- very	,
		by bhi	flow the	such cal	i continu	ally
	. •	air bet	$\beta$	Vicen		
	1110	turned	flort	to boo	2 The late	ori
		malion			- mar	
	1000	Carton	a circan	0.1011		
.	1100	GFC Con		g with the		
• .[		TD7 P	a a z i	gin par		<u> </u>
t t	1110	17/5 6	TDN'T	m.	111	
F		<u>sample -</u>	<u>LAZ-in</u>	There ?	SMEDIV	
┣		sampling	diff ) C.H			
ŀ	- 1330	(pent nation	C STOPPE	a premp	ving	
• –		<u>== 300qa'l</u>	senautid		<u>·</u>	
:  -		· · ·				
Ļ						
		•		•		
		-		•	· .	

### ARCADIS GERAGHTY & MILLER

## Well Construction Log (Unconsolidated)

<b>不</b> ft	Р
	т
ИИ .	с
inch diameter	P
	L
ИК	
Well casing,	
inch diameter,	In
M _ Sth 40p/C	Di
Backfill	
Grout Krtland	Di
ft*Choke-Sand	Dr
7/2 10/20 27.8	
	 De
Bentonite Islurry	
ft* pellets	
the Choke Sand	
111 74	
ft* Top of Screen	Flu
	Wa
	Sta
Well Screen.	Pur
FineCRC . C.Ol slot	Pur
	Y L
	Tie
Gravel Pack	Spe
Sand Pack	
Formation Collaspse	Wel
49	
ft* Screen	Rem
L] _ <u><l <="" u="" }="">#*</l></u>	
Top of Well Casing	

x009100,003 roject R Well 2 `own/City 5 Mid ounty State ermit No. and-Surface Elevation and Datum:

	feet	Surveyed
	1.1	Estimated
Installation Date(s)	1.600	-
Drilling Method	niveth	24 Casing
	~ 1	
Drilling Contractor	Geolo	GIC
Drilling Fluid	Liter )	

Development	Technique(s)	and Date(	s)
-------------	--------------	-----------	----

Fluid Loss During Drilling		gallons
Water Removed During De	velopment _	gallons
Static Depth to Water		feet below M.P.
Pumping Depth to Water		_feet below M.P.
Pumping Duration	hours	
Yield	gpm	Date
Specific Capacity	gpm/ft	

ll Purpose

Grat 4 arks

Unless Otherwise Noted.

\* Depth Below Land Surface

Prepared by	by	ed	repar	
-------------	----	----	-------	--

J. Gorotrel

MS

East	Coast D	rillin	na. I	Inc.				8a .	<sup>k</sup> .g				
P. O. I	BOX 961 - 1	WALLI	- GFO	RD. C	ONN	06402		•		SmEE *		· e	-
•0	Haley & Ald	rich 1	ine.		1	ADDRESS	Cambr	idge, Y	ass. fig	DATE			
PROJECT NA	Hanscon I	Field				LOCATION	Bedro	ck, Mas	5.	HOLE NO	RAP!	<u></u>	
REPORT SEN	TTO_CILEN	et at s	ite		-	r	ROJ. NO	58	33	LINE & STA			
SAMPLES SE	ENT TO					o	UR JOB NC	)	231	OFFSET			
GROU	ND WATER OBS	ERVATIO	45	Ι	and the second secon	CASIN	G 8.44			ľ.			
N_4.7	atter	_0 H~~	3	T					were nort.	SURFACE ELEV	23.	<u>6</u>	**************************************
				Simi	n	гэч 4"			وي دو الشعر ويتيان الشع	DATE COMPL 11	/18/8	5	
At	- oft <b>er</b>	. House	-	Home	er Wil				A.T.	BORING FOREMAN D.	Qua	gliaro	1: 1:
				Horn	wr Fall				(per )	SOLS ENDE	<b>1</b> 100-9	5	
LOCATION	OF BORING			S	ce Pla	a							
WELL	Sample	T	-	rt pt. (		Moisture	T.	T					
INSTALLATION	Depths	~	m :	Sample	r •	Density	Sireta	Rug	terta include ce	LATION Hecgredation, Type of	ł	SAMPL	£
	Prom - To	Freed	0-51	6-12	12-18	Conerst	Fine	SPI Not	ntc. Rock-caller, N. Drilling time, s	type, cendinen, hard-		Ta	
		Ì					0.5	Too			100		Hec
							1	Gra	-Brown fine	Send. *	+		<u> </u>
		<b>├</b>							- Orthandh			<u> </u>	
	-	<b>├</b> · -	·+						oocoarres				
			· †-	CT BRUNE									
		<u> </u>						1				<u> </u>	
			·										
		f f.	•										
			þ.				1				┣ <b>─</b> ─┥		
		L.,	╾╺╾┝╾				13.5	' <del> </del>					
								Georgy	ally line :				
		<b>.</b> .											
		- +	•										
	ĺ		<u> </u>				[						
		··	·				]	Gray	Silt, little	trace coarse-fine			
								Sand.	•				
	1							1					
	!		<u> </u>								+		
			╴┯┠┯					Gray :	Silt, little	coarse-fine Sandt	+		
								1		[			
							29 01			-			-
					-	+	49.0	Grav	Silty men	e -	-+-	+	
						ł							
										Ĺ			
								[		ŀ			
								Brong-	Gray Silty or	ere-fine Send.			
				╺┈┽╴	+	5			- Till -	F			
						1				+			·····
						ł		* See	note on last	pege of this log	+		
GROUNO SUR	FACE TOUS OF			-al-									
Somple Type			Pr	oportio	~ Use		-JUNIC: HODW	1 meni 4 La 30" h			e		
UP Undeturbed	WEWEEDAD		tro	ce (	010109	6 Ca		Density	Comme Con	East East	rth Bo	rng 4	<u>.0'</u>
TPa Test Pit As	Auger Vr Vmee	Test	1 10	18 1 718 7	U 10 209 D to 3 4 9		10-30 Ma	LOOSE d. Danes	0-4 S	Heft 30 + Hard Ro /3htt	ce Coi	ring	
UT + Undisturbed	Thowall	-	on	a 3	5.0504		10-50 X0 + Ven	Dense Dense	0-15 15-10 V	Shift HOL	F R/	P1-6T	

•

							5	Rua t.	1 246	-	•	F
PROJECT N	A 1.45					ADDRESS	5		DATI		-	-
REPORT SE	TTO		•••			LOCATIO	N		HOL	END. RAPI		
SAMPLES S	ENT TO		*• ••••			1	ROJ.NO		LINE	& STA		******
080						/ 0	AN BOL NU	)	orr	ET		
	HE HAILE OR	S. r.yet	X:45	1.		CASIN	13 P.m.					
	- ofter		<b>1.11</b>	1				FLER OPPE MAR	SURFACE EL	EV		
				S.m.			-		DATE STARTE			
N	- after	He	njur mj	Itane	ther Write				DATE COMPL.			
				Hom				BIT		MAN		
LOCATION	OF BORING	•			Contraction of the local division of the loc				SOLS DIER.			-
WELL	Sometie	Time	-									
HETALLATION	Depths	M	600	Servers	6 7	Maisters	Strate	SOL RENTER	ATION			
	From- To	horas	- rom		To	Density	Change	Resperta include pe	tre production, 1	ype of	SAME	. r
		1	2-6	6-12	12-14	Consist.	Eme	Reas, Drilling time, a	Type, condition,	NETO-		<u>-</u>
3		ł				·		Brown-Gray Silter			o Pen	
16 E		t					1	duy billy	concae-rine :	send.		
≥≊⊑		1-1					1				+	
												$\bot$
<u> </u>		L ]			-					<u>⊢</u>	+	+
S –											+	+
F		h	·								1	┢
		<u> </u> -	~ - +-									T
		+						- Till -				
		· - †-	[			4	46.0'					
<u> </u>			t					Bottom of boring	t 46 01		+	
<u> </u>		[										-
		ļ_	,			1						-
		<b> </b>	•					Note:			I	
								No see les tales				
						1		RAPI-6R for detail	ad soils info			
		·	···••			1		atica.		~~~		
			1	fra				Thetalled 2"		}		
		[		1				44.7!	LCOT WELL AL			
	······································							15.0' SCIMED				
						1		30.2' riser				
					-			1501bs. Send				
			-+	~		1		101bs. Bantonita	pellets	_ <b>├</b> ──┥		
		1_	1					1 screw plug				-
		!						t prochective sthe	l sleve with			-
		_ <u> </u>				.			lock.			
			+		~	1						
		-1										
		<u> </u>	+				1					
			T									
							1					
		-	1									
		··	·		_							
UNO SURFAC	Em											
е Туре			Fran	45500 67 h ana	110-0	"CASU	10: THE	N				
C=Cored W	Washed		Inoce	1000 S	U340	Com M	08 WI. 1 3	O" fall on 2"0.0. Somele	, 1	C:#1		
Osturbed Pist	00		ist Tim	1014				TY Contains Conside	ency (		MARY	
47 D.1 A.A.			•			1 0-1	0 [.684			COLUMN BOL	·····	-

East Coast Drilling	g, Inc.		8	Bit # Fig				
P. O. BOX 961 - WALLING	FORD, CONN	1. 06492	•	lia e	SHEET		_ )F _	
TO Haley & Aldrich,	Inc.	ADDRESS	Cambrid	ge, Mass. Hg	DATE 2/14/	86		
REPORT SENT TO Client		LOCATION	Bedford	. Mass.	HOLE NOE	AP1-	5R	
SAMPLES SENT TOTaken at si	te	M	10J. NO	8 <u>33</u> 85-221	UNE & STA			
				0721	[urrse1			
GROUPED WATER ORSERVA: KHE	5	CASING	SANPI	LER CORE BAR.	SURFACE ELEV.	23.	6	
after Hours	Туре	HW	<u>5-5</u>		DATE STARTED 11/	13/8	5	
Al Hours	Size I.D.	4"	$-\frac{13}{13}$	/8"	DATE COMPL. 11/1	3/85		
	HUMENER WY.		- 1401	DS. NT	NEPECTOR J.S. C	. no	55	
LOCATION OF BODING	See Plan				SOLS ENDR.			
		1	7	Y				
WELL Depths of	(m) Sanging	Canaita	Strate	SOIL IDENTIFIC	ATION			
From - To Compose -		Gr	Change	enil Mc. Rass-calar,	ype, condition, hard-	┣	Same (	±
0.0'-2.0' D	4 7 10			There is a series of the serie	tens and dec.	No	Pen	Rec
	13	M/Danae	<u> </u>	Topsoil. Brown fire Sand		μ	2.0'	2.0
		]	1		•	┣	<u> </u>	
		}						
5.0'-7.0' D 7	99	Moist	·	Comer Fine / Sand		_		
	14	H/Dease		GERY LIFF SHILL.		2	2.0	1.5
<u>10.0'-12.0'</u> <u>0</u> <u>7</u>	7 8	Hoist		Brown fine Sead,	trace Silt.	3	2.01	1.5
		n/Dense						
			13.5'					
	12	Moist v/Sriff		Gray Silt, trace	little fine Sand.	4	2.0'	2.0
		a, seri				_		
			1		}			
20.0'-22.0'	5 8-	Moist		Gray Silt trans	little mene			
	10	75tiff		Sand.		-54	2.0'	1.4
	1				ļ			
25.0'-27.0' D_6_	16 7	Hoist		Gerry Silt, Little	course-fine Send	6	2.01	2.0
		SUIT			Ľ			
					F			
					+	-+	+	
		bist /Terres	0	key Silty concern	fine Send.	7	2.0'	1.0
					ļ.			
			34.01		E	$\pm$		
35.0'-35.9i b 59	1027.2	triar 1	2	my coarse fine sa	adySilt to sile			
		Danas	þo	erse-fine Seod, co	arso-fine Gravel	8-44	<u>,,,</u> ,+	<u>.s</u> .
	++	.		obble and Poulders av Silty medium-fi	. Sand trace	i	3.5'	1.5
40.0'-42.0' p 37	j 42 55 5	Dense	Fi	ttle Gravel, coars	e Send.	-+-	0.1	1 6
Sonole Type		H CA	SING: T	ENCored to 7	5.01	ž		<u></u>
D=Dry C=Cored W=Woshed	mece Clains	L   Com	HODWI.s	30" full on 2"0.0. Ser	etter le	SU	MARY	
UP: Undisturbed Piston TP: Test Pit As Austral United Test	10 to 209	6 9	0.10	0-4 8	th 30 + Herd Roc	m BCI th Cor	ing 28	5.
UT : Undisturbed Thinwall	Prime 2010359		0-50 De	1-8 M/	Shift Sor	noies	_10_	
		N X	y + yery (	15-30 V-	Shill HOLL	N	C RAPI	-61

P. O.	BOX 961 -	WALI	INGEC	DRD, (	ONN	. 04407		-	-	_ [	SHEFT			-
TO						IADORE		Bit # .	Fig				JF	
PROVECT N	LAME			-		LOCATION	·			[	HOLENN	RAP:	-6R	<b></b> .
SAMPLE C	NT TO						ROJ. NO			I	LINE & STA			
						la	UR JOB N	0		[c	OFFSET_			
GR	AND WATER OBS	ERATX	345	T		C				I				
N	ofter		-			CR311	G <b>SA</b>	MPLER	COME BAR.	SURFACE	ELEV			
				Type				_		DATE ST	WITED			-
N	- ofter			See 1.	D.					DATE CO	M			
				Hanne	ar art. ar Failt				. #IT	NOPEL TO	"UREMAN _ M			
LOCATIO	N OF BORING									SOLS EN				
	S. DOMING	T 1				r								-
WELL	Depths	~	(10) (1) (1)	st per 5 Semain		Maishere	Strate		SOL DENTIN	CATION		T		
	From- To		From	I	Q	Density	Charge		mertes involvede of	rior, prodeti	en, Type of		SAMP	E
		┝──┥	0-61	6-12	12-18	Congist,	Eler	-	te, Dritting time, 1	and e	<b>nan<sub>a</sub> hard-</b> IC.	No	P	Т
		┝──┫					1		Free 5 - 5 - 12 - 4 - 4 - 4 - 4 - 4 - 6 - 6 - 6 - 6 - 6					╞
		<u>├</u> ──┤										$\vdash$		┞
							1							H
		ĻŢ					I							F
	42.0'-45.8'	-D .	<u>56  </u> 1	7/3"		Moist	1	Brow	n Gene Sile			<u>├</u> ──		
1 1 1 1						/Degae	[	trac	e-little Grav	el. trace	ue silicot, Cl <i>ure</i>	10	-B'-	hi
夏日二							Į	min	•		Camy,			-
I III E						ct	49.0	1 70	D of Beimck					-
	50.0'-55.0'					6		1						
3.						5		Gen	mite.			<u>a</u> 1	<u>.0'</u>	4.
3 2 P				···.	-	5		1				-**		53
						4					l			
∃ a  -	<u></u>	<u>C</u>	·			s		-		**			_	_
3						4		Bro	ma Grani ta 11	th march	us com			
					(	6		20000	, scan filled	l with Sil	t.	~+	2.0.	د
	59.8'-63.8'	<u>c</u>	1			5		м						-
		-+-								**	ß	4 2	.0' [	
	53.8'-68.8'	~-						1			ŀ			-
EL		× †	·· †					}	17		le le		012	_
												σ ·		Ť
			_	1										
E h	8.8-73.8	e-+-				ł					F			
e l			-+			1				**	a	5 5	01	17
				1							R	D-	h	$\hat{\boldsymbol{v}}$
e: L						1								_
	3.8'-75 01 0	-+-			-1								_	
		-+-	-1			l	1	11		**	10		211	
ž į			1			.	<b>x</b> a.				E			
-Ig			1	T			·3.0 <sup>.</sup>	her-						
· ·/								JUCION 1. Inet	or porring 75	.0'	_, ]_	+	1	
F		-+	~h	-+	_		ľ	71.7	natoria EVG M	WILLIOT V		+	-+-	
				+				20.2	ACCARD			1		
POUND SURP	ACE TO		- 	USED				53.6	RIPEL			1		
V CICANA	Washington	1	Prop	ortions	Used	(as		1.30°4	m #"00 -					
ngelinded t	w = vrashed		frace	01	10%	Cohe		Jeneity	Cohrente Care	NOT CY	5~~~	SUMM	ARY	
est Pit AsA	uger Villang Tay	. 1	9775@	10 1	20%		-10 U	0000	0-4 50	ft 30+	Herd Rock	Corm	·	
100.00		1		- 4UN	マココア	1 10				n//	ا فسب			



P. O. BOX 961 - WAILINGFORD, CONN. 06492         Is a		Eas	t Coast I	Drilli	nα.	Inc				Bit ø _	Hg				
NO         NO<		P. O.	BOX 961 -	WALL	er NAGE(	DPD,	CONN	. 06492	1	<b>B</b>		SHEET			
PROJET I MAE	١	ro			<del>.</del>			ADDRESS	·	1947 <b>#</b>	<sup>1</sup> 'g			-	-
CAUN SENT TO         Use 5 stat	F	ROJECT	NAME					LOCATIO	۷			HOLE NC R	AP1-1	0 R	
CRCLAD WATER DESERVIENCY         CASING SAMPLER         CORE M.         SAMPLER         CORE M.           A         ofte         Norr         Tage         OFFSET         DATE CORE M.         SAMPLER         CORE M.         SAMPLER         CORE M.         SAMPLER         DATE CORE M.         SAMPLER         SAMPLER <td></td> <td>EPONT S</td> <td>ENT TO</td> <td></td> <td></td> <td></td> <td></td> <td>  1</td> <td>ROJ. NO.</td> <td></td> <td></td> <td>LINE &amp; STA</td> <td></td> <td></td> <td></td>		EPONT S	ENT TO					1	ROJ. NO.			LINE & STA			
CASING SAMPLER CONTINUES         Type         CASING SAMPLER CONTINUES         Marries That TD           M		MPLES	SENT TO					(0	ur job n	0		OFFSET			·
A		GR	OUND WATER OF	SERVATI	(145 C	1.	-	CARIA				· · · · · · · · · · · · · · · · · · ·		-	
Al	-			Han				UR3(I		PLER	CORE MAR.	SUMPACE ELEV.			
Au     Other						TH	28			-		DATE STARTED			
Image: Full         Bit         Page: Full           LOCATION OF BORNOS:         Terming Full         Social State         Social State           E         Duping         Terming Full         Display         Social State         Social State           Figl:         Ford         Contract Full         Social State         Social State         Social State           E         Display         Ford         Contract Full         Social State         Social State           Image: Ford         Ford         Contract Full         Social State         Social State         Social State           Image: Ford         Contract Full         Contract Full         Social State         Social State         Social State           Image: Ford         Contract State         Social State         Social State         Social State           Image: Ford         Contract State         Social State         Social State         Social State           Image: Ford         Contract State         Social State         Social State         Social State           Image: Ford         Contract State         Social State         Social State         Social State           Image: Ford         Contract State         Contract State         Social State         Social State <td< td=""><td>A</td><td></td><td> other</td><td>Ha</td><td></td><td>Sun</td><td>e I D.</td><td></td><td></td><td></td><td></td><td>DATE COMPL.</td><td>Andrew Street and</td><td></td><td></td></td<>	A		other	Ha		Sun	e I D.					DATE COMPL.	Andrew Street and		
LOCATION OF BORING:         Social press         Social						150-	inni Eal		· · · · · · · · · · · · · · · · · · ·		<b>BIT</b>	NEPECTOR			
Control OF BORNACE         The Theorem prof.         Notating Deeping True         The Theorem prof.         Notating Deeping True         Sort of the theorem prof.         Sort of the theorem prof.         Sort of theorem prof. <t< td=""><td></td><td></td><td></td><td></td><td>*****</td><td>1.12</td><td></td><td></td><td></td><td></td><td></td><td>SOLS DIER.</td><td></td><td></td><td></td></t<>					*****	1.12						SOLS DIER.			
E       Birms       Druc pri 6'       Malabari pri from - To       Malabari pri from - To       Strate pri 6'       Strate pri 6'         1011       From - To       Common To       Common To       Strate pri 6'       Strate pri 6'         1011       From - To       Common To       Common To       Strate pri 6'       Strate pri 6'         1011       From - To       Common To       Common To       Strate pri 6'       Strate pri 6'         1011       From - To       Common To       Common To       Strate pri 6'       Strate pri 6'         1011       From - To       Common To       Common To       Strate pri 6'       Strate pri 6'         1011       From - To       From - To       From - To       Strate pri 6'       Strate pri 6'         1011       From - To       From - To       From - To       Strate pri 6'       Strate pri 6'         1011       From - To       From - To       From - To       Strate pri 6'       Strate pri 6'         1011       From - To       From - To       From - To       Strate pri 6'       Strate pri 6'         1011       From - To       From - To       From - To       Strate pri 6'       Strate pri 6'         1011       From - To       From - To       Strate	=	LUCAIL	IN OF BORING	<u>G</u>											
Bits         Dealty         Dealty <thdealty< th=""> <thdealty< th=""> <thdealty< th=""></thdealty<></thdealty<></thdealty<>	Ξ	Canna	Somele	Tyme	0	care be	< 6 <b>"</b>	Hoishere		1	SOL DENTE	CATION	T		
Image: Transmit in the second seco	1	per	Erom To	_ ~	Errom	्यसल्ह	Te	Density	Charmen	Re	norta incluse p	eter, predetion, Type o	1	SAMP	LE
Chow ter presere tasts con- dated in Baircot. Construction Construct		1001			Tõ ā	6-	21 12-14	Consist	Elm	· 1986	n, Drilling Hung 1	, Type, condition, light-	-		
CASMG: Trep: Construct Strates conduction Conduction Conduction Construct Strates conduction Co					1	]	-j-								Rec
Control Surface Tro UseD Control Trep:	1				ļ			]	1				<b>—</b>	╂────	
CROMO SURFACE TO CASHIG: THEN CONTROL OF THE CONTR		<b> </b>			ļ		-ļ	]						<u>+</u>	<del> </del>
dacted in betrock.     dacted in betrock	1		<u> </u>					1		2.	Two water pr	wasure tests cop-		<u>+</u>	╉────
			1					4	1.	1	ducted in Be	drock.			
Samola Type     Proprince Used       Change Proprince     Proprince Used <tr< td=""><td></td><td></td><td></td><td>1</td><td></td><td></td><td></td><td>{</td><td></td><td>3.</td><td>Genutad bore</td><td>iole above complet</td><td></td><td></td><td></td></tr<>				1				{		3.	Genutad bore	iole above complet			
GROUND SURFACE TO         USCD         CASHIG:: THEN           GROUND SURFACE TO         USCD         CASHIG:: THEN           Diff         Consumery         Consumery           Diff	ł	ļ					1	1			well install	ation.	<b>T</b>		
GROUND SURFACE TO         USED         CASMG: THEN           GROUND SURFACE TO         USED         CASMG: THEN           Somble Type         Proprinten Used         THEN           D'Dry Clored Wilvested         Proprinten Used         THEN           U'r Underlived Patient         Proprinten Used         Chasme Gradinary (Chasme Grad								]	1						<b> </b>
GROUND SURFACE TO         USED         "CASING:         THEN           D'Dry Criticad Pation         Use Barger         Sample Trype           D'Dry Criticad Pation         Use Barger         Use Barger         Sample Trype           D'Dry Criticad Pation         Use Barger         Use Barger         Sample Trype           D'Dry Criticad Pation         Use Barger         Use Barger         Use Barger         Use Barger <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>+</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>┢───</td>							+								┢───
GROUND SUMACE TO         USCD         "CASING: THEN           D'Dry Clored Winkosted         Proprintes Used         MOD With 30" and 20"				-+					1						
GROWD SURACE TO         USED         CASING: THEN           Songle Type         Proteines Used         THEN         Construction of 20.0 Service           D'Dry ClGred Witword         Proteines Used         The Unit States         Construction of 20.0 Service           D'Dry ClGred Witword         Proteines Used         The Unit States         States         States           D'Dry ClGred Witword         Proteines Used         The Unit States         Construct Construct         States           D'Intractional Witword         Proteines Used         The Unit States         Construct Construct         States           D'Intractional Witword         States         Construct Construct         States         States           UT : Undesturbed Throughi         Cross States         States         States         States           States         Cross States         States         Word States         States						**			1						
GROUND SURFACE TO GROUND SURFACE TO D'DOT CLÉGRIE MUNGENE UP: Undersurbed Patient UT: Undersurbed Throwdii Construing Construction Construing Construction Construction Construction Construing Construction Construing Construction Construing Construction Construing Construction Construing Construction Construing Construction Construing Construction Construing Construction Construing Construction Cons							-1		1						
GROUND SURFACE TO			~		]					1					<u> </u>
GROUND SURFACE TTO GROUND SURFACE TTO Somple Type D'Dy Cl Cored Wavegard UP: Underlarged Properties UP: Under															
GROUND SURVACE TO GROUND SURVACE TO Songle Type D: Dry C:Cond Wiwoshed UP: Undeflurged Pation UT: Undeflurged Thrwell UT: Undeflurged Thrwell	}			+			· }		1						<b> </b>
GROUND SURFACE TO GROUND SURFACE TO Somole Type D' Undelsturbed Pation TP: Trent In Adager Varyona Top: UT: Undelsturbed Thruedii UT: Undelsturbed Thr				╉╂					1						
GROUND SURFACE TO	Ľ			1	†		- <u> </u>		1						
GROUND SURFACE TO	Ļ														
GROUND SURFACE TO	ŀ			<b>_</b>	[·				1				┝──┥		
GROUND SURFACE TO	ŀ			┫			<u>+</u>		1				┝╼╼╉		
CASING: THEN     CASING: THEN     CASING: THEN     CASING: THEN     CASING: THEN     CONSIDER TYPE     CONSIDER TYP	E			╉┈╼╋			+		1						
GROUND SURFACE TO	E					7. I	1		1						
GROUND SURFACE TO	Ĺ								1				┝╾╌╄		
GROUND SURFACE TO     USED     CASING: THEN       Sample Type     Proportions Used     MOB WILLS OF Net on 2"0.0 Sensitive       D: Dry C:Cored Wewsined     Proportions Used     MOB WILLS OF Net on 2"0.0 Sensitive       D: Dry C:Cored Wewsined     Proportions Used     MOB WILLS OF Net on 2"0.0 Sensitive       D: Dry C:Cored Wewsined     Proportions Used     MOB WILLS OF Net on 2"0.0 Sensitive       D: Dry C:Cored Wewsined     Proportions Used     MOB WILLS OF Net on 2"0.0 Sensitive       D: Dry C:Cored Wewsined     Proportions Used     MOB WILLS OF Net on 2"0.0 Sensitive       D: Dry C:Cored Wewsined     Proportions Used     MOB WILLS SO" Net on 2"0.0 Sensitive       D: Dry C:Cored Wewsined     Proportions Used     MOB WILLS SO" Net on 2"0.0 Sensitive       UP: Undesturbed Pasten     Proportions Used     MOB WILLS SO" Net on 2"0.0 Sensitive       UT: Undesturbed Thinwall     Proportions Used     Net on 2"0.0 Sensitive       UT: Undesturbed Thinwall     Proportions Used     NO RAP1-t	H				$-\mathbf{f}$								┝──╉		
GROUND SURFACE TD     USED     "CASHIG: THEN       GROUND SURFACE TD     USED     "CASHIG: THEN       GROUND SURFACE TD     USED     "CASHIG: THEN       Sample Type     Proportions Used     MOB Wits 30" Hell on 2".0.0 Serveer       D: Dry C:Cored Wewgshed     Proportions Used     MOB Wits 30" Hell on 2".0.0 Serveer       UP: Underlarded Pasion     Proportions Used     MOB Wits 30" Hell on 2".0.0 Serveer       UP: Underlarded Pasion     Constructed Pasion     SUMMARY       UT: Undisturbed Throwolit     ord 33:030%     SO + Very Denne     8:15       SO + Very Denne     10:-SO Versitif     MOLE NO RAP1-t	F	+		<b>}</b> → <u>+</u>			<u> </u>								
GROUND SURFACE TD	F			t+						1					
GROUND SURFACE TD     USED     "CASHIG: THEN       Somple Type     Proportions Used     MOB W1: 3 50" New on 2"0.0 Sensitier       D: Dry C:Cored Witwoshed     Proportions Used     MOB W1: 3 50" New on 2"0.0 Sensitier       UP: Undisturbed Piston     Proportions Used     MOB W1: 3 50" New on 2"0.0 Sensitier       UP: Undisturbed Piston     Vitre 10 to 20%     0-to Loses       UT: Undisturbed Thinwall     ord 35:050%     30 + Very Dense	L			<u>t</u> †											
GROUND SURFACE TO     USED     "CASHING: THEN       Sample Type     Proportions Used     HOB W2 3 30" fail on 2" 0.0 Service       D: Dry C:Cored Wawashed     Proportions Used     HOB W2 3 30" fail on 2" 0.0 Service       UP: Undesturbed Piston     Proportions Used     HOB W2 3 30" fail on 2" 0.0 Service       TP: Test Pit AsAuger V:Voris Trat:     arms 0'is K0%     0-i0 Losse       UT: Undisturbed Thinwalt     org 33:0:50%     30 + Very Dense					T					1					
GROUND SURFACE TO     ISED     "CASING: THEN       Sample Type     Proportions Used     MOB Wt.s 30" hall on 2"0.0 Sensity       D: Dry C:Cored Wewoshed     Proportions Used     MOB Wt.s 30" hall on 2"0.0 Sensity       D: Dry C:Cored Wewoshed     Proportions Used     MOB Wt.s 30" hall on 2"0.0 Sensity       D: Dry C:Cored Wewoshed     Proportions Used     MOB Wt.s 30" hall on 2"0.0 Sensity       D: Dry C:Cored Wewoshed     Proportions Used     MOB Wt.s 30" hall on 2"0.0 Sensity       D: Dry C:Cored Wewoshed     Proportions Used     MOB Wt.s 30" hall on 2"0.0 Sensity       D: Dry C:Cored Wewoshed     Proportions Used     MOB Wt.s 30" hall on 2"0.0 Sensity       D: Dry C:Cored Wewoshed     Proportions Used     MOB Wt.s 30" hall on 2"0.0 Sensity       D: Dry C:Cored Wewoshed     Proportions Used     MOB Wt.s 30" hall on 2"0.0 Sensity       D: Dross     D: Dross     D: Dross     D: Dross       UP : Undisturbed Piston     Sono 200 Sense     3:0 Sensity     D: Sensity       UT : Undisturbed Thinwall     Ord 33:0 50%     3:0 + Yery Dense     15:30 Y-Stiff     MOLE NO RAP1-t	F												+-	+	
GROUND SURFACE TD     USED     "CASING: THEN       Somple Type     Proportions Used     MOB W1: 30" hall on 2" 0.0 Sensity       D: Dry C:Cored Wewoshed     Proportions Used     MOB W1: 30" hall on 2" 0.0 Sensity       UP: Undesturbed Piston     hime 10 to 20%     0-10     Losse       U7: Undesturbed Thinwall     ord 33:050%     30 + very Dense     30 + stiff	-									1					
GROUND SURFACE TD     USED     "CASING: THEN       Sample Type     Proportions Used     MOB W1: 30" hell on 2"0.0 Sension       D: Dry C:Cored Wewoshed     Proportions Used     MOD W1: 30" hell on 2"0.0 Sension       UP: Undesturbed Piston     http://document.com/document     Constructed Dension       TP: Test Pit A: Auger V:Vore Trp:     proce 201033%     30-50 Dense       UT: Undisturbed Thinwall     ord 33:050%     30 + Very Dense	F				+-										
GROUND SURFACE TD     USED     "CASING: THEN       Sample Type     Proportions Used     HOB W1: 3 30" hall on 2"0.0 Sampler       D: Dry C:Cored Wewosned     Proportions Used     HOB W1: 3 30" hall on 2"0.0 Sampler       UP: Undesturbed Piston     hime 10 to 20%     0-10 Losse     0-4 Sett       TP: Test Pit A:Auger V:Vore Tre:     some 20 to 35%     30-50 Dense     4-8 M/Stiff       UT: Undisturbed Thinwall     ord 33:0 50%     30 + Very Dense     15-30 V-51ff										I		ĺ			
GROUND SURFACE TD     ISED     *CASING: THEN       Sample Type     Proportions Used     MOB W1: 30" hall on 2"0.0 Sension       D: Dry C:Cored Wewoshed     Proportions Used     MOB W1: 30" hall on 2"0.0 Sension       UP: Undesturbed Piston     html     10 to 20%     0-10     Losse       UP: Undesturbed Piston     html     10 to 20%     0-10     Losse     0-4     Sett       UT: Undesturbed Thinwall     ond     33:050%     30+50     Dense     4-8     M/Bitt       UT: Undisturbed Thinwall     ond     33:050%     30+4     90+4     Stift     MOLE NO RAP1-t	F				=1			Í							
GROUND SUITACE TD       USED CASING: THEN         Sample Type       Proportions Used       HOB W1: 30" hell on 2"0.0 Sension       Summary         D: Dry C:Cored Wewashed       Proportions Used       HOB W1: 30" hell on 2"0.0 Sension       Summary         UP: Undesturbed Piston       Hitle i0 to 20%       0-10       Losse       0-4       Sett 30 + Herd         TP: Test Pit A: Auger V: Vons Trp:       pone 20to 35%       30-50       Dense       8-15       Striff         UT: Undisturbed Thinwall       ond 33:050%       50 + Very Dense       15-30 V-Striff       MOLE NO RAP1-t	<u> </u>					[					-	ł	+-		
D::     D:: <thd::< th=""> <thd::< th=""> <thd::< th="">     D::</thd::<></thd::<></thd::<>	ري سيري	ncuno Su	PERFACE TO		·	Ĩ		*C	ASING:	THEN .					
UP: Undisturbed Piston TP: Test Pit A: Auger V:Vone Test UT: Undisturbed Thinwall Dra 35:050% 0-10 Losse 0-30 Med. Dense 30-50 Dense 30-50 Dense 15-30 V-5111 MOLE NO RAP1-t	D= Dr	y C+Core	Wewgenad			rroport	NONE UN		HODW	12. : 30"h	Won 2"0.0 Se	noier I	SU	MMARY	٤
TP : Test Pit A: Auger V: Vone Test some 201035% 30-50 Danse 4-8 M/Bitt Somoles UT : Undisturbed Thinwall ond 33:050% 50 + Very Danse 15-30 V-Stift HOLE NO RAP1-t	UP:	Indisturber	Pision			·v··w •He	10 to 20		0-10	Lossa	0-4		rm Ba	- 6	
Ur + Undisturbed Thinwall and 35:050% 50 + Very Dense 15-30 V-Still HOLE NO RAPI-t	TPil	fest Pit A	Auger Vavor	a Toat		one	201035	¥	10-30 M	N. Danes	4-1 K	Anti S	moles	·····¥	
	0110	ndisturbe	d Thimeall			2	35:050	%	50 + Vw	y Dense	15-30 V	Smil HOL	EN	O RAF	21-1





DEPTH	CASING	SAMPLER	SAMFLE	SAMPLE	ELEV./		
(FT)	PER FT	PER 6 IN	REI OVFRY	(FT)	(FT)		VISUAL DESCRIPTION AND REMARKS
- 25 -	<u> </u>	17		12			FID (HDSP) = 0.6/3.6 ppm
		16	}	ra			
			1				-GLACIOFLUVIAL DEPOSITS-
			-		1		
			1			-	
30 -					95.8 30.0		Potton of Evaluation of 70.0.4
					30.0		Bottom of Exploration at SULU Ht.
							Located at site 1 recharge basin.
							Installed monitoring well of 28.0 ft.
							PID (HDSP): PID headspace reading collected
							from sample jars. Results presented as background reading.



Remarks: Plumbness test positive. Located at site 1 recharge basin.

WI06GOZ.FRP

PROJECT CLIENT CONTRAC	HANS U.S. FOR M &	COM A.F.R. ARMY COPP R ENVIROUM	LONG TER IS OF ENGI VENTAL DPT	M SAMPLIN NEERS LUING	IG PROGRAM	AM, BERFORD, MA FILE NO. 05833-065 SHEET NO. 1 of 3 LOCATION N 537,553
	ITEM		CV21AB	DRIVE SAUPLER	CORE BARREL	DRILLING EQUIPMENT & PROCEDURES
TYPE INSIDE I HAMMER I HAMMER I	DIAMETER VEIGHT FALL	(IN) (LR) (IN)	FH/HW 5.14.0 300 24	s 1 3/8 140 30	HQ 2.5 - -	RIG TYPEB-61 Mobile Trick rigELEVATION125.7RIT TYPERoller bitDATUMNGVDDRILL MUD NoneSTARTMay 16, 1996OTHERCathead/safetyFINISHMay 18, 1996hammer; PW to 14.0 ft.; HW toDRILLERP. Thornsbury52.5 ft.H & A REPS. Goldkamp
DEPTH (FT)	CASING BLOWS PER FT	SAME: ER BLOWS DER ( 14	SAMFLE NUMBER & REMOVERY	SAMPLE DEPTH (FT)	ELEV./ DEPTH (FT)	VISUAL DESCRIPTION AND REMARKS
		1 	\$1 *2" 51A 72" \$2 24"	0.0 1.0 2.0 2.0 4.0	124.7 1_0	<pre>Stiff black silty iOAH, trace gravel, sand, roots, asphalt specks -TOPSOIL- PID (HDSP) = 0.6/3.2 ppm Medium dense brown medium to fine SAND, little gravel, trace silt</pre>
E		- 21 -	₹ <u>3</u> 15"	-2.0	121.7 4.0	PID (HDSP) = 0.6/3.0 ppm D.0. except very dense PID (HDSP) = 0.6/3.0 ppm -TILL-
<b>,</b> -		3 10 12 12 10	\$7 <u>7</u> - 2# - 54 - 12**	5.5 6.0 6.0 7.0	120.2 5.5	Stiff black fine sondy LOAM, trace roots PID (HDSP) = 0.6/2.8 ppm -TOPSOIL- Medium dense gray silty fine SAND
10 -		1: 1: 1: 1:	54A 34 54B 411 55	7.0 7.5 7.5 8.0 8.0	117.7 8.0	-GLACIOFIDVIAL DEPOSITS- -GLACIOFIDVIAL DEPOSITS- Medium dense brown medium to fine SAND, little silt, trace coarse sand PID (HDSP) = 0.6/2.4 ppm
		14 1, 19 6 8	56 24" 3 <b>7</b> 24"	10.0 12.0 12.0 12.0 14.0		Ssand PID (HDSP) = 0.6/2.6 ppm S5: Very stiff gray clayey SILT PID (HDSP) = 0.6/2.4 ppm P.O.
15 -		9 14 3 5 4	ं है 12"	14.0 16.0		PID (HDSP) = 0.6/2.8 ppm -GLACIOLACUSTPINE DEPOSITS- Very stiff gray laminated SILT, little clay in seams PID (HDSP) = 0.6/3.0 ppm
		8 20 37 19	59 6"	16.0 18.0	109.2 16.5	S9: Very dense gray silty SAND, trace gravel PID (HDSP) = 0.6/2.6 ppm PID (HDSP) = 0.6/3.8 ppm
20		12 15 15	13"	20.0		PiD (HDSP) = 0.6/3.2 ppm
						-GLACIOFLUVIAL DEPOSITS-
25		<u>18</u> 12	<u>\$11</u> 12"	24.0 26.0		Dense rusty brown coarse to fine SAND, trace gravel
ATE	TIME	HATCH LEVE ELAPSED THE (HR)	EL DATA DEPT FOTTOM	H (FT) TO BOTTOM	UATER	SAMPLE     IDENTIFICATION     SUMMARY       0     IDENTIFICATION     OVERBURDEN (LIN FT)     54.0       1     IDENTIFICATION     IDENTIFICATION     IDENTIFICATION

с × х

.

.

۰.

. -

- -

~ --

------

- ----

----

------

A	HALEY & CA MASS	ALDRICH, MBRIDGE ACHUSETIS	INC.	-	TE	ΞS'	г вс	DR	ING	REPOR	रम	BORING NO. FILE NO. SHEET NO.	B240-MW 05833-065 2 of 3
DEPTH (FT)	CASING BLOWS PER FT	SAMPLER BLOVS PER 6 IN	SAMP NUMBF PECOV	IE R& ERY	SAI DE (	MPLE PTH FT)	ELEV./ DEPTH (FT)			VISUAL DES	CRIPTION	AND REMARKS	•
- 25 -		19 22							<b>T</b> . <b>TMMMMMMMMMMMMM</b>		ÞID (HDS	P) = 0.6/6.	2 ppm
- 30 -		15 17 19 22		2-	N.S. S.	29.0 31.0			Dense gi gravel,	ray brown sil trace clay	ty fine S/ PID (HDS	ND, little (P) = 0.6/5.	.4 ppm
		<u> </u>	si 91	3		34.0 36.0			Very de	-GLACIOFL	Y SAND, SO PID (HDS	ome gravel SP) = 0.6/4	.0 ppm
- 35 -		29 33		-			86.7						
- 40 -		15 - 41 - 74 - 63 	51		N.S.	39.0 41.0	39.0	100 00 00 00 00 00 00 00 00 00 00 00 00	Layered	hard gray CL	AY and SIL PID (HDS	.T P) ≈ 0.6/1.	.2 ppm
- 45 -		<u>27</u> 56 87 191	"61! 24'	5-		44.0		\$2,40,04,04,04	Very dei gravel,	nse gray coar trace fine s	er sandy S and PID (HDS	SILT, little SP) = 0.6/1.	e .2 ppm
		26	-516	6	77	49.0		10 00 Ti To .	D.O.	-GLACIAL	TILL DEPO	SITS-	•
- 50 -		44				50.5	73.2	111111		Top of Bed	PID (HDS	er) = 0.0/1. 2.5 ft.	, c ppm
			-				52.5	5	NOTE: S Descript	Roller bit	taken to 5 ng Report	6.0 ft.	ς.
											BORING N	p. B2	240-MW

A	<b>A</b> "	M	L / CAM ASS/	ALDRI( BRIDG ACHUSE	CH, INC E, ETTS		CO	RE	BO	RI	ING REPORT
DEP (FT		TE /FT.	RUN NO.	DEFT	H N N	OVERY/I	ROD X	ÆATH- ERING	ELEV./ DEPTH (FT)		VISUAL DESCRIPTION AND REMARKS
- 50	-										See pages 1 and 2 for Overburden Soils
									73.2 52.5		Top of Bedrock at 52.5 ft.
- 55	5		CT	54.0 59.0	,60 42	100	Di	scol.			Hard, slightly weathcred, extremely fractured to sound, mottled white-gray-brown, medium to very coarse grained PORFHYRITIC GRANITE. Joints very to moderately close, open, smooth to rough, planar and dipping of shallow angles. Joint surfaces weathered, oxidized. Observed healed joints dipping at high angles with numerous pits and vugs along them.
- 60 -	6 5 6 6 6	c	2 5	59.0 54.0	- 70 56	100 93	Dis	scot.			- REDROCK -
- 65	6 8 7 7 5	C3	69	2.0	46	160 76	Dis	col .	56.7		
									59.0	Lo In	Bottom of Exploration at 69.0 ft. .ocated at site 1 recharge basin. Installed monitoring well at 66.0 ft.
				-						PIC San rea	1D (HDSP): PID headspace reading collected from ample Jars. Results presented as background eading/headspace reading.

.



02.FRP F.IDS83300

<i>K</i> ∋	н	CAMBRIDGE SSACHUSETT	s	TES	TB	DRING REPORT	BORING NO. B242-MW
PROJECT CLIENT CONTRAC	HA U.	NSCOM A.F.R S. ARMY COR & R ENVIRON	E. LONG TE PS OF ENG MENTAL DR	PH SAHPLI SINEERS	NG PROGRA	M, BEDFORD, MA	FILE NO. 05833-065 SHEET NO. 1 of 2 LOCATION N 537,516
	ITEM		CASING	DRIVE SAMPLER	CORE BARREL	DRILLING EQUIPMENT & PROCEDURES	E 658,420
TYPE INSIDE HAMMER HAMMER	DIAMETER WEIGHT FALL	(IN) (LB) (IN)	HW 4.0 300 24	S 1 3/8 140 30	-	RIG TYPE B-61 Mobile BIT TYPE Roller bit DRILL MUD None OTHER Safety hammer/Cathead; HW to 49.0 ft.	ELEVATION 122.1 DATUM NGVD START May 30, 1996 FINISH May 30, 1996 DRILLER P. Thornsbury H & A REP W. Rubik
DEPTH (FT)	CASIN BLOWS PER F	G SAMPLER BLOWS PER 6 IN	SAHFLE NUHBER & PECOVERY	SAMPLE DEPTH (FT)	ELEV./ DEPTH (FT)	VISUAL DESCRIPTION	AND REMARKS
- 0 -					171 1	-TOPSOIL-	
5						Note: Cluster well with monit B243-MW. No sampling required B243-MW for soil description. -GLACIOFLUVIAL DEPO	toring well 1. See boring log SITS-
20	TIME	WATER LEVE	DATA	(FT) TO:	105.1	-GLACIOLACUSTRINE DEPO	SITS-
		IME (HP) B	CASING D	F HOLE	ATER 1	THIN WALL TUBE	(LIN FT) 49.0 (LIN FT) -
					s	SPLIT SPOON	15
						BORING NO.	B242-MW

•

n þ

ASA	HALEY & CA MASS	ALDRICH, MBRIDGE ACHUSETTS	ÎNC.	TES	г вс	RING	REPO	RT	BORING NO. FILE NO. SHEET NO.	B242-MW 05833-065 2 of 2
DEPTH (FT)	CASING BLOWS PER FT	SAMPLER BLOWS PER 6 IN	SAMPLE NUMBFR & RECOVERY	SAMPLE DEPTH (FT)	ELEV./ DEPTH (FT)		VISUAL DE	SCRIPTION A	ND REMARKS	
- 30 -			st 10"	45.0	83.2 38.9 73.1 49.0	Very den some sil	-GLACIOLAC -GLACIOFI se gray grav t, trace coa it tom of Expl in northeast d monitoring >): PID hea ole jars. R d reading/h	LIIVIAL DEPO velly medium rese sand FID (HDSF FID (HDSF Loration at ern area. well at 48 dspace read esults pres ead <pace read<="" th=""><th><pre>&gt;OSITS- SITS- n to fine SA y) = 0.4/0.8 49.0 ft. ing collecto ented as ading.</pre></th><th>ND, ppm</th></pace>	<pre>&gt;OSITS- SITS- n to fine SA y) = 0.4/0.8 49.0 ft. ing collecto ented as ading.</pre>	ND, ppm
							-	BORING NO.	B24	2-MW

-



Remarks: Plumbness test positive. Located at northeastern area.
ASA	HALEY & A CAN MASSA	LDRICH, I BRIDGE CHUSETTS	INC.	TEST	- BO	RING REPORT	BORING NO. B243-MW
PROJECT CLIENT CONTRACTO	HANSCO U.S. /	FILE NO. 05833-065 SHEET NO. 1 of 3 LOCATION N 537,515 E 658 426					
ITEM CASING			DRIVE SAMPLER	CORE BARREL	DRILLING EQUIPMENT & PROCEDURES	ELEVATION 122-2	
TYPE INSIDE DIAMETER (IN) HAMMER WEIGHT (LB) HAMMER FALL (IN)			rw/HW 5.0/4.0 300 24	S HQ 1 3/8 2.5 140 - 30 -		RIG TYPE 8-61 Mobile BIT TYPE Roller bit DRILL MUD None OTHER Safety hammer/Cathead; PW to 25.0 ft.; HW to 50.5 ft.	DATUM NGVD START Hay 28, 1996 FINISH May 29, 1996 DRILLER P. Thornsbury H & A REP W. Rubik
DEPTH (FT)	CASING BLOWS PER FT	SAMPLER BLOWS PER 6 IN	SAMPLE NUMBER & RECOVERY	SAMPLE DERTH (FT)	ELEV./ DEPTH (FT)	VISUAL DESCRIPTION	AND REMARKS
- 5 -		$ \begin{array}{r}   2 \\   2 \\   4 \\   6 \\   5 \\ $	s1 9" 51A 12" 52 14" 53 10" 53 10" 55 11" 55 11" 57 13" 57 13" 57 13" 57 13" 57 13" 57 13" 57 13" 57 13" 57 13" 57 13" 57 14" 57 14" 57 14" 57 57 57 14" 57 57 57 14" 57 57 14" 57 57 14" 57 57 14" 57 57 57 14" 57 57 14" 57 57 57 14" 57 57 57 14" 57 57 57 14" 57 57 57 57 14" 57 57 57 14" 57 57 57 57 57 57 57 57 57 57 57 57 57	0.0 1.0 7.0 2.0 4.0 4.0 6.0 8.0 10.0 10.0 10.0 12.0 14.0 14.0 14.0 14.0 14.0 14.0 14.0 15.0 15.0 16.0 17.0 18.0 16.0 17.0 17.0 10.0 12.0 14.0 10.0 12.0 10	121.2 1.0 105.2 17.0	Soft black LOAM, little fine -TUPSOIL- PID(SS) =0.4/0.4 ppm PID(H STA: Loose brown medium to f root fibers PID (H S2: Loose to medium donse me in partings PID ( FID (HD -GLACIOFLUVIAL DEP S3: D.O. PID ( PID (HD S4: Loose gray fine SAND in PID (HD S5: D.O. PID ( Loose gray-brown medium to fi partings PID ( PID (HD Loose gray-brown medium to fi partings, little silt PID (HD Loose rusty brown medium to fi partings, little silt PID (HD S7A: Medium dense gray to rusty br S10: D.O. except loose FID (HD -GLACIOLACUSTPINE D Stiff gray S1LT in partings,	sand, root fibers DSP) =0.4/0.4 ppm ine SAND, trace SS) = 0.4/0.4 ppm dium to fine SAND SS) = 0.4/0.4 ppm dium to fine SAND SS) = 0.4/0.4 ppm SS) = 0.4/0.4 ppm SP) = 0.4/0.5 ppm partings SS) = 0.4/0.4 ppm SS) = 0.4/0.4 ppm SP) = 0.4/0.5 ppm SS) = 0.4/0.4 ppm SS) = 0.4/0.4 ppm SS) = 0.4/0.4 ppm SS) = 0.4/0.8 ppm own fine SAND in SS) = 0.4/0.8 ppm ine SAND in SS) = 0.4/0.8 ppm ine SAND in SS) = 0.4/0.8 ppm SS) = 0.4/0.4 ppm
- 25		5	\$11 20"	26.0		frequent seams	
		WATER LI	EVEL DATA			SAMPLE IDENTIFICATION	SUMMARY
DATE 5-28-96	TIME 07:30	ELAPSED TIME (HR	DE CASINO	PTH (FT) FOTTOM OF HOLE 6.0	TO: WATER 4.0	O CONTROL OVERBUR T CALL TUBE U CALL TUBE U DISTURBED SAMPLE S SPLIT SPOON OVERBUR ROCK CO SAMPLES	DEN (LIN FT) 54.0 RED (LIN FT) 15.0 165, 30
						BORING	NO. B243-MW



Æ	P	HALEY	& AL CAMBR	DRICH, LIDGE, HUSETI	1NC.	CC	RE	BOR	ING REPORT BORING NO. B243 FILE NO. 05833-0 SHEET NO. 3 OF 3	8 - MW 165
DEP1 (FT	H N	RILLING RATE IN./FT.	RUN	DEPTH (FT)	RECOVE	RY/R9D	WEATH- ERING	ELEV./ DEPTH (FT)	VISUAL DESCRIPTION AND REMARKS	
- 45									See Pages 1-2 for overburden soils	
- 50								<b>Fa</b> <sup>1</sup> ,	Top of Bedrock at 50.4 ft.	
- 55		5 6 6	CT	54.0 59.0	-tg 58	97 97	Discol.	50.4	-BEDROCK- Hard, very slight weathered, slighty fractured sound, mottled white-gray-pink, medium to coar grained GRANITE. Joints, close to wide, open, to rough, planar and dipping at shallow angles Joint surfaces slightly weathered, oxidized, s with very thin clay coating.	to se smooth ome
- 60		6 6 7 6 6 8	cz	59.0 64.0	58 58	-97 97	Discol.			
- 65	5	6 4 4 5 4	C3	<u>69.0</u>	30 45	100 75	Discol		C3: D.O. except slightly weathered, extremely fractured to sound. Joints very close to clos open, weathered and oxidized. Some joints wit infillings. Note: Rapidly loosing water below 64.0 ft.	e, h sand
		4		-	-			69.0	Bottom of Exploration at 69.0 ft. Located in northeastern area. Installed monitoring well at 68.0 ft. PID (SS): PID reading collected from open spl spoon. Results presented as background readin spoon reading. PID (HDSP): PID headspace reading collected f sample jars. Results presented as background reading/headspace reading.	it g/split rom



F:\058337085\A04330\Al dbf

25 July 1996

DWI08GOZ.FRP

# Appendix A-4b

Authorization to Leave Monitoring Wells in Place

### Frizzell, Angela

From:Best Thomas Civ 66 CES/CEVR [Tom.Best@hanscom.af.mil]Sent:Friday, January 31, 2003 2:47 PMTo:'Lutes, Chris'; Best Thomas Civ 66 CES/CEVRCc:Hansen Jerry E Civ AFCEE/ERS; Gordon, Glen; Frizzell, AngelaSubject:RE: Hanscom IRZ Report/Wells

Sorry I haven't got back to you sooner - in general I have little to add in the way of comments - in regards to de-mob, yes we want to keep the wells as is - I do not foresee any requirements of ARCADIS in the transfer of the wells. Just walk away. Would like to get a type version of the boring & well logs but this is not imperative.

Comment - page 12, para 1.3.2 next to last sentence of 1st subpara - add "a former fire training area" after site 1. You do identify it as such later but I think it helps a reader to introduce it here, especially since most in the Air Force know exactly what was done at a fire training area.

Page 83, para 7.1 - change last 2 sentences of 1st para to read: A pre-NPL "Remedial Action Plan" for what is now NPL Operable Unit 1 (which includes Site 1) was approved and implemented in 1887. Subsequently, in October 2000 an Interim Record of Decision was issued for NPL Operable Unit 1 to continue operation of the existing dynamic groundwater collection and treatment system. MADEP concurred with the Interim Record of Decision.

last para, 1st sentence - change to .... Hanscom Field/Hanscom AFB boundaries....

That's all I have.

-----Original Message-----From: Lutes, Chris [mailto:CLutes@arcadis-us.com] Sent: Wednesday, January 22, 2003 11:17 AM To: Best Thomas Civ 66 SPTG/CEVR Cc: Hansen,Jerry (E-mail); Gordon, Glen; Frizzell, Angela Subject: Hanscom IRZ Report/Wells

Dear Tom -

Have you had a chance to review the draft Hanscom report we issued in November? We have received ESTCP and AFCEE comments and are revising the report now, so it would be a good time to incorporate any comments you might have.

Thinking ahead to demobilization of the IRZ demonstration project at Hanscom after the rebound round we plan to do in 6 months or so, we are assuming you would like to keep and maintain the one injection and five monitoring wells, rather then having them decommissioned by us after the rebound round. If you would like to keep them, do you foresee any other requirements of ARCADIS involved in the transfer of the wells to you? We need to include our plans for these wells in a new demobilization section of the final report, so your comments will be appreciated.

Thanks, Chris

# Appendix A-5

Survey Data



VIA: U.S. MAIL

October 16, 2002

Mr. Brian Therriault Arcadis G & M, Inc. 175 Cabot Street Suite 400 Lowell, Massachusetts 01854

### Re: Arcadis Project No. RN009901.0012 Hanscom Airforce Base

Dear Brian:

This letter is relative to Arcadis project No. RN009901.0012, more specifically well elevations for well IRZ-INJ at Hanscom Air Force Base.

Top of outer casing = 123.91

Top of inner casing = 123.67

Ground elevation @ well =  $123.9 \pm$ 

As requested all above elevations are based on the National Geodetic Vertical datum of 1929 (NGVD29).

Should you have any questions or need additional services please do not hesitate to call.

Sincerely MERIDIAN ENGENEERING, INC.

Christopher P. Nicholas

PS Invoice will follow

CPMjind\F:\~\3663\WORD\letters\Arcadis (021016).doc

en al de la surder de la servición de la servic

DANVERS

98 High Street Danvers, MA 01923-3189 P: (978) 739-9130 • F: (978) 739-9140

www.meridianengineering.com mei@meridianengineering.com

### WESTBOROUGH

69 Milk Street, Suite 302 Westborough, MA 01581-1227 P: (508) 871-7030 • F: (508) 871-7039

Printed on Recycled Paper

### HANSCOM A.F.B. OBSERVATION WELLS PREPARED FOR ARCADIS GERAGHTY & MILLER, INC. INSTALLATION RESTORATION PROGRAM

### May 14, 2001

DESCRIPTION	NORTHING	EASTING	GROUND ELEVATION	TOP OF PVC (INNERCASING) ELEVATION
IRZ-5	537428.9	658206.1	125.2+/-	127.98
B-239	537548.5	657673.2	125.9+/-	125.68
RAP 1-6-R	537420.7	658082.9	123.4+/-	125.64
RAP 1-6-S	537430.6	658081.2	123.5+/-	125.00*
RAP 1-6-T	537417.7	658074.4	123.7+/-	125.49*

#### NOTES:

- 1. ELEVATIONS DEPICTED ARE BASED UPON THE NATIONAL GEODETIC VERTICAL DATUM (N.G.V.D.) OF 1929.
- 2. HORIZONTAL LOCATIONS OF THE WELLS ARE BASED UPON THE MASSACHUSETTS STATE PLANE COORDINATE SYSTEM (1927 NORTH AMERICAN DATUM).
- 3. THE FIELD SURVEY TO LOCATE THE WELLS WAS PERFORMED ON THE GROUND BY MERIDIAN ENGINEERING, INC. ON MAY 9, 2001.
- \* ELEVATION OF CROSS-CUT SET ON OUTER CASING.

PAGE:02





### HANSCOM A.F.B. OBSERVATION WELLS PREPARED FOR ARCADIS GERAGHTY & MILLER, INC. INSTALLATION RESTORATION PROGRAM

### DECEMBER 11, 2000

DESCRIPTION	NORTHING	EASTING	GROUND ELEVATION	TOP OF PVC (INNERCASING) ELEVATION
IRZ-1	537439.3	658094.9	123.4+/-	122.82
IRZ-2	537369.1	658131.7	125.7+/-	127.62
IRZ-3	537397.3	658164.0	125.2+/-	128.16
IRZ-4	537396.2	658076.9	123.0+/-	126.00
IRZ-1NJ	537440.9	658048.9	124.0+/-	123.82

NOTES:

1. ELEVATIONS DEPICTED ARE BASED UPON THE NATIONAL GEODETIC VERTICAL DATUM (N.G.V.D.) OF 1929.

2. HORIZONTAL LOCATIONS OF THE WELLS ARE BASED UPON THE MASSACHUSETTS STATE PLANE COORDINATE SYSTEM (1927 NORTH AMERICAN DATUM).

3. THE FIELD SURVEY TO LOCATE THE WELLS WAS PERFORMED ON THE GROUND BY MERIDIAN ENGINEERING, INC. ON DECEMBER 1, 2000.



Ł

2

9EC-12.00 09:54

FROM: ARCADIS

PAGEISE

# Appendix A-6a

## Additional Discussion of Data Quality

### Appendix A-6a: Additional Discussion of Data Quality

### **Data Quality Assessment**

Much of the assessment of the data has been necessarily presented as part of the overall discussion of the data in Section 4.3. However, some specific data quality matters will be discussed in this section.

### **CAH Data Assessment**

As planned, a validation was performed on the critical data sets for technology verification – the CAH data from groundwater analyzed at the offsite laboratory (STL) from the 5 abbreviated and 3 full sampling rounds. These data were shown to be very reliable. Full details of this assessment are presented in individual data set validation memos compiled as Appendix A-6b. In addition, although it was not planned in the demonstration plan, numerous additional rounds of TCE and DCE analyses were provided courtesy of Tom Best of Hanscom AFB. These included samples collected on some occasions by the Base's contractor (Shaw/IT) and on other occasions by ARCADIS. The analyses were performed at Hanscom by IT. As discussed in Section 4.3 the data sets with and without the on-site data lead to virtually identical conclusions, suggesting that the two laboratories are in rough agreement. There were essentially no "missing values" in the CAH data set. However as discussed in Section 4.3.3.3.4 we were not able to provide an ideal upgradient well data set since the demonstration zone turned out to be in a source area and the area immediately upgradient of the injection well was under the runway out run.

As discussed in Section 4.3.3.3.5, the wells outside the reactive zone showed little or no evidence of biodegradation. This strengthens the conclusions that substrate availability is linked with improved biodegradation. Since these wells were above, below, upgradient and side gradient from the wells that showed clear evidence of biodegradation (IRZ-1 and RAP1-6T) this strengthens the conclusion that contaminant removal at those three wells was due to enhanced biodegradation rather than displacement or dilution.

Further, enhanced degradation rates were calculated as discussed in Section 4.3.3.5. They were shown to be substantially enhanced over the pretreatment rates at this site. These degradation rates were shown to be higher than typical natural attenuation results previously observed in the field at other sites. TCE degradation rates were shown to be in the range expected for enhanced in-situ bioremediation systems based on results at other sites.

Finally, the trends in CAH and dissolved gas concentrations observed at this site and discussed in Section 4.3 were consistent with theory and the literature (including data collected at other sites and in the laboratory). Specifically, TCE degradation occurred before and under less reducing conditions than DCE degradation. All of these factors suggest that the CAH data set is quite reliable.

### Assessment of Other Portions of the Data Set

Some difficulties encountered in well installation were described in Section 3.5.7.1.1 and difficulties encountered during injections in Section 3.5.1. However, neither of these problems is expected to materially affect overall data quality.

Other problems that were encountered during various sampling rounds are discussed in detail in Appendix A-6c. Most of these problems occurred with the field data or the TOC/DOC data. In particular the high levels of organic carbon in the injection wells caused problems with field measurements. However, these parameters are used more for process operation and interpretation than to assess the effectiveness of the technology. Therefore, it was judged that these problems did not materially affect the overall demonstration results.

# Appendix A-6b

Validation Memos

### ORGANIC DATA SUMMARY

Validation of Method 8260B Analysis (VOC's) Hanscom AFB, Bedford, Massachusetts

Analyses Performed by Severn Trent Laboratories

### \* SAMPLE DESCRIPTION

The following six groundwater samples and associated QC samples were collected at the Hanscom AFB site and submitted to Severn Trent Laboratories in Arvada, Colorado for analysis by USEPA Method 8260B.

STL Lot #: B0F170112

> HAN-GW-B239-1 HAN-GW-RAP1-6S-1 HAN-GW-RAP1-6R-1 HAN-GW-IRZ-3-1 HAN-GW-IRZ-2-1 HAN-GW-IRZ-4-1 HAN-GW-FB-1

#### \* FIELD SAMPLING

The sampling documentation was reviewed. The sampling date, team members, location, depth, technique and field preparation techniques were appropriate and properly documented. There were no notations in the field notes or sample custody forms of the pH of the sample. Method 8260B requires that HCl be added as a preservation to a pH<2. It was noted that HCl was added, but the volume or pH was not recorded.

#### \* SAMPLE SHIPPING/RECEIVING

Samples were received by STL June 17, 2000. Sample custody forms were reviewed. The temperature of the as received samples was 40 C, but no notation of the pH or headspace conditions of the samples.

### **\* HOLDING TIME COMPLIANCE**

Samples were taken on June 16, 2000 (?), shipped and received by the laboratory on June 17, 2000 and prepared and analyzed on June 27, 2000. Method 8260B stipulates a maximum hold time of 14 days. All samples were analyzed within the 14 days hold time.

### \* GC/MS INSTRUMENT PERFORMANCE CHECK

Laboratory report contains BFB tune data for June 27, 2000, the date the samples were analyzed. All tune criteria specified in the Functional Guidelines were met.

### \* GC/MS CALIBRATION

e . .

ر •

The initial calibration was performed on May 15, 2000. Acceptance criteria established for initial calibration is that relative response factors should be greater than or equal to 0.05 and the percent relative standard deviation (%RSD) between calibration levels should be less than 30%. All compounds in the target list (Table 1) met acceptance criteria.

Continuing calibration check standards were analyzed on June 26 and 27, 2000. The acceptance criteria established for the CCC is that the percent difference between RRF from the initial calibration must be within 25% and the actual calculated RRF must be greater than or equal to 0.05. All compounds in the target list meet these criteria.

#### \* METHOD BLANK

One method blank was analyzed with the batch on June 27, 2000. There were no compounds detected in the method blank.

### \* EQUIPMENT BLANK

The field crew submitted an equipment blank and a field blank to the laboratory for analysis. No compounds were detected on the equipment blank

### **\* SURROGATE RECOVERY**

Surrogate recoveries for the target compounds in the samples were all within established control limits.

#### \* MATRIX SPIKE ANALYSIS

MSD recovery for matrix spike samples were within the recovery limit of 80 - 120%.

### ? LABORATORY CONTROL SPIKE

All laboratory control samples were within established control limits.

### ? INTERNAL STANDARDS

Internal standards acceptance criteria requires that area counts for internal standards be within a factor of 2 (-50% to +100%) of the area counts associated with the calibration standard and that the retention time be within 30 seconds of the associated calibration standard. All internal standard areas and retention times met these criteria for all samples.

### ? COMPOUND IDENTIFICATION AND QUANTITATION

Functional guideline criteria require that relative retention times of identified compounds agree within 0.06 units of the associated standard. All retention times of the identified compounds were within 0.06 units of the continuing calibration standard retention times. The guidelines also require that the sample mass spectra match the standard mass spectra. Mass spectra data for the continuing calibration standard was not included in the laboratory data, so these could not be compared. Library matches of a few compounds were reviewed and compound identifications appear to be valid.

Reported concentrations were verified by hand-calculating the results using raw data reports and RRF's reported by the laboratory in the continuing calibration report. Concentrations reported by the laboratory were reproduced and are valid. However, the equation shown on the raw laboratory reports for the calculation of concentrations is not clear. The value referred to as *amt*. in the equation actually represents several values (i.e., the peak area of the compound of interest multiplied by the amount of internal standard divided by the area of the internal standard multiplied by the relative response factor of the target compound). In addition, dilution factors shown on the raw reports are inconsistent with those reported formally. The reviewer believes that all concentrations were reported correctly in the analytical report prepared for ARCADIS, but there are areas on the raw data reports that could lead to some confusion when the data is audited.

### FIELD DUPLICATE

No field duplicate in this lot.

### **OVERALL DATA ASSESSMENT**

The Method 8260B data for Lot # B0F170112 included in this report have been validated according to the criteria presented in the *National Functional Guidelines for Organic Data Review* and are valid.

### Table 1. Compounds Target List

r Li r

> Carbon Tetrachloride Chlorobenzene chloroethane 2-chloroethyl vinyl ether chloroform chloromethane 1,2 dichlorobenzene 1,3 dichlorobenzene 1,4 dichlorobenzene trans-1,4- dichloro-2-butene dichlorodifluromethane 1,1-dichloroethane 1,2-dichloroethane cis-1.2-dichloroethene trans-1,2-dichloroethene 1.1-dichloroethene 1,2-dichloroethene(total) 1.2-dichloropropane cis-1.3-dichloropropene trans-1,3-dichloropropene trichloroethene 1,1,1 trichloroethane vinyl chloride tetrachloroethane methylene chloride



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

#### MEMO

Subject: Validation of Water Samples Collected at Hanscom AFB 11/6/00

Date: 4/15/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Four samples were collected on November 6, 2000 at Hanscom AFB, Massachusetts. These samples and a trip blank were submitted on November 7, 2000 to Severn Trent Laboratories (STL) Tampa East Facility in Tampa, Florida for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest.

It is the reviewer's opinion that the quality of the data was acceptable with the qualifiers discussed below.

### Sample Delivery Group B0K070141

### **Volatile Organic Data**

### I Preservation

The samples arrived at the lab promptly and were within the temperature criteria. The samples pH were not recorded as checked although preservation was indicated on the COC. The samples were analyzed within the Acid-Preserved hold time criteria of 14 days.

# II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibration and Analysis days. 8260B criteria are more stringent that CLP and therefore the 8260B criteria were used. The criteria for ion 174 in the CLP guidelines is 50-120% of ion m/z 95. The 8260B criteria is "greater than" 50% and that 95 is the base peak. The 174 ion was at 110% of 95 and therefore 95 was not the base peak. Therefore this may (subject to interpretation) have failed the 8260B criteria but the reviewer does not believe it would adversely effect the data.

### III Initial Calibration

All of the Chloroethenes and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 10 to 625 ng introduced to the column. Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs.

### IV Continuing Calibration

The Samples were run in the same 12 hour window as the IC and therefore would not have been required.

### V Blanks

All blanks reported non-detect for all compounds of interest.

### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples. The DMC compounds were mixed with the internal standard compounds and spiked as one solution. This prevents the DMC solution from being used to check to see if the proper amount of internal standard solution was added. This is not specifically forbidden in 8260B but the method does speak in different areas about spiking 10µl of one and then later of spiking the other solution. The reviewer believes that separate spiking provides more and different information than when spiking in one solution.

### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

No matrix spikes were run with this batch.

### VIII Regional QA/QC

Not Applicable

### IX Internal Standards

All criteria were met using the first level of the IC to compare.

### X Target Compound Identification

The very high concentration samples showed poor ratio comparison in the non-diluted run. The compounds saturated the detector. They were fine in the diluted runs.

# XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The sample quantitation in the diluted runs had a gap in the quantitation ranges. The range of quantitation in the IC was 0.4 to  $25\mu g/l$ ; the 1:500 diluted runs had a range of 200 to  $12500\mu g/l$ . This allows compounds to be over range in the initial run and under range in the diluted run. This may have been a known and accepted but is out of the reviewer's knowledge. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples. The dilutions were run immediately after the samples. The reviewer would have preferred to see the dilutions run prior to higher level samples and at least on blank after a high sample to show no carryover.

### XII Tentatively Identified Compounds (TICs)

Not Applicable

### XIII System Performance

No problems noted.

### XIV Overall Assessment of Data

The data appears to be very good, with only minor problems (dilutions).



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

ΜΕΜΟ

Subject: Validation of Water Samples Collected at Hanscom AFB 11/7/00

Date: 4/12/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Six samples were collected on November 7, 2000 at Hanscom AFB, Massachusetts. These samples and a trip blank were submitted on November 8, 2000 to Severn Trent Laboratories (STL) Tampa East Facility in Tampa, Florida for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest.

It is the reviewer's opinion that the quality of the data was acceptable with the qualifiers discussed below.

### Sample Delivery Group B0K090260

### **Volatile Organic Data**

### I Preservation

The samples arrived at the lab promptly and were within the temperature criteria. The samples pH were not recorded as checked although preservation was indicated on the COC. The samples were analyzed within the Acid-preserved hold time criteria of 14 days.

# II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibration and Analysis days. 8260B criteria are more stringent that CLP and therefore the 8260B criteria were used.

### III Initial Calibration

All of the Chloroethenes and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 10 to 625 ng introduced to the column. Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs.

### IV Continuing Calibration

The Continuing Calibration (CC) was run in the 12-hour analysis window before any of the samples were analyzed. The CC had RFs above 0.05 for the Chloroethenes and the DMCs. The Relative Percent Deviations (RPDs) from the IC were below 15% for the Chloroethenes and the DMC

### V Blanks

All blanks reported non-detect for all compounds of interest.

### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples.

### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

Criteria were met for the MS but one compound was one percent outside of criteria for one compound. Data quality would not be significantly affected.

### VIII Regional QA/QC

Not Applicable

### IX Internal Standards

All criteria were met using the first level of the IC to compare.

### X Target Compound Identification

The very high concentration samples showed poor ratio comparison in the non-diluted run. The compounds saturated the detector. They were fine in the diluted runs.

# XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The sample quantitation in the diluted runs had a gap in the quantitation ranges. The range of quantitation in the IC was 0.4 to  $25\mu g/l$ ; the diluted runs had a range of 200 to  $12500\mu g/l$ . This allows compounds to be over range in the initial run and under range in the diluted run. This may have been a known and accepted but is out of the reviewer's knowledge. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples. The dilutions were run immediately after the samples. The reviewer would have preferred to see the dilutions run prior to higher level samples and at least on blank after a high sample to show no carryover.

### XII Tentatively Identified Compounds (TICs)

Not Applicable

### XIII System Performance

No problems noted.

### XIV Overall Assessment of Data

The data appears to be very good, with only minor problems (MSD and dilutions).



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

ΜΕΜΟ

Subject: Validation of Water Samples Collected at Hanscom AFB 4/05-06/2001

Date: 4/16/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Ten samples were collected on April 5 and 6, 2001 at Hanscom AFB, Massachusetts. These samples and a trip blank were submitted on April 7, 2001 to Severn Trent Laboratories (STL) Tampa East Facility in Tampa, Florida for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest. Acetone and other ketones were of secondary interest.

It is the reviewer's opinion that the quality of the data was acceptable with the qualifiers discussed below, for all samples except HAN-GW-IRZ-INJ. The differences between the diluted and undiluted values are more than are expected therefore the higher value should be used.



### Sample Delivery Group B1D090101

### Volatile Organic Data

### I Preservation

The samples arrived at the lab promptly and were within the temperature criteria. The samples pH were indicated as checked at the lab. Preservation was indicated on the Chain-of-Custody (COC) for three of the ten samples. The samples were analyzed within the Acid-preserved hold time criteria of 14 days. The trip blank was not on the COC. It is assumed that the lab made contact with the project personnel to get direction for the trip blank and two samples which had no analysis requested on the COC.

## II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibration and Analysis days. 8260B criteria are more stringent that CLP and therefore the 8260B criteria were used.

### III Initial Calibration

All of the Chloroethenes and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 25 to 625 ng introduced to the column. Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs. The light ketones were low on response factor and too high in variability. The Ketone data should not be used for precise quantitation.

### IV Continuing Calibration

The Continuing Calibration (CC) was run in the 12-hour analysis window before any of the samples were analyzed. The CC had RFs above 0.05 for the Chloroethenes and the DMCs. The Relative Percent Deviations (RPDs) from the IC were below 15% for the Chloroethenes and the DMC. The ketones were also low on response factor and high on variability in the CC.

### V Blanks

All blanks reported non-detect for all compounds of interest.

### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples.

### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

MS/MSD appeared in the run log but was not reported or mentioned anywhere else in the report. This is of high concern. It may be that the MS/MSD was not requested and the analyst accidentally ran it, but if so the lab would have avoided a suspicion by reporting it anyway.

The laboratory control spikes indicate that the ketones should not be used for precise quantitation. Order of magnitude or detect/nondetect decisions could be supported.

### VIII Regional QA/QC

Not Applicable

### IX Internal Standards

All criteria were met.

### X Target Compound Identification

The very high concentration samples showed poor ratio comparison in the non-diluted run. The compounds saturated the detector. They were fine in the diluted runs. The reference spectrum for 1,2-Dichloroethane includes the 78 ion for the co-eluting benzene. The ions for styrene are in the reference spectrum for o-xylene. The ions for butylbenzene are in the reference spectrum for 1,2-dichlorobenzene. This many extra peaks in reference spectra indicate that the reference spectra were taken from a mix of compounds and the reference spectra for the co-eluting compounds have not been removed. This could be a problem leading to misidentification. These were not compounds of interest for this project and therefore the reviewer does not see a problem. But these compounds were detected and if the data is used for tertiary purposes at a later time it could be a problem. No misidentifications were noticed.

# XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The sample quantitation in the diluted runs had a gap in the quantitation ranges. The range of quantitation in the IC was 1 to  $25\mu g/l$ ; the 500X diluted runs had a range of 500 to  $12500\mu g/l$ . This allows compounds to be over range in the initial run and under range in the diluted run. This occurred for at least one compound of interest in each run that required high dilution (7 of the 8 diluted samples). In one sample an overrange was diluted to a non-detect. This may have been a known and accepted but is out of the reviewer's knowledge. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples. The dilutions were run immediately after the samples. The reviewer would have preferred to see the dilutions run prior to higher level samples and at least one blank after a high sample to show no carryover.

One sample had very poor agreement between the diluted and undiluted runs. HAN-GW-IRZ-INJ seemed to have more difference than had been noted in other samples.

### XII Tentatively Identified Compounds (TICs)

Not Applicable

### XIII System Performance

No problems noted.

### XIV Overall Assessment of Data

The data appears to be very good, with only minor problems (MS/MSD, reference sprectra and dilutions).



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

МЕМО

Subject: Validation of Water Samples Collected at Hanscom AFB 5/3/2001

Date: 7/26/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Two samples were collected on May 3, 2001 at Hanscom AFB, Massachusetts. These samples and a trip blank were submitted on May 4, 2001 to Severn Trent Laboratories (STL) Tampa East Facility in Tampa, Florida for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest. Acetone and other ketones were of secondary interest.

It is the reviewer's opinion that the quality of the data was acceptable with the qualifiers discussed below, for all samples.



### Sample Delivery Group B1E040122

### **Volatile Organic Data**

### I Preservation

The samples arrived at the lab promptly and were within the temperature criteria. The samples pH were indicated as checked by the analyst using the footnote "All aqueous samples had a pH < 2 unless otherwise noted" the reviewer feels an actual entry indicating the pH would be a more comforting indicator that it was actually checked. Preservation was indicated on the Chain-of-Custody (COC) for the five samples. The samples were analyzed within the Acid-preserved hold time criteria of 14 days. The trip blanks ware not on the COC. The lab made contact with the project personnel to get direction for the trip blank it was analyzed with this set of samples.

# II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibration and Analysis days. 8260B criteria are more stringent that CLP and therefore the 8260B criteria were used.

### III Initial Calibration

All of the Chloroethenes and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 25 to 625 ng introduced to the column. Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs. The ketones were low on response factor and too high in variability. The Ketone data should not be used for precise quantitation.

### IV Continuing Calibration

The ICAL and Samples were run in the same 12 hr window. No CCC was needed but the midpoint of the curve was treated as a CCC and was already discussed.

### V Blanks

All blanks reported non-detect for all compounds of interest.

### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples.

### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

MS/MSD appeared in the run log in the batch but was not project specific and therefore only the summary sheet is in the report. It looked fine.

The laboratory control spikes were well reported and were acceptable.

### VIII Regional QA/QC

Not Applicable

### IX Internal Standards

All criteria were met.

### X Target Compound Identification

The very high concentration samples showed poor ratio comparison in the non-diluted run. The compounds saturated the detector. They were fine in the diluted runs. The reference spectrum for 1,2-Dichloroethane includes the 78 ion for the co-eluting benzene. The ions for styrene are in the reference spectrum for o-xylene. The ions for butylbenzene are in the reference spectrum for 1,2-dichlorobenzene. This many extra peaks in reference spectra indicate that the reference spectra were taken from a mix of compounds and the reference spectra for the co-eluting compounds have not been removed. This could be a problem leading to misidentification. These were not compounds of interest for this project and therefore the reviewer does not see a problem. But these compounds were detected and if the data is used for tertiary purposes at a later time it could be a problem. No mis-identifications were noticed.

# XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The sample quantitation in the diluted runs had a gap in the quantitation ranges. The range of quantitation in the IC was 1 to  $25\mu g/l$ ; the 200X diluted runs had a range of 200 to  $5000\mu g/l$ . This allows compounds to be over range in the initial run and under range in the diluted run. This occurred for at least one compound of interest in each run that required high dilution. This may have been a known and accepted but is out of the reviewer's knowledge. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples. The dilutions were run immediately after the samples. The reviewer would have



preferred to see the dilutions run prior to higher level samples and at least one blank after a high sample to show no carryover.

### XII Tentatively Identified Compounds (TICs)

Not Applicable

### XIII System Performance

No problems noted.

### XIV Overall Assessment of Data

The data appears to be very good, with only minor problems (reference spectra and dilutions).



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

МЕМО

Subject: Validation of Water Samples Collected at Hanscom AFB 5/4/2001

Date: 7/26/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Five samples were collected on May 4, 2001 at Hanscom AFB, Massachusetts. These samples and trip blanks were submitted on May 5, 2001 to Severn Trent Laboratories (STL) Tampa East Facility in Tampa, Florida for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest. Acetone and other ketones were of secondary interest.

It is the reviewer's opinion that the quality of the data was acceptable with the qualifiers discussed below, for all samples.


#### Sample Delivery Group B1E050110

#### **Volatile Organic Data**

#### I Preservation

The samples arrived at the lab promptly and were within the temperature criteria. The samples pH were indicated as checked by the analyst using the footnote "All aqueous samples had a pH < 2 unless otherwise noted" the reviewer feels an actual entry indicating the pH would be a more comforting indicator that it was actually checked. Preservation was not indicated on the Chain-of-Custody (COC) for the five samples. The samples were analyzed within the Acid-preserved hold time criteria of 14 days. The trip blanks ware not on the COC. The lab made contact with the project personnel to get direction for the trip blank it was not to be analyzed with this set of samples.

## II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibration and Analysis days. 8260B criteria are more stringent that CLP and therefore the 8260B criteria were used.

#### III Initial Calibration

All of the Chloroethenes and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 25 to 625 ng introduced to the column. Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs. The ketones were low on response factor and too high in variability. The Ketone data should not be used for precise quantitation.

#### IV Continuing Calibration

The ICAL and Samples were run in the same 12 hr window. No CCC was needed but the midpoint of the curve was treated as a CCC and was already discussed.

#### V Blanks

All blanks reported non-detect for all compounds of interest.

#### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples.

#### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

MS/MSD appeared in the run log in the batch but was project specific. It looked fine.

The laboratory control spikes were well reported and were acceptable.

#### VIII Regional QA/QC

Not Applicable

#### IX Internal Standards

All criteria were met.

#### X Target Compound Identification

The very high concentration samples showed poor ratio comparison in the non-diluted run. The compounds saturated the detector. They were fine in the diluted runs. The reference spectrum for 1,2-Dichloroethane includes the 78 ion for the co-eluting benzene. The ions for styrene are in the reference spectrum for o-xylene. The ions for butylbenzene are in the reference spectrum for 1,2-dichlorobenzene. This many extra peaks in reference spectra indicate that the reference spectra were taken from a mix of compounds and the reference spectra for the co-eluting compounds have not been removed. This could be a problem leading to misidentification. These were not compounds of interest for this project and therefore the reviewer does not see a problem. But these compounds were detected and if the data is used for tertiary purposes at a later time it could be a problem. No mis-identifications were noticed.

## XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The sample quantitation in the diluted runs had a gap in the quantitation ranges. The range of quantitation in the IC was 1 to  $25\mu g/l$ ; the 500X diluted runs had a range of 500 to  $12500\mu g/l$ . This allows compounds to be over range in the initial run and under range in the diluted run. This occurred for at least one compound of interest in each run that required high dilution. This may have been a known and accepted but is out of the reviewer's knowledge. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples. The dilutions were run immediately after the samples. The reviewer would have preferred to see the dilutions run prior to higher level samples and at least one blank after a high sample to show no carryover.

#### XII Tentatively Identified Compounds (TICs)

Not Applicable

#### XIII System Performance

No problems noted.

#### XIV Overall Assessment of Data

The data appears to be very good, with only minor problems (reference spectra and dilutions).



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

ΜΕΜΟ

Subject: Validation of Water Samples Collected at Hanscom AFB 5/7/2001

Date: 5/6/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Three samples were collected on May 7, 2001 at Hanscom AFB, Massachusetts. These samples and a trip blank were submitted on May 8, 2001 to Severn Trent Laboratories (STL) Tampa East Facility in Tampa, Florida for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest. Acetone and other ketones were of secondary interest.

It is the reviewer's opinion that the quality of the data was acceptable with the qualifiers discussed below, for all samples.



#### Sample Delivery Group B1E080187

#### **Volatile Organic Data**

#### I Preservation

The samples arrived at the lab promptly and were within the temperature criteria. The samples pH were indicated as checked by the analyst using the footnote "All aqueous samples had a pH < 2 unless otherwise noted" the reviewer feels an actual entry indicating the pH would be a more comforting indicator that it was actually checked. Preservation was indicated on the Chain-of-Custody (COC) for the three samples. The samples were analyzed within the Acid-preserved hold time criteria of 14 days. The trip blank was not on the COC. It is assumed that the lab made contact with the project personnel to get direction for the trip blank.

## II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibrations and Analysis days. 8260B criteria are more stringent that CLP and therefore the 8260B criteria were used.

#### III Initial Calibration

All of the Chloroethenes and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 25 to 625 ng introduced to the column. Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs. The ketones were low on response factor and too high in variability. The Ketone data should not be used for precise quantitation.

#### IV Continuing Calibration

The Continuing Calibration (CC) was run in the 12-hour analysis window before any of the samples were analyzed. The CC had RFs above 0.05 for the Chloroethenes and the DMCs. The Relative Percent Deviations (RPDs) from the IC were below 15% for the Chloroethenes and the DMC. The ketones were also low on response factor and high on variability in the CC.

#### V Blanks

All blanks reported non-detect for all compounds of interest.

#### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples, except one Dibromofluoromethane in one undiluted run. Due to the remoteness of this DMC to the compounds of interest, this one "out" is of no concern.

#### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

MS/MSD appeared in the run log in the batch but was not project specific and therefore only the summary sheet is in the report. It looked fine.

The laboratory control spikes were well reported and were acceptable.

#### VIII Regional QA/QC

Not Applicable

#### IX Internal Standards

All criteria were met.

#### X Target Compound Identification

The very high concentration samples showed poor ratio comparison in the non-diluted run. The compounds saturated the detector. They were fine in the diluted runs. The reference spectrum for 1,2-Dichloroethane includes the 78 ion for the co-eluting benzene. The ions for styrene are in the reference spectrum for o-xylene. The ions for butylbenzene are in the reference spectrum for 1,2-dichlorobenzene. This many extra peaks in reference spectra indicate that the reference spectra were taken from a mix of compounds and the reference spectra for the co-eluting compounds have not been removed. This could be a problem leading to misidentification. These were not compounds of interest for this project and therefore the reviewer does not see a problem. But these compounds were detected and if the data is used for tertiary purposes at a later time it could be a problem. No mis-identifications were noticed.

## XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The sample quantitation in the diluted runs had a gap in the quantitation ranges. The range of quantitation in the IC was 1 to  $25\mu g/l$ ; the 500X diluted runs had a range of 500 to  $12500\mu g/l$ . This allows compounds to be over range in the initial run and under range in the diluted run. This occurred for at least one compound of interest in each run that required high dilution. In one sample an overrange was diluted to a non-detect. This may

have been a known and accepted but is out of the reviewer's knowledge. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples. The dilutions were run immediately after the samples. The reviewer would have preferred to see the dilutions run prior to higher level samples and at least one blank after a high sample to show no carryover.

#### XII Tentatively Identified Compounds (TICs)

Not Applicable

#### XIII System Performance

No problems noted.

#### XIV Overall Assessment of Data

The data appears to be very good, with only minor problems (reference spectra and dilutions).



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

мемо

Subject: Validation of Water Samples Collected at Hanscom AFB 5/8/2001

Date: 6/17/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Two samples were collected on May 8, 2001 at Hanscom AFB, Massachusetts. These samples and two trip blanks were submitted on May 9, 2001 to Severn Trent Laboratories (STL) Tampa East Facility in Tampa, Florida for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest. Acetone and other ketones were of secondary interest.

It is the reviewer's opinion that the quality of the data was acceptable with the qualifiers discussed below, for all samples.



#### Sample Delivery Group B1E090142

#### **Volatile Organic Data**

#### I Preservation

The samples arrived at the lab promptly and were within the temperature criteria. The samples pH were indicated as checked by the analyst using the footnote "All aqueous samples had a pH < 2 unless otherwise noted" the reviewer feels an actual entry indicating the pH would be a more comforting indicator that it was actually checked. Preservation was indicated on the Chain-of-Custody (COC) for the three samples. The samples were analyzed within the Acid-preserved hold time criteria of 14 days. The trip blanks were not on the COC. The lab contacted the project leader and they were not analyzed.

## II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibrations and Analysis days. 8260B criteria are more stringent that CLP and therefore the 8260B criteria were used.

#### III Initial Calibration

All of the Chloroethenes and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 25 to 625 ng introduced to the column. Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs. Acetone was low on response factor and too high in variability.

#### IV Continuing Calibration

The Continuing Calibration (CC) was run in the 12-hour analysis window before any of the samples were analyzed. The CC had RFs above 0.05 for the Chloroethenes and the DMCs. The Relative Percent Deviations (RPDs) from the IC were below 15% for the Chloroethenes and the DMC. The ketones were also low on response factor in the CC.

#### V Blanks

All blanks reported non-detect for all compounds of interest.

#### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples. The listed criteria are not 8260B or CLP and are wider than either. The lab should note in the case narrative if in-house criteria are being used instead of method listed criteria.

#### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

MS/MSD appeared in the run log in the batch but was not project specific and therefore only the summary sheet is in the report. 1,1-Dichloroethene was out on one of the seven spiked samples. Trichloroethene was in range and looked good for all spiked samples.

The laboratory control spikes were well reported and were acceptable.

#### VIII Regional QA/QC

Not Applicable

#### IX Internal Standards

All criteria were met.

#### X Target Compound Identification

The reference spectrum for 1,2-Dichloroethane includes the 78 ion for the co-eluting benzene. The ions for styrene are in the reference spectrum for o-xylene. The ions for butylbenzene are in the reference spectra for 1,2-dichlorobenzene. This many extra peaks in reference spectra indicate that the reference spectra were taken from a mix of compounds and the reference spectra for the co-eluting compounds have not been removed. This could be a problem leading to misidentification. These were not compounds of interest for this project and therefore the reviewer does not see a problem. But these compounds were detected and if the data is used for tertiary purposes at a later time it could be a problem. No mis-identifications were noticed.

## XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The sample quantitation in the diluted runs had a gap in the quantitation ranges. The range of quantitation in the IC was 1 to  $25\mu g/l$ ; the 500X diluted runs had a range of 500 to  $12500\mu g/l$ . This allows compounds to be over range in the initial run and under range in the diluted run. This may have been a known and accepted but is out of the reviewer's knowledge. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples. The dilutions were run immediately after the samples. The reviewer would have preferred to see the dilutions run prior to higher level samples and at least one blank after a high sample to show no carryover.

#### XII Tentatively Identified Compounds (TICs)

Not Applicable

#### XIII System Performance

No problems noted.

#### XIV Overall Assessment of Data

The data appears to be very good, with only minor problems (reference spectra and dilutions).



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

мемо

Subject: Validation of Water Samples Collected at Hanscom AFB 7/11-12/2001

Date: 5/6/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Eleven samples were collected on July 11and 12, 2001 at Hanscom AFB, Massachusetts. These samples were submitted on July 13, 2001 to Severn Trent Laboratories (STL) Tampa East Facility in Tampa, Florida for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest. Acetone and other ketones were of secondary interest.

It is the reviewer's opinion that the quality of the data was acceptable with the qualifiers discussed below, for all samples.



#### Sample Delivery Group B1G130213

#### **Volatile Organic Data**

#### I Preservation

The samples arrived at the lab promptly and were within the temperature criteria. The samples pH were indicated as checked by the analyst using the footnote "All aqueous samples had a pH < 2 unless otherwise noted" the reviewer feels an actual entry indicating the pH would be a more comforting indicator that it was actually checked. Preservation was indicated on the Chain-of-Custody (COC) for all the samples. The samples were analyzed within the Acid-preserved hold time criteria of 14 days. Sample HANGW-IRZ-INJ had air bubbles in the sample collection vials. This could adversely affect the data.

## II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibrations and Analysis days. 8260B criteria are more stringent that CLP and therefore the 8260B criteria were used.

#### III Initial Calibration

All of the Chloroethenes and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 25 to 625 ng introduced to the column. Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs. The ketones were low on response factor and too high in variability. The Ketone data should not be used for precise quantitation.

#### IV Continuing Calibration

The Continuing Calibration (CC) was run in the 12-hour analysis window before any of the samples were analyzed. The CC had RFs above 0.05 for the Chloroethenes and the DMCs. The Relative Percent Deviations (RPDs) from the IC were below 15% for the Chloroethenes and the DMC. The ketones were also low on response factor and high on variability in the CC.

#### V Blanks

All blanks reported non-detect for all compounds of interest.

#### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples. The specified criteria on the reports is neither CLP nor 8260B. The recoveries look fine but if in-house recovery limits are being used it should be noted in the case narrative.

#### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

MS/MSD appeared in the run log in the batch but was project specific. It looked fine. The trichloroethene was unusable because of the high levels in the native samples.

The laboratory control spikes were well reported and were acceptable.

#### VIII Regional QA/QC

Not Applicable

#### IX Internal Standards

All criteria were met.

#### X Target Compound Identification

The very high concentration samples showed poor ratio comparison in the non-diluted run. The compounds saturated the detector. They were fine in the diluted runs. The reference spectrum for 1,2-Dichloroethane includes the 78 ion for the co-eluting benzene. The ions for styrene are in the reference spectrum for o-xylene. The ions for butylbenzene are in the reference spectrum for 1,2-dichlorobenzene. This many extra peaks in reference spectra indicate that the reference spectra were taken from a mix of compounds and the reference spectra for the co-eluting compounds have not been removed. This could be a problem leading to misidentification. These were not compounds of interest for this project and therefore the reviewer does not see a problem. But these compounds were detected and if the data is used for tertiary purposes at a later time it could be a problem. No mis-identifications were noticed.

## XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The sample quantitation in the diluted runs had a gap in the quantitation ranges. The range of quantitation in the IC was 1 to  $25\mu g/l$ ; the 500X diluted runs had a range of 500 to  $12500\mu g/l$ . This allows compounds to be over range in the initial run and under range in the diluted run. This occurred for at least one compound of interest in each run that required high dilution. This may have been a known and accepted but is out of the reviewer's

knowledge. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples. The dilutions were run immediately after the samples. The reviewer would have preferred to see the dilutions run prior to higher level samples and at least one blank after a high sample to show no carryover.

#### XII Tentatively Identified Compounds (TICs)

Not Applicable

#### XIII System Performance

No problems noted.

#### XIV Overall Assessment of Data

The data appears to be very good, with only minor problems (reference spectra and dilutions).



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

ΜΕΜΟ

Subject: Validation of Water Samples Collected at Hanscom AFB 11/19-20/2001

Date: 8/8/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Eleven samples were collected on November 19<sup>th</sup> and 20<sup>th</sup>, 2001 at Hanscom AFB, Massachusetts. These samples and a trip blank were submitted on November 20<sup>th</sup>, 2001 to Severn Trent Laboratories (STL) Savannah Facility in Savannah Georgia for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest. Acetone and other ketones were of secondary interest.

It is the reviewer's opinion that the quality of the data was barely acceptable with the qualifiers discussed below.



#### Sample Delivery Group: HAFB02

#### **Volatile Organic Data**

#### I Preservation

There is no indication in the report that the samples were checked for Temperature, headspace or pH upon arrival at the lab. The run log has a note "see log in" under the pH column. No login sheet with pH information was found. Preservation was indicated on the Chain-of-Custody (COC) for all the samples. The samples were analyzed within the Acid-preserved hold time criteria of 14 days.

## II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibrations and Analysis days. The lab used CLP BFB abundance criteria not 8260B. The tune checks passed the CLP Criteria but not the 8260B. 8260B criteria are more stringent that CLP. This difference is minor and would not cause question on the data. The lab should be more careful to use the criteria from a method if it is quoted as the method being done.

#### III Initial Calibration

All of the Chloroethenes, Ketones, and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 5 to 1000 ng introduced to the column. (ketones 10 to 2000) Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs. Some ketones were too high in variability. The Ketone data should not be used for precise quantitation.

#### IV Continuing Calibration

The Continuing Calibration (CC) was run in the 12-hour analysis window before any of the samples were analyzed. The CC had RFs above 0.05 for the Chloroethenes, Ketones, and the DMCs. The Relative Percent Deviations (RPDs) from the IC were below 15% for one of the DMC and half of the Chloroethenes. Trichloroethene was at 26% deviation. Looking at the trends it appears that the standard may have been poorly made up. Most of the compounds are in the 15 to 25% deviation range. The CCC compounds from 8260B passed the 20% requirement. But since the compound of main interest was trichloroethene the calibration should have been rerun.

#### V Blanks

All blanks reported non-detect for all compounds of interest.

#### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples. The specified criteria on the reports are neither CLP nor 8260B. The recoveries look fine but if in-house recovery limits are being used it should be noted in the case narrative. The stated limits are much wider than 8260B and are greater than what was being used in the previous laboratory on this project.

#### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

MS/MSD appeared in the run log in the batch and was project specific. There were apparent compounds that had been spiked and they were of similar concentration, but there was no information on spike levels or recovery so evaluation could not be done. The matrix spike and Dup have several compounds listed that appear to be off of retention time. They are not the usual spiked compounds so they may have just not been deleted during QC. These off-time compounds will not effect the usefulness of the data.

#### VIII Regional QA/QC

Not Applicable

#### IX Internal Standards

All criteria were met.

#### X Target Compound Identification

No mis-identifications were noticed.

## XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples.

#### XII Tentatively Identified Compounds (TICs)

Not Applicable

### XIII System Performance

No problems noted.

#### XIV Overall Assessment of Data

The data appears to be good, but with some problems.



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

мемо

Subject: Validation of Water Samples Collected at Hanscom AFB 3/27-28/2002

Date: 7/26/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Eleven samples were collected on March 27<sup>th</sup> and 28<sup>th</sup> 2002 at Hanscom AFB, Massachusetts. These samples were submitted on March 29<sup>th</sup> 2002 to Severn Trent Laboratories (STL) Savannah Facility in Savannah Georgia for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest. Acetone and other ketones were of secondary interest.

It is the reviewer's opinion that the quality of the data was acceptable with the qualifiers discussed below, for all samples.



#### Sample Delivery Group: HAFB06

#### **Volatile Organic Data**

#### I Preservation

There is no indication in the report that the samples were checked for Temperature or pH upon arrival at the lab. Preservation was indicated on the Chain-of-Custody (COC) for all the samples. The samples were analyzed within the Acid-preserved hold time criteria of 14 days. Two samples did not have analysis requested on the original COC but it is assumed that the lab discussed this with the project officer (but not noted) because the copy of the COC has indications that are not on the original COC. Both were in the report.

## II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibrations and Analysis days. The lab used CLP BFB abundance criteria not 8260B. The tune checks passed the CLP Criteria but not the 8260B. 8260B criteria are more stringent that CLP. This difference is minor and would not cause question on the data. The lab should be more careful to use the criteria from a method if it is quoted as the method being done.

#### III Initial Calibration

All of the Chloroethenes, Ketones, and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 5 to 1000 ng introduced to the column. (ketones 10 to 2000) Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs. Most ketones were too high in variability. The Ketone data should not be used for precise quantitation.

#### IV Continuing Calibration

The Continuing Calibration (CC) was run in the 12-hour analysis window before any of the samples were analyzed. The CC had RFs above 0.05 for the Chloroethenes, Ketones, and the DMCs. The Relative Percent Deviations (RPDs) from the IC were below 15% for the DMC and half of the Chloroethenes. Half of the Chloroethenes were between 15 and 20%. The ketones were also high on variability in the CC with only half of the target ketones better than 15%.

#### V Blanks

All blanks reported non-detect for all compounds of interest. One lab blank had 123trichloropropane at one half the lowest calibration level but that compound is not one of the compounds of interest.

#### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples. The specified criteria on the reports are neither CLP nor 8260B. The recoveries look fine but if in-house recovery limits are being used it should be noted in the case narrative. The stated limits are much wider than 8260B and are greater than what was being used in the previous laboratory on this project. The recoveries would have failed about half the time if the method specified limits were used.

#### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

MS/MSD appeared in the run log in the batch but was project specific. It looked fine. The laboratory control spikes were well reported and were acceptable.

#### VIII Regional QA/QC

Not Applicable

#### IX Internal Standards

All criteria were met.

#### X Target Compound Identification

No mis-identifications were noticed.

## XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project. The lab is to be commended for reporting all detected values in both the initial and diluted runs.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples.

#### XII Tentatively Identified Compounds (TICs)

Not Applicable

XIII System Performance

No problems noted.

#### XIV Overall Assessment of Data

The data appears to be very good, with only minor problems.



ARCADIS G&M, Inc. 4915 Prospectus Dr. Suite F Durham North Carolina, 27713 Tel 919 544 4535 Fax 919 544 5690

мемо

Subject: Validation of Water Samples Collected at Hanscom AFB 10/15/2002

Date: 11/12/02

From: Dennis Tabor

Department:

Drafted by:

ARCADIS Project No.: RN009900.0017.00001

To: Chris Lutes Copies:

Five samples were collected on October 15<sup>th</sup>, 2002 and Two samples were collected on October 16<sup>th</sup>, 2002 at Hanscom AFB, Massachusetts. These samples and a trip blank were received on October 16<sup>th</sup> and 17<sup>th</sup>, 2002 to Severn Trent Laboratories (STL) Savannah Facility in Savannah Georgia for analysis. This validation covers the samples submitted for Volatile organics by United States Environmental Protection Agency (USEPA) Method SW-846 8260B. Other aliquots were analyzed for other parameters that were not discussed in this memo.

Validation of this data was performed following the quality assurance/quality control (QA/QC) criteria set forth in the "USEPA Contract Laboratory Program (CLP) National Functional Guidelines for Organic Data Review", revised in June 2001. Method 8260B has slightly different criteria for some parameters than the CLP. Therefore, when the criteria were different and the 8260B criteria were more stringent the 8260B criteria were used.

This project is focused on the chlorinated ethenes. Trichloroethene was the primary compound of interest. Acetone and other ketones were of secondary interest.

It is the reviewer's opinion that the quality of the data was very good. With the exception that ketones in one sample were misidentified as present. The compounds do not appear to be the targets in question but



are there and large. When requested the laboratory evaluated the data again and agreed that the compounds were not the targets and removed them from the report.

#### Sample Delivery Group: HAFB10

#### **Volatile Organic Data**

#### I Preservation

There is no indication in the report that the samples were checked for Temperature, headspace or pH upon arrival at the lab. No login sheet with pH information was found. Preservation was indicated on the Chain-of-Custody (COC) for all the samples. All but one of the samples were analyzed within the non-acid-preserved hold time criteria of 7 days. The last sample was analyzed within the Acid-preserved hold time of 14 days.

## II Gas Chromatography/ Mass Spectrometer (GC/MS) Instrument Performance Check

Bromofluorobenzene checks were done successfully on the Initial Calibrations and Analysis days. The lab used CLP BFB abundance criteria not 8260B. The tune checks passed the CLP Criteria and the 8260B. 8260B criteria are more stringent that CLP. This difference is minor and would not cause question on the data. The lab should be more careful to use the criteria from a method if it is quoted as the method being done or specify what criteria are being used.

#### III Initial Calibration

All of the Chloroethenes, Ketones, and the Deuterated Monitoring Compounds (DMCs) had Response Factors above 0.05in the Initial Calibration (IC). The range was from 5 to 1000 ng introduced to the column. (ketones 10 to 2000) Relative Standard Deviations (RSDs) were below 15% for the Chloroethenes and the DMCs.

#### IV Continuing Calibration

The Continuing Calibration (CC) was run in the 12-hour analysis window before any of the samples were analyzed. The CC had RFs above 0.05 for the Chloroethenes, Ketones, and the DMCs. The Relative Percent Deviations (RPDs) from the IC were below 15.

#### V Blanks

All blanks reported non-detect for all compounds of interest.

#### VI Deuterated Monitoring Compounds (DMCs)

All criteria were met for all samples. The specified criteria on the reports are neither CLP nor 8260B but would has easily passed them. The recoveries look fine but if in-house recovery limits are being used it should be noted in the case narrative. The stated limits are much wider than 8260B.

#### VII Matrix Spike/ Matrix Spike Duplicates (MS/MSDs)

The MS/MSD was not project specific. So evaluation could not be done.

#### VIII Regional QA/QC

Not Applicable

#### IX Internal Standards

All criteria were met.

#### X Target Compound Identification

2-Butanone and 4-methyl-2-Pentanone were mis-identified in Sample HAN-GW-IRZ-INJ From 10/15/02. When requested the laboratory evaluated the data again and agreed that the compounds were not the targets and removed them from the report. No other mis-identifications were noticed.

## XI Compound Quantitation and Reported Contract Required Quantitation Limits (CRQLs)

Reporting limits were fine for this project.

The lab appears to have had prior knowledge of which samples would require dilution or screened the samples.

#### XII Tentatively Identified Compounds (TICs)

Not Applicable

#### XIII System Performance

No problems noted.

#### XIV Overall Assessment of Data

The data appears to be good. With the exception that ketones in one sample were misidentified as present. The compounds do not appear to be the targets in question but are there and large.

# Appendix A-6c

QC Notes

## Appendix A-6c: Summary of QA/QC Observations – ESTCP/AFCEE IRZ Project

#### Hanscom AFB

#### Well Installation May 2000

*Field Duplicates in soil sampling:* The field crew initially failed to collect the required field duplicate. The laboratory was given instructions to analyze two of the multiple containers collected for one sample to constitute the field duplicate.

#### First Full – Background Sampling Round – June 2000

*Hydrogen:* The lab (Vaportech) did not receive any vial labeled HAN-SG-IRZ-1-1 as indicated on the custody sheet. They received 2 vials labeled HAN-SG-IRZ-INJ-1. There were no further clues to identity on the vials. Both of the vials had the same value when analyzed >50nM/l, therefore the error was inconsequential.

*Field blank* (HAN-GW-FB-1) 6/12/00 15:15 gave a DOC at 9.5 mg/l and COD at 185 mg/l suggesting the presence of some organics. This field blank was prepared using distilled water purchased at a supermarket, which may be subject to some organic leaching from the container.

*Ferrous Iron:* Field kits were used for the 0-1 and 1-10 mg/l ranges, however many wells were reported as >10 mg/l. Note that a laboratory measurement is available however so this is not a major impediment to data interpretation.

*DO*: Problems were experienced with the original YSI probe for DO leading to a switch to the Horiba and repeat of some measurements. We also asked Vaportech to repeat this measurement offsite as a backup.

*ORP, Injection Well:* Two instruments were used. The Horiba data was 200 mv and the YSI –33.4. The Horiba data is reported for consistency with other measurements but the YSI value is more consistent with other wells measured at that time. For other wells the Horiba and YSI values agree fairly well for this parameter.

#### December 2000

A discussion was held with the offsite analytical laboratory when it was noted that the cis and trans DCE did not add up to the total DCE. They explained that unique calibration curves were used for these three analyses so they might not add up exactly. In short the response factor for "Total" is a function of the "cis" and "trans" response factors. Since our reporting is almost always in terms of the individual isomers this was judged to be unimportant

#### April 2001 abbreviated monitoring

A discussion was held with the offsite analytical laboratory in May when a difference of more then an order of magnitude was noted between two analyses of the same CAH analyte at differing dilutions in an injection well sample. The laboratory reviewed the situation and reported that one of the analyses was being effected by a positive interference which was then taken into account in data analysis.

#### May 2001 Full Monitoring Round

A non-detect for bromide at an elevated detection limit in an injection well sample was discussed with the offsite analytical laboratory. It was determined that a high chloride value made dilution necessary.

#### July 11, 2001 Process Monitoring

An unusual conductivity reading was investigated and determined to be due to a failure by the field staff to note an automatic change in units from ms/cm to s/m on the instrument in bright sunlight, the data was revised.

#### **October 12, 2001 Process Monitoring**

Tom Best of Hanscom reported that some samples from another area of the facility analyzed in the same batch with our samples showed anomalous results based on his detailed site knowledge. IT Corporation resampled some of the wells on October 19<sup>th</sup>. Since these results were more consistent with expectations they were used.

TOC results from the offsite lab were only 2.9 mg/l although the well was described in the field as "light brown" and with an odor of molasses. The laboratory pulled out the samples and determined that the TOC fraction was clear and colorless. The offsite lab was instructed to measure the TOC on the fraction that was originally collected for Bromide analysis. Although this sample was colored a TOC of only 3.4 mg/l was seen. The lab then attempted to determine if the TOC preservation procedure was the cause of the problem. To do this the acidified the Br sample but it did not change color. It was inferred that biodegradation may be continuing in the sampled, preserved TOC samples.

#### December 2001

The offsite lab was requested to confirm results after recent DOC values were observed to be higher then TOC. The results were confirmed by reanalysis.

#### Process Monitoring January 22, 2002:

Upon review by the project engineer and manager pH data were higher by 1-2 units then the expected value and one reading was far off scale. Since records indicated that the injection well was measured first and had a very high TOC loading it was suspected that it fouled the electrode. The results were discarded and the field crew was advised on methods to avoid this problem in the future.

#### **Process Monitoring June 6, 2002:**

An abnormally high DO value measured in this round in the injection well was investigated and attributed to fouling by the very high concentration of organics present.

#### Final Full Sampling Round October 14-16, 2002

The field staff initially failed to report a complete set of visual and olfactory data as had been requested. However they were able to provide it in an email several days later.

Results for DO and pH for this monitoring round collected in the field appeared suspect upon review by the project engineer and manager because they fell out of expected ranges. DO data also did not agree with the data from the off-site laboratory within reasonable tolerances for some samples that were measured with one particular meter. It was then determined that the field staff had failed to adequately follow calibration procedures on this occasion, a failure they attributed to a vendor not providing expected materials. The field staff were reminded of proper procedures and ARCADIS repeated these measurements at its expense on October 29, 2002. During that event calibrations were properly performed and values recorded within expected ranges for nearly all parameters. However the injection well water was visibly effervescing and gave a reading that was off-scale even after meter recalibration.

# Appendix A-7

Additional Discussion of Hydrogeologic Evaluations

#### Appendix A-7: Additional Discussion of Hydrogeological Evaluations

#### Changes in Groundwater Flow Direction Due to Changes in Pumping Rates

The ERD at Hanscom exhibited changes in groundwater flow that were not expected and led to complex patterns in substrate delivery and CAH concentration as discussed above. The most probable explanation for these changes is a combination of multiple sources and changing pumping patterns imprinted on top of a complex geology with relatively thin aquifer zones.

The geology of the area is a glacial valley fill on top of granitic bedrock. The general sediment sequence at the area of the IRZ/ERD demonstration from top down is a layer of fine sand on top a discontinuous layer of clay on top of a sandy till (see Figures 3-3 and 3-4). This sequence is on top of fractured granite, with some fractures filled with silt. The most hydraulically conductive layer is the lower sandy till with a hydraulic conductivity of 1 to 25 feet per day. The active IRZ/ERD is within this layer of sandy till. The fine sand above and the fractured bedrock are about a tenth as conductive as the till. The discontinuous clay is considerably less conductive than the till. All three of the sedimentary units are not present over the entire site, and all have areas of zero thickness (Haley and Aldrich, September, 1998). The granitic bedrock is present over the entire area of interest. The general flow direction in the three conductive units, the upper sand, the basal sandy till, and the fractured bedrock, is basically east (CH2M Hill, 1997).

Even this generalization about the "sandwich" of three sediment types is oversimplified. While at the area of the IRZ/ERD the general sequence is found, Haley and Aldrich, 1998, report that at Hartwell's Hill, the upper two units are absent and only the lower sandy till is present. Given the changes in thickness and in continuity of all three layers, CH2M Hill (1997) chose to model the area in horizontal layers with considerably different hydraulic conductivities in cells in the same layer. While this approach is reasonable, it does not allow the interpretation of the results of the Hill modeling to be taken to site-specific instances within the framework of the three recognized layers in the unconsolidated and the one bedrock layer. In simple terms, the output of the Hill model is not easily understood as to the hydrogeological unit that is presented. Nonetheless the Hill model is useful tool for a gross understanding.

The important features of the hydrogeology are:

- The most productive unit, the sandy till, is semi-confined over the entire area and in the area of active IRZ/ERD, and
- it is at most about 20 feet thick.

The confinement means that changes in withdrawals are expected to show effects much more quickly than in unconfined conditions. The quickness of reaction over larger areas is the result of the fact that in confined aquifers, unlike unconfined aquifers, the removal of large amounts of water is not needed to affect changes in head over relatively large areas because head is being transmitted, not water. There are two recovery trenches, numbers one and two, and five recovery wells, numbers BIW1 through BIW-4 and BIW-6, in the area of the IRZ/ERD (see Figure 3-1 for the location of these features and Figure 4-28 for the pumping rates). Both recovery trenches are in the upper sand only. Recovery Trench 1 is just to the northwest of the IRZ/ERD area, and Recovery Trench 2 is to the southeast. But, as both penetrate only the upper sand and not the lower till, their only effect is to deny possible leakage through the confining clay in the case of Trench 1. Trench 2, which is basically downgradient, would have little or no effect on the demonstration. Little impact is discernable from the two to three-fold changes in recovery at the two trenches in 2000 and 2001.

Four of the five recovery wells are arranged in an arc to the north and east of the IRZ/ERD area (Figure 3-1). BIW-1 is just about 1300 feet due north of the IRZ/ERD area. BIW-2 is about 1250 feet northeast. BIW-4 is about 1300 feet east, and BIW-3 is about 1500 feet southeast. In all of these wells (BIW-1 through 4), the sand is well confined by lacustrine clay that is apparently reasonably widely distributed in the areas of the wells. All of these wells, except BIW-1 extract about two-thirds of their water from the lower sandy till and the rest from the bedrock. In BIW-1, this ratio is reversed (CH2M Hill, 1997). In BIW-1, the lower unconsolidated productive zone is 12 feet thick, and the screen is set with 8 feet open to that zone and 53 feet open to the bedrock. Additionally, the contact of till and granite is at the 100-foot elevation, 30 feet higher than at the IRZ/ERD area. In contrast, at BIW-2 through 4, the lower productive unconsolidated zone is much lower in elevation, ranging from 60 feet to below 40 feet, or 10 to 30 feet lower than in the IRZ/ERD area. Therefore, because of the distance from the IRZ/ERD and the elevation of the base of the unconsolidated deposits, the effect of pumping at BIW-1 will be less than at BIW-2 through 4 on groundwater flow in the IRZ/ERD area. BIW-6 is about 500 feet west-northwest of the demonstration zone. It was installed in 1997, after the data collected for the CH2M Hill modeling work, and withdraws at a relatively constant flow rate from the bedrock aquifer (Hanscom AFB, 2002; see also Figure 4-28).

Groundwater withdrawal from BIW-2 has been relatively constant at about 275,000 gallons per month respectively. Pumpage at BIW-1 has not been as constant, but generally is about 750,000 to 850,000 gallons per month. Pumping from BIW-4 was fairly constant at 500,000 gallons per month until the pump was changed and the flow increased to around 1,100,000 gallons per month in November 2001. It stayed at this higher level through the end of the demonstration except for August and September 2002 when it was much lower. The withdrawal from BIW-3, however, due to a variety of operational reasons, has ranged from 0 gallons per month in September and October 2000, and again in May 2001, to over 2,000,000 gallons per month for all of the demonstration period in 2002 except for slight dips in July and August. It is thus likely that the production changes in BIW-3 and BIW-4 had the biggest effect on the direction of groundwater flow in the basal sandy till in which the demonstration occurred.

In 2000, withdrawal from BIW-3 from January to late August averaged approximately 850,000 to 900,000 gallons per month. In September and October 2000, no water was withdrawn. Around 600,000 gallons were withdrawn in November 2000 and more than 2,000,000 gallons in each of the next four months. If the potentiometric map from mid-June 2000 (Figure 4-9) is compared with those of December 2000 (Figure 4-10) and January 2001 (Figures 4-13 and 4-14), a marked shift in flow direction is seen from east to southeast and back to east. Since no potentiometric surface is available for September and October 2000, it is likely that even more marked changes occurred that were not observed.

In 2001, BIW-3 generally pumped between 1,700,000 and 2,500,000 gallons per month but was off in May, and the potentiometric surface observed in May (Figure 4-11) clearly changed from that seen in April (Figure 4-16). In April, the head contours were basically aligned north-south and the flow was east or east by northeast. In contrast in May, the flow was to the northeast, when the pump at BIW-3 was not operating. After returning the pump to operation, the flow eventually returned to a more easterly direction (Figures 4-17 through 4-19 and 4-12).

Finally, from the pumping perspective, the production from well BIW-4 was increased markedly starting in October 2001. The production at the well jumped from roughly 500,000 to 600,000 gallons per month to 1,120,000 and 1,240,000 gallons per month in November and December 2001, respectively. This increase also would swing the flow more directly easterly, as BIW-4 is directly east of the IRZ/ERD area.

In 2002, pumping rates were quite constant from January to June and flow appeared to be consistent in the easterly direction from (Figures 4-20 through 4-22). The rate in BIW-3 decreased somewhat in July and August due to maintenance problems. The rate in BIW-4 was sharply lower in August and September. As shown in Figure 4-23, this appears to have increased the relative influence of the upgradient well BIW-6, resulting in a flow divide between easterly and westerly flow being located somewhere under the runway outrun between B239 and IRZ-INJ for a time. Flows had returned to a normal easterly direction in September through October 2002 (Figures 4-24 through 4-27).

These changes in flow direction no doubt affected substrate delivery and thus treatment efficiency. The basic assumption made in setting up the injection and monitoring wells is that the direction of groundwater flow and hence contaminant and substrate transport is relatively stable in a southeasterly direction. As discussed, the maps of head in the basal sandy till show that the orientation of the head contours has not been stable. In fact, that orientation has responded to the changes in pumpage in the basal sandy till, especially changes in BIW-3 and to a lesser degree in BIW-4. As discussed previously while the reduction in the parent compound, TCE, has generally been favorable in the two monitoring wells that fell within the reactive zone, there have been excursions from that trend. Two of the most notable were in late 2000, when a precipitous drop was followed by a sharp rise (see Figure 4-37), and in fall and winter 2001. The late 2000 changes came immediately after shutting off BIW-3 followed by very high rate of withdrawal from BIW-3 (0 in September and October, 655,000 gallons in November, and 2,080,000

gallons in December). The second occurred primarily in fall and winter 2001 and coincided with the increase in pumpage in from BIW-4 (from less than 630,000 gallons each month before September 2001 to more than 1,100,000 gallons per month from October 2001 to July 2002).

In summary, the changes in pumpage from wells BIW-3 and BIW-4 have induced changes in the flow regime at the IRZ/ERD area in the basal unit that help explain the variations in substrate delivery and treatment efficiency.

#### **Groundwater Velocity Estimation**

Based on preexisting data we estimated the groundwater flow velocity at the site to be 0.8 ft/day (ARCADIS Geraghty & Miller 2000). Groundwater velocity observed in the demonstration area was calculated using three methods:

- a) Based on average bromide concentrations observed at the first line of observation wells (IRZ-1 and RAP1-6T) and the mass bromide loading rate at the injection well, a volumetric flow rate through the lower aquifer was estimated and used to derive a flow velocity. In this calculation the average bromide concentrations at IRZ-1 and RAP1-6T since their first appearances (1.9 mg/l and 5.9 mg/l, respectively) were used to calculate an average bromide concentration (3.9 mg/l) at the "flow window" described by these two wells. The "flow window" of 30 feet wide by 20 feet deep (the thickness of the lower aquifer) has an estimated effective porosity of 15%, giving a flow area of 30 feet x 20 feet x 0.15 = 90square feet. In the course of the demonstration, 4732 grams of bromide were injected with 20,075 gallons of liquid, for an average injection concentration of 62.3 mg/l bromide. The dilution ratio was calculated as 62.3 mg/l  $\div 3.9$  mg/l = 16. Therefore, 20,075 gallons x 16 = 320,685 gallons of liquid (groundwater plus injection fluid) passed through the flow window during the demonstration, over a period of 734 days. The daily flow rate through the window was 320,685 gallons  $\div$ 734 days = 436 gallons per day, or 58 cubic feet per day. This amount of flow through a 90 square foot flow area would occur at a velocity of 0.64 feet/day.
- b) The bromide arrival time at the first line of observation wells was used to calculate flow velocity. The bromide data show initial increases in concentration at these wells within a month of the first injection, and consistently low concentrations until early 2001, when substantial increases occur. Thus, both dates were used to estimate a range of velocity values.

The first bromide detections at IRZ-1 and RAP1-6T occurred within 27 days after the first injection (see November 7, 2000 data). This interpretation of arrival time is supported by the simultaneous increase in TOC at IRZ-1 and decreases in DO at IRZ-1, RAP1-6T and other monitoring wells. Alternately, the first arrival of higher bromide concentrations occurred between 112 and 166 days after the first injection. The midpoint of approximately 140 days was assumed to be the arrival time for this calculation. The distances of the downgradient wells from the injection well are approximately 45 feet (INJ-1) and 40 feet (RAP1-6T). Using the average travel distance of 42.5 feet, the maximum bromide velocity was at least 42.5 feet  $\div$ 27 days = 1.57 feet/day, and the minimum velocity was 42.5 feet  $\div$  140 days = 0.30 feet/day.

c) Measured or estimated values of hydrogeologic characteristics were used to calculate flow velocity based on a variation of Darcy's Law. The average linear groundwater flow velocity in the direction parallel to flow is given by:

$$v_x = \frac{KI}{n_e}$$

where:

 $v_x$  = average linear groundwater flow velocity (ft/day);

K = hydraulic conductivity (ft/day) – for the lower aquifer, ranges from 3 to 35 ft/day, based on slug test results (CH2M Hill, 1997)

I = hydraulic gradient (unitless) – typically 0.006 over the course of the demonstration; and

 $n_e$  = effective porosity (unitless) – estimated to be 0.15.

Using an average hydraulic conductivity of 19 feet/day, the flow velocity was calculated by this method to be 0.76 feet/day.

Thus these three methods provided estimates of velocity that agree fairly closely:

- a) 0.64 ft/day
- b) 0.30 to 1.57 ft/day
- c) 0.76 ft/day

Method b) is the most direct measure of velocity during the test. However, it may overestimate velocity during normal conditions because of the force of the initial injections. On the other hand it could underestimate velocity if the bromide tracer was not fully conservative (i.e., was adsorbed to soil particles or taken up in biomass or if the initial injection rate of molasses/ bromide was too low and thus became too diluted to provide a significant increase over the background concentration. Method a) probably overestimates velocity because it assumes all of the injected water travels through a 20x30 window. Method c) may also overestimate normal velocity because slug tests typically overestimate hydraulic conductivity. The values derived from the three methods are reasonably close, thus an average velocity during the demonstration period of approximately 0.80 ft/day is assumed.
# Appendix A-8

Additional Discussion of Secondary Water Quality Issues

#### Appendix A-8: Additional Discussion of Secondary Water Quality Issues

#### **Secondary Water Quality Issues**

We recognize that while the substrate injected (molasses) and its breakdown products are generally nontoxic, it may elevate certain parameters in the water within the reactive zone. For example, by definition, any substrate used to enhance anaerobic bioremediation will elevate the biological oxygen demand (BOD), a traditional measure of water quality. Furthermore, since we are intentionally creating reducing conditions within the reactive zone, this will necessarily alter the geochemistry of the reactive zone. This will make some soil mineral metals more mobile (more dissolved) and others less mobile (more inclined to the solid phase). Further information about these matters can be found in Sections 1.2, 2.1, 3.1 and 7 of the protocol document (Suthersan, 2002).

In general, it is believed that enhanced anaerobic in-situ bioremediation processes will reduce the mobility of many metals (indeed it has been successfully used for the treatment of many) but it will solubilize some other naturally occurring metals in the reactive zone (for example iron, manganese, and arsenic). This creates a potential secondary water quality impact. Other parameters of interest with regard to secondary water quality impacts are COD, BOD, TDS, and sulfides. However even in solubilized form under anaerobic conditions metals such as arsenic are substantially retarded by adsorption to the aquifer matrix. Furthermore it is generally believed that they will be reprecipitated/immobilized down-gradient of the reactive zone when the conditions return to their preexisting state (which for the purposes of this discussion is assumed to be aerobic). Similarly reprecipitation/immobilization will occur within the IRZ area some time after system shutdown. Finally, we note that these reducing conditions are by no means unique to IRZ systems – they occur for example at sites of TPH releases and landfills as well.

Finally, the substrate itself has been mentioned as a potential source of metals that may cause secondary water quality impacts. Molasses in its pure form contains concentrations of several metals that may exceed water quality criteria. In a dilute mixture, as is typically used in IRZ applications, the concentrations have been below regulatory standards. However, this is a potential issue that should be considered in the design phase.

Therefore, we agreed with ESTCP on a multi-step process to evaluate these issues in the context of the Hanscom demonstration:

1. <u>Review existing base monitoring data:</u> This was found to be of very limited utility since the only metals data available according to the restoration manager, Tom Best, are copper, lead and hardness in the upper aquifer, which is above our target zone. Furthermore, all of the available data was collected before the beginning of system operation. Thus it can only be an imperfect indicator of background for site groundwater.

- 2. <u>Review mineralogy:</u> Results of this effort are discussed below. This provides guidance for parameter selection and data interpretation but cannot be definitive.
- 3. <u>Review data gathered to date in ESTCP Demonstration on iron, manganese, COD</u> <u>and BOD:</u> This is discussed below.
- 4. <u>Expand the final monitoring round for this project</u> to include 12 total and 12 dissolved priority pollutant metals (including mercury) plus TDS in all 10 wells normally in use for this project plus well B242 (roughly 500 days downgradient of the injection point. The results of this approach are discussed below.
- 5. <u>Incorporate analysis of data gathered on a related project:</u> We are currently conducting a large pilot scale study and a related simultaneous bench scale column study for Fort Devens Massachusetts, an Army site. Fort Devens and Hanscom are approximately 25 miles apart but their mineralogy may be somewhat different. A key issue for the Fort Devens project, and the primary focus of the bench scale column study, is the mobilization of arsenic during treatment of CAHs. The column study is primarily focused on observing the rate of arsenic reprecipitation/immobilization under various conditions. The results of this analysis are discussed below.
- 6. Expand the rebound monitoring to include 12 total and 12 dissolved priority pollutant metals (including mercury) plus TDS in two wells (IRZ-1 and RAP1-6T) in the heart of the treatment zone (4 total and 4 dissolved metals samples with QA/QC).
- 7. <u>Review available data for metals in molasses</u>: Available data for molasses and molasses-water mixtures such as those used at the site were reviewed in conjunction with the review of subsurface metals impacts. The results of this review are discussed below.

## Secondary Water Quality – Implications of Mineralogy for Metals Available to be Released

Our system is operating in the lower sandy till aquifer, which is also known as the sublacustrine unit. The sublacustrine unit is primarily composed of granite, with lesser amounts of quartz diorite and gneiss (Haley and Aldrich, 1998). This unit has substantial background flows of moderately aerobic water. Based on this mineralogy we can conclude that in the bulk solid matrix:

- Fe, Mn are likely present in percent quantities
- Ba, Zn, Cr, V, Pb are likely present at 10-100 ppm (mg/kg)
- Cu, As, Co are likely present at 1-10 ppm (mg/kg)

This information was then used as guidance in developing strategies for further testing.

## Secondary Water Quality – Iron and Manganese – Time Series during Demonstration

Total iron and manganese were measured at the off-site laboratory in the full monitoring rounds (background, midpoint and final). The data in Figures 4-66 and 4-67 clearly show that these metals were elevated vs. the background well (B239) within the reactive zone (IRZ-1, RAP1-6T and the injection well) but were not elevated downgradient (i.e., IRZ-5).

#### Secondary Water Quality: All Metals – Final Full Sampling Round

Total and dissolved metals data for groundwater were collected for the purpose of evaluating the possibility of secondary water quality impacts from ERD implementation. The metals data are summarized in Tables 4-12 and 4-13. Samples for this evaluation were collected on October 14-16, 2002, approximately two years after the first molasses injection. Wells are listed in the table approximately in order from upgradient to downgradient, starting with background well B239-MW. However, it is noted that the wells are screened at different levels within the aquifer system. The majority of the listed wells, including B239-MW, the six IRZ wells, RAP1-6T and B242-MW, are screened in the lower aquifer, in which injections occurred. Screened intervals for RAP1-6S and RAP1-6R are in the upper aquifer and in bedrock, respectively.

In the summary tables, metals concentrations are compared to Federal drinking water standards or goals. These standards are not necessarily ARARs at this facility, but are provided to put the results in perspective.

A comparison of the total and dissolved metals results indicates little difference between the two. Their similarity suggests that the metals contained in the samples were predominantly dissolved, with minor amounts contained in suspended solids. Because of the similarity between the two data sets, the following discussions refer to both collectively, with exceptions as noted. In some cases, dissolved metals levels exceeded total levels slightly. These discrepancies are attributed to normal sample and analytical variability. Since manganese was collected primarily as a process monitoring parameter, the discrepancy is not considered to be critical.

A comparison of results for the background well to the injection well shows that concentrations of several metals were higher in the injection well [arsenic (As), beryllium (Be), chromium (Cr), copper (Cu), iron (Fe), lead (Pb), manganese (Mn), nickel (Ni), selenium (Se), zinc (Zn)], suggesting solubilization in the reactive zone. The magnitude of the increase must be presented with a note of caution, since many of the background concentrations were non-detects or otherwise qualified. With this in mind, metals levels at IRZ-INJ were as much as 90 times background levels.

However, concentrations of the majority of the (potentially) solubilized metals fell to background levels within a few months' travel time from the injection well (i.e., at IRZ-1 and RAP1-6T). At these two wells, still within the strongly reducing zone (or reactive zone), the only metals remaining at levels higher than background were As, Fe, Mn and possibly Cr. Among these, the magnitude of the difference was usually less than fivefold. Then, by the time the groundwater reached the wells directly downgradient from the reactive zone (IRZ-5 and B-242), metals levels were not elevated over the upgradient well (B239), indicating that they had successfully been reprecipitated or sorbed at the edges of the reactive zone.

A detailed discussion by metal is provided below:

- Arsenic levels were two to five times the background level at IRZ-1 and RAP1-6T, and at one further downgradient well (IRZ-3). (Interpretation of the dissolved arsenic data is inconclusive because of blank contamination). This distribution may indicate an elevation of arsenic mobility within the reactive zone, but the effect does not appear to be widespread or pronounced at greater distances (see also Figures 4-68 and 4-69). Importantly, the arsenic concentration downgradient of the reactive zone does not exceed the standard and in the two directly downgradient wells (IRZ-5 and B242) is not elevated over background. The range of arsenic detections may also reflect normal sampling and analytical variability. Additional background groundwater samples would be required to make this determination.
- Iron levels were three to five times higher at IRZ-1 and RAP1-6T than at the background well. Further downgradient, iron levels were comparable to background, suggesting that increased iron mobility was limited to the reactive zone.
- Manganese levels were slightly higher than background within the reactive zone, and below background at downgradient wells. As with iron, any increased manganese mobility appears to have been limited to the reactive zone.
- The average chromium concentration at RAP1-6T is slightly higher than the background concentration, but lower than at downgradient wells IRZ-4 and B242-GW (total chromium only). The differences likely reflect normal sampling variability since chromium is typically less mobile under reducing conditions.

At other levels of the aquifer above and below the injection zone, no background data are available for comparison. However, arsenic and iron levels at RAP1-6S were generally higher than in the lower aquifer, which may indicate some influence by the injections, or merely reflect differences in the background water quality between the aquifers.

In summary, the metals results indicate that secondary water quality impacts may occur within the reactive zone during implementation of ERD as a result of increased mobilization/ solubilization of some metals. However, the effect appears to be limited to the injection area and to the extent of the strongly reducing zone. The metals data supports the concept that the affected metals, including arsenic, iron and manganese, are reprecipitated/immobilized downgradient of the reactive zone when conditions return to preexisting (less reducing) state. Similarly, it is expected that reprecipitation/immobilization will occur within the IRZ area some time after system shutdown.

## Secondary Water Quality – Arsenic – Field and Bench Scale Observations from another DoD IRZ Site

Similar issues have been discussed during the implementation of pilot tests performed for DoD at another Massachusetts site. At the Devens Reserve Forces Training Area, ARCADIS conducted bench-scale and field-scale ERD pilot tests that were designed in part to test for arsenic mobilization. The following discussion is summarized from the Devens ERD Pilot Test Evaluation Report (ARCADIS, November 2002).

There appear to be three primary triggers that can cause the release/solubilization of geogenic arsenic, including development of high pH (greater than 8.5), the presence of high concentrations of competing anions (such as phosphate, bicarbonate, or silicate), and development of reducing conditions at circumneutral pH.

Within the anaerobic and reducing IRZ created by ERD technology, there is evidence that some control on arsenic solubility can be realized through the formation of low-solubility arsenic sulfide compounds. However, it is expected that the primary control on arsenic solubility will be provided by adsorption to and co-precipitation with hydrous ferric (iron) oxides under ambient oxidizing conditions.

Under the Devens site's normal aerobic groundwater conditions, both dissolved-phase arsenic and iron concentrations were below laboratory detection limits. In the field pilot, arsenic was solubilized in the pilot study area at levels greater than both the current and proposed MCLs for arsenic. However, field tests supported the expectation that the presence of soluble arsenic will be limited to the boundaries of reducing zones created by the ERD technology. Once the original aerobic and oxidizing poise of those reducing zones is restored, it is expected that dissolved arsenic will decrease to non-detectable levels. ERD application was therefore considered appropriate for treatment of CAHs provided the temporary presence of arsenic was appropriately monitored and managed.

In the bench-scale treatability study (flow-through column study), the initial aerobic poise of each of three soil columns was overcome by passing reduced groundwater containing dissolved concentrations of arsenic, iron, and manganese through the columns. Measurements of the three metals/metalloids and DO and ORP were recorded at intervals as the water was applied to the columns. After reducing conditions had been achieved, the aerobic poise of two columns was restored using two different oxidation techniques (air injection and hydrogen peroxide injection). Based on the treatability study results, the following observations were made:

- even under reducing conditions, the aquifer materials provided a significant level of control on arsenic solubility.
- the injection of air or hydrogen peroxide in the field can create an aerobic environment (most suitable for controlling arsenic solubility).

Thus, both empirical data from the Devens site and published research indicate that arsenic solubility as it relates to the use of ERD can be controlled, mitigating concerns associated with use of the technology.

#### Secondary Water Quality – BOD, COD and Sulfide

Time-series data for BOD and COD during the demonstration are presented in Table 4-4. In general, the data confirm that as expected elevated BOD and COD occurred at the injection well after injections began (BOD and COD are measures of injected molasses and its metabolic products just as TOC is). However, levels of both parameters were reduced by several orders of magnitude at the first line of observation wells (IRZ-1 and RAP1-6T), and returned to background levels at the second line of wells.

As shown in Figure 4-40, hydrogen sulfide shows a similar pattern – it is substantially elevated as would be expected under anaerobic (sulfate reducing) conditions in the injection well and RAP1-6T. Although spotty detections occur in other wells it stays below 0.1 mg/l in the two wells monitored most directly downgradient of the reactive zone (IRZ-5 and IRZ-3). This suggests that sulfide production is confined to the reactive zone and decreases rapidly downgradient.

#### Secondary Water Quality – Metabolic Byproduct VOCs

VOCs other than the target species for treatment (chlorinated ethenes and ethanes and trace carbon tetrachloride) that were detected during three full monitoring events are summarized in Table 4-11 along with potentially applicable regulatory standards. Among these VOCs of secondary interest are petroleum constituents (EDB, MTBE, toluene, xylenes) that are probably attributable to the historical fire training activities at the site, since they do not show a strong association with the reactive zone. Another species detected was methylene chloride, which is a common laboratory contaminant and rarely exceeded its MCL in the samples collected for this project.

Chloroform and carbon tetrachloride were also detected. Chloroform is a known disinfection byproduct and carbon tetrachloride is commonly found in tap water. Thus, they were likely introduced with the tap water used for injections (California Department of Health Services, Howard 1990). They were only rarely detected and never above their MCLs.

Dichlorodifluoromethane and styrene were each detected once, but well below their regulatory standards and thus are not of concern. 1,2,3-trichloropropane was also detected once, substantially above its regulatory standard. However given the rarity of its appearance it is probably not of great concern.

However the ketones (acetone, 2-butanone, 2-hexanone, MIBK) that were detected are probably byproducts of molasses biodegradation. Of these:

- Acetone and 2-butanone were present within the reactive zone, consistent with the idea that both are metabolic by-products. Neither persisted further downgradient and even within the reactive zone the concentrations were below regulatory levels.
- 2-Hexanone and MIBK, also possible metabolic byproducts, were present only at IRZ-5 directly downgradient of the zone. However they were well below their regulatory standards.

Thus, in summary, the risks posed by these expected metabolic byproducts of the degradation of food grade carbon sources are very low in comparison to the risks posed by the chlorinated constituents that are targeted for remediation (note that these metabolic byproducts rarely exceeded their regulatory levels, and as shown in Section 4.3.3.3.6, the chlorinated compounds, especially vinyl chloride, were present at several hundred times their regulatory levels!

#### Secondary Water Quality – TDS

TDS data for October 2002 are presented in Table 4-10. The data show that two years into the demonstration, with injections ongoing, TDS is elevated in the injection area. The October 2002 TDS concentration at IRZ-INJ exceeds the Federal Secondary Drinking Water standard for TDS of 500 mg/l. At the first transect of observation wells downgradient from the injection area (IRZ-1 and RAP1-6T), TDS levels are lower by two orders of magnitude. TDS levels at the farthest downgradient IRZ wells and MW-B242 range from 150 to 220 mg/l and are comparable to the background level of 130 mg/l. Thus like many of the other secondary water quality parameters TDS is elevated in the reactive zone but dramatically decreased downgradient.

The ratio of TDS to specific conductance, which in natural waters is expected to fall within the range of 0.55 to 0.86 (Friedman and Erdman, 1982), is elevated at IRZ-INJ and RAP1-6S. At the injection well, and possibly at RAP1-6S, the ratio is likely disrupted by the presence of the substrate. Since both TDS and specific conductance were collected primarily as process monitoring parameters, the discrepancy is not considered to be critical.

#### Secondary Water Quality – Metals in Molasses

Molasses in its pure form contains concentrations of several metals that may exceed water quality criteria. Published analyses of blackstrap molasses (US Sugar, 2001) and analyses of metals and chloride in molasses/water mixtures by ARCADIS are presented in Tables 4-14 and 4-15. The ARCADIS metals sample was from a commercial remediation site in Ohio where a different molasses source was used than at Hanscom, but the results should be similar for Hanscom. Also note that the water-to-molasses mixture used at the Ohio site was slightly more dilute than the Hanscom mixture. None of the metals detected exceeded available Federal MCLs, and would not if adjusted to match the more concentrated Hanscom mixture. In addition, the site metals data discussed above encompasses any solute quality issues. On the basis of this evidence, we would not typically expect to see water quality impacts from the molasses injectate. However, this is a potential issue that should be considered in the design phase. The paucity of available data suggests that further work should be done in this area.

# Appendix A-9

Data Archiving and Demonstration Plans

#### **Appendix A-9. Data Archiving and Demonstration Plans:**

#### **1.0 Data Formats**

There are four broad classes of data that were collected in this project.

- 1. The field measurements of groundwater parameters (generally for biogeochemical conditions) and other field observations
- 2. Laboratory measurements of soil and groundwater parameters, both contaminants and indicators of biogeochemical conditions
- 3. Engineering data on the design and operation of treatment systems
- 4. Economic data on the treatment systems.

**1.1 Class I Data.** Class I data was recorded on standardized field forms, including groundwater sampling form, photograph log, daily log, well construction log, sample/core log, water sampling log, reagent injection log, chain of custody record, Hach analytical log, and soil core/sampling log as found in Appendix D of the demonstration plan.

Chain-of-custody procedures were followed as described in Sections 5.4.3.3 and 6.1 of the demonstration plan. A field log may also be used to supplement the forms with notes and drawings describing the location, field conditions, and method of sample collection and identification.

**1.2 Class II Data.** Class II data will generally be received in the form of formal reports from the analytical laboratories. Note however, ARCADIS also received almost all of the analytical data on concentrations in the form of electronic deliverables.

**1.3 Class III Data.** Class III data is of two types. Engineering designs were documented in AutoCAD files and printouts. Field operating data such as reagent doses, flow rates, and concentrations will be documented on standard forms in a manner similar to Class I data.

**1.4 Class IV Data.** Procedures for collection of class IV data are discussed in Section 7 of the demonstration plan.

#### 2.0 Data Storage and Archiving Procedures

**2.1 Class I Data Storage**. Class I data form originals will be retained in the office local to the site (the Andover office near Boston, Massachusetts). The field note originals will be maintained at the location of our Andover, Massachusetts, office in the building file storage locker. Copies of these documents are maintained in the file storage warehouse associated with the Durham, North Carolina, office.

**2.2 Class II Data Storage**. Class II data, as discussed above, were received and stored in both paper and electronic formats, initially at a central project archive to be maintained in the Durham, North Carolina, office of ARCADIS. It is also anticipated that the

analytical laboratories involved will maintain their own copies of this data set for a period of years. However, this cannot be relied upon since firms in the analytical laboratory business have a history of rapid change.

**2.3 Class III Data Storage.** Class III engineering design data will be archived to the central project file following preparation. Copies will also be maintained in any office preparing engineering designs. Field operating data will be handled in the same manner as class I data.

**2.4 Class IV Data Storage.** Class IV data will be collected as discussed in Section 7 of the demonstration plan and archived to the central project file in the Durham, North Carolina, office of ARCADIS.

2.5 Archiving Procedures. The central hardcopy project archives at the Durham, North Carolina, office of ARCADIS will be maintained largely on site until the final reports are finalized for this project. This archive will include all data, documentation, records, protocols, reports, and correspondence. The archive will be transferred off-site at the completion of the project and stored for at least five years in a commercial file storage warehouse operated by Iron Mountain, Inc., 130 Nova Drive, Morrisville, North Carolina. The masonry and steel construction of this facility protects from most natural and human threat. Iron Mountain is the sole tenant of the facility, thus eliminating any conflicts associated with a multi-tenant facility. The facility can only be accessed by card key entry. Only those on the authorized list have access to the facility. The facility is monitored 24 hours a day, 365 days a year, by Sonitrol Security Systems. They employ the following type of security measures: motion, sound, smoke and heat detectors, as well as laser-trigger alarms. The facility is protected against fire by an Early Suppression Fast Response (ESFR) 6 ln. CSC Central sprinkler system. Their current operating system for records management is Total Recall by DHS Associates, Inc., of Orange Park, Florida. A back up tape for the Iron Mountain Facilities records is created daily and sent off site to a secure vault location to ensure that the data is protected and can be restored in the event of an emergency. The property the facility is on has been determined by FEMA, as of March 3, 1992, to be located in the 500-year flood plain, Zone X on map number 37183CO284E, community number 370242 and 550 feet from the 100-year flood plain.

The central electronic project archive will also be maintained in the Durham, North Carolina, office of ARCADIS on the central office server. The directory that will be used is accessible only to the project manager, system administrator, and a small group of his direct reports. This server is backed up to tape daily by the system administrator; these tapes are maintained for at least three weeks. The server is backed up to tape monthly and these tapes are permanently retained. Tape storage takes place in an on site fire proof cabinet. At the completion of this project ARCADIS anticipates placing the primary data tables on CD for ease of storage and access.

**2.6 Data Availability Following Key Personnel Changes.** In order to ensure data availability following key personnel changes, the project manager will be notified of any change in the employment status of that employee either by the employee or their direct

supervisor (such as an office manager). The project manager will immediately take action as appropriate in conjunction with operations management to ensure the integrity and readability of all data. Should the transition affect the project manager himself for some unanticipated reason, the principal investigator and quality assurance officer would work together to ensure the integrity and readability of all data.

#### 3.0 Demonstration Plan Availability

These are available as needed from Chris Lutes, the ARCADIS Project Manager, whose contact information is in Appendix A.

# Appendix A-10

Microbial Characterization Report



2340 Stock Creek Blvd. Rockford TN 37853-3044 Phone (865) 573-8188 Fax: (865) 573-8133 Email: microbe@microbe.com

## **Microbial Analysis Report**

Client:	Glen G Arcadis 175 Ca Suite 4 Lowell,	Gordon S Abot Street 00 MA 0185	4	Phone: Fax:	978-937-9999 978-937-7555	
MI Identi	fier:	6agm	Date Rec.:	10/15/02	Report Date:	12/9/02
Analysis	Reque	sted:	PLFA, VFA, DGGE/DH	E		
Proiect:						

#### **Comments:**

All samples within this data package were analyzed under U.S. EPA Good Laboratory Practice Standards: Toxic Substances Control Act (40 CFR part 790). All samples were processed according to standard operating procedures. Test results submitted in this data package meet the quality assurance requirements established by Microbial Insights, Inc.

Reported by:

**Reviewed by:** 

**NOTICE:** This report is intended only for the addressee shown above and may contain confidential or privileged information. If the recipient of this material is not the intended recipient or if you have received this in error, please notify Microbial Insights, Inc. immediately. The data and other information in this report represent only the sample(s) analyzed and are rendered upon condition that it is not to be reproduced without approval from Microbial Insights, Inc. Thank you for your cooperation.



2340 Stock Creek Blvd. Rockford TN 37853-3044 Phone (865) 573-8188 Fax: (865) 573-8133 Email: microbe@microbe.com

### **Microbial Analysis Report**

#### **Executive Summary**

Three samples were collected on 10/14/02 to evaluate changes in the microbial communities after a molasses injection. Samples were collected from an upgradient control location (HAN-GW-B239) and downgradient (HAN-GW-1R21), and crossgradient (HAN-GW-RAP1-6T) of the injection area. The following analyses were used for this characterization:

- Phospholipid Fatty acid analysis
- Volatile Fatty Acids
- Denaturing Gradient Gel Electrophoresis
- Targeted Gene Detection for *Dehalococcoides ethenogenes*.(DHE)

Results from this study revealed the following key observations:

- Although minimal, biomass estimates were higher in the samples collected downgradient and crossgradient of the molasses injection.
- Compared to the upgradient sample, conditions in the downgradient (HAN-GW-RAP1-6T) and crossgradient (HAN-GW-1RZ1) samples appeared to be more anaerobic with increased biomarkers associated with the firmicutes (clostridium-like bacteria), and detectable VFA. Additionally, DGGE results identified three anaerobic bacteria within HAN-GW-1RZ1.
- A targeted gene detection for DHE confirmed the presence of this bacterium in HAN-GW-RAP1-6T and HAN-GW-B239.

#### **Overview of Approach:**

#### **Phospholipid Fatty Acid Analysis**

Determination of the phospholipid fatty acids (PLFA) in environmental samples is an effective tool for monitoring microbial responses to their environment. They are essential components of the membranes of all cells (except for the Archea, a minor component of most environments), so their sum includes all important actors of most microbial communities. There are three different types of information in PLFA profiles – biomass, community structure, and physiological status.

**Biomass:** PLFA analysis is the most reliable and accurate method available for the determination of viable microbial biomass. Since phospholipids breakdown rapidly upon cell death (21, 23) the PLFA biomass does not contain 'fossil' lipids of dead cells. The sum of the PLFA, expressed as picomoles (1 picomole =  $1 \times 10^{-12}$  mole) is proportional to the number of cells. The proportion used in this report, 20,000 cells/pmole, is taken from cells grown in laboratory media, and varies somewhat with type of organism and environmental conditions. Starving bacterial cells have the lowest cells/pmol, and healthy eukaryotic cells have the highest.

**Community Structure:.** The PLFA in an environmental sample is the sum of the microbial community's PLFA, and reflects the proportions of different organisms in the sample. PLFA profiles are routinely used to classify bacteria and fungi (19), and are one of the characteristics used to describe new bacterial species (25). Broad phylogenic groups of microbes have different fatty acid profiles making it possible to distinguish between them (4, 5, 22, 24). Table 1 describes the six major structural groups employed in this report.

PLFA Structural Group	General classification
Monoenoic (Monos)	Abundant in Proteobacteria (Gram negative bacteria), typically fast growing, utilize many carbon sources, and adapt quickly to a variety of environments.
Terminally Branched Saturated (TerBrSats)	Characteristic of Firmicutes (Low G+C Gram-positive bacteria), and also found in Bacteriodes, and some Gram-negative bacteria.
Branched Monoenoic (BrMonos)	Found in the cell membranes of micro-aerophiles and anaerobes, such as sulfate- or iron- reducing bacteria
Mid-Chain Branched Saturated (MidBrSats)	Common in Actinobacteria (High G+C Gram-positive bacteria), and some sulfate-reducing bacteria.
Normal Saturated (Nsats)	Found in all organisms.
Polyenoic	Found in Eukaryotes such as fungi, protozoa, algae, higher plants, and animals.

 Table 1. Description of PLFA structural groups.

**Physiological status:** The membrane of a microbe must adapt to the changing conditions of it's environment, and these changes are reflected in the PLFA. Toxic compounds or environmental conditions which disrupt the membrane cause some bacteria to make trans fatty acids from the usual cis fatty acids (7). Many Proteobacteria and others respond to starvation or highly toxic conditions by making cyclopropyl (7) or mid-chain branched fatty acids (20). The physiological status biomarkers for Toxic Stress and Starvation/Toxic conditions are formed by dividing the amount of the stress-induced fatty acid by the amount of it's biosynthetic precursor.

PLFA were analyzed by extraction of the total lipid (21) and then separation of the polar lipids by column chromatography (6). The polar lipid fatty acids were derivatized to the fatty acid methyl esters, which were quantified using gas chromatography (15). Fatty acid structures were verified by chromatography/mass spectrometry and equivalent chain length analysis.

#### **Volatile Fatty Acids**

The volatile fatty acids (VFA) pyruvate, lactate, formate, acetate, propionate, and butyrate are used as biomarkers of anaerobic metabolism. Anaerobic bacteria produce these compounds by fermentation, while under aerobic conditions, these compounds are rapidly oxidized for carbon and energy by aerobic bacteria. The VFA are analyzed by ion chromatography.

#### Denaturing gradient gel electrophoresis (DGGE)

Denaturing gradient gel electrophoresis (DGGE) is a powerful tool for detection and identification of organisms from environmental samples (1, 18, 12). In this method, sample microbial DNA is first isolated and purified. The DNA sequence for the Bacterial 16S ribosomal RNA (rDNA) is then amplified (many copies are made) using the polymerase chain reaction. The 16S rDNA gene is used for bacterial identification since it is common to all bacteria, and there are large databases of sequences available for comparison. The amplified sequences are separated into bands using a denaturing gradient gel. Numerically dominant members of the microbial community (>1 to 2% of the community) can be detected, so the bacterial identifications reported are examples of abundant members of the microbial community. For each sequenced DNA band, the closest described relative of each is reported. Phylogenetic affiliations are determined by comparing the rDNA sequences from samples to known bacterial sequences in the National Center for Biotechnology Information database (GenBank) (13). Recent progress in classifying Bacteria has caused many of the names used for bacteria and groups of bacteria to be changed. This can be a source of confusion since most scientists and engineers were trained when the earlier nomenclature was used. Table 2 shows the current names used in this report, and the corresponding obsolete terminology.

Current Names	Obsolete Names
Phylogenic Groups	
Actinobacteria	High G+C Gram positive bacteria such as Actinomycetes, Mycobacterium, Rhodococcus
Eukaryotes	Fungi, protozoa, algae, flowering plants, and animals
Firmicutes	Low G+C Gram positive bacteria such as Bacillus and Clostridia
Proteobacteria	Gram-negative bacteria

Table 2. Names for bacteria and bacterial groups used in this report and the corresponding obsolete forms.

#### **Targeted Gene Detection**

Specific DNA primers for a conserved region of the 16S rDNA gene were used to detect Dehalococcoides ethenogenes, based upon the method of Loffler et. al. (10). The sensitivity is  $\sim 10^3$  cells per milliliter or gram of sample. Two amplification samples were used to ensure the validity of the results: negative control; E. coli and positive control; D. ethenogenes.

#### **Results and Discussion**

#### **Phospholipid Fatty Acid Analysis**

Biomass estimates (expressed as the total concentration of PLFA) were fairly similar in all three samples, however estimates were slightly higher in the samples collected downgradient (HAN-GW-RAP1-6T) and crossgradient (HAN-GW-1RZ1) of a molasses injection point (see Figure 1 and Table 3).



**Figure 1.** Biomass content is presented as the total amount of phospholipid fatty acids (PLFA) present in a given sample. PLFA comprise a large proportion of the membranes of all living cells, but decompose quickly upon cell death.

**Table 3.** Viable microbial biomass expressed as picomoles PLFA per mL and as cells per mL, fatty acid structural groups as percent of total PLFA, and physiological status biomarkers as mole ratio. "-" indicates data not available. Detection of Volatile fatty acids is expressed as mg/L. Results for the DHE amplification were performed using primers directed to a variable region of the 16S rRNA gene of *Dehalococcoides ethenogenes*. Presence is noted with a plus sign and the relative abundance determined by the number of plus signs.

Sample Name	HAN-GW-1R21	HAN-GW-RAP1-6T	HAN-GW-B239
Sampling Date	10/14/2002	10/14/2002	10/14/2002
<u>Biomass</u>			
pmols PLFA/ml filtered	10	13	8
Cells/ml filtered <sup>1</sup>	1.91E+05	2.59E+05	1.61E+05
Community Structure: (% of Total PLFA)			
Firmicutes (TerBrSats)	17.2	16.6	9.9
Proteobacteria (Monos)	53.3	55.3	53.3
Anaerobic metal reducers (BrMonos)	2.4	2.7	3.9
Actinomycetes (MidBrSats)	3.2	2.2	6.1
General (Nsats)	22.5	20.8	19.5
Eukaryotes (polyenoics)	1.5	2.4	7.3
Physiological Status			
Starvation, Cy/cis	0.14	0.16	0.42
Membrane Stress, trans/cis	0.16	0.18	0.13
Volatile Fatty Acids (mg/L)			
Pyruvic	<4	<4	<4
Lactic	< 1	< 1	< 1
Formic	<1	<1	<1
Acetic	111	57	<1
Proprionic	18	6	<1
Butyric	<u>14</u>	<u>7</u>	<u>&lt; 1</u>
Total VFA	143	70	ND
Dehalococcoides ethenogenes (DHE)			
Present/Absent	(Absent)*	Present (+++)	Present (+)

\* No PCR product was obtained for either DNA analysis for this sample. Therefore, it is possible that DHE was present, but that we were unable to detect it because of PCR inhibiting compounds being present.

The PLFA profiles for these samples revealed relatively diverse community structures at all three sampling locations. All three communities were dominated by proteobacteria (indicated by percentage of monoenoic PLFA), which often rapidly reproduce to take advantage of available organic carbon. Terminally branched PLFA were higher in the samples collected downgradient (HAN-GW-RAP1-6T) and crossgradient (HAN-GW-1RZ1) of a molasses injection point. These are most commonly due to Firmicutes (clostridia-like Gram positive bacteria). An increase in terminally branched PLFA is often seen in environmental transects from more aerobic to more anaerobic conditions, so this may signal in increase in anaerobes (as compared to the upgradient location HAN-GW-B239) due to the injection of molasses (see Figure 2). Volatile fatty acids (VFA) were also

<sup>&</sup>lt;sup>1</sup> The cell equivalent value is calculated from experiments with typical bacteria isolated from soil and water. This value is based on 2.0 x 10<sup>12</sup> cells per gram dry weight of cells and 10<sup>8</sup> picomoles of phospholipid/gram dry weight of cells. The number of cells/gram of dry weight may vary and is dependent on the environmental conditions from which the microorganisms were recovered.

detected in the downgradient (HAN-GW-RAP1-6T) and crossgradient (HAN-GW-1RZ1) samples, which further indicates that conditions are anaerobic.

Physiological Status markers for starvation (cy/cis) showed that the Gram negative bacteria in the samples collected downgradient (HAN-GW-RAP1-6T) and crossgradient (HAN-GW-1RZ1) of a molasses injection point were less starved than those in the upgradient location HAN-GW-B239. This is likely a positive response from the molasses injection (see Figure 3).



Figure 2. Relative percentages of total PLFA structural groups in the samples analyzed. Structural groups are assigned according to PLFA chemical structure, which is related to fatty acid biosynthesis. See Table 1 for detailed descriptions of structural groups. An "X" indicates samples with insufficient biomass for community structure to be determined.



**Figure 3.** Microbial physiological stress markers. Starvation biomarker for the Gram-negative community is assessed by the ratio cyclopropyl fatty acids to their metabolic precursor. Adaptation of the Gram-negative community to toxic stress is determined by the ratio of  $\omega$ 7t/ $\omega$ 7c fatty acids. Gram-negative bacteria generate *trans* fatty acids to minimize the permeability of their cellular membranes as adaptation to a more hostile environment. Ratios (16:1 $\omega$ 7t/16:1 $\omega$ 7c and 18:1 $\omega$ 7t/18:1 $\omega$ 7c) greater than 0.1 have been shown to indicate an adaptation to a toxic or stressful environment resulting in decreased membrane permeability. An "X" indicates samples with insufficient biomass for metabolic status markers to be determined.

#### Denaturing gradient gel electrophoresis (DGGE)

DGGE profiles were only obtained from samples HAN-GW-RAP1-6T and HAN-GW-B239. It is likely some type of inhibition made it difficult to obtain DNA results from HAN-GW-IRZ1. Only the bacterial profile for sample HAN-GW-RAP1-6T produced a distinct banding pattern in which identifications could be obtained. These results identified three anaerobic bacteria, two Bacteriodes, and a Clostridium. Members of the genus bacteriodes are obligate anaerobic Gram negative bacteria, which produce high levels of acetate and succinate as metabolic end products. Information about the presence bacteria affiliated with *Bacteriodes* at contaminated sties is limited, but does indicate conditions are anaerobic. Members of the genus *Clostridium* are anaerobic counterparts to the Bacilli. Some members of this genus are notable pathogens, but other members can be found in environmental samples where oxygen is absent (or at least restricted where they are growing). Clostridia are commonly found in soils, sewage, marine sediments, and decaying vegetation. As a group, they have a wide pH tolerance range (4.5 - 8.5), can use a variety of compounds as energy sources, and form drought- and oxygen-resistant spores. Some Clostridia can fix nitrogen, taking nitrogen gas ( $N_2$ ) to form ammonium. While there have been some recent publications (see appendix) suggesting the involvement of *Clostridium bifermentans* in the dehalogenation of PCE, a review article published in 1999 in Bioremediation Journal listed several members of *Clostridium* that were unable to dechlorinate PCE.

Given the limited information obtained from the DGGE profiles, a targeted gene detection for a known dechlorinating bacterium, *Dehalococcoides ethenogenes* (DHE), was used to determine its presence. Bacteria belonging to the *Dehalococcoides* assemblage are currently the only known organisms capable of converting all chlorinated ethenes to ethene. Given this, detection of this organism provides insight into the genetic potential of a given site. Within these three samples, DHE was detected in HAN-GW-RAP1-6T and HAN-GW-B239. The presence of DHE in HAN-GW-IRZ1 is inconclusive at this point given the difficulty with amplification.



Figure 4. DGGE gel image of the bacterial domain. Banding patterns and relative intensities of the recovered bands provide a measure of change in the community. Dominant species must constitute at least 1-2% of the total bacterial community to form a visible band. Labeled bands were excised and sequenced. Results from sequencing can be found in the following table.

Table 4. Sequence results from bands excised from Figure 5.	Identifications are based upon the Ribosomal Database Project (RDP)
Similarity indecies above .800 are considered excellent, .600	700 are good and below .500 are considered to be unique sequences.

Band	Closest Match	Similarity Index	Phylogenetic Affiliation	Habitat	Ref
А	Uncultured Bacteriodes sp.	0.865	Bacteriodetes	Anaerobic	-
В	Uncultured Bacteriodes sp.	0.865	Bacteriodetes	Anaerobic	-
С	Clostridium sp.	0.995	Firmicutes	Anaerobic	-

#### References

- 1. Amann, R. I., W. Ludwig, and K.-H. Schleifer. 1995. Phylogenetic identification and in situ detection of individual microbial cells without cultivation. Microbiological Reviews 59:143-169.
- 2. Cottrell, MT and David L. Kirchman. Appl Environ Microbiol. 2000 April; 66 (4): 16921697.
- Gillis, M., V. Tran Van, R. Bardin, M. Goor, P. Hebbar, A. Willems, P. Segers, K. Kerstens, T. Heulin, and M. P. Fernadez. 1995. Polyphasic taxonomy in the genus Burkholderia leading to an amended description of the genus and proposition of Burkholderia vietnamiensis sp. nov. for N2-fixing isolates from rice in Vietnam. Int. J. Syst. Bacteriol. 45:274-289.
- 4. Dowling, N. J. E., F. Widdel, and D. C. White. 1986. Phospholipid ester-linked fatty acid biomarkers of acetate-oxidizing sulfate reducers and other sulfide forming bacteria. Journal of General Microbiology 132:1815-1825.
- Edlund, A., P. D. Nichols, R. Roffey, and D. C. White. 1985. Extractable and lipopolysaccharide fatty acid and hydroxy acid profiles from Desulfovibrio species. Journal of Lipid Research 26:982-988.
- Guckert, J. B., C. P. Antworth, P. D. Nichols, and D. C. White. 1985. Phospholipid ester-linked fatty acid profiles as reproducible assays for changes in prokaryotic community structure of estuarine sediments. FEMS Microbiol. Ecol. 31:147-158.
- Guckert, J. B., M. A. Hood, and D. C. White. 1986. Phospholipid ester-linked fatty acid profile changes during nutrient deprivation of Vibrio cholerae: increases in the trans/cis ratio and proportions of cyclopropyl fatty acids. Appl. Environ. Microbiol. 52:794–801.

- Hedrick, D.B., A Peacock, J.R. Stephen, S.J. Macnaughton, Julia Brüggemann, and David C. White. 2000. Measuring soil microbial community diversity using polar lipid fatty acid and denatured gradient gel electrophoresis data. J. Microbiol. Methods, 41, 235-248.
- 9. ITRC Internet Training on Natural Attenuation of Chlorinated Solvents in Groundwater: Principles and Practices, Apr 00.
- 10. Löffler, F. E., Q. Sun, et al. (2000). "16S rRNA gene-based detection of tetrachloroethene-dechlorinating Desulfuromonas and Dehalococcoides species." Appl Environ Microbiol 66(4): 1369-1374.
- 11. Maymo-Gatell X, Chien Y, Gossett JM, Zinder SH. 1997. Isolation of a bacterium that reductively dechlorinates tetrachloroethene to ethene. Science 276(5318):1568-71.
- Muyzer, G., E. C. De Waal, and A. G. Uitterlinden. 1993. Profiling of complex microbial populations by denaturing gradient gel electrophoresis analysis of polymerase chain reaction-amplified genes coding for 16S rRNA. Applied and Environmental Microbiology 59:695-700.
- 13. National Center for Biotechnology Information. (http://www.ncbi.nlm.nih.gov/)
- 14. Overman, J., "Family Chlorobiaceae," in M. Dworkin et al., eds., The Prokaryotes: An Evolving Electronic Resource for the Microbiological Community, 3rd edition, release 3.7, November 2, 2001, Springer-Verlag, New York, www.prokaryotes.com.
- 15. Ringelberg, D. B., G. T. Townsend, K. A. DeWeerd, J. M. Sulita, and D. C. White. 1994. Detection of the anaerobic dechlorinating microorganism Desulfomonile tiedjei in environmental matrices by its signature lipopolysaccharide branch-long-chain hydroxy fatty acids. FEMS Microbiol. Ecol. 14:9-18.
- 16. Schlötelburg, C. 2001. Mikrobielle Diversität und Dynamik einer 1,2-Dichlorpropan dechlorierenden Mischkultur (Microbial Diversity and Dynamics in a 1,2-Dichloropropane Dechlorinating Mixed Culture). Dissertation, Humbolt University, Berlin, Germany. In German: http://edoc.hu-berlin.de/dissertationen/schloetelburg-cord-2001-12-07/PDF/Schloetelburg.pdf
- 17. Sharp, R., D. Cossar, and R. Williams. 1995. Physiology and metabolism of Thermus. Biotechnol. Handb. 9:67-91.
- Stephen, J. R., Y.-J. Chang, Y. D. Gan, A. Peacock, S. Pfiffner, M. Barcelona, D. C. White, and S. J. Macnaughton. 1999. Microbial characterization of a JP-4 fuel-contaminated site using a combined lipid biomarker/polymerase chain reaction-denaturing gradient gel electrophoresis (PCR-DGGE) based approach. Environmental Microbiology 1:231-241.
- Tighe, S.W., de Lajudie, P., Dipietro, K., Lindström, K., Nick, G. & Jarvis, B.D.W. (2000). Analysis of cellular fatty acids and phenotypic relationships of Agrobacterium, Bradyrhizobium, Mesorhizobium, Rhizobium and Sinorhizobium species using the Sherlock Microbial Identification System. Int J Syst Evol Microbiol 50, 787-801.
- 20. Tsitko, I.V. Gennadi M. Zaitsev, Anatoli G. Lobanok, and Mirja S. Salkinoja-Salonen. 1999. Applied and Environmental Microbiology 65(2) 853-855.
- 21. White, D. C., W. M. Davis, J. S. Nickels, J. D. King, and R. J. Bobbie. 1979. Determination of the sedimentary microbial biomass by extractable lipid phosphate. Oecologia 40:51-62.
- White, D. C., H. C. Pinkart, and D. B. Ringelberg. 1997. Biomass measurements: Biochemical approaches, p. 91-101. In C. J. Hurst, G. R. Knudsen, M. J. McInerney, L. D. Stetzenbach, and M. V. Walter (ed.), Manual of Environmental Microbiology. ASM Press, Washington.
- 23. White, D. C., and D. B. Ringelberg. 1995. Utility of signature lipid biomarker analysis in determining in situ viable biomass, community structure, and nutritional / physiological status of the deep subsurface microbiota. In P. S. Amy and D. L. Halderman (ed.), The microbiology of the terrestrial subsurface. CRC Press, Boca Raton.
- 24. White, D. C., J. O. Stair, and D. B. Ringelberg. 1996. Quantitative comparisons of in situ microbial biodiversity by signature biomarker analysis. Journal of Industrial Microbiology 17:185-196.
- Vandamme P, Pot B, Gillis M, de Vos P, Kersters K, Swings J. Polyphasic taxonomy, a consensus approach to bacterial systematics. Microbiol Rev 1996 Jun;60(2):407-38.

# Appendix C

Quality Assurance Project Plan (QAPP)

This QAPP (Appendix C) was taken from the following document:

### Technology Demonstration Plan Hanscom Air Force Base

### TECHNOLOGY DEMONSTRATION *IN-SITU* SUBSTRATE ADDITION TO CREATE REACTIVE ZONES FOR TREATMENT OF CHLORINATED ALPHATIC HYDROCARBONS

Prepared by:

ARCADIS Geraghty & Miller, Inc. P.O. Box 13109 Research Triangle Park, NC 27709

March 2, 2000

### **Table of Contents**

Exec	utive	Summary	. 1
1.	Intro	duction	. 2
	1.1	Background Information	. 2
	1.2	Official DoD Requirement Statement(s)	. 3
	1.3	Objectives of the Demonstration	. 4
	1.4	Regulatory Issues	. 5
2.	Tech	nology Description	. 7
	2.1	Background and Applications	. 7
	2.2	Advantages and Limitations of the Technology	12
	2.3	Comparison of <i>In-situ</i> Reactive Zone Technology to other Electron Donor Supply Methods	14
3	Pre-I	Demonstration Activities	17
2.	31	Selecting Sites	17
	3.2	Pre-Demonstration Sampling and Analysis	18
4.	Site/	Facility Descriptions	20
	4.1	Site/Facility History	20
	4.2	Site/Facility Characteristics	21
	4.3	Site/Facility Maps and Photographs	22
	4.4	Conceptual Site Model	22
5.	Dem	onstration Approach	24
	5.1	Performance Objectives.	24
	5.2	Demonstration Setup, Commencement, and Operation	26
		5.2.1 Site Preparation	26
		5.2.1.1 Equipment Set-up	26
		5.2.1.2 Analytical Instrumentation	27
		5.2.1.3 Utilities	27
		5.2.1.4 Injection Well Installation Procedures/Methods	28
		5.2.2 Operating Parameters and Requirements	28
	5.3	Technical Performance Criteria	29
		5.3.1 Contaminants	29
		5.3.2 Process Waste	29
		5.3.3 Factors Affecting Technology Performance	30
		5.3.3.1 Biological Factors Affecting IRZ Performance	30
		5.3.3.2 Geological and Hydrogeological Factors Affecting IRZ	
		Performance	31
		5.3.3.3 Chemical Factors Affecting IRZ Performance	32

		5.3.4 Reliability	
		5.3.5 Ease of Use	35
		5.3.6 Versatility	
		5.3.7 Off-the-Shelf Procurement	
		5.3.8 Maintenance.	
		5.3.9 Scaleup Issues	39
	5.4	Sampling Plan	40
		5.4.1 Selection of Analytical Laboratory	40
		5.4.2 Selection of Analytical Method	41
		5.4.3 Sample Collection	41
		5.4.3.1 Groundwater Sampling for Technology Verification	42
		5.4.3.2 Groundwater Sampling	
		5.4.3.3 Soil Sampling	
		5.4.3.4 Sample Shipment	
		5.4.3.5 Gas Sampling (when necessary)	48
		5.4.4 Experimental Controls.	
		5.4.5 Sample Analysis	49
	5.5	Demonstration Schedule	50
6.	Data	Collection, Storage, and Archiving Procedures	
	6.1	Data Format	51
		6.1.1 Class I Data	
		6.1.2 Class II Data	52
		6.1.3 Class III Data	52
		6.1.4 Class IV Data	52
	6.2	Data Storage and Archiving Procedures	53
		6.2.1 Class I Data Storage	53
		6.2.2 Class II Data Storage	53
		6.2.3 Class III Data Storage	53
		6.2.4 Class IV Data Storage	53
		6.2.5 Archiving Procedures	53
		6.2.6 Data Availability Following Key Personnel Changes	54
7.	Cost	Performance Criteria	55
	7.1	Startup Costs	55
	7.2	Operations and Maintenance Costs	56
	7.3	Demobilization	56
	7.4	Life-Cycle Costs	57
8.	Reg	ulatory Issues, Permits & Site Access	59
	8.1	Regulatory Issues – General	59
	8.2	Hanscom Regulatory Issues, Permits & Site Access	59
9.	Qua	lity Assurance Plan	61
	9.1	Purpose and Scope of the Plan	61

	9.2	Quality Assurance Responsibilities	61
	9.3	Data Quality Parameters	63
	9.4	Calibration Procedures, Quality Control Checks, and Corrective Action	64
		9.4.1 Dissolved Gas Methods – Fixed Facilities	64
		9.4.2 Chemical Fixed Facility Measurements	64
		9.4.3 Geotechnical Fixed Facility Measurement	66
		9.4.4 Field Measurements	66
		9.4.5 Data Validation	66
	9.5	Demonstration Procedures	66
		9.5.1 Start-up	66
		9.5.2 Technology Maintenance	66
		9.5.3 Corrective Actions	67
	9.6	Calculation of Data Quality Indicators	67
	9.7	Performance and System Audits	67
		9.7.1 Performance and Systems Audits	67
		9.7.2 Contingency Laboratory	68
	9.8	Quality Assurance Reports	68
10.	10. Health and Safety Plan		70
11.	References		
12.	Date	d Signature of Project Lead	72
	12.1	Project Team Roles and Responsibilities	72
	12.2	Dated Signatures	75
List o	of Aci	onyms	İV
List o	of Tab	bles (Attachment)	Vİ
List o	of Fig	ures (Attachment)	V1
Appe	endix	A Points of Contact	
Appe	endix	B Laboratory Standard Operating Procedures	
Appe	endix	C Laboratory Detection Limits	
Appendix D Field Forms		D Field Forms	
Appendix E Standard Operating Procedures For Field Instruments			

Contract No.: F41624-99-C-8032 Contractor Name: ARCADIS Geraghty & Miller Contractor Address: 1099 18th Street, Suite 2100, Denver, CO 80202 Expiration: Receipt of final payment by Contractor upon completion of the Contract, including all modifications, *or* two years after Contract Award date, whichever occurs at the latest date.

The Government's rights to use, modify, reproduce, release, perform, display, or disclose these technical data are restricted by paragraph (b)(3) of the Rights in Technical Data--Noncommercial items clause contained in the above identified contract. Any reproduction of technical data or portions thereof marked with this legend must also reproduce the markings. Any person, other than the Government, who has been provided access to such data must promptly notify the above named Contractor.

### 9. Quality Assurance Plan

#### 9.1 Purpose and Scope of the Plan

This QA Plan delineates our approach for monitoring the demonstration to ensure that the facilities, equipment, personnel, methods, practices, records, and controls are in conformance with ESTCP-approved data quality objectives. In addition to preparing this QAPP, ARCADIS Geraghty & Miller's Research Triangle Park Office, the lead office for this project, operates under a quality system that is described in an office Quality Management Plan written according to ANSI/ASQC E4-1994 Specifications and Guidelines for Quality Systems for Environmental Data Collection and Environmental Technology Programs.

#### 9.2 Quality Assurance Responsibilities

The ARCADIS Geraghty & Miller Program Manager/Principal Investigator has the following QA responsibilities:

- Review Demonstration QA Project Plan, Sampling Plans, Test Plans, etc.
- Serve as the project's liaison with senior corporate management to ensure the assignment of adequate resources
- Review decisions about major corrective actions
- Review Final Report and Cost & Performance Report
- Serve as the primary quality assurance reviewer and authority for engineering design and geologic matters. Will be assisted in this regard by personnel licensed in states where Mr. Palmer is not licensed

The ARCADIS Geraghty & Miller Project Manager has the following QA responsibilities:

- Coordinate preparation of Demonstration QA Project Plan, Sampling Plans, Test Plans, etc.
- Ensure personnel assigned to project are adequately trained
- Ensure activities are carried out as planned and deviations are documented
- Ensure equipment and instrumentation is calibrated and in good working condition
- Initiate corrective action procedures
- Communicate any problems or deviations from plan to the QA Officer
- Coordinate preparation of Final Report and Cost & Performance Report

Contract No.: F41624-99-C-8032 Contractor Name: ARCADIS Geraghty & Miller Contractor Address: 1099 18th Street, Suite 2100, Denver, CO 80202 Expiration: Receipt of final payment by Contractor upon completion of the Contract, including all modifications, *or* two years after Contract Award date, whichever occurs at the latest date.

The Government's rights to use, modify, reproduce, release, perform, display, or disclose these technical data are restricted by paragraph (b)(3) of the Rights in Technical Data--Noncommercial items clause contained in the above identified contract. Any reproduction of technical data or portions thereof marked with this legend must also reproduce the markings. Any person, other than the Government, who has been provided access to such data must promptly notify the above named Contractor.

The ARCADIS Geraghty & Miller QA Officer for this demonstration has the following responsibilities:

- Review and approve Demonstration QA Project Plan, Sampling Plans, Test Plans, etc.
- Perform periodic audits to ensure demonstration is conducted as planned and any deviations from plan or standard methods are adequately documented
- Report any audit findings or problems to the Project Manager
- Review laboratory data and ensure it is supported by appropriate QA/QC information
- Review Final Report and Cost & Performance Report to ensure that is accurately describes the methods and standard operating procedures, and that the reported results are supported by raw data

It is the responsibility of the Project Manager and Deputy Project Manager to ensure that required QA/QC documentation is in place before any demonstration activity is begun and that documented QA/QC activities are carried out in all phases of the demonstration. Christopher Lutes is the Project Manager and Don Kidd is the Deputy Project Manager for this demonstration. Project Management is responsible for ensuring that staff members are adequately trained to perform assigned duties.

Ms. Laura Beach is the Data Quality Assurance Officer for the IRZ demonstration. Ms. Beach is the QA Manager for ARCADIS Geraghty & Miller's Technology Services Division. She has more than 10 years of experience in providing QA support to government contracts (for example, USEPA-APPCD, USEPA Environmental Technology Verification Program, NFESC Innovative Technology Project and USAF Environics Directorate Support) and is very familiar with the QA/QC activities required to support them. The Data Quality Assurance Officer will assume responsibility for, or assign an on-site QA representative to perform QA support activities during the demonstration. Any designated QA representative will report regularly to Ms. Beach and will be jointly responsible for ensuring that QA tasks meet contractual requirements as well as the requirements that are established in the ARCADIS Geraghty & Miller Raleigh RTP Office Quality Management Plan referenced earlier. The Data Quality Assurance Officer's responsibilities include support in the preparation and review of this work plan, conducting internal systems and/or performance audits, QA/QC reporting, and involvement in the correction of any issues leading to data quality concerns.

As Project Managers, Mr. Lutes and Mr. Kidd will openly communicate with both Ms. Beach and the senior project advisors. The assigned Engineering and Biogeochemical Assessment Team Leaders for the IRZ demonstration are Mike Hansen and Jeffrey Burdick respectively. Project personnel including both Team Leaders, other technical staff, and field technicians are expected to work closely with the Data Quality Assurance Officer to ensure

Contract No.: F41624-99-C-8032 Contractor Name: ARCADIS Geraghty & Miller Contractor Address: 1099 18th Street, Suite 2100, Denver, CO 80202 Expiration: Receipt of final payment by Contractor upon completion of the Contract, including all modifications, *or* two years after Contract Award date, whichever occurs at the latest date.

The Government's rights to use, modify, reproduce, release, perform, display, or disclose these technical data are restricted by paragraph (b)(3) of the Rights in Technical Data--Noncommercial items clause contained in the above identified contract. Any reproduction of technical data or portions thereof marked with this legend must also reproduce the markings. Any person, other than the Government, who has been provided access to such data must promptly notify the above named Contractor.

that QA/QC activities are adequate and that any problems are identified and corrected. Corrective actions are initiated by the Team Leader Project Manager and reported to both the Principal Investigator and the Data Quality Assurance Officer.

It is the intention of ARCADIS Geraghty & Miller that communications about data quality flow freely both up and down the organizational chart during the demonstration. Past experience in ARCADIS Geraghty & Miller suggests that it is imperative to include field level personnel in communications pertinent to data quality. This open communication to and from field staff will aid in ascertaining the quality of the data generated during the effort.

#### 9.3 Data Quality Parameters

Table 6 contains goals for the data quality parameters accuracy, precision, and completeness for the analytical measurement process. The table incorporates data quality goals for field analysis (temperature, ORP, pH, and dissolved oxygen) and for fixed laboratory based analysis. Thus, this table has been compiled with input from ARCADIS Geraghty & Miller as well as from the two analytical laboratories that will conduct laboratory based analyses for analytes stable enough to be shipped.

ARCADIS Geraghty & Miller and its subcontract analytical laboratories will rely primarily on timely servicing and appropriate calibration of analytical instruments to attain the accuracy goals listed in Table 6. With the possible exception of hydrogen analysis, the analyte list contains parameters that have been chemically quantified for many years in environmental media. As a result, correctly performed analysis of these parameters is capable of generating the accuracy needed to guarantee the success of this demonstration.

It is important to attain the accuracy goals contained in Table 6 for the listed analytes so as to facilitate inter-comparison of analytical results from multiple collection points at individual demonstration sites. The IRZ technology is expected to affect/generate trends in the analytes listed in Table 6 within each reactive zone. The trends generated by molasses injection are important in determining the size and bacterial community characteristics of the reactive zone over time.

ARCADIS Geraghty & Miller routinely checks the precision of its analytical field instruments as a course of collecting data during low flow well sampling activities. The procedure universally utilized is to purge the well being sampled and then begin pumping the groundwater through a low-flow, flow-through sample cell where it comes into contact with probes that are calibrated for the parameters of interest. The groundwater is pumped through the sample cell until the readings for the parameters of interest stabilize with the precision guidelines found in Table 6. Thus, precision is determined at the completion of the period required for the parameter readings to stabilize. Likewise, the contract analytical laboratories chosen for this project are accustomed to goals similar to those shown in Table 6. Accuracy, Precision, and Completeness Goals for Analytical Parameters

Contract No.: F41624-99-C-8032 Contractor Name: ARCADIS Geraghty & Miller Contractor Address: 1099 18th Street, Suite 2100, Denver, CO 80202 Expiration: Receipt of final payment by Contractor upon completion of the Contract, including all modifications, *or* two years after Contract Award date, whichever occurs at the latest date.

The Government's rights to use, modify, reproduce, release, perform, display, or disclose these technical data are restricted by paragraph (b)(3) of the Rights in Technical Data--Noncommercial items clause contained in the above identified contract. Any reproduction of technical data or portions thereof marked with this legend must also reproduce the markings. Any person, other than the Government, who has been provided access to such data must promptly notify the above named Contractor.

routinely determining the precision of their analyses in keeping with their commitment to quality control.

Representativeness of groundwater samples is assured by careful well placement and through purging of each well prior to sample collection. ARCADIS Geraghty & Miller's assessment of site geology and hydrogeology is essential to verification of appropriate well placement. ARCADIS Geraghty & Miller has commenced this assessment using available data from individual site characterization efforts and will complete the effort in the course of establishing initial site conditions during the first round of biogeochemical analyses. The purging of groundwater wells is a part of the sampling procedures to be utilized at the demonstration sites. Purging of the wells insures that the chemical properties of the groundwater collected for analysis has not altered as a function of residence time within the well casing itself.

The use of identical analytical methodologies during the conduct of work at four demonstration sites will support the comparability of the data gathered during this project. This standardization of analytical methods is important so that the economics of IRZ implementation at the four sites can be delineated and reported to ESTCP/AFCEE.

#### 9.4 Calibration Procedures, Quality Control Checks, and Corrective Action

Calibration procedures for the standard EPA and ASTM methods are covered fully in those methods. Copies of the methods are available upon request.

9.4.1 Dissolved Gas Methods – Fixed Facilities. The dissolved gas methods referred to above are provided in full Appendix B. In short the light hydrocarbon method calls for a three point external calibration with calibration standards prepared from commercial certified gas standards traceable to the National Institute of Standards of Technology standards. The carbon dioxide method uses triplicate external calibration points with calibration standards prepared from commercially available certified gas standards traceable to National Institute of Standards of Technology standards. The hydrogen method uses a 7 point external calibration with calibration standards prepared from commercial certified gas standards. The laboratory data package provided by VaporTech includes initial calibration, continuing calibration check results (control limit is +/-20%), case narrative, chain of custody and laboratory blank results. Their are no established holding times for these analyses. Vapor Tech uses the 7 day VOA unpreserved holding time for the light hydrocarbons and carbon dioxide, and has demonstrated the adequacy of a 28 day holding times for hydrogen. Since the calibration is performed with gas standards, and analyses from liquid samples, it is generally not possible for Vapor Tech to report laboratory control spikes or matrix spikes. Vapor Tech will provide Excel compatible electronic deliverables which can be directly used by ARCADIS Geraghty & Miller to prepare final data tables following validation.

Contract No.: F41624-99-C-8032 Contractor Name: ARCADIS Geraghty & Miller Contractor Address: 1099 18th Street, Suite 2100, Denver, CO 80202 Expiration: Receipt of final payment by Contractor upon completion of the Contract, including all modifications, *or* two years after Contract Award date, whichever occurs at the latest date.

The Government's rights to use, modify, reproduce, release, perform, display, or disclose these technical data are restricted by paragraph (b)(3) of the Rights in Technical Data--Noncommercial items clause contained in the above identified contract. Any reproduction of technical data or portions thereof marked with this legend must also reproduce the markings. Any person, other than the Government, who has been provided access to such data must promptly notify the above named Contractor.

#### 9.4.2 Chemical Fixed Facility Measurements. Quanterra's QC protocols include the following:

- Minimum of one method blank is analyzed per 20 samples to detect contamination during preparation and/or analysis
- Laboratory Control Sample (LCS) consisting of target analytes spiked into a inert matrix is analyzed every 20 or fewer investigative samples. The LCS is used to monitor the laboratory's day to day as well as ongoing performance of the applicable analytical methods
- Matrix spikes and matrix spike duplicates (MS/MSD) for organic analyses and matrix spikes and matrix duplicates (MS/DU) or MS/SD's for inorganic analyses will be analyzed every 20 or fewer samples to determine the affect of the matrix on the method performed. Due to the potential variability of the matrix, the MS/SD results may have bearing on the specific sample spiked and not all samples in the batch
- Internal and surrogate standards will be added where appropriate to quantitate results, determine recoveries and to account for sample-to-sample variation

Calibration of instrumentation will be determined according to the appropriate EPA methods.

The Quanterra data reports will contain the following items:

- 1. Case Narrative
  - a. Date of issuance
  - b. Laboratory analysis performed
  - c. Any deviations from intended analytical strategy
  - d. Laboratory batch number
  - e. Numbers of samples and respective matrices
  - f. Quality control procedures utilized and references to the acceptance criteria
  - g. Laboratory report contents
  - h. Project name and number
  - i. Condition of samples received
  - j. Discussion of whether or not sample holding times were met
  - k. Discussion of technical problems
  - 1. Signature of Laboratory Project Manager
- 2. Chemistry Data Package
  - a. Case narrative for each analyzed batch of samples
  - b. Cross reference of laboratory sample to project sample identification numbers
  - c. Sample results with sample preparation and analysis dates
  - d. Raw data for sample results and laboratory quality control samples

Contract No.: F41624-99-C-8032 Contractor Name: ARCADIS Geraghty & Miller Contractor Address: 1099 18th Street, Suite 2100, Denver, CO 80202 Expiration: Receipt of final payment by Contractor upon completion of the Contract, including all modifications, *or* two years after Contract Award date, whichever occurs at the latest date.

The Government's rights to use, modify, reproduce, release, perform, display, or disclose these technical data are restricted by paragraph (b)(3) of the Rights in Technical Data--Noncommercial items clause contained in the above identified contract. Any reproduction of technical data or portions thereof marked with this legend must also reproduce the markings. Any person, other than the Government, who has been provided access to such data must promptly notify the above named Contractor.

e. Initial and continuing calibration checks, GC/MS tunes f. Matrix spike and matrix spike duplicate recoveries, laboratory control samples, method blank results, calibration check compounds, system performance check compounds g. Labeled and dated chromatograms and spectra of sample results and laboratory quality control checks

The data package will include a full CLP-like deliverable package without CLP forms.

Quanterra will provide an ASCII comma delimited electronic deliverable.

**9.4.3 Geotechnical Fixed Facility Measurement.** Calibration procedures for the ASTM particle size methods are included in the method. Essentially the only required calibration is on a balance. The balance calibration will be reported with that data set. The sieve screens are 'calibrated' by the manufacturer and inspected by the laboratory for tears before use.

**9.4.4 Field Measurements.** Calibration procedures for field instruments are included in Appendix E ARCADIS Geraghty & Miller field instrument standard operating procedures manual.

**9.4.5 Data Validation.** Formal validation will be performed on measurements of the contaminants but only a rapid review of data quality indicators will be performed for the other biogeochemical parameters. Site by site final reports and a cost and performance report will be prepared in accordance with ESTCP formats. It is anticipated that these reports will include both tabular and graphical depictions of the data collected.

#### 9.5 Demonstration Procedures

**9.5.1 Start-up.** Start-up activities for the demonstration will be limited. Initiation of the demonstration will begin with the collection of the baseline groundwater monitoring data (as outlined in section 5.4). This will be followed by the initial reagent solution injection in the injection well network. Upon completion of the baseline data collection and initial reagent injection the demonstration will move into the technology maintenance phase.

**9.5.2 Technology Maintenance**. Please see section 5.3.9.

**9.5.3 Corrective Actions.** Corrective action with regard to analytical measurements has been discussed in section 6.1. Corrective action with regard to system operation is covered in section 5.3.8.

Contract No.: F41624-99-C-8032 Contractor Name: ARCADIS Geraghty & Miller Contractor Address: 1099 18th Street, Suite 2100, Denver, CO 80202 Expiration: Receipt of final payment by Contractor upon completion of the Contract, including all modifications, *or* two years after Contract Award date, whichever occurs at the latest date.

The Government's rights to use, modify, reproduce, release, perform, display, or disclose these technical data are restricted by paragraph (b)(3) of the Rights in Technical Data--Noncommercial items clause contained in the above identified contract. Any reproduction of technical data or portions thereof marked with this legend must also reproduce the markings. Any person, other than the Government, who has been provided access to such data must promptly notify the above named Contractor.

#### 9.6 Calculation of Data Quality Indicators

Accuracy: Accuracy can be expressed as percent bias from a known standard or percent recovery based upon known spiked amounts. Percent bias is calculated using the following equation:

%Bias =[known value – obtained value)/known value] \* 100

Percent recovery is calculated by:

%Recovery = [measured value/spiked amount] \* 100

Precision: Precision, expressed as relative standard deviation (RSD) between replicate measurements can be determined using the formula:

%RSD = standard deviation of replicate measurements/average \*100

Completeness: Completeness is defined as the number of acceptable measurements compared to the number of total measurements taken expressed as percent. Acceptable measurements are defined as measurements that fall within data quality indicator goals for accuracy and precision.

Comparability and Representativeness: Comparability is defined as the degree to which different methods, data sets, and/or decisions agree or can be represented as similar. The methods used to obtain data and the manner in which data is presented will be consistent throughout this program to ensure comparability between data sets.

Representativeness is defined as the degree to which data accurately and precisely represent the frequency distribution of a specific variable in the population. A great deal of thought will be spent by the Principal Investigators at each site to ensure that the data obtained is representative. Issues that will be assessed are the number and location of wells with regard to the plume at each site, the number of samples taken, and the analytes present at each site.

#### 9.7 Performance and System Audits

**9.7.1 Performance and Systems Audits**. The ARCADIS Geraghty & Miller QA Officer, or her designee, routinely performs audits to ensure that projects are performed according to plan and that acquired environmental data is of a known and defensible quality. Audits performed by ARCADIS Geraghty & Miller on ARCADIS Geraghty & Miller projects are considered internal audits. Audits performed by a third party or by EPA are considered external audits.

An internal technical systems audit (TSA) for at least one site will be performed during the early stages of this demonstration. The QA Officer will use this QA Project Plan as a basis for the TSA

Contract No.: F41624-99-C-8032 Contractor Name: ARCADIS Geraghty & Miller Contractor Address: 1099 18th Street, Suite 2100, Denver, CO 80202 Expiration: Receipt of final payment by Contractor upon completion of the Contract, including all modifications, *or* two years after Contract Award date, whichever occurs at the latest date.

The Government's rights to use, modify, reproduce, release, perform, display, or disclose these technical data are restricted by paragraph (b)(3) of the Rights in Technical Data--Noncommercial items clause contained in the above identified contract. Any reproduction of technical data or portions thereof marked with this legend must also reproduce the markings. Any person, other than the Government, who has been provided access to such data must promptly notify the above named Contractor.

checklist, in addition to the standard methods used for sampling and analysis. The purpose of the technical systems audit is to ensure that the project is carried out as planned and that any deviations from the methods or plan are adequately documented. To reduce costs, the QA Officer may assign a Deputy QA Officer from an office located in close proximity to the sampling site to perform the field audit of sampling procedures. This Deputy QA Officer will be independent of the project and technically qualified to carry out this duty.

Analytical activities may be audited by providing the subcontracted laboratories with a performance evaluation audit (PEA) sample. The laboratories that are being used are routinely audited under EPA's contract laboratory program and through ARCADIS Geraghty & Miller's internal laboratory approval program. The date and results from the last audit performed at each laboratory will be requested. If they have not been audited within the last year, by EPA or ARCADIS Geraghty & Miller, analysis of PEA samples supplied blind by the ARCADIS Geraghty & Miller QA Officer will be required.

**9.7.2 Contingency Laboratory.** During the competitive bidding process discussed above contingency laboratories were identified. STL and IES can serve as contingency labs in case Quanterra is unable to perform. Microseeps can serve as a contingency laboratory in case vapor Tech is unable to perform. ARCADIS Geraghty & Miller can perform the particle size analysis in its own facility if WEA is unable to perform.

#### 9.8 Quality Assurance Reports

Quality related problems will be addressed in monthly progress reports prepared by the Project Manager if data quality is compromised. Reports will detail any limitations on the data and any corrective actions that were implemented to resolve the problem.

Any findings, problems, or observations found through internal audits by the QA Officer will be reported directly to the Project Manager. Major concerns will be expressed on the day of the audit if immediate corrective actions are necessary. The QA Officer will submit an audit report to the Project Manager within 15 days of completion of any internal audit.

The final report for each site will contain a QA section that will specify the QA activities that were conducted at the site and the quality of data achieved. It will provide sufficient information to enable users to have confidence in the data. If the data have limitations, the QA section will detail those limitations. The QA section of the final report will also relate data to the established data quality objectives and data quality indicator goals and explain any significant differences.

It is anticipated that significant quality assurance issues will be discussed in monthly status reports and quarterly performance and cost reports as well as annual presentations. Therefore interested parties will be aware of these issues if any arise before completion of the final report.